# EXPRESSION OF GABA<sub>A</sub> RECEPTOR SUBUNIT GENES IN THE AVIAN SONG SYSTEM AND THEIR ROLE IN LEARNING AND MEMORY

**AMY POOLE** 

A thesis submitted in partial fulfillment of the requirements of Nottingham Trent University for the degree of Doctor of Philosophy

March 2010

This work is the intellectual property of the author, and may also be owned by Nottingham Trent University. You may copy up to 5% of this work for private study, or personal, non-commercial research. Any re-use of the information contained within this document should be fully referenced, quoting the author, title, university, degree level and pagination. Queries or requests for any other use, or if a more substantial copy is required, should be directed in the first instance to the author.

Data from the following thesis will contribute to three papers currently in preparation and has been presented at conferences held by the British Neuroscience association (Poster; Harrogate, 2007), Federation of European Neuroscience Societies (Poster; Vienna, 2008), Edinburgh Napier University (Seminar, 2010) and Nottingham Trent University (Poster and Oral presentations; Nottingham, 2006, 2010).

# ABSTRACT

 $\gamma$ -aminobutyric acid type A (GABA<sub>A</sub>) receptors are the primary mediators of inhibitory neurotransmission in the brain. In avian systems, 14 GABA<sub>A</sub> receptor subunits ( $\alpha$ 1-6;  $\beta$ 2- $\beta$ 4;  $\gamma$ 1, 2 and 4,  $\delta$  and  $\pi$ ) have been identified. These assemble into pentameric transmembrane structures with an intrinsic chloride-selective pore and are involved in the modulation of learning and memory. Following imprinting training in the one-day old chicken, mRNA encoding the GABA<sub>A</sub> receptor  $\gamma$ 4 subunit is significantly reduced in learning-relevant brain regions indicating a role for receptors comprising this subunit in learning and memory. The zebra finch (Taenopygia guttata) song system has long since been used as a paradigm for studying the underlying molecular mechanisms of learning and memory due to the discrete nature of song, the song system and established stages in song development. The avian brain displays many comparable structures and pathways to mammalian systems and there are striking parallels between birdsong and speech production in humans hence the fundamental neuronal mechanisms are similar. Despite major developments towards understanding the anatomical and electrophysiological properties of various song-system nuclei, the nature of the underlying molecular and biochemical/genetic architecture remains largely unknown. Electrophysiological and pharmacological techniques have localised GABA<sub>A</sub> receptors in the song system and more recently the spatial distribution of y4-subunit mRNA has been mapped, producing striking results. This had inspired this study (the first of its kind in zebra finch) to isolate all complementary DNAs for zebra finch  $\alpha 1-6$ ;  $\beta 2-\beta 4$ ;  $\gamma 1$ ,  $\gamma 2$ ,  $\gamma 4$ ,  $\delta$  and  $\pi$  subunits and characterise their expression in this learning and memory paradigm. Radioactive in situ hybridisation was employed to quantitatively map all GABAA receptor subunits within four important nuclei of the song system (LMAN, Area X, HVC and RA) and revealed regionspecific transcription of genes; each subunit exhibiting an individual expression profile in brain, indicative of a variety of major and minor subtype assemblies. Real-time RT-PCR confirmed developmentally-associated subunit mRNA levels in all song-system nuclei. y4-subunit mRNA exhibited the most robust expression and coordinated peaks of mRNA expression at specific developmental time points in the relevant nuclei, subsequently expression of the  $\gamma$ 4-subunit mRNA was examined in a behavioural context. Corresponding mRNA was down-regulated in response to acquisition and production of song (not evident with  $\gamma$ 2-subunit mRNA), indicative of a specialised role for these receptors in cognitive processes. Preliminary in vitro RNA interference experiments indicated that  $\gamma$ 4-subunit gene expression could be reduced in chicken neurons, paving the way for *in vivo* gene silencing experiments in zebra finch.



-from the Edwin Smith Surgical Papyrus, (Egypt, 1700 BC), the oldest written record of the word "brain"

'After the first glass of you see things as you wish they were. After the second you see them as they are not. Finally, you see things as they really are, and that is the most horrible thing in the world' - Oscar Wilde on Absinthe (a modulator of GABA<sub>A</sub> receptors)

# ACKNOWLEDGEMENTS

First and foremost I would like to thank Mark Darlison for giving me such a great opportunity and sharing with me his invaluable knowledge and enthusiasm for research and molecular biology.

To my second supervisor Christian Thode I thank for his guidance in the laboratory and his good advice over the years.

I am grateful to our collaborators, Manfred Gahr, Carolina Frankl and Falk Dittrich at the Max-Plank-Institut für Ornithologie, Seewiesen for their generous gifts of zebra finches, and their help with the real-time RT-PCR experiments.

Many thanks to all my colleagues at Nottingham Trent University, most importantly Tina Schwabe, Fiona Freeman, David Munoz, Wayne Harris and Anne Schneider-Darlison for all their help and support and making our laboratory such a friendly place. I would also like to acknowledge the PhD students at Edinburgh Napier University, namely Jen Behan, Matt Boyles and Julie Watson for being so welcoming and making my write-up *almost* enjoyable!

Thank you to my family, Mum, Dad, Lucy, Vicky and Laura for your unfailing support (both emotional and financial!) and for keeping me smiling, this one's for you.

Lastly, but not least I wish to thank Ash Belsham, who always believed in me even when I didn't believe in myself.

To you all, I couldn't have done it without you.

1.	GENERAL INTRODUCTION	1
1.1 '	y-aminobutyric acid (GABA)	1
1.2 GABA <sub>A</sub> receptors		2
1	.2.1 GABA <sub>A</sub> receptor-mediated neurotransmission	2
1	.2.2 GABA <sub>A</sub> receptor function during development	3
1	.2.3 GABA <sub>A</sub> receptor structure	4
1	.2.4 GABA <sub>A</sub> receptor subunits	7
1	.2.5 $GABA_A$ receptor subtypes	9
1	2.5.6 Clinical importance of GABA <sub>A</sub> receptors	10
1	.2.7 Enaogenous modulators of GABA <sub>A</sub> receptors .2.8 GABA <sub>A</sub> receptor trafficking	13 14
1.3	GABA <sub>C</sub> Receptors	16
1.4 '	The avian song system	18
1	.4.1 History of songhird research	19
1	.4.2 Structure of the song system	22
1	.4.3 Sexual dimorphism	25
1	.4.4 Stages of song development	27
1	.4.5 Song structure	30
1	.4.6 Functions of song	30
1	.4.7 Seasonal plasticity	32
1	.4.8 Song production and human speech	34
1	.4.9 $GABA_A$ receptors in the avian song system	35
1.5	Project Aims	37
2. 1	MATERIALS AND METHODS	38
2.1	Animals	38
2.2	Isolation of total RNA from tissues	38
2	.2.1 Isolation of total RNA	38
2	.2.2 Determining concentration of nucleic acids	39
2.3	Synthesis of complementary DNA (cDNA) from isolated RNA	39
2	.3.1 Removal of genomic DNA	39
2	.3.2 Phenol/chloroform extraction	39
2	.3.3 Ethanol precipitation	40
2	.3.4 cDNA synthesis: reverse transcription reaction	40
2	.3.5 Precipitation and washing of cDNA	40
2.4	Amplification of $\mathbf{GABA}_{\mathbf{A}}$ and $\mathbf{GABA}_{\mathbf{C}}$ receptor subunit cDNAs	41
2	.4.1 Reverse transcription-polymerase chain reaction (RT-PCR) primer design	41
2	.4.2 Reverse-transcription polymerase chain reaction (RT-PCR)	42
2	.4.3 Nested RT-PCR	43
2	.4.4 Agarose gel electrophoresis	44
2	.4.5 KT-PCR product purification	44
2.5	Cloning of partial cDNAs	45
2	.5.1 Ligation	45
2	.5.2 Synthesis of competent <u>Escherichia coli</u> XL1-Blue and TB1 cells	45
2	.5.3 Transformation of competent <u>E.coli</u> XL1-Blue cells	46

2.5.4 Overnight cell culture	46
2.5.5 Isolation of plasmid DNA	47
2.5.6 Restriction digestion	47
2.6 Padioactive in situ hybridisation	47
2.6 1 Oligonucleotide prohe design	48
2.6.1 Ougonactional prove aesign 2.6.2 Silanisation of microscone slides	40 /10
2.6.2 Summation of meroscope states 2.6.3 Rearing of zehra finches and sectioning of hrains	50
2.6.5 Rearing of zeora functies and sectioning of orains 2.6.4 Firation of tissue sections	51
2.6.4 I mainted by ussue sections 2.6.5 Labelling of the aligonucleatide probe	51
2.6.5 Labelling of the organic centre probe 2.6.6 Hybridisation of the probe	52
2.6.8 Hybraisation of the prove	52
2.6.7 Washing of sections 2.6.8 Densitometric analysis of autoradiographs	53
	54
2.7 Real-time quantitative RT-PCR	54
2.7.1 Animals and tissue dissection	54
2.7.2 Isolation of RNA	54
2.7.3 Real-time RT-PCR primer design	55
2.7.4 Relative quantification of mRNA levels	56
<b>2.8</b> Generation of $\gamma$ 4-subunit-specific antibodies	56
2.8.1 pMAL <sup>TM</sup> protein fusion and purification system	57
2.8.2 Design and construction of the fusion plasmid	57
2.8.3 Amplification and cloning of the fusion protein	58
2.8.4 Pilot experiment	59
2.8.4.1 Transformation	60
2.8.4.2 Subculturing	60
2.8.4.3 IPTG induction	60
2.8.4.4 Protein extraction from cells	61
2.8.5 Discontinuous SDS-PAGE	62
2.8.5.1 Preparation of the resolving gel	62
2.8.5.2 Preparation of the stacking gel	63
2.8.5.3 Staining of the protein gel	64
2.8.6 Western Blotting	64
2.8.6.1 Assembly of the blot sandwich	64
2.8.6.2 Blocking of the membrane (reduction of non-specific binding)	64
2.8.6.3 Immunoprobing with primary antibody	65
2.8.6.4 Immunoprobing with secondary antibody	65
2.8.6.5. Immunodetection of proteins	65
2.8.7 Affinity chromatography	66
2.8.7.1 Subculturing	66
2.8.7.2 Protein extraction	66
2.8.7.3 Protein estimation assay	67
2.8.7.4 Affinity chromatography	67
2.8.7.5 Dialvsis	68
2.8.8 Generation of a peptide anti-GABA <sub>A</sub> receptor $\gamma$ 4-subunit antibody	69
2.9 RNA interference (RNAi)	70
2.9.1 Short-hairpin (sh)RNA construct design	70
2.9.2 Ligation of shRNAs into pSilencer $^{TM}$ 4.1-CMV neo vector	74
2.9.3 Primary neuronal culture	75
2.9.3.1 Animals	75
2.9.3.2 Establishment of primary neuronal culture	75
2.9.3.3 Transfection of cells	76
2.9.4 RNA extraction from cells	, 0 77
2.9.5 Real-time RT-PCR	77

<b>3. SEQUENCES ENCODING ZEBRA FINCH GABAA AND GABAC</b>	
RECEPTOR SUBUNITS	80
3.1 Introduction	80
3.1.1 Isolation of GABA <sub>A</sub> receptor subunit sequences	80
3.1.2 GABA <sub>A</sub> receptor subunit sequences	80
3.1.3 GABA <sub>A</sub> receptor subunit splice variants	81
3.1.4 GABA <sub>C</sub> receptor subunit splice variants	82
3.1.5 GABA <sub>C</sub> receptors	83
3.1.6 Sequencing of the zebra finch genome	85
3.2 Results	85
3.2.1 Sequences encoding zebra finch $GABA_A$ receptor subunits	85
3.2.1 Sequences encoding zebra finch $GABA_C$ receptor subunits	95
3.4 Discussion	100

100

137

#### 4. EXPRESSION OF GABAA AND GABAC RECEPTOR SUBUNIT GENES IN THE SONG SYSTEM 109

4.1 Introduction	109
4.1.1 Heteromerisation of $GABA_A$ and $GABA_C$ receptors	110
4.1.2 Zebra finch song system	110
4.1.3 GABA <sub>A</sub> receptors in zebra finch song system	111
4.2 Results	113
4.2.1 Expression of GABA <sub>A</sub> receptor $\alpha$ -subunit genes in the song system	114
4.2.2 Expression of GABA <sub>A</sub> receptor $\beta$ -subunit genes in the song system	116
4.2.3 Expression of $\gamma$ - and $\delta$ -subunit genes in the song system	118
4.2.4 GABA <sub>4</sub> receptor $\pi$ subunit	122
4.2.5 Expression of GABA <sub>C</sub> receptor subunit mRNAs in song system	123
4.2.6 Expression of $GABA_A$ receptor subunit genes in the zebra finch song system during	
development	124
4.3 Discussion	128
4.3.1 Spatial expression of GABA <sub>A</sub> receptor subunit genes in the song system	128
4.3.2 Expression of GABA <sub>A</sub> receptor $\pi$ -subunit gene in avian brain	133
4.3.3 Spatial expression of $GABA_C$ receptor subunit genes in the song system	134
4.3.4 Developmental expression of $GABA_A$ receptor subunit genes in the song system	135

# 4.3.4 Developmental expression 4.3.5 Concluding remarks

# 5. THE EFFECT OF AUDITORY LEARNING ON THE EXPRESSION OF THE GABA<sub>A</sub> RECEPTOR $\gamma$ 4-SUBUNIT GENE IN THE SONG SYSTEM 139

5.1 Introduction	139
5.2 Results	144
5.2.1 Internal control region (Nidopallium)	145
5.2.2 Spatial expression of $GABA_A$ receptor $\gamma$ 4-subunit gene in aviary-reared and tutor	-isolated
male zebra finch brains	147
5.2.3 Quantitative expression of $GABA_A$ receptor $\gamma$ 4-subunit gene in aviary-reared and	tutor-
isolated male zebra finch brains	148
5.2.3.1 Internal control (74-subunit gene expression in nidopallium)	148
5.2.3.2 GABA <sub>A</sub> receptor $\gamma$ 4-subunit mRNA expression in song-system nuclei	149
5.2.4 Spatial expression of GABA <sub>A</sub> receptor $\gamma$ 2-subunit gene in aviary-reared and tutor-isolated	
male zebra finch brains	151

5.2.5 Quantitative expression of GABA <sub>A</sub> receptor $\gamma$ 2-subunit gene in tutor isolated and non-tu	tor
isolated male zebra finch brains	153
5.2.5.1 Internal control (12-subunit gene expression in nidopallium)	153
5.2.5.2 GABA <sub>A</sub> receptor $\gamma$ 2-subunit gene expression in song-system nuclei	154
5.2.6 Effects of tutor-isolation on song production	155
5.2.7 Spatial expression of the GABA <sub>A</sub> receptor $\gamma$ 4-subunit gene in aviary-reared, cage-reared	l and
tutor-isolated zebra finches	156
5.2.8 Quantitative expression of the GABA <sub>A</sub> receptor y4-subunit gene in aviary-reared, cage-	
reared and tutor-isolated zebra finches	158
5.2.8.1 Internal control (74-subunit gene expression in nidopallium)	158
5.2.8.2 GABA <sub>A</sub> receptor $\gamma$ 4-subunit mRNA expression in song-system nuclei	159
5.3 Discussion	161
5.3.1 Effects of tutor-isolation on song	162
5.3.2 Effects of environment on song production	163
5.3.3 Spatial expression of 1/4- and 1/2-subunit genes in aviary-reared and tutor-isolated birds	164
5.3.4 Down-regulation of GABA <sub>A</sub> receptor 74-subunit transcript in response to song learning	166
5.3.5 Temporal expression of $GABA_A$ receptor $\gamma$ 4-subunit gene	170
5.3.6 Regulation of gene expression	171
5.3.7 Lateralisation of function	173
5.3.9 Concluding remarks	173

# 6. PRELIMINARY WORK TO CONFIRM THE ROLE OF $\gamma 4\mbox{-}SUBUNIT-CONTAINING GABA_A RECEPTORS IN THE SONG SYSTEM$

175

6.1 Introduction	175
6.1.1 Immunohistochemical/immunocytochemical studies	175
6.1.2 Co-precipitation studies	177
6.1.3 Generation of GABA <sub>A</sub> receptor 14-subunit-specific antibodies	178
6.1.4 Reverse-genetic strategies employed for assessing $GABA_A$ receptor function	179
6.1.5 RNA interference (RNAi)	181
6.1.5.1 Mechansims of RNAi	181
6.1.5.2 Functions of RNAi	185
6.1.5.3 Interferon response	185
6.1.5.4 Induction of RNAi in cells	185
6.1.5.5 Design of RNAi experiments	186
6.1.5.6 Applications of RNAi	188
6.1.5.7 RNAi in zebra finch brain	189
6.2 Results	190
6.2.1 Generation of a γ4-subunit specific antibody	190
6.2.1.1 Amplification of a partial region of the zebra finch $GABA_A$ receptor $\gamma$ 4-subunit	100
transcript	190
0.2.1.2 Fuol experiment	192
6.2.2 KIVA Interference (KIVAI)	190
6.2.2.1 Optimisation of real-time R1-PCR primers	190
6.2.2.2 Fillinally culture and transfection	197
0.2.2.5 Quantitative real-time K1-r CK	197
6.3 Discussion	199
6.3.1 $\gamma$ 4-subunit-specific antibody generation	200
6.3.2 Silencing of the GABA <sub>A</sub> receptor $\gamma$ 4-subunit gene	201
7. DISCUSSION	205
7.1 Isolation of partial cDNAs encoding $\text{GABA}_{A}$ and $\text{GABA}_{C}$ receptor subunits	205
7.2 Spatial distribution of $GABA_A$ receptor subunit mRNAs in song system	206

7.3 Potential association of $\mathbf{GABA}_{\mathbf{C}}$ and $\mathbf{GABA}_{\mathbf{A}}$ receptor subunits	209
7.4 Importance of the $GABA_A$ receptor $\gamma$ 4-subunit	210
7.5 Regulation of $GABA_A$ receptor subunit gene expression	212
7.6 Concluding remarks	214
BIBLIOGRAPHY	218
APPENDIX	250

# LIST OF FIGURES

Figure 1.1. Toplogy of the GABA <sub>A</sub> receptor.	6
Figure 1.2. Schematic detailing anatomical organisation of the song system nuclei	23
Figure 1.3. Schematics of male zebra finch, non-song bird and female zebra finch	26
Figure 1.4. Timeline detailing stages of song learning in the zebra finch	29
Figure 1.5. Audiospectrograms showing zebra finch song at different developmental stages	31
<b>Figure 2.1</b> . Nucleotide and peptide alignments of GABA <sub>A</sub> receptor $\gamma$ 4-subunit	70
Figure 2.2. shRNA construct design	73
Figure 3.1. Schematic diagram depicting $GABA_A/GABA_C$ receptor subunit topology	86
<b>Figure 3.2a.</b> Alignment of zebra finch and chicken $\alpha$ 3-subunit sequences	86
<b>Figure 3.2b.</b> Alignment of zebra finch and chicken $\alpha$ 4-subunit sequences	87
<b>Figure 3.2c.</b> Alignment of zebra finch and chicken $\alpha$ 6-subunit sequences	87
Figure 3.2d. Alignment of zebra finch and chicken γ1-subunit sequences	88
<b>Figure 3.2e.</b> Alignment of zebra finch and chicken $\delta$ -subunit sequences	88
<b>Figure 3.2f.</b> Alignment of zebra finch and chicken $\pi$ -subunit sequences	91
<b>Figure 3.3</b> . Exonic arrangement of zebra finch GABA <sub>A</sub> receptor $\pi$ -subunit gene	92
<b>Figure 3.4</b> . Spidey alignment GABA <sub>A</sub> receptor $\pi$ subunit with zebra finch chromosome 13	93
Figure 3.5. Alignment of all cloned zebra finch GABA <sub>A</sub> receptor subunit sequences	95
Figure 3.6a. Alignment of zebra finch and chicken p1-subunit sequences	96
Figure 3.6b. Alignment of zebra finch and chicken p2-subunit sequences	97
Figure 3.6c. Alignment of zebra finch and chicken p3-subunit sequences	97
Figure 3.7. Alignment of all zebra finch GABA <sub>C</sub> receptor subunit sequences	98
Figure 3.8. Chromosomal of arrangements $GABA_A$ and $GABA_C$ receptor subunit genes	99
Figure 4.1. Inverse autoradiographs of GABA <sub>A</sub> receptor $\alpha$ -subunit transcripts	114
Figure 4.2. Densitometric quantification of $GABA_A$ receptor $\alpha$ -subunit gene expression	115
Figure 4.3. Inverse autoradiographs of GABA <sub>A</sub> receptor $\beta$ -subunit transcripts	116
Figure 4.4. Densitometric quantification of $GABA_A$ receptor $\beta$ -subunit gene expression	117
Figure 4.5. Inverse autoradiographs of GABA <sub>A</sub> receptor $\gamma$ - and $\delta$ -subunit transcripts	118
<b>Figure 4.6</b> . Densitometric quantification of $\gamma$ - and $\delta$ -subunit gene expression	119
<b>Figure 4.7.</b> Amplification of $\pi$ -subunit partial cDNAs from different tissues of zebra finch	122
Figure 4.8. Inverse autoradiographs of GABA <sub>c</sub> receptor $\rho$ 1-, $\rho$ 2-, and $\rho$ 3-subunit transcripts	123
<b>Figure 4.9</b> . Developmental expression of GABA <sub>A</sub> receptor $\alpha$ -subunit genes	125
<b>Figure 4.10</b> . Developmental expression of GABA <sub>A</sub> receptor $\beta^2$ -, $\beta^3$ - and $\gamma^4$ -subunit genes	126
Figure 5.1. Sagittal schematic illustrating layers within the zebra finch brain	145
Figure 5.2. Schematics detailing the location of the selected nidopallial internal control areas	146
Figure 5.3. Quantification of GABA <sub>A</sub> receptor $\gamma$ 4-subunit gene expression in the nidopallium of	isolated
and unisolated zebra finches	148

Figure 5.4. Quantification of $GABA_A$ receptor $\gamma$ 4-subunit gene expression in the song system of i	solated	
and unisolated zebra finches	149	
Figure 5.5. Quantification of $GABA_A$ receptor $\gamma$ 4-subunit gene expression in left and right hemisp	oheres of	
zebra finch brain	150	
<b>Figure 5.6.</b> Inverse autoradiographs of GABA <sub>A</sub> receptor $\gamma$ 2-subunit transcript	152	
Figure 5.7. Quantification of $GABA_A$ receptor $\gamma$ 4-subunit gene expression in the nidopallium of is	solated	
and unisolated zebra finches	153	
Figure 5.8. Quantification of $GABA_A$ receptor $\gamma$ 4-subunit gene expression in the song system of i	solated	
and unisolated zebra finches	154	
Figure 5.9. Quantification of $GABA_A$ receptor $\gamma 2$ -subunit gene expression in left and right hemisp	oheres of	
zebra finch brain	155	
Figure 5.10. Inverse autoradiographs of GABA <sub>A</sub> receptor $\gamma$ 4-subunit transcript	157	
Figure 5.11. Quantification of $GABA_A$ receptor $\gamma$ 4-subunit gene expression in the nidopallium	n of three	
groups of zebra finches	158	
Figure 5.12. Quantification of $GABA_A$ receptor $\gamma$ 4-subunit gene expression in the song system	n of three	
groups of zebra finches	159	
Figure 5.13. Quantification of $GABA_A$ receptor $\gamma$ 4-subunit gene expression in left and right hemi	spheres of	
zebra finch brain	160	
Figure 6.1. Mechanisms of RNA interference (RNAi).	184	
Figure 6.2. Ligation of shRNAs into the pSilencer <sup>TM</sup> 4.1-CMV neo vector	190	
<b>Figure 6.3.</b> DNA fragment encoding a region of the GABA <sub>A</sub> receptor $\gamma$ 4-subunit transcript, amplified from		
zebra finch brain by RT-PCR	191	
Figure 6.4. Alignments of cloned construct with original zebra finch GABA <sub>A</sub> receptor $\gamma$ 4-subunit		
sequences	191	
Figure 6.5. Results of pilot experiment with 10% (w/v) SDS-PAGE and western blot analysis to d	letermine	
activity of the MBP-fusion protein	193	
Figure 6.6. Analysis of eluted MBP-74 fusion protein following affinity purification with an amyle	ose	
column	194	
Figure 6.7. Calibration graph following a BCA assay determining the final protein concentration of	of the	
MBP-y4 fusion protein following dialysis	195	
Figure 6.8. Silencing of $GABA_A$ receptor $\gamma$ 4-subunit and GAPDH genes in primary chicken cerebellar		
neurons	198	

# **ABBREVIATIONS AND SYMBOLS**

α	Alpha
$\alpha^{35}S$	Deoxyadenosine 5'(alpha-thio)
u b	trinhosnhate
A A	A mino agid
	Antino actu
AFP	Anterior forebrain pathway
ATP	Adenosine triphosphate
BLAST	Basic local alignment search tool
ΒZ	Benzodiazepine
bp	Base pair
β	Beta
CACA	Cis-4-aminocrotonic acid
cDNA	Complementary deoxyribonucleic acid
CNS	Central nervous system
δ	Delta
	Deoxyribonucleic acid
	DeloxyIIDolluciele acid
$un_2O$	Defoilised water
DEPC	Diethylpyrocarbonate
DTT	Dithiothreitol
DLM	Medial nucleus of the dorsolateral
	thalamus
EDTA	Ethylenediaminetetraacetic acid
3	Epsilon
γ	Gamma
σ	Acceleration due to gravity
8 GARA	Gamma (y)-aminobutyric acid
GARAD	Camma (y) aminobutyric acid recentor
CAD	Chatemia and deperturned
GAD	Glutamic acid decarboxylase
HA	Apical part of the hyperpallium
HD	Dorsal part of the hyperpallium
HVC	Formal name
IPTG	Isopropyl $\beta$ -D-1-thiogalactopyranoside
IEG	Immediate early gene
Kb	Kilobase
LGIC	Ligand-gated ion channel
LMAN	Lateral magnocellular nucleus of the
	anterior nidopallium
1	Litre
M	Molar
	Maltaas hinding metain
MBP	Mattose binding protein
m	Milli
μ	Micro
min	Minute
mRNA	Messenger ribonucleic acid
Mst	Medial striatum
ng	Nanogram
nm	Nanometer
NCBI	National centre for biotechnology
	information
NIf	Interfacial nucleus of the nidonallium
NVIIto	Trachaosyringaal part of the hypoglossal
INAIIts	malaus
PHD	Post natch day
qPCR	Quantitative PCR
π	Pi
R	Registered trademark
RA	Robust nucleus of the arcopallium
ρ	Rho

	RNA	Ribonucleic acid
	RNAi	Ribonucleic acid interference
	RPM	Revolutions per minute
	rTdT	Terminal deoxynucleotidyl transferase,
		recombinant
RT-PCI		Reverse transcription polymerase chain
		reaction
	SDS	Sodium dodecyl sulphate
	SDS-	Sodium dodecyl suphate-polyacrylamide
	PAGE	gel electrophoresis
	shRNA	Short hairpin ribonucleic acid
	siRNA	Short interfering ribonucleic acid
	SSC	Standard sodium citrate
	TEMED N,N.N',N'-tetramethylethylenediamine	
	θ	Theta
	ТМ	Trademark
	Tm	Melting temperature
	TM	Transmembrane domain
	UV	Ultraviolet
	VMP	Vocal motor pathway
	V/V	Volume per volume
	W/V	Weight per volume
	XGAL	5-bromo-4-chloro-3-indolyl-β-D-
		galactoside

# **1. GENERAL INTRODUCTION**

# **1.1** γ-aminobutyric acid (GABA)

Successful functioning of a neuronal network relies upon efficient signaling between individual neurons. The central nervous system (CNS) is composed of a hugely intricate network of neurons whereby sensory information is interpreted, analysed, stored and responded to. Predominantly, chemical neurotransmitters carry nerve impulses across the majority of synapses (Eccles, 1982). In the vertebrate brain the two most prevalent neurotransmitters are the amino acids, L-glutamate and  $\gamma$ -aminobutyric acid (GABA); which upon binding to corresponding post-synaptic receptors elicit excitatory and inhibitory (except in the developing nervous system) responses respectively. This balance between excitatory and inhibitory neurotransmission results in a finely tuned neuronal network (Hablitz et al., 2009). Unlike the excitatory neurotransmitter glutamate, GABA is a non-protein amino acid; it is rarely used as a component of proteins, but is functional in its own right (Li and Xu, 2008). GABA was first reported in the brain in 1950 during a study of free amino acids (Roberts and Frankel, 1950). Further research concluded GABA to act at synapses eliciting an inhibitory effect (for a review see Brioni, 1993) and it is now known to be the most widely distributed inhibitory neurotransmitter in the CNS (Li and Xu, 2008). Due to localisation of GABAergic interneurons in hypothalamic and brain stem regions, GABA (via its receptors), also acts in other capacities. It is involved in regulation/modulation of blood pressure (Hayakawa et al., 2002), hormone release (Zemkova et al., 2008), respiration (Fujii et al., 2007), circadian rhythms (Liu and Reppert, 2000; Mignot et al., 2002), food/drink intake (Turenius et al., 2009), locomotion (Ménard et al., 2007), anxiety (Rudolph and Möhler, 2006) and learning and memory (Chapouthier and Venault, 2002; 2004; Kalueff, 2007; McNally et al., 2008). However, its predominant effects are within the brain, mediating inhibitory neurotransmission.

Biosynthesis of GABA occurs in GABAergic interneurons where it is packaged into synaptosomes in the nerve terminal ready for release (Hablitz *et al.*, 2009). Rapid  $\alpha$ -decarboxylation of L-glutamic acid by GAD (glutamic acid decarboxylase), with coenzyme pyridoxal phosphate (a form of vitamin B6) produces GABA. There are two isoforms of

GAD in vertebrate brain; GAD<sub>65</sub> (65 kDa protein) and GAD<sub>67</sub> (67 kDa protein). The larger form is found in its active state alongside cofactor pyridoxal phosphate distributed throughout the neuron, and is responsible for basal GABA levels. The smaller form is localised to axon terminals, where only a heightened demand for GABA will cause a transient activation (Kaufman *et al.*, 1991). Breakdown of GABA is catalysed by a mitochondrial aminotransferase, GABA-transaminase (GABA-T). Along with  $\alpha$ ketoglutarate, reversible transanimation of GABA produces succinic semialdehyde and glutamic acid. Both GAD and GABA-T are found throughout inhibitory neurons; GAD is more highly concentrated at the pre-synaptic terminal and GABA-T is confined to the mitochondria. It is GAD that governs the steady state of endogenous GABA rather than GABA-T (Buddhala *et al.*, 2009).

# **1.2 GABA**<sub>A</sub> receptors

GABA is the endogenous agonist of GABA receptors which have been identified based on molecular, electrophysiological and receptor binding studies. There are three recognised types: GABA<sub>A</sub>, GABA<sub>B</sub> and GABA<sub>C</sub>. GABA<sub>A</sub> and GABA<sub>C</sub> receptors are ionotropic and possess intrinsic chloride-selective channels. GABA<sub>B</sub> receptors are metabotropic heterodimers that are coupled to G-proteins, through which they trigger second messenger pathways (Chebib and Johnston, 1999; Bormann, 2000). GABA<sub>A</sub> and GABA<sub>B</sub> receptors are located in the CNS, whereas the majority of GABA<sub>C</sub> receptors are confined to the retina (Albrecht and Darlison, 1995; McCall *et al.*, 2002; Rozzo *et al.*, 2002). Although mounting evidence has revealed some GABA<sub>C</sub> receptor expression in the rat brain; it is only at low levels (discussed further in section 1.3. Boue-Grabot *et al.*, 1998; Ogurusu *et al.*, 1999; Rozzo *et al.*, 2002; Milligan *et al.*, 2004; Alakuijala *et al.*, 2005). Within the CNS, GABA<sub>A</sub> receptors far outnumber GABA<sub>B/C</sub> receptors in their distribution and so are universally regarded as the major mediators of fast inhibitory synaptic transmission (Bormann, 2000).

## 1.2.1 GABA<sub>A</sub> receptor-mediated neurotransmission

When a pre-synaptic GABAergic interneuron is activated, the ensuing action potential travels along the neuron until it reaches the nerve terminal, where GABA is stored in membrane-bound vesicles. Depolarisation of the nerve-terminal plasma membrane

transiently opens voltage-gated Ca<sup>2+</sup> channels which causes an influx of Ca<sup>2+</sup> into the terminal and increases cytosolic [Ca<sup>2+</sup>]. This results in calcium-dependant exocytosis of GABA which rapidly diffuses across the synaptic cleft (< 1 ms). Two molecules of GABA simultaneously bind to post-synaptically located GABA<sub>A</sub> receptors, which become activated and undergo a conformational change, opening the Cl<sup>-</sup> -specific pore. A Cl<sup>-</sup> influx follows down the electrochemical gradient (~10<sup>6</sup> ions/sec), into the post-synaptic neuron, hyperpolarising the membrane. IPSP (inhibitory post-synaptic potential) is induced and the influx of Ca<sup>2+</sup> becomes inhibited, resulting in a decrease neuronal excitability. Excess GABA in the synaptic cleft is taken up into the pre-synaptic neuron and surrounding glial cells by high affinity neuronal uptake systems.

In addition to the rapid (phasic) inhibitory action of GABA<sub>A</sub> receptors, they can also mediate a persistent tonic current, via receptors located extrasynaptically and perisynaptically (Semyanov *et al.*, 2004). Tonic inhibition is a result of micromolar GABA levels which persist extracellularly and are contributed to by 'overspill' following fast phasic inhibitory transmission. Extrasynaptic receptors are constantly activated and exhibit a high GABA affinity (Mody and Pierce, 2004; Farrant and Nusser, 2005). More recent work has elucidated that the subunit composition of extrasynaptic GABA<sub>A</sub> receptors is also different to those located at the synapse (refer to section 1.2.4; Glykys *et al.*, 2008).

## 1.2.2 GABA<sub>A</sub> receptor function during development

As previously outlined, excitatory neurotransmitters (e.g. glutamate) induce  $Ca^{2+}$  influx which increases the excitability of the post-synaptic neuron; inhibitory neurotransmitters (e.g. GABA) reduce  $Ca^{2+}$  influx into the neuron, thereby decreasing neuronal excitability. However, in early development of the mammalian CNS (in restricted brain regions), release of GABA *increases* the  $Ca^{2+}$  concentration in the post-synaptic neuron, eliciting an excitatory effect (Lin *et al.*, 1994; Ganguly *et al.*, 2001). During CNS development, GABA<sub>A</sub> receptors appear first (before glutamate receptors) and are widely distributed. Accordingly, they mediate the majority of signal transduction (Li and Xu, 2008), by acting in an excitatory capacity. The paradoxical excitatory action mediated by GABA<sub>A</sub> receptors early in development can be attributed to a high intracellular Cl<sup>-</sup> concentration (Stein, 2004). It is purported that this temporal GABAergic excitatory signal transduction is assuming the role of the relatively undeveloped excitatory inputs, in order to provoke increased intracellular Ca<sup>2+</sup> and stimulate neuronal growth (Laurie *et al.*, 1992b; Li and Xu, 2008). When the GABA<sub>A</sub> receptors are activated, the central ion channel opens and Cl<sup>-</sup> effluxes resulting in depolarization of the post-synaptic membrane. As development of the CNS ensues, intracellular Cl<sup>-</sup> concentration progressively decreases until it is lower than that of the extracellular matrix and at this point 'GABA switching' terminates (Herlenius, 2004). Intracellular Cl<sup>-</sup> concentration is primarily governed by cation-cotransporters: KCC1-4 (K<sup>+</sup>-dependent Cl<sup>-</sup> cotransporters); NCC (Na<sup>+</sup>-dependent Cl<sup>-</sup> cotransporters and NKCC1 and 2 (K<sup>+</sup>- and Na<sup>+</sup>-dependent Cl<sup>-</sup> cotransporters). In situ hybridisation experiments have mapped mRNA expression of the two primary cotransporters found in the mammalian CNS, NKCC1 (Cl<sup>-</sup> into cell) and KCC2 (Cl<sup>-</sup> out of cell) (Kahle *et al.*, 2005; Giménez, 2006). NKCC1 mRNA levels are high in the embryonic and early developing brain where KCC2 transcript levels are low; however, <1 week postnatal the mRNA levels for NKCC1 were decreased accompanied by an increase in the distribution of mRNA encoding of KCC2 (Clayton et al., 1998; Li et al., 2002; Yamada et al., 2004).

#### 1.2.3 GABA<sub>A</sub> receptor structure

GABA<sub>A</sub> receptors are members of the Cys-loop pentameric ligand-gated ion channel (LGIC) superfamily. A group named according to the characteristic 15 AA conserved motif in the extracellular N-terminus with two terminal cysteines which form a disulphide bridge (Simon *et al.*, 2004). Other members include, the nicotininc acetylcholine receptors (nAChRs), glycine receptors, 5-hydroxytryptamine type 3 (5-HT<sub>3</sub>) receptors and more recently a  $Zn^{2+}$ -activated ion channel (Davies *et al.*, 2003). These channel proteins form tiny hydrophilic pores in the membrane through which solutes can pass through by diffusion. They are selective and gated, the selectivity of the channel depends upon its diameter, shape and the distribution of charged amino acids in its lining (Jensen *et al.*, 2002; Karlin, 2002). nAChRs, 5-HT3Rs, and Zn<sup>2+</sup>-activated channels are selective for cations (thereby eliciting an inhibitory response); GABA<sub>A</sub> and Glycine receptors are selective for anions, eliciting an inhibitory response. In keeping with the characteristics of the other group members, the GABA<sub>A</sub> receptor is composed of five membrane-spanning

subunits arranged around a central pore (Fig. 1.1b) into a heteropentameric glycoprotein of ~275 kDa. The absolute structure of LGICs cannot be defined due to limitations of the various techniques. For example, crystal structures may be contorted due to the process of purification and crystallisation, and the loss of information regarding interactions with other proteins and the lipid environment (Bogdanov et al., 2005). To date, no X-ray structure or nuclear magnetic resonance (NMR) data is available for LGICs in mammals, however recent studies have resolved the structure of bacterial LGIC homologues (of which there are more than 20: Hilf and Dutzler, 2008, Bocquet et al., 2009. The determination of the crystal structure of acetylcholine-binding protein (AChBP) from Lymnea stagnalis has provided some comprehension pertaining to the structure of the extracellular domain of ACh receptor subunits and subsequently serves as a template for other LGICs (Brejc et al., 2001). In silico analysis has most often been employed for the determination of protein structure, modeled on the hydrophobic nature of the constituent amino acids, though these present only 60-70% accuracy in predicting protein topology (Taylor *et al.*, 2003). Insights into TM segment orientation and structure of the pore have been further eluded by using site-directed mutagenesis and substituted-cysteine-accessibility-methods (SCAM; Karlin and Akabas, 1998; Bogdanov et al., 2005). In brief, protein analysis in this manner requires the removal of all native cysteine residues and subsequent systematic mutation of individual residues to cysteine where the thiol side-chains react with a range of charged and polar MTS (methanethiosulfonate) reagents developed by Karlin and colleagues (1998). By characterisation of the reaction and accessibility (i.e. buried or exposed) of the substituted cysteine with the thiol-specific reagents (which can be either membrane-permeable, or impermeable dependent on whether residues to be labelled are buried within the protein, exposed to the lipid bilayer or exposed to water within the channel), features such as channel-lining residues, secondary structure, physical size, electrostatic potential, location of selectivity filters, gates, binding sites of channel blockers can be identified (Karlin and Akabas, 1998).



**Figure 1.1 Schematic representation of the toplogy of the GABA**<sub>A</sub> receptor. (a) The GABA<sub>A</sub> receptor is a heteromeric transmembrane structure, the five subunits most commonly consist of 2 α's, 2 β's and a single  $\gamma$  subunit (Pirker *et al.*, 2000; Whiting, 2003a) with the stoichiometry α-β-α-β- $\gamma$  (Baumann *et al.*, 2002) assembled around a chloride pore. (b) GABA<sub>A</sub> receptor subunits are individual polypeptides of ~450 amino acids (AA) in length (50-60 kDa). They comprise a short signal peptide (7 AA) and a long amino-terminal (NH<sub>2</sub>) extracellular domain (220 to 250 AA) which contains sites for glycosylation and the well conserved characteristic 15 AA di-cysteine loop (depicted in pink; Jones and Sattelle, 2008). There are four highly conserved hydrophobic transmembrane α-helices (TM1-4, ~20 AA each); it is TM2 (highlighted in green) which forms the lining of the central chloride anion pore in conjunction with TM1 (Sine and Engel, 2006). The third and fourth transmembrane domains are linked by a large variable intracellular loop which contains numerous consensus sites for phosphorylation (highlighted in yellow) and an α-helical domain within this loop which is thought to influence ion conduction (Peters *et al.*, 2005), followed by a short extracellular carboxyl-terminal (COOH). All the subunits share a similar tertiary structure and are of a similar size (30 Å x 40 Å x 160 Å). (d) They assemble into pentamers with their long axes perpendicular to the lipid bilayer (Unwin, 2005).

A combination of the above described methods (alongside cryo-electron microscopy and site-directed mutagenesis studies) has resulted in resolution of the structure of the central ion-conducting channel of GABA<sub>A</sub> receptors. It is composed primarily of five  $\alpha$ -helical TM2 regions which are slightly 'bent' at the centre with TM1, TM3 and TM4 separating the TM2 domains from the lipid bilayer with a surrounding ring of  $\alpha$ -helices (Unwin, 2005). The TM2 domains form a funnel structure which narrows towards the intracellular side (Karlin, 2002). A selectivity filter (selecting ion charge and size) is located nearer to the intracellular end (Xu and Akabas, 1996; Karlin, 2002). Selectivity can be attributed to individual pore-lining residues; mutation of five amino acids of the GABA<sub>A</sub> receptor  $\beta 3$ subunit TM2 domain resulted in a cation-selective GABA<sub>A</sub> receptor. Interestingly, no such effect was observed for the  $\alpha 1$  and  $\gamma 2$  subunits indicating the importance of  $\beta$  subunits in influencing ion-selectivity (Jensen *et al.*, 2002). When the channel is closed or in a resting state, there is a gate which occludes ion passage; this gate is opened upon agonist binding to the extracellular ligand-binding domain (in N-terminus), allowing solutes to pass into the cell (Akabas, 2004). With prolonged exposure to agonist, the receptor becomes desensitised and conduction of ions is suspended. There is some considerable debate as to the location of the gate within TM2 or the TM1-TM2 intracellular loop (Peters et al., 2005; Bali and Akabas, 2007) and whether the gating method is the same for closed and desensitised receptors (Purohit and Grosman, 2006).

#### 1.2.4 GABA<sub>A</sub> receptor subunits

Complexity within the GABA<sub>A</sub> receptor system can be attributed to a vast combination of available subunits which co-assemble into heteropentameric structures. In the mammalian CNS, 16 individual GABA<sub>A</sub> receptor subunit polypeptides have been identified ( $\alpha$ 1-6,  $\beta$ 1-3,  $\gamma$ 1-3,  $\delta$ ,  $\varepsilon$ ,  $\theta$  and  $\pi$ ) each encoded for by a separate gene. Detailed analysis of the human genome database has revealed no further subunit sequences exist (Simon *et al.*, 2004). Subunits are classified exclusively upon their sequence homology with other members of a group. Subunits within the same class share ~70% sequence identity, whereas with subunits from different classes share only ~30% sequence identity (Darlison *et al.*, 2005). Despite the plethora of subunits, they all share a highly conserved structure (Fig. 1.1c).

In lower vertebrates and avian species, there are notable differences in the subunit repertoire. Most notably, two additional subunit polypeptides, termed  $\beta$ 4 and  $\gamma$ 4 (Lasham et al., 1991; Harvey et al., 1993). Coupled with the absence of  $\theta$  and  $\varepsilon$  subunits in these animals, it has been subsequently concluded that the  $\beta$ 4 and  $\gamma$ 4 subunits are the lower vertebrate orthologues of  $\theta$  and  $\varepsilon$  respectively, with the  $\varepsilon$  subunit sharing 50% sequence identity with the  $\gamma$ 4 polypeptide (Harvey *et al.*, 1998) and  $\beta$ 4 and  $\theta$  sharing 56% sequence similarity (Bonnert et al., 1999), suggesting evolution from a common ancestor. This degree of identity is slightly higher than between subunit classes but lower than between isoforms of the same class (Darlison *et al.*, 2005).  $\theta$ - and  $\varepsilon$ -subunit genes are located in cluster with  $\alpha 3$  on the human X chromosome (Simon *et al.*, 2004), occupying the same positions and transcriptional orientation as the  $\beta 4\alpha 3\gamma 4$  subunit gene cluster on chromosome 4A of the zebra finch genome (Fig. 3.8 Chapter 3). In silico analysis of the human genome has revealed no sequences encoding  $\beta 4$  and  $\gamma 4$  subunits (Simon *et al.*, 2004).  $\gamma 4$ -subunit encoding mRNA had been detected in chicken (Gallus gallus domesticus), zebra finch (Taenopygia guttata), canary (Serinus canaria) and red-eared slider turtle (Trachemys scripta elegans), and the sequences were highly similar (C. Thode and M. G. Darlison, unpublished results). High sequence similarity was also observed for the B4 subunit transcript, which was detected chicken, zebra finch, clawed frog (Xenopus borealis), lizard (*Podarcis muralis*) and fish (species unspecified). Such high sequence homology points to conservation of physiological function (I. Pahal, C. Thode, and M. G. Darlison, unpublished results). Data suggests that the mammalian  $\theta$  subunit is unable to replace the  $\beta$ -subunit polypeptide in functional receptors (Bonnert *et al.*, 1999); however the  $\varepsilon$  subunit may substitute for a  $\gamma$  subunit. This confers different electrophysiological and pharmacological properties for the receptor, most notably these receptors are insensitive to benzodiazepines and open spontaneously (Davies et al., 1997; Whiting et al., 1997a). Some more recent evidence suggests the  $\varepsilon$  subunit may be able to substitute a  $\beta$  subunit (particularly  $\beta$ 3). In this case, potentially two  $\varepsilon$ -subunits can assemble into a single receptor complex (Jones and Henderson, 2007) agreeing with previous data (Thompson et al., 2002), although this is contradicted by data suggesting an  $\varepsilon$  can replace an  $\alpha$  or  $\beta$  subunit but only one  $\varepsilon$  subunit exists per complex (Bollan *et al.*, 2008).

The heterogeneity of the GABA<sub>A</sub> receptor family is furthered by the existence of splice variants for many of the subunits ( $\alpha 2$ ,  $\alpha 4$ ,  $\alpha 5$ ,  $\alpha 6$ ,  $\beta 2$ ,  $\beta 3$ ,  $\beta 4$ ,  $\gamma 2$ ,  $\gamma 3$  and  $\epsilon$ ), usually resulting in a short and long form, further details and references in Chapter 3, section 3.1.3.

#### 1.2.5 GABA<sub>A</sub> receptor subtypes

Despite a multitude of potential GABA<sub>A</sub> receptor subtypes (some 360,000 in mammals, although a maximum of 800 subtypes was postulated by Barnard et al. (1998)), preferred assemblies of subunit polypeptides are observed, with the majority of receptors comprising  $2\alpha$ 's  $2\beta$ 's and a single  $\gamma$ ,  $\delta$  or  $\varepsilon$  subunit (Pirker *et al.*, 2000; Sieghart and Sperk, 2002; Whiting, 2003a), typically with the stoichiometry  $\alpha$ - $\beta$ - $\alpha$ - $\beta$ - $\gamma$  (Baumann *et al.*, 2002). Distinct GABA<sub>A</sub> receptor subtypes exist in different neuronal populations. Furthermore, individual neurons can express several different subtypes (for more details refer to Chapters 4 and 6). Techniques including: reverse-transcription polymerase-chain reaction (RT-PCR), real-time RT-PCR, in situ hybridisation, immunohistochemistry, immunocytochemistry, co-precipitation and recombinant expression systems assaying pharmacological and electrophysiological characteristics have all been applied to answering the elusive question of native GABA<sub>A</sub> receptor stoichiometry (for review see Sieghart and Sperk, 2002; Olsen and Sieghart, 2008). In fact, despite the vast volumes of experimental work completed on the GABA<sub>A</sub> receptors, due to the stringent criteria for confirming the stoichiometry of native GABAA receptors (described by Olsen and Sieghart, 2008), and the promiscuous nature of the subunits, to date only 11 receptor subtypes are formally classified as native GABA<sub>A</sub> receptor subtypes (Table 1).

Category A	Category B	Category C
Confirmed native GABA <sub>A</sub> receptor subtypes; <i>all</i> <i>criteria met</i>	High probability as native GABA <sub>A</sub> receptor subtypes; <i>most criteria met</i>	Tentative as GABA <sub>A</sub> receptor subtypes; <i>some</i> <i>criteria met</i>
α1β2γ2*	α1β3γ2	ρ1
$\alpha 2\beta \gamma 2$	α1 <mark>β</mark> δ	ρ2
α3βγ2	α5β3γ2	ρ3
$\alpha 4\beta \gamma 2$	αβ1γ	<mark>αβ</mark> γ1
$\alpha 4\beta 2\delta$	<mark>α</mark> β1δ	αβγ3
α4β3δ	αβ	αβθ
α5βγ2	α1α6 <mark>βγ</mark>	αβε
α6βγ2	α1α6 <mark>β</mark> δ	αβπ
α6β2δ		αχαγβγ2
α6β3δ		
ρ		

**Table 1**. Classification of native mammalian GABA<sub>A</sub> receptor subtypes adhering to stringent experimental criteria as described by Olsen and Sieghart (2008), adapted from Olsen and Sieghart (2009). There are a total of 28 GABA<sub>A</sub> receptor subtypes of which only 11 are confirmed (category A). Text highlighted in red designates an unspecified subunit. (\*) The most abundant GABA<sub>A</sub> receptor subtype in the mammalian CNS (Whiting, 2003a).

## 1.2.6 Clinical importance of GABA<sub>A</sub> receptors

GABA<sub>A</sub> receptors are of significant clinical importance due to their contribution to the etiology of a number of neurological and psychological diseases. These include anxiety disorders (Rudolph and Mohler, 2004), Epilepsy (Benarroch, 2007), Huntington's disease (Thompson-Vest *et al.*, 2003), Fragile X syndrome (D'Hulst and Kooy, 2007), Angelman syndrome (DeLorey and Olsen, 1999), Autism (Fatemi *et al.*, 2009), alcoholism (Krystal *et al.*, 2006), stroke (Costa *et al.*, 2004) and Alzheimer's disease (Marcade *et al.*, 2008). In addition they are the site of action for a number of clinically important exogenous and endogenous compounds which are able to interfere with the delicate physiological balance of the nervous system (Reynolds, 2008). Substances can mimic the natural ligand (GABA) and bind in its place (at the  $\alpha\beta$  interface), bind to the receptor at some other modulatory site where they can elicit inhibition or over-stimulation of the natural activity, or bind to the ion channel. To this end, the pharmaceutical industry puts much effort into the development of substances that can exert a precisely defined effect by specific binding. GABA<sub>A</sub> receptors

are of clinical interest due to their sensitivity to benzodiazepines (BZs), barbiturates, volatile anaesthetics, neurosteroids and ethanol (Reynolds, 2008). Potentiation of GABAA receptors produces inhibition of neuronal firing, dependent on the compound and the dose, the result of inhibition can range from anxiolysis to complete sedation of a patient. Attenuation of GABAergic transmission elicits anxiety, insomnia, arousal, restlessness, exaggerated reactivity and seizures (Sieghart, 1995). The stoichiometry of the receptor which determines the precise pharmacological profile. The primary problem is that many GABA<sub>A</sub> receptor-targeting compounds were developed before the stoichiometry and varying pharmacological properties of the receptors were elucidated, leading to many unwanted side effects (Nemeroff, 2003) such as dependence, cognitive impairments, unwanted sedation, tolerance and ataxia (loss of muscle coordination; Reynolds, 2008). A further caveat is the widespread distribution of GABA<sub>A</sub> receptors in the CNS, particularly the brain, which results in generalised effects of non-selective agonists and antagonists on CNS function. Complicated further by the high degree of similarity between subunits (Wingrove et al., 2002), and the fact that any subunit can be present in more than one subtype combination (refer to Table 1). However, with increasing understanding of native GABA<sub>A</sub> receptor stoichiometry and localisation of individual subtypes within the brain, the heterogeneity and wide-spread expression can become advantageous in subtype-specific drug design.

Ligands of the GABA site include the agonist muscimol and the antagonists bicuculline and gabazine; GABA<sub>A</sub> receptors are un-competitively blocked by picrotoxin. Benzodiazepines (BZs) e.g. diazepam (Vallium) and alprazolam (Xanax) are amongst the most commonly prescribed classes of drugs, producing anxiolytic, anticonvulsant, sedative, myorelaxant and antipsychotic effects by allosterically modulating GABA-activated currents by binding at an extracellular surface at the  $\alpha/\gamma$  subunit interface (Smith and Olsen, 1995). Therefore sensitivity of the receptor to BZs is governed by these subunits (Pritchett *et al.*, 1989; Günther *et al.*, 1995). Conventional BZs are primarily active at receptors comprising  $\alpha 1$ ,  $\alpha 2$ ,  $\alpha 3$  or  $\alpha 5$  subunits in partnership with a  $\gamma 2$  subunit (Pritchett *et al.*, 1989; Sieghart, 1995). They are unable to modulate receptors containing  $\alpha 4$  and  $\alpha 6$  subunits, and demonstrate low affinity for receptors with  $\gamma 1/\gamma 3$ , or  $\alpha/\beta$  pentamers (Sieghart, 1995). In

contrast, GABA<sub>A</sub> receptors comprising the  $\alpha$ 4 and  $\alpha$ 6 subunits are highly sensitive to modulation by imidazobenzodiazepines. For example Ro15-4513, flumazenil and bretazenil which posses an imidazo ring and lack the 5-phenyl substituent common to classic BZs (Korpi *et al.*, 2002), and thus bind slightly differently to BZs, but within the same binding pocket (Kucken *et al.*, 2003).

Different BZs also demonstrate variable affinities for GABA<sub>A</sub> receptors dependant upon the subtype (Atack, 2003). Subtype-specific receptor activity by BZs is further complicated by subtype-specific effects, for example, diazepam (Vallium), which is a common drug prescribed primarily for anxiety, insomnia, muscle spasm, seizures (although not long term) and symptoms of acute alcohol withdrawal. A single residue mutation (Arg  $\rightarrow$  His) within the  $\alpha$ -subunit sequence can render the receptor diazepam-insensitive (Wieland *et al.*, 1992; Kleingoor *et al.*, 1993; Benson *et al.*, 1998). Sedative, anticonvulsant and amnesic effects of diazepam are mediated by receptors containing the  $\alpha$ 1 subunit (Rudolph *et al.*, 1999), whereas anxiolytic effects are mediated by  $\alpha$ 2-subunit-containing receptors (Löw *et al.*, 2000) and myorelaxant effects are mediated by  $\alpha$ 2-,  $\alpha$ 3- and/or  $\alpha$ 5-subunit containing receptors (Rudolph *et al.*, 1999).

The BZ site also binds a large selection of non-BZ compounds, such as the hypnotic compound zolpidem (an imidazopyridine), which has a high affinity for  $\alpha$ 1-subunit-containing receptors, low affinity for  $\alpha$ 2- and  $\alpha$ 3-subunit-containing GABA<sub>A</sub> receptors and virtually no sensitivity at receptors assembled with the  $\alpha$ 5 subunit (Pritchett and Seeburg, 1990). As with classical BZs, zolpidem sensitivity is only exhibited when a  $\gamma$ 2 subunit is present, regardless of  $\alpha$  subunits (Sanna *et al*, 2002). Although they share the same binding pocket, different residues are involved in BZ and zolpidem binding within the  $\gamma$ 2 subunit (Sancar *et al.*, 2007), thus there is differentiation between two types of ligands.

It is important to elucidate the structural determinants of ligand binding in order to develop more pharmacologically (and thus behaviourally) sensitive ligands. Anaesthetics induce reversible unconsciousness at low concentrations. They are structurally diverse and include volatile agents such as isoflurane and high dose ethanol, and intravenous compounds such

as neuroactive steroids, etomidate, propofol and barbiturates, all of which allosterically bind and enhance GABA<sub>A</sub> receptor currents (there are different binding sites, not all of which have been resolved; for a review see Franks, 2006). Barbiturates such as pentobarbital, phenobarbital and thiopental are positive allosteric modulators of GABA<sub>A</sub> receptors. At high concentrations they can directly activate the receptor and at extremely high concentrations they can completely block Cl<sup>-</sup> conductance (Jackson et al., 1982). Such compounds bind at a site located ~50 Å below the GABA binding site (also a similar site to general anaesthetics; Mercado and Czajkowski, 2008). They potentiate GABA<sub>A</sub> receptormediated inhibition by inducing conformational changes in the GABA binding site, increasing the receptors affinity for GABA and stabilising an open state structure; subsequently increasing the duration of channel opening, eliciting sedative, hypnotic, anaesthetic and anticonvulsant effects (Olsen, 1982; Mercado and Czajkowski, 2008). However, the prolonged use of barbiturates results in physical dependency and so BZs are generally preferred (Reynolds, 2008). Ethanol has long been known to induce sedative effects via GABA<sub>A</sub> receptors (Wafford et al., 1991), but the precise mode of action and particular subtypes involved was less clear. More recently it has been resolved that extrasynaptic  $\delta$ -subunit-containing receptors (in combination with  $\beta$ 3 and  $\alpha$ 1,  $\alpha$ 4 or  $\alpha$ 6 subunits), display greatest sensitivity to ethanol at low concentrations (10 mM). Only at higher concentrations (100 mM), do synaptic  $\gamma$ 2-subunit-containing receptors exhibit sensitivity (reviewed by Olsen, 2007; Santhakumar et al., 2007).

## 1.2.7 Endogenous modulators of GABA<sub>A</sub> receptors

There are several endogenous modulators of GABA<sub>A</sub> receptors including protein kinases, Zn<sup>2+</sup> and neurosteroids. Zn<sup>2+</sup> ions serve to allosterically inhibit GABA<sub>A</sub> receptor function. Three individual binding sites have been identified; the first within the lining of the ion channel and the other two at the  $\alpha/\beta$  interfaces (Hosie *et al.*, 2003). Interestingly, GABA<sub>A</sub> receptors can also be modulated by protons and these appear to bind at the same position as Zn<sup>2+</sup> ions (H267 on the TM2 channel lining of  $\beta$  subunits). In contrast to Zn<sup>2+</sup>, H<sup>+</sup> potentiates GABA<sub>A</sub> receptor function. It is postulated that these paradoxical effects are attributed to the different way in which cations interact with imidazole side chains of the histidine residue (Wilkins *et al.*, 2002). Redox reagents such as glutathione and ascorbic

acid can modulate GABA<sub>A</sub> receptor function; reducing agents potentiate GABA-evoked currents and oxidising agents inhibit these currents (Amato et al., 1999; Pan et al., 2000). There is some speculation as to whether or not they act at the disulphide bridged region within the N-terminus, as homomeric p1 GABA<sub>C</sub> receptors (which have this Cys-loop) are not modulated by redox reagents, although there is new evidence to the contrary (Calero and Calvo, 2008). Instead, a cysteine residue within TM3 of  $\beta$  subunits was proposed as a potential binding site (Pan et al., 2000; Wilkins and Smart, 2002). In part, the potentiation of GABA-evoked currents by reducing agents at a low concentration is likely to be accredited to chelation of inhibitory Zn<sup>2+</sup> from H267 (Wilkins and Smart, 2002). The most important endogenous GABA<sub>A</sub> receptor modulators are neurosteroids (Belelli and Lambert, 2005). Until recently the position of a neurosteroid binding site remained unclear, with studies indicating a binding site located at TM domains with residues in TM1 (Q241) and TM4 (N407 and/or Y410) playing a role, although in which subunits remained unclear (Hosie et al., 2006, 2007). More recently further confirmation of a well conserved binding site within the TM1 domain of  $\alpha$ -subunits was reported (Q241; Hosie *et al.*, 2009), and the two previously reported TM4 residues were found not to be involved in the formation of hydrogen bonds between the steroid and the receptor but instead are involved in potentiation of the channel (Li et al., 2009). At nanomolar concentrations neurosteroids allosterically increase GABA-induced chloride flux by increasing the open state of the receptor. However, at micromolar concentrations they are able to directly activate the receptor in the absence of GABA, similar to barbiturates (Shu et al., 2004). Synthetic neurosteroid analogues (e.g. alphaxolone, hydroxydione) are generally used for sedation, however, currently under development is ganaxolone (a synthetic analogue of allopregnanolone) which is currently being tested in clinical trials for the treatment of epilepsy (D'Hulst et al., 2009).

Determining the native stoichiometry of GABA<sub>A</sub> receptors and their relative distribution within brain would allow the development of more specialised therapeutic strategies, which could be highly effective without the undesirable side effects.

#### 1.2.8 GABA<sub>A</sub> receptor trafficking

Regulation of neuronal excitability in the brain is partly dependent on synaptic inhibition (Hablitz et al., 2009). Strength of synaptic inhibition is attributable to the number of synaptic GABA<sub>A</sub> receptors. Surface GABA<sub>A</sub> receptor numbers are in turn determined by a variety of mechanisms such as rate of assembly, trafficking between the synapse and intracellular structures (either for degradation or recycling), and their stability at the membrane (Arancibia-Cárcamo and Kittler, 2009). GABAA receptor subunits are assembled into their pentameric structure in the endoplasmic reticulum (ER) via interactions of subunit N-terminal assembly domains (Connolly and Wafford, 2004). Assembly occurs within five minutes of translation, but only at 25% efficiency (Gorrie et al., 1997). Chaperone molecules including calnexin and immunoglobulin heavy chain binding protein (BiP), which ensure correct folding. Plic-1 also interacts with intracellular domains of all  $\alpha$ - and  $\beta$ -subunit polypeptides where it aids receptor stability in the ER, primarily by preventing proteosomal degradation (Bedford et al., 2001). Single subunits, homomers and  $\alpha\gamma$  or  $\beta\gamma$  heteromers are generally retained within the ER and subsequently degraded (Kittler et al., 2002), thus it is in the ER where subunits compete for their preferential assembly (Jacob et al., 2008). Within the golgi apparatus, GABA<sub>A</sub> receptor associated protein (GABARAP) and golgi-specific DHHC zinc finger domain protein (GODZ) interact with  $\gamma$  subunits and aid trafficking of assembled GABA<sub>A</sub> receptors to the plasma membrane (Kittler et al., 2001; Keller et al., 2004). In addition to their insertion and stability at the synapse, GABA<sub>A</sub> receptor number is dictated by internalisation processes. Generally these are clathrin- and dynamin-dependent (Kittler et al., 2000). Intracellular domains within  $\beta$ - and  $\gamma$ -subunit polypeptides interact with clathrin adapter protein (AP2) resulting in receptor internalisation, where they can be degraded or recycled back to the surface. Retention and clustering of GABA<sub>A</sub> receptors at the synapse is purported to be partly due to their association with gephyrin. Gephyrin is essential for clustering of glycine receptors, mediated by a direct interaction between the  $\beta$ -subunit loop and gephyrin which has been demonstrated in vivo (Meyer et al., 1995). Although data suggests that gephyrin accumulates at GABAergic synapses and there is some evidence of interaction with the  $\gamma 2$ subunit (Alldred et al., 2005) and  $\alpha 2$  subunit in vitro (Tretter et al., 2008), there is no concrete *in vivo* data of such an association.  $\alpha$ 5-subunit receptors, which are extrasynpatic,

are anchored by radixin to the actin cytoskeleton (Loebrich *et al.*, 2006). Receptors are able to laterally diffuse within the plasma membrane forming discrete populations which may be synaptic or extrasynaptic. For a full review concerning GABA<sub>A</sub> receptor trafficking consult Arancibia-Cárcamo and Kittler (2009).

To summarise, GABAergic neurotransmission is highly-tuned at many different levels i.e. two types of GAD, many different conformational states of GABA, Ca<sup>2+</sup>-dependant and independent release, re-uptake in neurons and glial cells, the presence numerous receptor subtypes which can be located pre-, post- and extra-synaptically, which are activated by a range of endogenous ligands. All of these factors contribute to a multifaceted system operating at a molecular, cellular and systemic level.

# **1.3 GABA**<sub>C</sub> Receptors

Like the GABA<sub>A</sub> receptors, GABA<sub>C</sub> receptors are a subfamily of the Cys-loop LGIC superfamily. Where GABA<sub>A</sub> receptors are the most complicated members, GABA<sub>C</sub> receptors are the simplest, with only three identified subunits in mammals,  $\rho 1$ ,  $\rho 2$  and  $\rho 3$ , and only p1 exhibiting alternative splicing (Martínez-Torres et al., 1998; Simon et al., 2004). There is considerable debate pertaining to the classification of these receptors as to whether they are a simply a sub-class of the GABA<sub>A</sub> receptors, or an individual sub-family of the LGIC superfamily (Olsen and Sieghart, 2008). The International Union of Pharmacology Committee on Receptor Nomenclature and Drug Classification (NC-IUPHAR) have recently recommended that the former GABA<sub>C</sub> receptor  $\rho$ 1- $\rho$ 3 subunits be designated as part of the GABA<sub>A</sub> receptor family but still retain the Greek subunit letters as designated by Barnard et al. (1998), and the term GABA<sub>C</sub> receptor should not be used (Collingridge *et al.*, 2009). The primary reasons for this controversy are manifold. Firstly, GABA<sub>C</sub> receptor subunits have an almost identical structure to GABA<sub>A</sub> subunits, sharing 30-38% sequence identity at an amino-acid level with GABA<sub>A</sub> receptor subunits (Bormann, 2000) although, in the TM2 region they share greatest homology with the glycine receptor  $\alpha$  subunits (Feigenspan *et al.*, 1993). They have an identical pentameric arrangement forming a chloride-specific channel of a similar diameter (~5.1 Å; Bormann and Feigenspan, 1995; Bormann, 2000). However, GABA<sub>C</sub> receptor subunits are thought to commonly exist in homomers (unlike the majority of GABA<sub>A</sub> receptor subunits; Enz and Cutting 1998, 1999), although there is some evidence demonstrating the assembly of  $\rho$ 1and  $\rho$ 2-subunit polypeptides into functional heteropentamers (Enz and Cutting, 1999), although this may not represent native receptor stoichiometry. This is in stark contrast to the apparent promiscuity of GABA<sub>A</sub> receptor subunits. Furthermore, there is no firm evidence for existence for native receptors comprising both GABA<sub>A</sub> and GABA<sub>C</sub> receptor subunits, despite reported assembly of  $\rho$  and  $\gamma$ 2 subunits in functional recombinant receptors (Ekema *et al.*, 2002; Milligan *et al.*, 2004; Pan and Qian, 2005).

The prevailing differences between  $GABA_A$  and  $GABA_C$  receptors are their pharmacological profiles and electrophysiological responses (Bormann, 2000). Unlike  $GABA_A$  receptors,  $GABA_C$  receptors are sensitive to the GABA analog CACA (*cis*-4aminocrotonic acid; Johnston, 1996) and (except for  $\delta$ -containing GABA<sub>A</sub> receptors; Brown *et al.*, 2002) have a significantly higher affinity (~10 x) for GABA. Upon binding, the response is slower and more sustained, with a weak desensitisation even at high GABA concentrations (Bormann and Feigenspan, 1995; Johnston, 1996; Qian & Pan, 2002).

Relative to GABA<sub>A</sub> receptors, GABA<sub>C</sub> receptors show much lower single-channel conductance only ~7 pS (picosiemens) and a pore opening time that is 6 x longer (Bormann and Feigenspan, 1995; Johnston, 1996). Further distinctions between receptor types include a comparable insensitivity to bicuculline and baclofen (4-amino-3-(4-chlorophenyl) butanoic acid, a GABA<sub>B</sub> receptor agonist; Curtis *et al.*, 1970; Johnston, 1996; Chebib and Johnston, 2000) and high sensitivity to TPMPA [(1, 2, 5, 6-tetrahydropyridine-4-yl) methylphosphinic acid] (Murata *et al.*, 1996). In addition, GABA<sub>C</sub> receptors are generally not modulated by benzodiazepines, barbiturates, neuroactive steroids and general anaesthetics at appropriate concentrations (Shimada *et al.*, 1992; Bormann and Feigenspan, 1995; Qian and Pan, 2002). Akin to GABA<sub>A</sub> receptors, the subunit stoichiometry of GABA<sub>C</sub> subtypes dictates the pharmacological (Chebib *et al.*, 1998; Enz and Cutting, 1999; Bormann, 2000; Chebib *et al.*, 2007; Xie *et al.*, 2008) and biophysical properties (Bormann, 2000; Zhu *et al.*, 2007). For example, GABA<sub>A</sub> receptors are sensitive to cyclothiazide (as are GABA<sub>C</sub> receptor  $\rho$ 2 homomers), yet  $\rho$ 1 homomers display no such sensitivity. This

sensitivity is attributed to a single residue at the 2' position of TM2 (Xie *et al.*, 2008), the same residue which regulates channel conductance (Zhu *et al.*, 2007). Picrotoxin exerts a similar effect by potently antagonising GABA<sub>A</sub> and  $\rho$ 1 homomeric GABA<sub>C</sub> receptors but  $\rho$ 2 homomers and  $\rho$ 1 $\rho$ 2 heteromeric receptors are insensitive (Borman and Feigenspan, 1995; Enz and Cutting, 1999).

Moreover, (and maybe most significantly), GABA<sub>C</sub> receptor subunits are predominately expressed in the bipolar and horizontal cells of the retina, where they mediate synaptic inhibition (Albrecht and Darlison, 1995; Chebib and Johnston, 1999; Rozzo *et al.*, 2002) although expression has been documented in some regions of mammalian brain (Boue-Grabot *et al.*, 1998; Ogurusu *et al.*, 1999; Rozzo *et al.*, 2002; Milligan *et al.*, 2004; Alakuijala *et al.*, 2006), chicken brain (Albrecht *et al.*, 1997) mammalian ovary and testes (Rozzo *et al.*, 2002) and gut (Jansen *et al.*, 2000).

Another disparity concerns the chromosomal location of the subunit genes. Within the human genome, GABA<sub>C</sub> receptor subunit genes are not closely associated with GABA<sub>A</sub> receptor genes.  $\rho$ 1- and  $\rho$ 2-subunit genes are clustered on chromosome 6, at position 6q13– q16.3 (Bailey *et al.*, 1999; Simon *et al.*, 2004), perhaps positioned for their possible co-assembly in receptors (Enz and Cutting, 1999); and the  $\rho$ 3-subunit gene is localised to chromosome 3 in the human genome at position 3q11.2 (Bailey *et al.*, 1999; Simon *et al.*, 2004), along with no other GABA<sub>A</sub> or GABA<sub>C</sub> receptor genes. Previously, it has been suggested that native GABA<sub>A</sub> receptors may be composed of their chromosomal partners (Barnard *et al.*, 1998), thus the isolated genomic positioning of the  $\rho$ -subunit genes would therefore point to homomeric assemblies. However, there are numerous exceptions to this rule (especially in GABA<sub>A</sub> receptors), suggesting chromosomal partnering is not an important determinant in receptor assembly (Olsen and Sieghart, 2008).

# **1.4 The avian song system**

It may be assumed that studies on mammalian brain would be most appropriate for understanding the mechanisms of human cognitive functions, from an evolutionary perspective. However, humans and songbirds are members of a small group of animals which have to learn from a conspecific tutor to produce their complex vocal signals (Thorpe, 1958; Konishi, 1965; Marler, 1970). Other members include cetaceans, such as whales and dolphins, and some species of bat. Unlike primates, there are many species of vocal-learning songbirds (~4000). Birdsong is a learned vocal behaviour ranging in complexity and repertoire between different species (DeVoogd, 2004; DeVoogd *et al.*, 1993; Brenowitz *et al.*, 1997). Songbirds, as a term encompasses primarily passerine (perching) birds of which the sub-order oscines are able to produce learned vocalisations. There are also some non-songbirds which are able to mimic vocalisations but are not classified as songbirds, such as parrots and hummingbirds (Marler, 1997).

#### 1.4.1 History of songbird research

Observations of songbird behaviour, such as their requirement for a conspecific (same species) tutor during early development, were noted as early as Charles Darwin (1871). However, experimental research on birdsong commenced with the pioneering work of British ethologist William Thorpe some 80 years later (1958, 1961). He observed that juvenile chaffinches (Fringilla coelebs) reared in isolation produced abnormal songs, but if they were exposed to song of wild chaffinches they eventually produced songs reminiscent of these, confirming the need for a tutor. However, adult chaffinches exposed to the same songs did not imitate these songs, which was indicative of a critical period of learning. Thorpe's work was continued by Marler, who proposed the notion that songbirds possessed an innate auditory template, that was speculated to be genetically encoded. This was attributed to the observation that two different species of sparrows, when isolated from tutors, produced abnormal and completely different songs, but the individual songs still retained some characteristic species-specific elements (Marler, 1970, 1976). Work was continued by students of Marler; Masakazu Konishi who demonstrated the need for auditory feedback during the sensorimotor phase for normal song learning (Konishi, 1965) and Fernando Nottebohm who identified the neuronal circuits controlling the production of song (i.e. the song system; Nottebohm et al., 1976). Fifty years after Thorpe's original research, work continues in the field with earnest, providing us with numerous, well characterised avian models to understand motor function and learning and memory processes. Many of these early observations are still upheld today.

#### 1. Birds require a tutor during sensory acquisition

Akin to speech acquisition in humans, the learning of song is a process of imitation whereby a period of perceptual learning is followed (or in some cases, overlapped) by a rehearsal period requiring auditory feedback (Konishi 1965, 2004; Marler, 1997; Mooney, 1999). Birds reared in isolation from tutors produce highly abnormal songs (Thorpe, 1958; Immelmann, 1969; Marler and Sherman, 1983; Eales, 1987; Searcy and Marler, 1987; Tchernichovski and Nottebohm, 1998; Kojima and Doupe, 2007; Fehér *et al.*, 2009) as do humans (Fromkin *et al.*, 1974; Lane, 1976). If juveniles are reared in the presence of heterospecific tutors (different species), they will learn/attempt to learn the song of the foster tutor (Immelmann, 1969). However, if presented with a choice of hetero- or conspecific (same species) tutors, juveniles will always preferentially learn the song of the conspecific tutor, suggesting an innate pre-disposition for conspecific song (Immelmann, 1969; Marler and Peters, 1988; Riebel *et al.*, 2002). Furthermore, isolate songs still retain some species-specific characteristic features (Marler and Sherman, 1985; Zann, 1996; Konishi, 2004; Kojima and Doupe, 2007), which has been attributed (at least in part) to an innate auditory template (Marler, 1970, 1976).

#### 2. Birds require auditory feedback during song learning

Birds must be able to hear themselves sing in order to produce normal song (Konishi 1965; 2004). Birds deafened as juveniles who cannot hear their own vocalisations produce highly unstructured songs (Konishi, 1965, 1985, 2004; Price, 1979). Despite harboring a species-specific innate auditory template, they cannot match their vocal output to it, thus producing random noises (Konishi, 2004). This is also apparent in the human population, whereby deaf children cannot vocalise normally (Kuhl, 2004). If birds are deafened after sensory acquisition (listening to tutor sing), but before sensorimotor learning (the rehearsal stage), they also develop highly abnormal songs, even though the tutor song no longer needs to be heard at this stage (Konishi, 1965). Birds raised in isolation and then deafened before the sensorimotor phase, produce songs that are the same as birds which have been deafened from the start with no identifiable features at all (Konishi, 1965, 1985).

### 3. Innate template

One of the main features of birdsong is that it is characteristic to each species. Both humans and birds posses versatile vocal apparatus, so motor constraints are not considered a limiting factor of vocal learning. Marler was the first to propose the notion of an innate auditory template (which is present from hatching), pre-disposing birds to sing a song characteristic of their own species (1970). Evidence supporting this arose from several behavioural studies. Firstly, isolated birds produced songs with characteristic elements despite no auditory input from a tutor (Marler and Sherman, 1985; Searcy and Marler, 1987; Zann, 1996; Konishi, 2004; Kojima and Doupe, 2007). Physiological responses (such as begging calls and increased heart rate) of young birds are much greater in response to conspecific song (Dooling and Searcy, 1980; Nelson and Marler, 1993) and when presented with a choice of tutor, juveniles preferentially learn the song of a conspecific (Immelmann, 1969; Marler and Peters, 1988; Riebel et al., 2002). Thus songbirds posses an innate (possibly genetically encoded) auditory template coupled with a pre-disposition for perceptual learning which precludes song production. However, in addition to these, normal song production also relies on both sensory input (tutor song) and auditory feedback (birds own song).

#### 4. Neural substrate for song learning

Perception of song, acquisition of an auditory template, motor production of song and modification of vocal output via auditory feedback are processes that all require specialised neuronal circuits. The neural substrates for vocal learning are referred to as the song system (Nottebohm *et al.*, 1976), further details given in section 1.4.2. Birds which do not sing (learned song), are able to discriminate between different sounds but are unable to mimic what they hear and comparative anatomical analysis has shown the nuclei responsible for motor control to be absent in non-learners (Gahr, 2000). However, in avian species where only the males sing (e.g. zebra finch, *Taenopygia guttata*) females posses some of the song-system nuclei but they are significantly smaller than their male counterparts (Nottebohm and Arnold, 1976; MacDougall-Shackleton and Ball, 1999).

One of the most frequently studied songbirds is the zebra finch. These are small birds native to the semi-arid regions of Australia and Timor (Zann, 1996). They are social animals nesting in groups of 20-1000 birds, living up to 3 years in their natural environment and 5-7 years in captivity. Zebra finches are close-ended learners (they can only learn as juveniles and after sexual maturity cannot change their song, discussed in section 1.4.4) and their development process and songs have been extensively characterised. Thus they provide an excellent model for studying the underlying molecular mechanisms involved in the acquisition and production of song. This study exclusively utilises male zebra finches, as female zebra finches do not sing (Nottebohm and Arnold, 1976).

## 1.4.2 Structure of the song system

A discrete group of interconnected nuclei within in the telencephalon (in both hemispheres) are responsible for memorisation and production of learned song; referred to as the song system (Nottebohm *et al.*, 1976). The avian song system is broadly divided into two functionally and anatomically distinct pathways, the anterior forebrain pathway (AFP; primarily involved during sensory acquisition) and vocal motor pathway (VMP; control of the motor production of song) although more recent evidence indicates the two circuits are far more interconnected than originally postulated (Fig. 1.2).

HVC (formal name, formally higher vocal centre) is the site of auditory input into the song system. Auditory information originates from Uva (nucleus uvaeformis, a thalamic nucleus) via NIf (interfacial nucleus of the nidopallium) and Field L (auditory area in nidopallium). In turn the HVC projects to nuclei located in both pathways of the song system (RA and Area X). HVC is involved in the timing of song syllables (both their duration and the length of the interval between them; Long and Fee, 2008). HVC composes three different types of neurons; projection neurons which innervate the pre-motor area (HVC<sub>RA</sub>), projection neurons which innervate the basal ganglia circuit (HVC<sub>X</sub>) and interneurons (Wild *et al.*, 2005; Mooney and Prather, 2008), of which there are two classes (Scott and Lois, 2007). Area X is the first nuclei of the AFP. Located within the striatum, it is composed of a mixture of striatal and pallidal cells, with large GABAergic interneurons sparsely

distributed throughout (Luo and Perkel, 1999; Farries and Perkel, 2002). Area X demonstrates the greatest levels of neurogenesis in post-natal songbird brains in the form of medium spiny neurons (Scott and Lois, 2007). Medium spiny neurons contribute a large portion of the mammalian basal ganglia, where they are under much scrutiny due to their loss in Huntington's disease (Mitchell et al., 1999). Area X is the equivalent of the mammalian basal ganglia whereby inputs (from HVC and LMAN) are processed in a similar way and it comprises many equivalent cell types (Luo and Perkel, 1999; Farries and Perkel, 2002). From Area X, the AFP pathway descends to the medial nucleus of the dorsolateral thalamus (DLM) and then to the lateral portion of the magnocellular nucleus of the anterior nidopallium (LMAN), which is the output of the AFP pathway (Nottebohm et al., 1976). LMAN is the one song system nuclei which is sexually monomorphic in zebra finch brains (Nixdorf-Bergweiler, 2001), although mRNA expression is generally lower in females than that of males (Thode et al., 2008; Tomaszycki et al., 2009). Projections from the LMAN connect to Area X and RA, thus the LMAN in the final stage where auditory information can be processed before being transmitted to the VMP or back into the AFP (potential feedback mechanisms; Fig. 1.2b), these projection neurons are innervated by GABAergic interneurons (Rosen and Mooney, 2000).



**Figure 1.2**. (a) Schematic representation of the song system nuclei within zebra finch brain. (b) Flow diagram illustrating the flow of information through the song system. HVC (formal name) projects to both the AFP
and VMP, but its connections within the VMP are far more robust; projecting neurons from HVC to RA are more prevalent over those projecting to Area X (4:1; Wild *et al.*, 2005). RA is the output of the song system and there are considerably more HVC synapses at RA in comparison to the number of LMAN synapses (20:1; Canady *et al.*, 1988). Despite being classified as separate pathways there is a link between LMAN and RA and circuits within the AFP enabling performance-based feedback (Brainard and Doupe, 2002). Abbreviations: Am, ambigualis; DLM, medial nucleus of the dorsolateral thalamus; DM, dorsal medial nucleus; HVC, formal name; L, field L; LMAN, lateral magnocellular nucleus of the nidopallium; NIf, interfacial nucleus of the nidopallium: nXIIts, tracheosyringeal part of the hypoglossal nucleus; PN, projection neuron; RA, robust nucleus of the arcopallium; Ram, retroambigualis; rVRG, rostro–ventral respiratory group; X, area X

The VMP is more direct (Fig. 1.2). Projection neurons originating exclusively in the ventricular region, dorsal to the HVC, descend to the RA during song learning in juvenile animals (Scott and Lois, 2007). The sexually dimorphic RA represents the output of the song system, projecting to hyperglossal motor neurons which innervate the syrinx (nXIIts) and Ram (nucleus retroambigualis), which is the site of vocal and respiratory muscle coordination (Nottebohm *et al.*, 1976). Ram is comprised of five types of projection neurons and is the functional equivalent of the mammalian NRA (nucleus retroambigualis; Wild *et al.*, 2009). The VMP is important for song production. Ablation of RA or HVC effectively halts normal song production (Nottebohm *et al.*, 1976; Simpson and Vicario, 1990), resulting in highly abnormal song or muteness. Although, it is now recognised that the AFP also plays a role in adult song production, as both Area X and LMAN are active during singing (Hessler and Doupe, 1999) and microlesions in HVC can be recovered by ablation of LMAN (Scott *et al.*, 2007; Thompson *et al.*, 2007b). This may be because inactivation of LMAN disrupts the feedback mechanisms involved in the control of song (Fig. 1.2b; Brainard and Doupe, 2000)

Disruption of the AFP pathway does not completely impair the bird's ability to produce learned vocalisations, however, it stops further advancement of vocal learning and new sounds can no longer be imitated. Lesions of LMAN in juvenile birds impact on song development, producing over-simplified, highly repetitive song with premature crystallisation, as soon as one day post-surgery (Bottjer *et al.*, 1984; Scharff and Nottebohm, 1991). Conversely, in adults, LMAN lesions have little effect on crystallised

song (Bottjer et al., 1984). Lesions of Area X (rendering it non-functional) in juveniles produce plastic (changeable) songs which appear not to crystallise (become stable), a sharp contrast to the premature crystallisation induced by LMAN lesions (Scharff and Nottebohm, 1991). Accordingly, it was deduced that the AFP was not necessary for song production, but was important in song acquisition. However, further work demonstrated that previously crystallised song of adult birds is subject to degradation if birds are deafened or auditory feedback is perturbed (Nordeen and Nordeen, 1992; Woolley et al., 1997; Leonardo and Konishi, 1999; Funabiki and Konishi, 2003; Zevin et al., 2004), indicating the potential for plasticity even after crystallisation. Degradation of stable adult zebra finch song is proportional to the age of deafening in that the song of younger adults deteriorates faster than that of older animals following deafening (Lombardino and Nottebohm, 2000). Furthermore, if perturbations are discontinued, some recovery of normal song is observed (Leonardo and Konishi, 1999; Woolley and Rubel, 2002; Funabiki and Konishi, 2003) although, to varying degrees (Zevin et al., 2004). It is therefore proposed that there is ongoing comparison of vocal output with the sensory template (whether this template be the tutor song or a memory of their own crystallised vocalisations remains unknown), beyond the critical period of learning (Woolley, 2004). However, these experiments do not allow for modification of song by the bird in response to perturbation of auditory feedback. Recent work by Sober and Brainard (2009) has revealed that by mechanically changing the pitch of song during auditory feedback (by use of headphones) resulted in adult Bengalese finches readjusting the pitch of their song in attempts to bring it in line with their original sensory template. In addition, after temporary deafening, when hearing was restored, birds were able to learn the songs of cage-mates (Woolley and Rubel, 2002). It can therefore be postulated that song remains plastic into adulthood, perhaps by similar mechanisms utilised by juvenile songbirds prior to crystallisation, indicating a potential role for the AFP after sensory acquisition, but the exact mechanism remains unknown (Thompson et al., 2007b). It is only the integration of both pathways (AFP and VMP) which can result in the normal production of learned song.

Chapter One

#### 1.4.3 Sexual dimorphism

Zebra finch behaviour is highly dimorphic, with only male birds learning to sing. This is reflected in the song system where females exhibit significantly smaller nuclei and less interconnected circuitry (Nottebohm and Arnold, 1976; Konishi and Akutagawa, 1985; Fig. 1.3). Female zebra finches treated with estradiol as juveniles, then with testosterone as adults can produce basic song (Arnold, 1997); although the growth of the HVC appears to be independent of steroid hormones (Gahr and Metzdorf, 1999), perhaps an attributing reason as to why hormonally-treated female zebra finches can never produce male-quality songs. It has been speculated by Bolhius and Macphail (2001) that the sex differences (i.e. different nuclei volumes) are actually related to song production and not song acquisition. This is of interest as, female zebra finches, which do not sing, appear to undergo a critical learning period at a similar developmental stage to male birds (Bailey and Wade, 2005), although the data is controversial. It is agreed that females show preference for conspecific song; this is also true of female zebra finches isolated from adult males (Clayton, 1988; Laulay et al., 2004). However it was originally postulated that this preference is not apparent in females isolated before PHD 25 (Clayton, 1988), but isolation at a later stage (PHD 35) resulted in females favouring their fathers song (Miller, 1979). In contrast, recent data shows auditory isolation as early as PHD 18 still resulted in females which preferred conspecific song (Laulay et al., 2004). However, females isolated at PHD 18 showed no preference for tutored (normal quality) or untutored (poor quality) song (Laulay et al., 2004). Furthermore, during the period of sensory acquisition, there are significant differences in gene expression between male and female zebra finches with many sexlinked, differentially expressed genes within song system nuclei (Tomazycki et al., 2009).





**Figure 1.3**. Sagittal schematics of male zebra finch (a), non-song bird, e.g. pigeon (b) and female zebra finch (c) brains. The zebra finch song system is sexually dimorphic; females typically have smaller nuclei (except for LMAN) and Area X is completely absent. Furthermore, connections between HVC $\rightarrow$ RA cannot be detected. All the major nuclei of the song system are absent in non-songbirds, but they are able to produce non-learned vocalisations, this is also the case with female zebra finches (Zann, 1996; Price, 1979; Bolhius and Gahr, 2006). Both females and non-songbirds have field L which is the primary auditory area which in males projects to the song system. Data obtained from Bolhius and Gahr, 2006 and Tomazycki *et al.*, 2009. Abbreviations: Am, ambigualis; DLM, medial nucleus of the dorsolateral thalamus; DM, dorsal medial nucleus; HVC, formal name; L, field L; LMAN, lateral magnocellular nucleus of the nidopallium; NIf, interfacial nucleus of the nidopallium: nXIIts, tracheosyringeal part of the hypoglossal nucleus; RA, robust nucleus of the arcopallium; Ram, retroambigualis; rVRG, rostro-ventral respiratory group; X, area X

#### 1.4.4 Stages of song development

Song learning is a process of imitation where a tutor model is required (Thorpe, 1958, Marler, 1970). The song learning process has been compared to imprinting, in that information is learned very quickly and then retained for long periods of time. Zebra finches can learn songs as juveniles with as little as 1 min tutor exposure per day (Tchnernichovski *et al.*, 1999), the tutor template is stored as a long-term memory (Funabiki and Konishi, 2003). In the production of song there are two types of learners, the age-limited (or close-ended) learners (e.g. the zebra finch (*Taeniopygia guttata*) and white-crowned sparrow (*Zonotrichia leucophrys*) and the open-ended learners (e.g. the canary (*Serinus canaria*) and the European starling (*Sturnus vulgaris*); Brainard and Doupe, 2002). With birds in the former category, song memorisation is restricted to the early stages of life, with no new songs being learned in adulthood (Brenowitz, *et al.*, 1997). In contrast, open ended learners are able to develop new songs into adulthood. Often this occurs on a seasonal basis, usually during the spring (Mooney, 1999). In both open- and close-ended learners there are 'critical' periods for song development.

Despite the presence of an innate auditory template, to produce normal song, juvenile birds must be exposed to conspecific tutor song during the critical phase of song learning (~PHD 20-40). This is essential to retain a more precise representation of the tutor song (Konishi, 1965, 2004; Immelmann, 1969; Marler, 1970). Similar to humans, long term memories are formed of the tutor song (Funabiki and Konishi, 2003), which are thought to be stored within the auditory system (Phan *et al.*, 2006). Once equipped with an auditory template, the bird enters the sensorimotor phase where it begins to sing (Fig. 1.4; Marler, 1997). The first utterances of song are relatively unstructured and do not really resemble the tutor song (termed subsong), much like the ramblings of a human infant (Thorpe, 1958). As the bird continues matching its own vocalisations with the sensory template, the song output becomes more structured and refined and is known as plastic song (Catchpole and Slater, 1995). With further practice the song becomes crystallized, and is a close replica to the song they were exposed to during sensory acquisition (Fig. 1.4). Song crystallisation coincides with sexual maturity at ~PHD 90 (Immelman, 1969; Mooney, 1999; Harding, 2004), where there is a large surge of circulating testosterone (Pröve, 1983). Close-ended

learners such as zebra finches generally do not modify their songs, which remain stable and stereotyped following crystallisation (Marler, 1997). However, there are circumstances when close-ended learners display adult-song plasticity. These include deafening (Nordeen and Nordeen, 1992), syrngeal denervation (Williams and McKibben), muting (Pytte and Suthers, 2000), delayed auditory feedback (Leonardo and Konishi, 1999) and perturbation of auditory feedback (Funabiki and Konishi, 2003; Sober and Brainard, 2009). In cases such as these, adult zebra finches may modify their crystallised song, although the effects are not immediate (Williams *et al.*, 2003). These changes are mediated by the AFP, also known as the basal ganglia circuit, which is primarily involved in the learning of song. Experimentally induced song plasticity does not occur in birds with LMAN lesions (Brainard and Doupe, 2000). Thus these 'critical' periods may be more flexible than originally postulated (Funabiki and Konishi, 2003).

The song system is not fully developed in males immediately after hatching (Kirn, 2009). Many of the nuclei involved in the acquisition/production of song are small in juvenile birds and increase in size in correlation with the song learning process (Fig. 1.4). HVC is the entry point of auditory information into the song system and is involved in the motor production of song. HVC projections to Area X develop at ~PHD 20 (onset of sensory acquisition) and HVC projections to RA develop at ~PHD 35 (onset of sensorimotor phase; Fig. 1.4); these projections are not detected in female zebra finches (Konishi and Akutagawa, 1985; Mooney and Rao, 1994; Scott and Lois, 2007). HVC reaches its full size in males at ~PHD 60 (Ward et al., 2001), but in females at ~PHD 30 neurons begin to die and the HVC reduces in size (Nixdorf-Bergweiler, 2001). The RA is also involved in the production of song by projecting to the respiratory tract and syrinx (Wild, 2004). In male zebra finches, RA is the first song system nuclei to begin increasing in size (~PHD 6), and decreasing in females at ~PHD 30 due to cell death (Tomaszycki et al., 2009). Area X is not visible in female zebra finches (Nottebohm and Arnold, 1976), but interestingly LMAN is sexually monomorphic, being of a similar size in males and females (Nixdorf-Bergweiler, 2001). Gene expression is also sexually dimorphic in zebra finches, with enhanced expression in nuclei of the song system in male birds relative to females (Thode et al., 2008; Tomaszycki et al., 2009).



**Figure 1.4.** Timeline detailing stages of song learning in the zebra finch, the entire process of song development takes ~90-120 days. There are distinct stages to song development (Marler, 1970), but in the zebra finch these stages overlap (Brainard and Doupe, 2002). The juvenile chick first listens to and memorises the song of a tutor (sensory acquisition) and then begins vigorous singing (sensorimotor phase) whereby the song output is gradually matched to the template acquired during sensory acquisition, at this stage the tutor no longer needs to be heard (Konishi 1965, 1985). As song learning progresses the nuclei increase in size and connections are formed between them (green boxes). A large increase in circulating testosterone at ~PHD (post-hatch day) 90 (Pröve, 1983) is thought to increase song stereotypy (Williams *et al.*, 2003). Data modified from Tomaszycki *et al.* 2009. Abbreviations: DLM, dorsolateral thalamus; HVC (formal name); LMAN, Lateral magnocellular nucleus of the anterior nidopallium; nXIIts, tracheosyringeal part of the hypoglossal nucleus; RA, Robust nucleus of the arcopallium; X, Area X; Uva, nucleus uvaeformis.

#### 1.4.5 Song structure

Song is easily quantified using an audiospectrogram (which measures vibrations per sec at individual frequencies (Hz)) and as aforementioned, is characteristic to each species, some

having a repertoire of thousands of songs (e.g. mocking bird, *Mimus polyglottis*) and some having only one (e.g. the zebra finch, *Taenopygia guttata*; DeVoogd *et al.*, 1993; Brenowitz *et al.*, 1997; Brainard and Doupe, 2002). Song has a well defined structure which can be broken down into clear structural components. The most basic sound produced by a bird is termed an element, several elements together are a syllable and multiple repeated syllables become a phrase or motif. If the phrases are broken up by variable length silent intervals, the sequence becomes a song 'bout' (Brenowitz *et al.*, 1997). Each species of song bird has its own characteristic core elements which are relatively well conserved, with the exception of species such as the Mockingbird which is able to mimic the songs and calls of other species (Marler, 1997). Zebra finch song is acoustically complex with syllables produced during inspiration and expiration (Goller and Daley, 2001; Leadbeater *et al.*, 2005), but shows remarkable stereotypy across the species.

#### 1.4.6 Functions of song

Song is not exactly like spoken language, which conveys complex meanings (Fitch, 2009); it is simple and is used to communicate defensive (although not zebra finches; Zann, 1996), attractive and warning signals. Zebra finches which live in large groups use song to identify individual members (Zann, 1996). Faithful reproduction of tutor song is therefore necessary to ensure a successful life for the bird. Strength of song can be an indication of fitness, whereby birds with a large repertoire or more complex songs attract females more easily (Catchpole and Slater, 1995) and juveniles will preferentially learn from the tutor with the largest repertoire, if given a choice, even if this is not the father (Soma et al., 2009). Male zebra finches have the smallest repertoire, with just one song (although there are directed and non-directed versions), so more emphasis is placed upon the singing rate (Houtman, 1992) and complexity of the song (Clayton and Pröve, 1989). The song itself is a repeated motif lasting 0.5-1.5 sec with a consistent number of syllables; of which the average is 6.7 (range 3-10 syllables; Spencer et al., 2003), preceded by a few introductory notes. Factors such as nutritional stress during development can significantly reduce the number of syllables in a song (Spencer et al., 2003) and the accuracy of learning, which is representative of a lower level of fitness during development (Brumm et al., 2009). Furthermore, increased levels of corticosterone (released under stressful conditions) can

facilitate or detrimentally affect learning and memory (Coburn-Litvak *et al.*, 2003; Martin *et al.*, 2009; Quirarte *et al.*, 2009). Corticosterone administration during development results in adult zebra finches producing shorter and less complex songs than their untreated counterparts (Spencer *et al*, 2003).



Figure 1.5. Audiospectrograms showing zebra finch song at different developmental stages, displayed as frequency vs. time. When zebra finches enter the sensorimotor phase their first vocalisations are rambling and soft, akin to the babbling of an infant (a). As the bird continues to practice and match its vocalisations to the sensory template, the song becomes more structured and syllables emerge (plastic song), although some sequencing errors still occur (b). Following sexual maturity, song has clear structure and is very similar to that memorised during sensory acquisition, once the song is crystallised it remains relatively unchanged throughout adulthood (c); shown here are a few introductory syllables followed by three renditions of the Audiospectrograms laboratory Florida song. adapted from Johnson at State University (www.psy.fsu.edu/~johnson/johnsonlab/johnson.htm).

The structure of song is unique to each species, although there is slight variation amongst different members (akin to regional dialects observed in the human population; Kuhl, 2004; Fehér *et al.*, 2009), during cultural transmission (Marler and Tamura, 1964; Marler and Slabbekoorn, 2004; Fehér *et al.*, 2009). The characteristic elements are retained throughout,

with females preferring the song of their local dialect (Clayton, 1989). These small diversities are thought to be advantageous in discriminating several individuals from the same clutch which have had an identical tutor (Böhner, 1983). The younger siblings tended to produce a crystallised song most reminiscent to that of their fathers (Tchernichovski and Nottebohm, 1998). An interesting finding was that if non-tutored zebra finches (i.e. produce abnormal song) are allowed to breed, the subsequent generations faithfully produce the isolate song, but with each generation, songs converged closer to that of a normal zebra finch, until the fifth generation when the song could not be easily distinguished from a normal zebra finch song (Féher *et al.*, 2009). These examples further support the idea of an innate auditory template, which may be in part genetically encoded (inducing species-typical constraints), but is subject to environmental influences (Marler, 1970).

#### 1.4.7 Seasonal plasticity

The song(s) of some birds exhibits seasonal plasticity which is accompanied by volumetric changes in the nuclei, most notably in the HVC, which increases during the spring (breeding season) and improved connections between the nuclei (DeGroof *et al.*, 2008). Changes in nuclei size may be attributable to an increase/decrease in neuronal number, or alterations in their relative size or spacing, these changes are not uniform across the nuclei and may be triggered by different environmental cues such as photoperiod, which causes an increase in circulating testosterone (Smith *et al.*, 1997), or social cues such as the presence of females (Tramontin *et al.*, 1999). Zebra finches are opportunistic breeders and are generally not considered to be photoperiodic (Zann, 1996), although there is some evidence to the contrary (Bently *et al.*, 2000). Age-limited learners such as zebra finches, which do not modify their song on a seasonal basis, still undergo seasonal changes in circulating hormones and nuclei size but their songs remain unaffected (Brenowitz *et al.*, 1991). Conversely, in the white crowned sparrow (close-ended learner), no seasonal fluctuations in nuclei volumes are observed (Baker *et al.*, 1984)

Song-stereotypy of both open- and close-ended learners is purported to be linked to circulating testosterone levels (Pröve, 1983). In canaries, testosterone levels are at their highest during the breeding season in the spring, and this is when the songs are most

stereotyped (Nottebohm, 1981). In contrast, in zebra finches (which are close-ended learners), there is a surge of testosterone towards the end of the sensorimotor phase which coincides with song crystallisation (Pröve 1983). Testosterone is detected in juvenile zebra finches during development (Nottebohm, 1989), but if levels are artificially raised, it induces premature song crystallisation (Korsia and Bottjer, 1991). Correspondingly, if testosterone is blocked, song development is also disrupted (Bottjer and Hewer, 1992), indicating that testosterone plays a role in song stereotypy. Furthermore, female zebra finches treated with estradiol during development, followed with testosterone as adults can produce basic song (Arnold, 1997). This is also seen in testosterone-treated female canaries (Nottebohm, 1980). In adult male zebra finches, testosterone administration results in a change in the frequency of song (Cynx et al., 2005). All this evidence points to an important role for testosterone in crystallization. However, a rise in circulating testosterone may not be the only trigger for crystallisation. Testosterone (androgen) induces singing, but it may be that the motor act of singing itself which serves to close the critical period (Eales, 1987; Brainard and Doupe, 2002). Evidence supporting this theory arises from the fact that adult zebra finches which had been raised isolation or with females are able to incorporate some new zebra finch syllables into already crystallised isolate song (Slater *et al.*, 1988; Livingston et al., 2000); suggesting a plasticity of adult song and inferring that crystallisation may be guided by experience and not only by hormones (Eales, 1987; Morrison and Nottebohm, 1993; Jones et al., 1996). Furthermore, zebra finches raised by Bengalese finches and then re-exposed to conspecifics as adults were able to add characteristic zebra finch song syllables into their songs after the end of the critical period. However, zebra finches raised normally with conspecifics were not able to modify their songs once they had crystallised (Immelmann, 1969; Slater et al., 1988). It can therefore be speculated that during the process of learning, singing, in addition to testosterone, may play a role in the closure of the critical period (reviewed by Doupe and Kuhl, 1999).

#### 1.4.8 Song production and human speech

Phylogenetically, mammals and birds separated some 300 million years ago (Bolhius and Macphail, 2001) and although not identical, there are some striking parallels between birdsong and speech production, consequently the underlying neuronal mechanisms are

similar (Doupe and Kuhl, 1999; Kuhl, 2004; Bolhius and Gahr, 2006). Both birdsong and human speech are complex vocalisations, requiring coordination of both vocal and respiratory apparatus and the presence of a suitable tutor early in life to imitate. Darwin (1871) actually speculated that singing came before speech, and such things as singing apes may have arisen, communicating in a way similar to birds, bats and whales today. Similar cell types are found in avian brain as are present in mammalian/human brain circuitry (Farries and Perkel, 2000; Mooney and Prather, 2005; Scott and Lois, 2007), and although the overall morphology of the telencephalic systems may somewhat vary, the core neuronal circuitry remains remarkably similar (Jarvis et al., 2005; Bolhuis and Gahr, 2006). The VMP is homologous to a mammalian motor pathway which originates in the cerebral cortex and descends through the brainstem (Nottebhom et al., 1976) and the mammalian equivalent of the AFP is a cortical loop pathway which traverses the basal ganglia and thalamus (Luo and Perkel, 1999). The striatal Area X is homologous to mammalian striatum, in that both comprise nearly identical physiological cell types (Farries and Perkel, 2002). Johnson and Whitney (2005) have shown that the telencephalon is the origin of a large majority of the processes involved in human speech and the production of song in zebra finches. When compared proportionally to a mammal (rat) and a non-singing bird (pigeon) they noted that a notably larger proportion of the human and zebra finch brain was occupied by the telencephalon (>50%) while other species only showed 35% occupancy. This is not necessarily a discrimination between vocal learners and non-learners, but it is nonetheless noteworthy. Both human speech and birdsong production occur during critical periods, and upon termination of these periods, learning becomes increasingly difficult (generally after sexual maturity; Marler, 1970; Konishi, 1985; Doupe and Kuhl, 1999). In the human population, cases of extreme social and auditory deprivation have a negative impact to the point where normal speech can never be learned (Lane, 1976), e.g. in the case of Genie (Fromkin et al., 1974). This phenomenon is also observed in close-end learners such as the zebra finch whereby isolation during the song-learning phase (PHD 35-90) results in production of abnormal song (Thorpe, 1958; Immelmann, 1969; Marler and Sherman, 1983; Eales, 1987; Searcy and Marler, 1987; Tchernichovski and Nottebohm, 1998; Kojima and Doupe, 2007; Fehér et al., 2009). Both birdsong and human speech require auditory feedback during the learning period and are subject to degradation if auditory feedback is disturbed later in life (Cowie and Douglas-Cowie, 1992; Nordeen and Nordeen, 1992; Leonardo and Konishi, 1999; Woolley, 2004; Zevin *et al.*, 2004; Funabiki and Funabiki, 2007; Sober and Brainard, 2009). All these factors contribute to the recognition of the songbird as a suitable paradigm for understanding the neuronal mechanisms of cognitive processes which may be extrapolated to humans, giving insights into common sensory and motor learning, as well as the acquisition of speech (Brainard and Doupe, 2002; Kuhl, 2004).

#### 1.4.9 $GABA_A$ receptors in the avian song system

As detailed earlier, there are 16 GABA<sub>A</sub> receptor subunits identified in mammals, each encoded for by an individual gene; no further genes of this family exist in the human genome (Simon et al., 2004). In avian species only 14 GABAA receptor subunits have been detected ( $\alpha 1$ - $\alpha 6$ ,  $\beta 2$ - $\beta 4$ ,  $\gamma 1$ ,  $\gamma 2$ ,  $\gamma 4$ ,  $\delta$  and  $\pi$ ); where the  $\gamma 4$  and  $\beta 4$  subunits are orthologues of the mammalian  $\varepsilon$  and  $\theta$  subunits respectively (Darlison *et al.*, 2005). Despite a multitude of studies investigating the expression of GABA<sub>A</sub> receptor subunit genes within rat (Laurie et al, 1992a, b; Wisden et al, 1992; Pirker et al., 2000; Pöltl et al., 2003; Wafford et al., 2004) and monkey (Macaca mulatta) brain (Kultas-Ilinsky et al., 1998; Huntsman et al., 1999), no such studies have been completed within the zebra finch brain. This may be considered unusual, due to its suitability as a model of cognitive function; although it may be attributable to the paucity of available zebra finch GABA<sub>A</sub> receptor isolated cDNAs (only the  $\gamma$ 4-subunit sequence was published at the time of this work). GABA is well distributed within the zebra finch song system (Pinaud and Mello, 2007). GABA<sub>A</sub> receptors have been detected electrophysiologically in song system nuclei (Bottjer et al., 1998; Dutar et al., 1998; Spiro et al., 1999; Farries et al., 2005; Mooney and Prather, 2005; Prather et al., 2008, for further details refer to Chapter four) and pharmacologically (for example, Carlisle et al., 1998; Vicario and Raksin, 2000; Farries et al., 2005; Mooney and Prather, 2005; Ölveczky et al., 2005), but this has yielded little information pertaining to the stoichiometry of receptor subtypes present, or indeed their individual physiological roles. Only the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit has been identified and the corresponding mRNA mapped throughout the entire zebra finch song system (Thode et al., 2008). Besides this, no other studies have thoroughly investigated the distribution of the  $GABA_A$  or  $GABA_C$  receptor subunit genes within the song system.

Song is a learned behaviour which can be quantified by use of a sound spectrogram (Mooney, 1999). Due to the discrete nature of song, the song system and the distinct stages in song development, the zebra finch song system provides an excellent neuronal model for studying the underlying mechanisms of learning and memory. As  $GABA_A$  receptor-mediated neurotransmission is involved in modulation of learning and memory processes, work herein aimed to fully characterise the expression of  $GABA_A$  and  $GABA_C$  receptor subunit genes within nuclei of the male zebra finch song system.

## **1.5 Project Aims**

- Isolation of partial cDNAs encoding GABA<sub>A</sub> and GABA<sub>C</sub> receptor subunit genes from zebra finch (α1-α6, β2-β4, γ1, γ2, γ4, δ, π and ρ1-ρ3), which had not been previously sequenced
- Real-time reverse-transcription polymerase chain reaction (RT RT-PCR) and radioactive *in situ* hybridisation were employed to quantitatively ascertain the spatial and temporal expression profiles of GABA<sub>A</sub> receptor subunit genes in brain.
- Behavioural experiments using male zebra finches raised in auditory isolation were then completed in an effort to confirm the role of  $\gamma$ 4-subunit-containing GABA<sub>A</sub> receptors.
- Subsequently, generation of  $\gamma$ 4-subunit-targetting shRNA constructs were developed for RNA interference experiments to further confirm the role of these receptors in cognitive processing in zebra finch, these were tested at an *in vitro* level.
- Finally, a γ4-subunit-fusion protein was constructed and purified ready for synthesis into a γ4-subunit-specific antibody for future protein analyses.

## 2. MATERIALS AND METHODS

## 2.1 Animals

Animals were raised and treated in accordance with Home Office guidelines. Male zebra finches (*Taeniopygia guttata*) at different post hatch days (PHDs) were obtained from a variety of locations according to availability and different experimental criteria; details of which are further specified where relevant. Fertilised chicken eggs (Leghorn; *Gallus gallus domesticus*) were obtained from Henry Stewart Ltd., Lincolnshire, UK.

Zebra finches and chicks were sacrificed according to the Code of Practice for Humane Killing of Animals under Schedule 1 of the Animals Act (1986), by dislocation of the neck. Brains and any other tissues were quickly removed and frozen over dry ice before being wrapped in foil and parafilm and stored at -80°C until further use.

## 2.2 Isolation of total RNA from tissues

#### 2.2.1 Isolation of total RNA

Tissue was weighed and chopped prior to homogenisation in a glass-Teflon homogeniser. RNA was extracted with cold Tri<sup>®</sup> reagent (Ambion<sup>®</sup>, USA), in a 100µl per 10mg tissue ratio. On complete homogenisation, tissue was kept at 4°C for 5 min and then centrifuged at 12,000*g* at 4°C for 20 min (Micro 22R; Hettich, Germany). 200µl chloroform was added per ml of original Tri<sup>®</sup> reagent. The sample was vortexed briefly, maintained at 4°C for a further 5 min and centrifuged at 12,000*g* for 5 min at 4°C. The aqueous layer containing RNA was recovered and transferred to a sterile 1.5ml microfuge tube where 500µl isopropanol per ml original Tri<sup>®</sup> reagent was added. The sample was briefly vortexed and kept at 4°C for 10 min. RNA was precipitated with ethanol as described in section 2.3.3 and the pellet resuspended in an appropriate volume of 0.1% (v/v) diethylpyrocarbonate (DEPC)-treated H<sub>2</sub>O (DNase- and RNase-free H<sub>2</sub>O). Unless specified, all following experiments utilised water treated in this way.

#### 2.2.2 Determining concentration of nucleic acids

The yield of nucleic acids (5 $\mu$ l sample) was determined spectrophotometrically at 260nm and the purity verified by determining the 260/280 ratio (DU<sup>®</sup> 530 UV Spectrophotometer; Beckman Coulter, USA). Concentration of nucleic acids (ng/ $\mu$ l) was calculated by the following equation (where A is absorbance):

```
\label{eq:20ng/ml} \begin{array}{l} 20ng/\mu l \mbox{ (single-stranded DNA)} \\ \mbox{ Concentration (ng/\mu l) = A_{260} \ x \ dilution factor \ x \ 40ng/\mu l \ (single-stranded RNA)} \\ & 50ng/\mu l \ (double-stranded RNA) \end{array}
```

During an RNA extraction, RNA concentration was adjusted to  $80\mu g/108\mu l$  with H<sub>2</sub>O. Any samples not used immediately were suspended in a 2.5x volume of absolute ethanol and stored at -80°C until required.

# 2.3 Synthesis of complementary DNA (cDNA) from isolated RNA

#### 2.3.1 Removal of genomic DNA

Extracted total RNA suspension ( $80\mu g$ ) was added to the following reaction mix:  $32\mu l$  5x transcription optimised buffer (40mM Tris-HCl, pH 7.9; 6mM MgCl<sub>2</sub>; 2mM spermidine and 10mM NaCl), RNasin (RNase inhibitor; 80U),  $10\mu l$  dithiothreitol (DTT; 100mM), RQ1 DNase (6U; all from Promega, UK), and incubated at  $37^{\circ}C$  for 15 min.

#### 2.3.2 Phenol/chloroform extraction

Isolated RNA (this method also applies to DNA) was then subjected to a phenol/chloroform extraction to eradicate any remaining protein contamination from the nucleic acid sample. An equal volume of cold PCI solution (phenol: chloroform: isoamyl alcohol 25:24:1; pH 4.5) was added to the sample and mixed thoroughly, then centrifuged at 12,000g at  $4^{\circ}$ C for 5 min creating a biphasic mixture. The upper aqueous phase (containing suspended RNA) was aspirated avoiding the interface and lower organic phase and transferred to a sterile 1.5ml microfuge tube. An equal volume of cold CI solution (chloroform: isoamyl

alcohol, 24:1) was added to the aqueous phase to remove any traces of phenol. The sample was mixed thoroughly and centrifuged at 12,000g at  $4^{\circ}C$  for 5 min. Again, the top aqueous phase was removed and added to a sterile 1.5ml microfuge tube. RNA was then recovered from the aqueous phase by precipitation with ethanol (section 2.3.3).

#### 2.3.3 Ethanol precipitation

When DNA/RNA was required to be further concentrated, washed or used after storage in absolute ethanol, the following protocol was observed. A 2.5x sample volume of absolute ethanol and  $1/10^{\text{th}}$  sample volume of 3M sodium acetate pH 5.2 (300mM final) were added to the sample. Contents were gently mixed and precipitated at -80°C (RNA) or -20°C (DNA) for a minimum of 30 min. Following precipitation, the sample was centrifuged at 12,000*g* (RNA) or 20,000*g* (DNA) for 20 min at 4°C. The supernatant was aspirated then discarded and 150µl 75% (v/v) ethanol was added. Contents were mixed gently to free the pellet from the wall of the tube and remove any salts bound to the DNA/RNA. The mixture was centrifuged at 12,000*g* (RNA) or 20,000*g* (DNA) for 5 min at 4°C and all residual ethanol was removed. The pellet was left to air dry for 5-7 min and resuspended in 108µl H<sub>2</sub>O.

#### 2.3.4 cDNA synthesis: reverse transcription reaction

 $80\mu$ g RNA solubilised in  $108\mu$ l H<sub>2</sub>O was heated at  $65^{\circ}$ C for 5 min to denature RNA secondary structures before being added to the following reaction mix: RNasin (80U; Promega), 20 $\mu$ l 5x M-MLV buffer (Molony murine leukaemia virus; 50mM Tris-HCl, pH 8.3; 75mM KCl; 3mM MgCl<sub>2</sub> and 10mM dithiothreitol (DTT); Promega), 40 $\mu$ l 5mM deoxynucleotide mix (dNTPs; 5mM each of : dATP, dCTP, dGTP and dTTP; Sigma-Aldrich<sup>®</sup>, UK), random primers (9mer) (2.16 $\mu$ g; Promega), M-MLV reverse transcriptase (800U; Promega) and incubated at 42°C for 90 min.

#### 2.3.5 Precipitation and washing of cDNA

cDNA was precipitated with an equal volume of 100% (v/v) isopropanol and  $1/10^{\text{th}}$  volume of 3M sodium acetate (pH 5.2) at -20°C for 1 h. cDNA was

subsequently centrifuged at 20,000*g* for 20 min at 4°C. Isopropanol was decanted and the pellet was subjected to two subsequent ethanol precipitations (section 2.3.3) with 80% and 75% (v/v) ethanol. cDNA was then solubilised in 50 $\mu$ l H<sub>2</sub>O and stored at -20°C ready for use.

# **2.4** Amplification of $GABA_A$ and $GABA_C$ receptor subunit cDNAs

For this section of work, adult male zebra finches were obtained from a breeding colony at Newcastle University, UK. All birds were PHD >100 and brains were extracted as described in section 2.1.

## 2.4.1 Reverse transcription-polymerase chain reaction (RT-PCR) primer design

Degenerate oligonucleotide primers were designed to amplify specific partial cDNAs of GABA<sub>A</sub> and GABA<sub>C</sub> receptor subunit genes of interest (i.e.  $\alpha$ 3,  $\alpha$ 4,  $\alpha$ 6,  $\gamma 1$ ,  $\delta$ ,  $\pi$  and  $\rho 1$ -3) from zebra finch. Sequences were already available for the remainder of the GABA<sub>A</sub> receptor subunits (i.e.  $\alpha 1$ ,  $\alpha 2$ ,  $\alpha 5$ ,  $\beta 2$ ,  $\beta 3$ ,  $\beta 4$ ,  $\gamma 2$  and  $\gamma 4$ (GenBank accession number AM086933); as these had been previously cloned in the laboratory (data unpublished). Primers were designed adhering to standard criteria: 50% G+C (guanine and cytosine) content, 18-25mer in length and no more than three identical consecutive bases. All forward primers were in the region of sequence encoding the N-terminus and the reverse primers corresponded to a sequence that encodes part of the large intracellular loop (between transmembrane domains (TM) 3 and 4) of the subunit gene (Fig. 1.1, Chapter 1). The primers were degenerate and based on highly conserved subunit sequence domains between species and non-homologous regions between different subunits. They were designed primarily with chicken and rat sequences which were the most relevant sequences available at the time (for accession numbers refer to Appendix). For this, sequences were all obtained from the National Centre for Biotechnology Information (NCBI) gene database (www.ncbi.nlm.nih.gov/) and DNA aligned using an online alignment tool available at http://www.justbio.com/aligner/index.php, details of primers are given in Tables 2.1 and 2.2.

Target GABA₄ receptor gene	Forward: 5'→3'	Reverse: 5'→3'	Anneal. temp. (ºC)	Cycle No.	Amplicon (bp)
$\alpha_3$	ATGGA <b>R</b> TA <b>Y</b> ACTCTAGA <b>Y</b> GT <b>N</b> TT <b>Y</b> TT	GGRTATGTNGTNCCAACKATGTTRTARG	53	35	882
$\alpha_4$	TTGATGGTTCGAAGGTATGGACTCC	GCAGAAGA¥AAÞGT¥TCTGTGCATT	60	40	980
$\alpha_6$	TGATGAGAGGTTGAGTTTGGTGGG	CGGAGTTTACTCGTTTCTTCAGGT	62	30	878
$\gamma_1$	TTCCGGAATCGAGGAAATCTGATGC	CCTTCTAGGCACTCATATCCATATC	62	30	788
δ	<u>N</u> GCCTG <u>R</u> TT <u>Y</u> CA <u>Y</u> GA <u>Y</u> GTGAC <u>N</u> GTGG	G <u>S</u> ACDATGGCRTTCYTNACRTCCATYT	55	40	715
π	TGGTCTTTCATGGCAACAAGAGC	TGACTTTTCATGGTCAAGTCACTGC	59	35	820

**Table 2.1.** RT-PCR primer combinations, detailing conditions required for successful GABA<sub>A</sub> receptor partial subunit cDNA amplification from zebra finch brain and expected amplicon sizes. Nucleotides at degenerate positions are represented by a single letter code: ( $\underline{N}$ ) A,C,G,T; ( $\underline{D}$ ) G, A, T; ( $\underline{H}$ ) A, T, C; ( $\underline{W}$ ) A, T; ( $\underline{M}$ ) A,C; ( $\underline{R}$ ) A,G; ( $\underline{K}$ ) G,T; ( $\underline{S}$ ) G, C; ( $\underline{Y}$ ) C, T.

Target GABAc receptor gene	Forward: 5'→3'	Reverse: 5'→3'	Anneal. temp. (°C)	Cycle No.	Amplicon (bp)
ρ1	tgga <b>y</b> gt <b>s</b> ca <b>r</b> gt <b>k</b> ga <b>r</b> ag <u>yy</u> tgg	GCMAGGCTSCACGCGCWGGGGAKCTTMT	57	30	822
ρ2 (outer)	tgga <b>y</b> gt <b>s</b> ca <b>r</b> gt <u>k</u> ga <b>r</b> ag <b>yy</b> tgg	TTCTTGGGTA <b>R</b> CC <b>W</b> GCCAGG	62	40	940
ρ2 (inner)	gaagatctgggtccc <b>x</b> ga <b>x</b> g	gaatga <b>r</b> g <b>y</b> at <b>y</b> ccacacgtgca	65	35	722
ρ3	tgg <b>r</b> tttggagg <b>n</b> tc <b>d</b> cca <b>r</b> t <b>r</b> cc	¥G¥TT¥CDCTCTTCMA¥TGTGG	61	40	1118

**Table 2.2.** RT-PCR primer combinations, detailing conditions required for successful  $GABA_C$  receptor partial subunit cDNA amplification from zebra finch brain and expected amplicon sizes. Denoted lettering as in Table 2.1. \*In the case of the  $\rho$ 2 subunit, a nested RT-PCR was carried out (section 2.4.3).

#### 2.4.2 Reverse-transcription polymerase chain reaction (RT-PCR)

Amplification of  $GABA_A$  and  $GABA_C$  receptor subunit cDNAs was carried out by RT-PCR. All 50µl reactions contained: 10µl 5x buffer (Promega); 3µl MgCl<sub>2</sub>

(1.5mM final concentration; Promega); 8µl dNTPs (1.25mM; Sigma-Aldrich<sup>®</sup>), zebra finch cDNA (160ng), 200ng each of forward and reverse primer (Eurofins MWG Operon, Germany), GoTaq<sup>®</sup> flexi DNA polymerase (1.5U; Promega) in storage buffer В (20mM Tris-HCl, pH8.0; 100mM KCl: 0.1mM ethylenediaminetetraacetic acid (EDTA), 1mM DTT; 50% (v/v) glycerol; 0.5% (v/v) Tween<sup>®</sup>20 and 0.5% (v/v) Nonidet<sup>®</sup>-P40; Promega), made up to 50µl with H<sub>2</sub>O. A negative control reaction was always completed concurrently, with the cDNA template omitted, to ensure products were amplified from the cDNA template and not any contamination.

Cycling conditions were as follows; 5 min *initial denaturation* (94°C) followed by, 30-40 cycles at: 1 min *denaturation* (94°C), 1 min *annealing* (Tables 2.1 and 2.2), 2 min *extension* (72°C) and a final 10 min extension at 72°C. RT-PCR reactions were carried out in thin-walled 0.5ml microfuge tubes in a thermocycler (Genius or TC-512; both Techne).

#### 2.4.3 Nested RT-PCR

A nested RT-PCR strategy was employed for amplification of the GABA<sub>C</sub> receptor  $\rho$ 2-subunit gene as there were problems with the specificity of the original primers and low levels of mRNA expression. In this case, two pairs of RT-PCR primers were used for a single locus (Table 2.2). The first pair (outer) amplified various products in a standard RT-PCR reaction as described above. In a second RT-PCR the second set of primers (inner) bind within the first PCR product, producing a shorter and more specific amplicon. This ensured that unwanted products amplified in the first reaction were not be amplified a second time, by a second set of primers. This strategy is essential in cases like this where the expression of the gene of interest is very low, and the sequence of the gene is unknown.

The first round of RT-PCR was completed as detailed in section 2.4.2. From the 50µl reaction mix after the RT-PCR, 10µl was removed and added to 10µl 5x buffer (Promega), 3µl MgCl<sub>2</sub> (1.5mM; Promega), 8µl dNTPs (1.25mM; Sigma-Aldrich<sup>®</sup>), 200ng each (inner) forward and reverse primer (Table 2.2), Go*Taq*<sup>®</sup>

flexi DNA polymerase (1.5U; Promega) and made up to  $50\mu$ l with H<sub>2</sub>O. Cycling conditions as above.

#### 2.4.4 Agarose gel electrophoresis

5µl of 6x loading dye (0.2% (w/v) bromophenol blue; 28% (w/v) sucrose; 26% (v/v) TE buffer (protects DNA/RNA from degradation); 10mM Tris-HCl, 1mM EDTA; pH 8) and 72% (v/v) H<sub>2</sub>O) was added to the 50µl RT-PCR product. Separation of PCR products was performed on 1.5-2.5% (w/v) agarose (Bioline, UK) gels (depending on the expected fragment size), which contained ethidium bromide (10µg/100ml; Sigma-Aldrich<sup>®</sup>), in 1x TAE buffer (Tris-acetate EDTA; 40mM Tris-base; Sigma-Aldrich<sup>®</sup>, 20mM glacial acetic acid; Fisher Scientific, UK, 5mM EDTA; pH 8). Gels were electrophoresed between 80-100V for 45-90 min depending on the expected amplicon size. Samples were run alongside molecular weight markers; Hyperladder IV (Bioline) and 1kb DNA ladder (stock:  $1.0\mu g/\mu l$  ladder in 10mM Tris-HCl, pH 7.5; 50mM NaCl; 0.1mM EDTA; Invitrogen<sup>TM</sup>, UK). PCR products were visualised using a UV (ultraviolet) transilluminator (Syngene<sup>TM</sup> tools software.

#### 2.4.5 RT-PCR product purification

Products of expected size were excised from gels with a sterile scalpel and stored on ice in pre-weighed 1.5ml sterile microfuge tubes. Each DNA gel slice was purified according to the standard GFX<sup>TM</sup> PCR DNA and gel band purification kit protocol (GE Healthcare, UK). The kit utilises a chaotropic agent in acetate to dissolve agarose and proteins, DNA binds selectively to the silica membrane within the spin-column where it is washed of salts and other contaminants by an ethanolic buffer (80% (v/v) ethanol; Tris-EDTA 10mM; Tris-HCl, pH 8.0; 1mM EDTA). DNA was then eluted with 50µl H<sub>2</sub>O and used immediately or stored at - $20^{\circ}$ C.

### 2.5 Cloning of partial cDNAs

#### 2.5.1 Ligation

PCR products were ligated into the pGEM<sup>®</sup>-T Easy vector (Promega; for full vector map see Appendix). The linearised vector has single thymine (T) base overhangs which covalently bind with complementary adenine (A) overhangs on the amplified product added during *Taq* polymerase RT-PCR. Each ligation reaction contained: 7.5µl 2x buffer T4 (60mM Tris-HCl, pH 7.8; 20mM MgCl<sub>2</sub>; 20mM DTT, 2mM ATP and 10% (v/v) polyethylene glycol), pGEM<sup>®</sup>-T Easy vector (50ng), T4 DNA ligase (3U; all Promega) and 5.5µl purified RT-PCR product. This was ligated overnight at 4°C.

## 2.5.2 Synthesis of competent <u>Escherichia coli</u> XL1-Blue and TB1 cells

Competent cells are able to take up naked exogenous DNA from the environment; E. coli XL1-Blue and TB1 cells were made artificially competent by chemical treatment with calcium chloride. Both types of competent cells used for different experiments were synthesised in the same way. A  $100\mu l$  aliquot of a previous E. coli XL1-blue/TB1 culture was taken from storage at -80°C, defrosted on ice and added to 2ml LB (Luria Bertani) medium (1% (w/v) peptone from casein; 0.5% (w/v) yeast extract; Merck, UK; 1% (v/w) NaCl; Sigma-Aldrich®; pH 7.0) supplemented with 0.1% (v/v) tetracycline (100µg/ml; Sigma-Aldrich<sup>®</sup>). The culture was incubated at 37°C, 200 rpm (revolutions per minute) overnight in an orbital shaker (S150 Orbital Incubator, Stuart Scientific, UK). 2ml of the overnight culture was then added to 100ml pre-warmed LB medium (100µg/ml tetracycline). Immediately, a 1ml aliquot of the inoculated culture was transferred to a cuvette and stored on ice to be used as a blank. The inoculated culture was then incubated as before. After an initial 2 h incubation, absorbance (A<sub>660</sub>) was measured every 30 min until it reached an optical density (OD) of ~0.5. At this stage the culture was divided between two 50ml tubes and centrifuged at 900g (Allegra® X-15R: Beckman Coulter) at 4°C for 15 min. The supernatant was discarded and each pellet resuspended in 20ml filter sterilised (0.2µm), cold TFB I buffer (transformation buffer 1; 30mM potassium acetate; 50mM MnCl<sub>2</sub>; 100mM KCl; 10mM CaCl<sub>2</sub>; 15% (v/v) glycerol; pH 5.8), and incubated on ice for 30 min. Cells were centrifuged again at 900*g* for 20 min at 4°C, the supernatant discarded and each pellet gently resuspended in 2ml filter sterilised (0.2 $\mu$ m) cold TFB II buffer (10mM 3-(N-Morpholino) propanesulfonic acid (MOPS); 76mM CaCl<sub>2</sub>; 10mM KCl; 15% (v/v) glycerol, pH 6.5; all Sigma-Aldrich<sup>®</sup>). Ensuring the cells were kept chilled at all time, 100 $\mu$ l aliquots were carefully prepared and stored at -80°C until use.

#### 2.5.3 Transformation of competent <u>E.coli</u> XL1-Blue cells

The pGEM<sup>®</sup>-T Easy vector contains the *lacZ* gene that encodes for the protein  $\beta$ galactosidase, which metabolises X-Gal (5-bromo-4-chloro-3-indolyl-β-Dgalactopyranoside) to produce bright blue colonies. If the open reading frame of this gene is interrupted by the presence of an inserted sequence,  $\beta$ -galactosidase is non-functional. By adding IPTG (isopropyl  $\beta$ -D-1-thiogalactopyranoside the *lacZ* gene inducer) and X-Gal (the substrate) to the agar plate, this chromogenic assay visually indicates the presence or absence of an insert. 15µl ligation mix was added to 100µl XL1-Blue cells and incubated on ice for 45 min (chilling in the presence of divalent cations prepared the cell walls to become permeable to plasmid DNA). Cells were then thermal shocked at 42°C for 1 min to incorporate the plasmid into the cells and immediately placed back on ice. 150µl cold SOC (formal name) medium (2% (w/v) bacto-tryptone; 0.5% (w/v) bacto-yeast extract; 10mM NaCl; 10mM KCl; 20mM Mg2+ and 20mM glucose) was added and cells were incubated at 37°C whilst shaking at 200 rpm for 30-45 min (sufficient time for expression of the antibiotic resistance gene prior to plating). Culture was then divided equally between three replicate LB agar plates (1% (w/v) peptone from casein; 0.5% (w/v) yeast extract; 1% (v/w) NaCl; 1.2% (w/v) agar-agar, pH 7.0; Merck) supplemented with 0.1% (v/v) ampicillin (100µg/ml; Sigma-Aldrich<sup>®</sup>), which were previously spread with 100µl IPTG (200mg/ml; Sigma-Aldrich<sup>®</sup>) and 40µl X-Gal (20mg/ml: Fluka Chemicals, Germany) and incubated overnight at 37°C in a humid environment.

#### 2.5.4 Overnight cell culture

Recombinant cells were differentiated by blue/white colony screening. Selected white colonies were sub-cultured using a sterile toothpick to gently transfer an

individual colony into 2ml LB medium supplemented with 0.1% (v/v) ampicillin ( $100\mu g/ml$ ; Sigma-Aldrich<sup>®</sup>). Cultures were incubated overnight at 37°C with agitation at 200 rpm, to ensure good aeration.

#### 2.5.5 Isolation of plasmid DNA

pGEM<sup>®</sup>-T Easy vectors containing cloned inserts were isolated from recombinant *E. coli* cells using a GenElute<sup>TM</sup> plasmid miniprep kit (Sigma-Aldrich<sup>®</sup>), according to the manufacturer's protocol. The kit contains a series of solutions whereby cells are lysed with a sodium dodecyl sulfate (SDS) mixture, the reaction is neutralised, DNA is adsorped onto silica, impurities are washed away with an ethanolic buffer and bound DNA is eluted with 50µl H<sub>2</sub>O.

#### 2.5.6 Restriction digestion

Each cloned PCR fragment was digested from the pGEM<sup>®</sup>-T Easy vector with an *Eco*RI restriction digest, the vector contains recognition sites for this enzyme either side of the insert in the multiple-cloning sites (see Appendix). Reaction mix contained; 1µl eluted plasmid DNA, 1.5µl 10x Buffer H (90mM Tris-HCl, pH 7.4; 10mM MgCl<sub>2</sub>; 50mM NaCl), *Eco*RI (6U; Promega), made up to 15µl with H<sub>2</sub>O. This was incubated at 37°C for 90 min. Gel electrophoresis was performed with digested clones (section 2.4.4), alongside an undigested plasmid DNA sample. Clones containing an insert of the anticipated size were sent for automated DNA sequencing (Eurofins MWG Operon, UK). Multiple clones were sequenced and aligned using online alignment software (section 2.4.1) to identify any errors brought about by the *Taq* DNA polymerase or sequencing inaccuracies.

#### 2.6 Radioactive in situ hybridisation

Radioactive *in situ* hybridisation utilises a radiolabelled nucleic acid probe to localise specific RNA transcripts within a tissue. Tissues are fixed and the radiolabelled complementary DNA probe is able to hybridise at an elevated temperature with target mRNA in the tissue. Excess probe is then washed away and the target mRNA bound to the radiolabelled probe in the tissue can be localised by autoradiography.

#### 2.6.1 Oligonucleotide probe design

Synthetic oligonucleotide probes were designed which were complementary to mRNA encoding the zebra finch GABA<sub>A</sub> and GABA<sub>C</sub> receptor subunit genes as detailed in Table 2.3. All probes adhered to the following criteria where possible: 45mer in length, between 45-55% G+C content, no more than 3 consecutive bases, bridging of an exon boundary (to ensure high specificity) and non-homologous (i.e. no sequence similarity) with other GABA<sub>A/C</sub> receptor subunit mRNAs. Two probes were designed to target each subunit gene to ensure an accurate mRNA distribution was observed. All sequences were analysed using BLAST (Basic local alignment search tool) as per section 2.4.1. Oligonucleotides were produced commercially at Eurofins MWG Operon (Germany), and delivered in a lyophilised form.

Target gene	Probe sequence (5'→3')	mRNA sequence location	%GC
GABA <sub>A</sub> rec	eptor subunit genes		
α1*	TGCTGGTTCCCGTGTCCACTCATACACAACCTCTGCTCTGGTATA	N-terminus	51
α1	TCTCACAGTCAATCTCATTGTGTATAGCAACGTCCCATCCTCTGT	N-terminus	44
α2	GCCTAAAAGATCATACTGGTTCAGTCTTGAGCCATCTGGTGCCAC	N-terminus	49
α2*	GAGCCGTCATTACTGTATATTCACCTGTGCTTGACTTAACTGTCT	N-terminus	42
α3	GTATTCGCCTGTGCTGGATCGAACCATCTCTGTCCCAACCACGTG	N-terminus	56
a3*	TTTTCTTCATCTCCTGGGCCTCCAGCACCTTCTTGCCATCCCAGG	Loop	56
α4*	ACACTCTGCACTTATTGTAAGCCTCATTGTGTACAGGATGGTGCC	N-terminus	47
α4	CCTTTGTACTGGAGCAACAGGAAATTCAAGAAGGGACTTGACCGT	Loop	47
α5*	TTCACCTGTACTGGTACTGATGTTTTCCGTGCCCACGGTTTGTCC	N-terminus	51
α5	CGTGAGCCATCTTCGGCCACCACCACAGACGTTGTGGTGCTGTTA	N-terminus	58
α6*	TTGGAGTCAGAATGCGAGACAATCTCGTCCTCTGCCGGTACAGTT	Loop	51
a6	TGGAGAAGACTGGAAGACTCCTGTGGTACTTCTACTGAATGCAAT	N-terminus	44
β2	GAGATGTCCACGGATGGGTTCCGACAGAGAGTCGAAATTTGAATT	N-terminus	47
β2*	TTGAAGAAGGAAAAAACAACGGGGAAGAATATGCGAGACCATCGA	Loop	42
β3*	TTCCACCTGAAGCTCAGTGAAAGTCTTGGATAGGCACCTGTGGCA	N-terminus	51
β3	AATGTGAATGGAAACACCATTCTTGACCATCTGTCGATGGGATTC	Loop	42

β4	CTTGTCAATGGTGCTGACGTCTGTCAGGTCTGGGATTTTGAGCTT	Loop	49
β4*	ACAGCAGAGTCATTCCCTTGCCAGAAGAAGACAATGTCATCCACC	N-terminus	49
γ1*	TTGGAATTAGTGTTGATCCAGGTCGAACAGCAATTGCAGGTGGTT	Loop	44
γ1	TTCTGGGATATCCATAGCTTGAAAACTCCAGAGGACAGGAGTGTT	N-terminus	44
γ2*	CATTTGAATGGTAGCTGATCGGGGGGGGGGATGTCAATAGTGGGTGC	Loop	53
γ2	TCTCCTGAAGTAGTCTTCACCACCTCAGTTGTGTTTCTTAGTCCT	N-terminus	44
γ4	TGCATGATGTGGTTGATGTTGATGGTGGTGAAGGTTGGCATCACC	Loop	49
γ4*	AGAGTGTAGAGCACTTTTCCATCGTTCCAGATGCGGAGGAGCTGG	N-terminus	53
$\delta^*$	AAGGTGGAAGTGGAGACTGAGCCTGGGAAACTGACCAGATTTGAA	N-terminus	49
δ	TCCTCTGAAGAGTAGCCATAGCTCTCCAAATCCAACATGCATTCC	N-terminus	47
GABA <sub>c</sub> receptor subunit genes			
ρ1*	TACCACCCTGTGCTGCTGTAAAAGGCAAGCTTTGTGGTAGTGTGA	N-terminus	49
ρ1	CAGCCAGATGGGAAGGTACTTTATAGCCTCAGAGTTACAGTAACA	N-terminus	44
ρ2*	CACTGTGATCCTCATGCTGTAGAGGACATGACCATCAGGGAACAC	N-terminus	51
ρ2	GTCTCAATTCTTGATACAGAAATTCCACACTACATCAAGACTGGC	N-terminus	40
ρ3	GCAAGTCCACTGGAAGCACTGAATTCCTCAATGAAGAACTGGGAG	N-terminus	49
ρ3*	AGCAGAAACAGTTATTCTCAGACTGAAGAGGACATTGCCATCGGG	N-terminus	47

**Table 2.3.** Sequences encoding 45mer synthetic oligonucleotide probes complementary to the subunit-sequences of the  $GABA_A$  and  $GABA_C$  receptors. Sequences shown are the reverse complement to the sense strand. Two probes targeting each subunit mRNA were designed and tested, and in all cases gave identical binding patterns with all genes. Probes denoted with an (\*) were utilised for all subsequent experiments.

#### 2.6.2 Silanisation of microscope slides

The majority of animal tissues carry a negative charge. To ensure optimal adherence of the tissues to microscope slides, slides were pre-treated with (3-Aminopropyl) triethoxysilane (Sigma-Aldrich<sup>®</sup>) which creates a positively charged surface and allows covalent bonding of tissue sections. Slides (twin-frost 0.8-1.0mm thick; VWR<sup>TM</sup>, UK) were initially sterilised at 180°C for 3 h and were then passed through a series of baths: 30 sec in 100% (v/v) acetone; 5 min in 2% (v/v) (3-aminopropyl)-triethoxysilane in acetone; 2x 2 min rinses in 100% (v/v) acetone (to remove (3-aminopropyl) triethoxysilane) and 3x 2 min rinses in H<sub>2</sub>O

(to remove acetone). Once slides were completely dry, they were sterilised for a further 3 h at 180°C.

#### 2.6.3 Rearing of zebra finches and sectioning of brains

For initial *in situ* hybridisation experiments (localisation of  $GABA_{A/C}$  receptor subunit transcripts within the song system), adult male zebra finches (all PHD >100) were obtained from a breeding colony at The University of Sheffield and sacrificed as in section 2.1.

A second set of *in situ* hybridisation studies were undertaken to investigate whether  $GABA_A$  receptor  $\gamma$ 4-subunit mRNA expression in zebra finches was affected by song learning, whereby birds were able to sing properly (aviary-reared, with tutor) or unable to produce 'normal song' (cage-isolated, without tutor). All birds were reared at sacrificed at Max-Plank-Institut für Ornithologie in Seewiesen, Germany. The work formed two separate parts. For the primary experiments there were two groups of zebra finches (all male; n = 5 per group); the first group were 'normally' reared birds in that they were raised in an aviary with their parents and other conspecifics; the second group were raised in individual isolation cages where they remained with their parents until PHD 5 and then were raised in the cage alone, in physical and auditory isolation. In the second set of experiments there were three groups of zebra finches (all male; n=5 per group). The first group were raised in an aviary with tutors and conspecifics, as before. A second group were raised in individual isolation cages with their parents (tutor present). A third group were raised in individual isolation cages, but instead of being alone (as in the preliminary work), they were accompanied by their parents until PHD 7, their mother until PHD 30 and then reared by two unrelated females (which do not sing) until PHD 80. Songs produced by birds in the experimental groups were recorded with a sonogram at PHD 80 in the presence of an unfamiliar female bird (directed song). All birds were sacrificed at PHD 80, their brains removed, immediately frozen over dry ice and shipped to the laboratory in the fastest possible time, where upon receipt, they were immediately stored at -80°C until use. Brains were sectioned on a cryostat (CM 1900; Leica, Germany) where 10µm coronal sections were cut and freeze-thaw mounted onto

pre-silanised slides (section 2.6.2) at -12°C. Mounted cryo-sections were then stored at -80°C until use.

#### 2.6.4 Fixation of tissue sections

Tissue fixation ensures the preservation of tissue morphology and protects the integrity of RNA in cells. Paraformaldehyde is a cross-linking fixative; the aldehyde group binds primarily with nitrogen groups on proteins in the tissue where it forms a methylene bridge. Nucleic acids are retained in the matrix of the cross-linked protein molecules and they themselves remain chemically unaffected. Prior to hybridisation, the sections were fixed for 5 min in 2% (w/v) paraformaldehyde (Sigma-Aldrich<sup>®</sup>) in phosphate buffered saline (PBS; 130mM sodium chloride, 7mM disodium hydrogen orthophosphate, 3mM sodium dihydrogen orthophosphate, pH 7.4) at 4°C. Sections were then washed twice (5 min each) with PBS at room temperature, dehydrated in 70% (v/v) ethanol followed by 100% (v/v) ethanol, and left to air dry.

#### 2.6.5 Labelling of the oligonucleotide probe

Each oligonucleotide probe was radioactively labelled at the 3' end with <sup>35</sup>S (deoxyadenosine 5' (alpha-thio) triphosphate). Terminal deoxynucleotidyl transferase, recombinant (rTdT) catalyses the repetitive addition of mononucleotides (radiolabelled adenosine) to the terminal 3'-OH of a DNA initiator, in this case the ssDNA probe. The mix was as follows: 5µl 5x TdT buffer (100mM cacodylate buffer pH 6.8; 1mM CoCl<sub>2</sub> and 1mM DTT; Promega); 5ng oligonucleotide probe (resuspended in H<sub>2</sub>O);  $\alpha^{35}$ S-dATP (15.625µCi; Perkin-Elmer<sup>®</sup> Life Sciences Ltd., USA); rTdT (30U; Promega) made up to 25µl with H<sub>2</sub>O. The mix was incubated at 37°C for 75 min. Following this, 1µl of the reaction was added to 5ml scintillation fluid (Ultima Gold<sup>TM</sup>; Perkin Elmer<sup>®</sup> Life Sciences Ltd.) to verify activity before purification of the probe.

Any unincorporated nucleotides from the labelling reaction were removed using commercially prepared spin columns (Sephadex G25 suspended in 1x sodium chloride–Tris–EDTA buffer; Roche Applied Science, UK) according to the manufacturer's protocol; the columns use gel-filtration chromatography to separate molecules according to their size whereby smaller molecules are retained (unincorporated nucleotides) and larger molecules such as oligonucleotides can pass through. Activity of the purified radiolabelled probe was quantitated by adding 1µl purified probe to 5ml scintillation fluid in a scintillation vial (Packard<sup>®</sup>, USA) and measuring on a liquid scintillation analyser (Tri-Carb<sup>®</sup> 2250CA: Packard<sup>®</sup>), along with the sample taken prior to purification. This determined the counts per minute (cpm) of 1µl purified probe and thus the volume of probe needed to hybridise slides at 100,000 cpm/slide could be calculated.

#### 2.6.6 Hybridisation of the oligonucleotide probe

Hybridisation (200µl of buffer per slide under parafilm cover slips) was performed in: 50% (v/v) deionised formamide (Fluka Chemicals), 4x standard saline citrate (SSC – the high ionic strength solution maximises the rate of probe annealing, 600mM NaCl (60mM Tri-sodium citrate dehydrate pH 7; Sigma-Aldrich<sup>®</sup>), 5x Denhardt's solution (this contains high molecular weight polymers capable of saturating non-specific binding sites), 25mM sodium phosphate (pH 7), 1mM sodium pyrophosphate, 20mM DTT (this is used in all solutions concerning in situ hybridisation as <sup>35</sup>S can give high background signals), 100µg/ml polyadenylic acid, 120µg/ml heparin (this increases specificity of the in situ hybridisation), 200µg/ml denatured, fragmented salmon sperm DNA (competitive blocking agent; all Sigma-Aldrich<sup>®</sup>), 10% (w/v) dextran sulphate (this increases probe binding to target mRNA by increasing the rate of re-association of nucleic acids; Fluka Chemicals) and 100,000 cpm/µl radiolabelled probe. Sections were hybridised in a humid atmosphere at 42°C overnight. Control slides for each experiment contained, in addition, a 200-fold excess of the same unlabelled oligonucleotide, this competitive hybridisation verified the hybridisation of the probe was specific.

#### 2.6.7 Washing of sections

Hybridised sections were initially rinsed in 1x SSC (150mM NaCl; 15mM Trisodium citrate dehydrate, pH 7.0) at room temperature, to remove the cover slips and the majority of the hybridisation buffer. Slides were then subjected to two high stringency washes for 30 min each at 55°C in a shaking water bath (Clifton, UK). The washes were followed by rinsing in 1x SSC then 0.1x SSC (30 sec each) at room temperature. Hybridised sections were then dehydrated in 70% (v/v) ethanol and 100% (v/v) ethanol (30 sec each), loosely covered and left to air dry for 2-3 h. Slides were secured (arranged randomly) in a standard X-ray film cassette where they were apposed to Kodak Biomax MR X-ray film (Perkin-Elmer<sup>®</sup> Life Sciences Ltd).  $\beta$ -electrons from the <sup>35</sup>S are completely absorbed by the film to produce a global image. Exposure time was usually one month unless otherwise stated. Two films per experiment were developed after two different exposure times to ensure that any densitometric measurements were taken in the linear range of the film. All exposures were completed at room temperature.

#### 2.6.8 Densitometric analysis of autoradiographs

Autoradiographs were developed (Compact X4 developing machine; Xograph Healthcare Ltd., UK) and individual images of sections were visualised with a light box and CCD camera (TM-6CN PULNiX Europe Ltd.) then captured using Easygrab Picolo<sup>TM</sup> software (Euresys<sup>TM</sup>, Belgium). mRNA expression was quantified by densitometric analysis of individual brain sections using Scion image software (Scion Corporation, USA). Brain regions (more specifically, the nuclei of the song system), were identified by use of a stereotaxic atlas of the zebra finch brain (available at www.ncbi.nlm.nih.gov/books/bookres.fcgi/ atlas/atlaspdf) in combination with a recent publication of a three-dimensional magnetic resonance imaging stereotaxic atlas of the zebra finch brain (Poirier et al., 2008). Autoradiographic signal in areas of interest was calculated as pixel density per brain area measured. Three consecutive brain sections containing the nucleus of interest were measured to eliminate the possibility of section-specific labelling. Densitometric readings of the background were also taken to remove any fluctuations in signal intensity which were not a result of probe binding. Furthermore, internal control measurements were taken from the Nidopallium which is a brain area not associated with learning and memory. This was verified by comparison of gene expression in this region which remained unaffected by the different parameters of the experiment (Figs. 5.3, 5.6 and 5.10, Chapter 5). These readings were considered as baseline (100% signal), and thus the autoradiographic signal (mRNA expression) from other brain regions could be expressed as a percentage of the baseline. Data is expressed as mean  $\pm$ SE and the significance of any differences between mRNA expression in any of the nuclei of the three groups of birds were analysed by one way analysis of variance (ANOVA).

#### 2.7 Real-time quantitative RT-PCR

Real-time RT-PCR was employed to fully quantify expression of mRNA encoding selected GABA<sub>A</sub> receptor subunit genes ( $\alpha$ 1,  $\alpha$ 2,  $\alpha$ 3,  $\alpha$ 4,  $\beta$ 2,  $\beta$ 3 and  $\gamma$ 4) within Area X, HVC and RA of the song system and any changes of mRNA expression during important developmental stages. Facilities for the microdissection of the individual song-system were not available to us and so the work was carried out by our collaborative group at the Max-Plank-Institut für Ornithologie in Germany.

#### 2.7.1 Animals and tissue dissection

Male zebra finches were used at PHD 30, 55, 80 and 100 and were reared in breeding aviaries (6 m<sup>3</sup>; 7 breeding couples) at the Max-Planck-Institut für Ornithologie, Seewiesen. The following light regime was used: 0.5 h dawn, 14 h light, 0.5 h dusk, 9 h darkness. Experimental birds were collected in the same season of the same year; they were sacrificed and their brains quickly removed and shock-frozen. Constant areas of HVC, Area X and RA tissue were micro-dissected from cryostat sections of both hemispheres by means of a laser interfaced into a research microscope. All experiments were performed with permission of the government of Oberbayern.

#### 2.7.2 Isolation of RNA

In brief, RNA was purified from the micro-dissected nuclei of zebra finches following the manufacturer's instructions of the RNeasy<sup>®</sup> Micro Kit (Qiagen, UK) and any remaining DNA was digested in the RNeasy MinElute spin column with RNase-free DNase I, the RNA was eluted with H<sub>2</sub>O. The quality and yield of RNA was digitally determined on a 2100 Bioanalyzer using the pico-chip (Agilent Technologies, USA). First-strand cDNA synthesis was carried out with the SuperScript<sup>™</sup> III First-strand synthesis system for RT-PCR (Invitrogen) following the standard protocol with random hexamers and equal amounts of RNA (10ng).

#### 2.7.3 Real-time RT-PCR primer design

Specific primers were designed against a selection of GABA<sub>A</sub> receptor subunit genes ( $\alpha$ 1,  $\alpha$ 2,  $\alpha$ 3,  $\alpha$ 4,  $\beta$ 2,  $\beta$ 3 and  $\gamma$ 4),  $\beta$ -actin, glyceraldehyde-3-phosphate dehydrogenase (GAPDH), and hypoxanthine-guanine-phosphoribosyl transferase 1 (HPRT1) gene sequences (Table 2.4) of zebra finch using the Primer3 software (http://frodo.wi.mit.edu/cgi-bin/primer3/primer3\_www.cgi). All primers adhered to the standard stringent real-time RT-PCR primer criteria: amplicon between 80-250bp and should span intron/exon boundaries, primer length between 15-25bp, no more than three consecutive bases. All primer pairs require the same melting temperature (T<sub>m</sub>) and typically the same annealing temperature to avoid differing reaction efficiencies. The efficiency (E) was 95 – 100 % for all primer pairs used.

Target gene	Forward: 5'→3'	Reverse: 5'→3'	Amplicon size (bp)
$GABA_AR\alpha_1$	ACTGTGAGAGCAGAATGTCCAA	TTCCCGTGTCCACTCATACA	123
$GABA_AR\alpha_2$	CCAATGCACTTGGAGGACTT	CAGTCTTGAGCCATCTGGTG	147
$GABA_AR\alpha_3$	CAGTCTTGAGCCATCTGGTG	ACCCTTGGCTACTTCCACAGATTT	155
$GABA_{A}R \ \alpha_{4}$	TGGTCATGCTTGTCCTCTCA	ATGCCCAAGCAGGTCATACT	142
$GABA_AR\beta_2$	GTGGGGGTGATAATGCAGTC	AAGCTGAGAGACAGCCTTGG	130
$GABA_AR\beta_3$	GTCACTGGCGTGGAGAGAAT	CCACCTGAAGCTCAGTGAAAG	120
$GABA_AR\gamma_4$	GGAACGATGGAAAAGTGCTC	CCTCCCGAGGGTATCCATAA	132
β-actin	AACCGGACTGTTTCCAACAC	CACCTTCACCGTTCCAGTTT	143
GAPDH	CCAACCCCTAACGTGTCTGT	TGTCATCGTACTTGGCTGGT	68
HPRT 1	ACGACCTGGATTTGTTCTGC	TTTCTCGTGCCAGTCTCTCC	108

**Table 2.4.** Real-time RT-PCR primers and amplicon sizes for various GABA<sub>A</sub> receptor subunits and the house-keeping genes  $\beta$ -actin (GenBank accession number XM\_002192780), glycerinaldehyd-3-phosphat-dehydrogenase (GAPDH; GenBank accession number AF255390) and Hypoxanthine-guanine-phosphoribosyl-transferase 1 (HPRT 1) of the zebra finch. All GABA<sub>A</sub> receptor subunit-specific primers were designed from sequences cloned within our laboratory and sent to our collaborators in Germany where the RT-PCR reactions were performed.

#### 2.7.3 Real-time RT-PCR

Reverse transcriptase (RT) reactions were performed in duplicate with a non-RT control for each RNA sample. For qPCR, 12.5µl *Power* SYBR<sup>®</sup> Green PCR master mix (Applied Biosystems, USA) and 2µl of each RT reaction (1:5 dilution) were used. qPCRs were run on separate 96-well plates for each song nucleus with RT and non-RT reactions in a final volume of 25µl, and a final primer concentration of 100 nM. Assays were performed on an Mx300 5P<sup>TM</sup>QPCR System (Stratagene). Cycling conditions were as follows; 10 min *initial denaturation* (95 °C), 45 cycles at: 30 sec denaturation (95 °C), 1 min *annealing* (60 °C) and 30 sec *extension* (72 °C). A melting curve was generated at the end of every run to ensure product uniformity. Product sizes were confirmed with electrophoresis on 3.0% (w/v) agarose gels stained with ethidium bromide (section 2.4.4).

#### 2.7.4 Relative quantification of mRNA levels

The mean number of cycles of threshold (CT) determined for the reference genes (ref),  $\beta$ -actin, GAPDH and HPRT1 and primer pair efficiencies (E) were used to calculate normalised mRNA expression levels of GABA<sub>A</sub> receptor subunits following the equation for MNE (mean normalised expression) (Muller *et al.*, 2002 and Simon, 2003)

$$MNE = (E_{ref, mean})^{CT ref, mean} / (E_{GABAAR})^{CT GABAAR}$$

To visualise expression of GABA<sub>A</sub> receptor subunit mRNAs at different developmental stages during song learning, MNE levels were expressed in relation to the level at PHD 30 (100%), this was the earliest developmental stage analysed. Comparisons of relative expression levels were conducted using two-way analyses of variance (ANOVA) followed by LSMeans Tukey's post hoc tests. For considering the problem of multiple comparisons (seven genes per song control nucleus) the sequential Bonferroni correction procedure was used.

#### **2.8** Generation of $\gamma$ 4-subunit-specific antibodies

Although real-time RT-PCR and *in situ* hybridisation were useful tools in providing a qualitative and quantitative insight into expression levels and spatial

distribution of the GABA<sub>A</sub> receptor subunit mRNAs, it was important to consider if this corresponded at a protein level, with particular reference to the GABA<sub>A</sub> receptor  $\gamma$ 4 subunit. Two different strategies were employed in order to generate specific antibodies (as none were commercially available), the first involved synthesis of a protein antibody using fusion-protein technology and the second was generation of a peptide antibody which was out-sourced commercially.

#### 2.8.1 pMAL<sup>TM</sup> protein fusion and purification system

A fusion protein was generated using the pMAL<sup>TM</sup> protein fusion and purification system (New England Biolabs<sup>®</sup> Inc., USA). In this system a cloned sequence, encoding a region of the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit sequence was inserted into the pMAL<sup>TM</sup>-c2X vector in the same translational reading frame, downstream from the *malE* gene, which encodes maltose-binding protein (MBP); this results in the expression of a MBP-fusion protein. The fusion protein was then purified by amylose-affinity chromatography (specific for MBP) and the target protein could then be cleaved from the fusion protein by a specific protease (factor Xa).

#### 2.8.2 Design and construction of the fusion plasmid

A 120bp region of the target  $GABA_A$  receptor  $\gamma$ 4-subunit nucleotide sequence (GenBank accession number: AM086993) was carefully selected, adhering to the following criteria:

- 1) Sequence must have no similarity with other GABA<sub>A</sub> receptor subunits
- 2) Sequence must have no internal restriction sites that are of relevance
- 3) Sequence must not start with an arginine residue
- 4) Sequence must contain a stop codon (TAG/TGA/TAA)
- 5) Sequence must not show homology to the factor Xa cleavage site (ATC GAG GGA AGG ATT)
- Sequence must ligate into polylinker site in the vector, i.e. have appropriate restriction sites (the *Eco*R1 restriction site was selected as this ensured that there was minimal vector derived sequence remaining after cleavage with factor Xa)
- 7) Sequence must be in the same reading frame as the *malE* sequence

A region of the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit nucleotide sequence was identified in the large intracellular loop (between TM3 and TM4), as this is the part of the subunit sequence which shares the lowest homology with other GABA<sub>A</sub> receptor genes. Once the region of sequence was selected, specific primers were designed to amplify this section with RT-PCR. Some single-base alterations were engineered into the primer design to ensure that a translational stop codon (TAG – highlighted in grey) and the desired recognition sites for the *Eco*R1 restriction enzyme (underlined and modified bases highlighted in bold) were present in the correct positions. Primer sequences were: 5'-CTGGTG<u>GAATTC</u>AAGAAACCA CTG-3' and 5'-GGGC<u>GAATTC</u>ATCTAGTGGTT GAT-3' (both primers had 46% G+C and T<sub>m</sub> 60°C). Primers were subjected to a BLAST search to verify there was no homology to the other GABA<sub>A</sub> receptor subunit mRNAs.

#### 2.8.3 Amplification and cloning of the fusion protein

RT-PCR was performed with single stranded cDNA synthesised from adult male zebra finch brain as described in section 2.4.2. Cycling conditions were as follows; 5 min *initial denaturation* (94°C) followed by 30 cycles at: 1 min *denaturation* (94°C), 30 sec *annealing* (56°C) and a 30 sec *extension* (72°C). RT-PCR reactions were carried out in thin-walled microfuge tubes in a thermocycler (TC-3000; Techne). The RT-PCR product was run on a 2.5% (w/v) agarose gel and visualised as per section 2.4.4. The amplicon of expected size (120bp) was excised and purified as per section 2.4.5.

The DNA fragment and pMAL<sup>TM</sup>-c2X vector (New England Biolabs<sup>®</sup> Inc.) were then subjected to *Eco*R1 restriction digestions to create complementary 'sticky ends' for ligation. Reaction mixes were as follows:  $5\mu$ l purified RT-PCR product,  $1\mu$ l 10x buffer H, *Eco*RI (6U; Promega), made up to  $10\mu$ l with H<sub>2</sub>O which was incubated at 37°C for 90 min. For linearisation of the plasmid: 500ng vector,  $2\mu$ l buffer H, *Eco*RI (6U; Promega) made up to  $20\mu$ l with H<sub>2</sub>O. Both samples were digested at 37°C for 90 min. Digested DNA/plasmid was then immediately purified by use of the GFX<sup>TM</sup> PCR DNA and gel band purification kit (GE Healthcare) protocol as per section 2.4.5. The linearised pMAL<sup>TM</sup>-c2X vector (40ng) and digested RT-PCR product (1µl) were then combined in a ligation reaction with T4 DNA ligase (6U; Promega),  $2\mu$ l 10x buffer (as per section 2.5.1; Promega) made up to 20 $\mu$ l with H<sub>2</sub>O and ligated at 4°C overnight.

The vector containing the insert was transformed into chemically competent E. coli TB1 cells (section 2.5.2-2.5.5). However, unlike with previous cloning, there was no X-Gal or IPTG added to the LB agar plates as the manufacturers (New England Biolabs<sup>®</sup> Inc.) suggested they would cause loss of part of the fusion gene or dramatically reduce the expression levels of the gene due to the strength of the P<sub>tac</sub> promoter. Therefore, the plates were incubated overnight on standard LB agar plates supplemented with 0.1% (v/v) ampicillin ( $100\mu g/ml$ ; Sigma-Aldrich<sup>®</sup>) at 37°C and the resulting colonies picked with a sterile toothpick and stabbed onto a master LB agar plate (0.1% (v/v) ampicillin) and then onto an LB agar plate (0.1% (v/v) ampicillin) which was layered with 100µl IPTG (200mg/ml) and 40µl X-Gal (20mg/ml). These were incubated at 37°C overnight and any positive 'white' colonies on the X-Gal/IPTG plate were recovered from the corresponding patch on the master plate. Colonies were subcultured as per section 2.5.4 and then screened for the presence of inserts by minipreparation (section 2.5.5) and subsequent EcoR1 digestion (section 2.5.6). The digested plasmid clones were electrophoresed on a 2% (w/v) agarose gel (section 2.4.4) and any samples containing the insert of expected size were sent for automated sequencing (Eurofins MWG Operon, UK) with the MalE sequencing primer (5' GGTCGTCAGACTGTCGATGAAGCC 3'; New England Biolabs<sup>®</sup> Inc.), which generates sequence data from inserts within the polylinker region of the pMAL<sup>TM</sup>c2X vector. The sequence was verified for any inaccuracies against the original sequence design and GABA<sub>A</sub> receptor y4-subunit mRNA sequence. The successful plasmid containing the insert was stored at -20°C ready for use.

#### 2.8.4 Pilot experiment

It was first necessary to determine the behaviour of the MBP-fusion protein before any larger-scale experiments were undertaken.
#### 2.8.4.1 Transformation

A 10ng sample of suspended plasmid DNA (from the minipreparation containing the successful fusion plasmid) was added to  $50\mu$ l of competent *E. coli* TB1 cells and placed on ice for 2 min. Following a 30 sec,  $42^{\circ}$ C thermal shock, the mixture was placed back on ice for 2 min, where  $150\mu$ l room temperature SOC medium was added.  $100\mu$ l of the cells were spread on an LB agar plate (0.1% (v/v) ampicillin) and incubated overnight at  $37^{\circ}$ C.

#### 2.8.4.2 Subculturing

Two colonies were subcultured into 2ml LB medium (0.1% (v/v) ampicillin) and incubated at 37°C overnight at 200 rpm in an orbital shaker with good aeration. The following day, 80ml of rich broth (1% (w/v) bacto-tryptone; 0.5% (w/v) bacto-yeast extract; 0.2% (w/v) glucose; 90mM NaCl) was inoculated with 1ml of the overnight culture. Cells were grown at 37°C with agitation at 200 rpm to 2x  $10^8$  cells/ml (A<sub>600</sub> ~0.5); this took approximately 3 h, (1ml samples were taken every hour and the absorbance read against a blank taken immediately after inoculation and stored on ice). A 0.5ml sample was withdrawn and microcentrifuged at 20,000*g* for 2 min, the supernatant was discarded and the cells were resuspended in 50µl 2x sodium dodecyl sulfate polyacrylamide gel electrophoresis (SDS-PAGE) sample buffer (Laemmli; 4% (w/v) SDS, 20% (v/v) glycerol, 10% (v/v) 2-mercaptoethanol, 0.004% (w/v) bromophenol blue and 0.125 M Tris HCl, pH 6.8; Sigma-Aldrich<sup>®</sup>), vortexed to mix and stored at -20°C. These were the uninduced cells (sample 1).

#### 2.8.4.3 IPTG induction

IPTG was added to the remaining culture to a final concentration of 0.3mM and incubation was continued as before for 2 h. A 0.5ml sample was withdrawn and treated as above (Sample 2, induced cells). Remaining culture was divided between two 50ml centrifuge tubes and to each tube, 0.5ml protease inhibitor cocktail was added (215mg cocktail: 1ml DMSO: 5ml dH<sub>2</sub>O). Protease cocktail is a mixture of protease inhibitors with broad specificity for the inhibition of serine, cysteine, aspartic and metallo-proteases, and aminopeptidases; containing 4-(2-aminoethyl) benzenesulfonyl fluoride (AEBSF), pepstatin A, E-64, bestatin, and

sodium EDTA; Sigma-Aldrich<sup>®</sup>). Cells were harvested by centrifugation at 4000*g* for 10 min (AvantiTM J-30; Beckman Coulter). Supernatants were discarded and each pellet was resuspended in 2.5ml column buffer (1M Tris-HCl, pH 7.4; 0.5M EDTA; 11.7% (w/v) NaCl and 0.154% (w/v) DTT; all Sigma-Aldrich<sup>®</sup>). The two samples were pooled and 2ml protease cocktail was added, this was frozen overnight at -20°C.

#### 2.8.4.4 Protein extraction from cells

Cells were defrosted slowly in cold water and were transferred to a beaker sat in a bath of ice. The suspension was sonicated (Soniprep 150 ultrasonic disintegrator; MSE UK Ltd.) at an amplitude of 9 microns, to release cellular proteins, in 15 sec pulses for two minutes. Optimum sonication duration and amplitude for maximum protein release was assessed by the bicinchoninic acid (BCA) assay (section 2.8.7.3). Cells were transferred back to a 50ml centrifuge tube and centrifuged at 9000*g* for 20 min at 4°C. The supernatant was decanted (crude extract) and stored on ice and the pellet resuspended in 5ml column buffer (insoluble matter). 10µl samples were taken from both the crude extract and insoluble matter and vortexed with 10µl 2x SDS-PAGE sample buffer and stored at  $-20^{\circ}$ C (Samples 3 and 4 respectively).

200µl amylose resin (a composite of amylose/agarose beads in 20% (v/v) ethanol; New England Biolabs<sup>®</sup> Inc.) was added to a microfuge tube and centrifuged briefly, the supernatant was aspirated and the resin was resuspended in 1.5ml column buffer. 50µl of amylase-resin slurry was mixed with 50µl of crude extract and kept for 15 min on ice. The sample was centrifuged at 20,000*g* for 1 min, the supernatant was removed and the pellet was washed with 1ml column buffer. After a further 1 min centrifugation at 20,000*g*, the supernatant was aspirated and the resin resuspended in 50µl SDS-PAGE sample buffer (Sample 5; protein bound to amylose).

 $1\mu$ l of each sample was diluted 1:10 with SDS-PAGE sample buffer (in preparation for a Western blot) and  $10\mu$ l of each of the samples (in preparation for SDS-PAGE) were boiled for 5 min.

# 2.8.5 Discontinuous SDS-PAGE

SDS-PAGE enables the separation of an assortment of proteins within a single sample by their size and molecular weight. Samples are first mixed with sodium dodecyl sulfate (SDS). SDS is an anionic detergent which, when heated to 100°C in the presence of protein, denatures the protein into individual polypeptides and binds to them, conferring a negative charge to the polypeptide in proportion to its length. The buffer system is discontinuous, whereby a large pore polyacrylamide gel (stacking gel), which is about 2 pH units lower than the electrophoresis buffer, is cast over a small pore acrylamide gel (resolving gel). The stacking gel serves to concentrate the SDS-coated proteins into a small area ready to migrate into the resolving gel where they separate according to their size.

#### 2.8.5.1 Preparation of the resolving gel

Throughout the experiments vertical mini gels (10cm) were cast in the omniPAGE gel casting system (VS10CBS; Geneflow, UK) using 2mm spacers and a 2mm, 12-well comb.

10% (w/v) acrylamide resolving gels were routinely used as these have a separating resolution of ~20-75kDa. This was selected as the molecular weight of MBP was ~42.4kDa and the GABA<sub>A</sub> receptor  $\gamma$ 4 peptide was ~4kDa, so the expected size of the fusion protein was ~46kDa. It was a high enough resolution to distinguish a 4kDa difference in product size.

10% Resolving gel	2x10cm gels	1x10cm gel
dH <sub>2</sub> 0	6.3ml	3.15ml
30% (w/v) Stock acrylamide	5.0ml	2.5ml
4x Resolving tris solution pH8.8*	3.75ml	1.875ml
10% (w/v) Ammonium persulphate (APS)	150µl	75µl

**Table 2.5.** Composition of the 10% (w/v) resolving gel for SDS-PAGE. All components were added in the order stated.\* (1.5M Tris-HCl; 0.4% (w/v) SDS). As APS is an initiator for gel formation, it was added last.

The solution was made as per Table 2.5. The final addition was  $7.5\mu$ l or  $15\mu$ l of the polymerising agent *N*,*N*,*N*',*N*'-tetramethylethyenediamine (TEMED; Sigma-

Aldrich<sup>®</sup>), for one or two gels respectively. APS is a strong oxidising agent and works in conjunction with TEMED to catalyse the polymersisation of acrylamide. The mixture was carefully mixed to avoid creating bubbles and loaded between two glass plates of the casting module, leaving a 3cm gap from the top. A 1ml overlay of isopropanol was layered on the top of the resolving gel to create an even surface and minimise contact with the air. After 30 min polymerisation, the isopropanol was carefully decanted, the top of the gel rinsed with dH<sub>2</sub>O and blotted dry with filter paper.

Stacking gel				
dH <sub>2</sub> 0	4.2ml			
30% (w/v) Stock acrylamide	0.65ml			
4x Stacking tris solution pH6.8*	1.6ml			
10% (w/v) APS	67µl			

2.8.5.2 Preparation of the stacking gel

**Table 2.6.** Composition of the stacking gel (enough for two gels), detailing components, the APSwas always added last. \* (0.5M Tris-HCl; 0.4% (w/v) SDS; pH 6.8).

After addition of 6.7µl TEMED and gentle mixing, the stacking gel solution was added directly on top of the resolving gel, a comb was inserted and the gel left to polymerise for a further 30 min at room temperature. When the gel was completely set, the comb was removed and any un-polymerised material was removed from the wells by rinsing them with Tris-glycine electrophoresis buffer (5mM Tris-base; 50mM glycine, pH 8.3; 0.02% (w/v) SDS; all Sigma-Aldrich<sup>®</sup>). The gel was removed from the casting chamber and placed in to the electrophoresis tank where it was submerged entirely in Tris-glycine electrophoresis buffer. As aforementioned, samples, including the ladder and marker, were heated at 100°C for 5 min before loading, and a maximum of 25µl was loaded per well. Samples were run against a pre-stained broad range protein marker (2-212kDa) (0.1-0.2mg/ml of each protein in 70mM Tris-HCl pH 6.8, 33mM NaCl, 1mM Na<sub>2</sub> EDTA, 2% (w/v) SDS, 40mM DTT, 0.01% (w/v) bromophenol blue and 10% glycerol; New England Biolabs<sup>®</sup> Inc) and a MBP2 standard which is a protein encoded for by the pMAL-c2X vector, with a stop codon inserted into the XmnI site and a molecular weight of 42,482Da (6mg/ml

protein; 20mM Tris-HCl; 0.2 M NaCl; 1mM EDTA; 50% (v/v) Glycerol, pH 7.2; New England Biolabs<sup>®</sup> Inc.). The gel was run at 50V for the first 15 min (to allow the molecules to concentrate in the stacking gel) then at 100V for a further 50 min.

# 2.8.5.3 Staining of the protein gel

The gel was gently removed from the glass plates after the dye line had run the full length of the gel and was immediately washed twice (10 min each) with dH<sub>2</sub>O on a slowly rotating platform (50 rpm) to remove any traces of the electrophoresis buffer. The gel was then immersed in 5x volume of the protein stain, Coomassie blue (0.25% (w/v) coomassie brilliant blue R250; National Diagnostics, USA; 45% (v/v) methanol; 45% (v/v) dH<sub>2</sub>O and 10% (v/v) glacial acetic acid; Sigma-Aldrich<sup>®</sup>) and placed on a slowly rotating platform (50 rpm) for 1 h. The stain was decanted and the gel was de-stained (45% (v/v) methanol; 45% (v/v) dH<sub>2</sub>O and 10% (v/v) glacial acetic acid) for 6-8 h; the de-staining solution was changed 3-4 times during this period. The gel was visualised under white light and the image captured using Syngene<sup>TM</sup> software.

# 2.8.6 Western Blotting

# 2.8.6.1 Assembly of the blot sandwich

Nitrocellulose membrane ( $0.2\mu$ m, Optiran BA-S 83) and 2mm thick filter paper (both Whatman International Ltd., UK) were cut to the size of the gel and then pre-soaked, along with two fibre pads, in 1x continuous transfer buffer (CTB; 25mM Tris; 192mM glycine; 20% (v/v) methanol; pH 8.3) for 15 min. Using forceps throughout, the blot sandwich was assembled, with the gel facing the negative pole. The tank was part filled with pre-chilled transfer buffer along with a magnetic stirrer. The module containing the blot sandwich was then inserted, followed by a cooling pack. The electrophoresis tank was then filled with the remainder of the chilled CTB and the tank placed on ice. The blot was run at 75V for 100 min.

# 2.8.6.2 Blocking of the membrane (reduction of non-specific binding)

Unbound sites in the nitrocellulose membrane filter were saturated by incubating the blot in 50ml 3% (w/v) non-fat dry milk (Bio-Rad Laboratories) in 1x Tris-

buffered saline (TBS; 50mM Tris-HCl pH 7.4; 0.9% (w/v) NaCl) for 1.5 h at 50 rpm at room temperature.

# 2.8.6.3 Immunoprobing with primary antibody

 $5\mu$ l of the primary antibody (anti-MBP antiserum (rabbit); New England Biolabs<sup>®</sup> Inc.) was added directly into the blocking solution (1:10,000 dilution). This was left at room temperature for 30 min to 1 h at 50 rpm to allow the antibody to mix, and then overnight at 4°C to bind. Any unbound primary antibody was removed by washing the membrane in 1x TBS/0.1% (v/v) Tween<sup>®</sup>20 (Sigma-Aldrich<sup>®</sup>) for 6x 10 min washes.

# 2.8.6.4 Immunoprobing with secondary antibody

 $25\mu$ l of the secondary antibody (goat anti-rabbit IgG, AP-linked; New England Biolabs<sup>®</sup> Inc.) was added into 50ml 3% (w/v) non-fat dry milk in 1x TBS, at a 1:2000 dilution and gently agitated (50 rpm) for 2 h at room temperature. Any unbound secondary antibody was washed off as described in section 2.8.6.3.

# 2.8.6.5. Immunodetection of proteins

Identification of proteins on the membrane was completed using the BCIP (5bromo-4-cholo-3-indoyl-phosphate)/NBT (nitro blue tetrazolium) chromogenic assay. This is a three component system whereby combination of BCIP, NBT and an alkaline buffer result in a substrate solution for alkaline phosphatase. When incubated with this enzyme, an insoluble NBT diformazan product is formed which is identified as a purple precipitate, thus allowing colorimetric detection of alkaline phosphate-labelled molecules. Firstly, the membrane was rinsed in 50ml alkaline phosphatase buffer (100mM Tris HCl, pH 9; 1mM MgCl<sub>2</sub>; 150mM NaCl) to equilibrate the pH of the blot.  $330\mu$ l NBT (50mg/ml; 70% (v/v) dimethylformamide; Promega) and  $165\mu$ l BCIP (50mg/ml; 100% (v/v) dimethylformamide; Promega) were added to a further 50ml of the alkaline phosphatase buffer, mixed and added to the membrane, this was left in the dark at room temperature. Colour development occurred in 3-6 h. The membrane was rinsed with dH<sub>2</sub>O to stop the colour reaction and was blotted dry with filter paper. The membrane was then visualised under white light and the image captured using Syngene<sup>™</sup> software.

# 2.8.7 Affinity chromatography

The fusion protein was purified from the cell lysate by affinity chromatography. A chromatographic method of separating individual proteins on the basis of reversible biological interactions between a protein (in this case MPB) and a specific ligand (amylose) which is immobilised within a solid matrix. Using a biorecognition technique such as this enables high specificity and recovery of target protein.

# 2.8.7.1 Subculturing

12 individual colonies were subcultured into 12ml rich broth and incubated at  $37^{\circ}$ C overnight at 200 rpm on an orbital shaker. Following this, 11 of pre-warmed rich broth was inoculated with all the overnight culture. Cells were grown at  $37^{\circ}$ C, 200 rpm for a further 3 h. IPTG was added to a final concentration of 0.375mM to induce protein expression within the cells and incubation was continued for 3.5 h. Culture was divided equally between two 750ml centrifuge tubes and 2ml protease inhibitor cocktail was added to each tube. Cells were harvested by centrifugation at 4000*g* for 20 min at 4°C (Allegra<sup>®</sup> X-15R; Beckman Coulter). Each pellet was resuspended in 24ml ice-cold column buffer and the two samples were pooled. 2ml protease inhibitor cocktail was added and cells were frozen overnight at -20°C

# 2.8.7.2 Protein extraction

Cells were thawed in cold water (30 min) and transferred to a sterile beaker. Cells were lysed by sonication in 15 sec pulses for 3 min. Cells were then transferred to tubes and centrifuged at 9000*g* for 30 min at  $4^{\circ}$ C (Avanti<sup>TM</sup> J-30I; Beckman Coulter). The supernatant was aspirated and kept on ice and the pellet discarded. A bicinchoninic acid (BCA) assay was completed to estimate the concentration of extracted protein in the crude extract.

#### 2.8.7.3 Protein estimation assay

The bicinchoninic acid (BCA) assay was employed to estimate the protein concentration after sonication. The assay is based on the principle that, under alkaline conditions proteins form complexes with  $Cu^{2+}$  which is then reduced to  $Cu^+$ . BCA itself is a highly specific chromogenic agent which forms a purple-blue complex with the reduced  $Cu^+$ , thus the proteins reduce the  $Cu^{2+}$  in a concentration-dependant manner and the more  $Cu^{2+}$  is reduced, the more  $Cu^+$  complexes with the BCA and the greater the colour change. Absorbance was read at 562nm and a standard curve created from bovine serum albumin (BSA) resuspended in TBS calibration standards of a known concentration. By taking readings from this, the deduced protein concentration in the original sample was obtained.

The assay was completed in a 96-well plate (Starstedt, Germany).  $25\mu$ l of each BSA standard (in 1x TBS) were added to the plate in triplicate (concentrations of  $0\mu$ g/ml,  $200\mu$ g/ml,  $400\mu$ g/ml,  $600\mu$ g/ml,  $800\mu$ g/ml and  $1000\mu$ g/ml). The crude extract was diluted 2x, 4x, 8x, 32x and 64x in 1x TBS,  $25\mu$ l of each of these was added to the plate in triplicate. In a separate beaker, 8ml of bicinchoninic acid was added to 160µl copper (II) sulphate solution (4% (w/v) copper (II) sulphate pentahydrate; both Sigma-Aldrich<sup>®</sup>) and mixed. Immediately, 200µl of this was added to each well containing either standards or samples. The plate was covered and incubated at 37°C for 30 min. Absorbance was read at 570nm (Expert 96; ASYS Hitech GmbH, Austria) and analysed with DigiRead software; Liberté, South Africa). A standard curve of absorbance vs. µg/ml protein was prepared and the equation of the curve was rearranged to calculate the concentration of protein in the unknown sample, taking into account the dilution factors (Fig. 6.7, Chapter 6). The crude extract was then diluted with ice-cold column buffer to approximately 2.5mg/ml ready for the affinity purification.

# 2.8.7.4 Affinity chromatography

A  $30\text{cm}^3$  column (20 x 1.5cm; Biosepra, USA) was packed with 8ml amylose resin slurry. Care was taken to ensure the resin was kept wet at all times and no air bubbles became trapped (this causes the resin to split and decreases binding efficiency). When the column was secure, it was attached to a peristaltic pump

and ddH<sub>2</sub>O was run through the column at a rate of 2ml/min to ensure the equipment worked properly and there were no leaks. When satisfied, the ddH<sub>2</sub>O was exchanged for column buffer; 8 column washes were passed through (64ml total). Approximately half-way through the first set of column washes, the pH of the flow-through was measured; if it was similar to that of the original column buffer then the column was deemed calibrated. Following the washes, the crude extract was added, this was pumped through the column at a rate of 1ml/min. A 0.5ml sample of the crude extract was withdrawn before purification and stored at  $-20^{\circ}$ C for diagnostic purposes later on. Flow-though from the crude extract contained unbound proteins, so this was collected and stored at  $-20^{\circ}$ C in case the target protein did not bind properly. A 0.5ml sample of this was taken and stored at -20°C. The column was then washed with 12 volumes of column buffer (94ml total) at 1ml/min. Fusion protein (bound to the amylose) was then eluted with 30ml elution buffer (column buffer +10mM maltose) and collected in 20x 1.5ml fractions (2110 fraction collector; Bio-Rad Laboratories). Which were stored on ice ready to be analysed.

A BCA assay was performed as in section 2.8.7.3, however it was noted that the maltose in the elution buffer interfered with the assay giving erroneous results. Therefore BSA standards were made up in undiluted, 10- and 20-fold diluted elution buffer and analysed. Results indicated that the 10- and the 20-fold dilutions of the elution buffer gave identical results (thus at these concentrations maltose no longer interefered with the assay) and so the BSA standards in the 10-fold diluted elution buffer could be used (10mM maltose final concentration). This was also fortuitous as all eluted samples were diluted 10-fold before analysis. Samples were loaded on a SDS-PAGE gel and subjected to western blot analysis to determine the purity of the eluted fusion protein.

# **2.8.7.5** *Dialysis*

In order to remove the residual maltose from the eluent, dialysis was employed. Utilising a cellulose semi-permeable membrane where larger protein molecules are retained inside the bag and the smaller maltose molecules diffuse out, down the osmotic gradient. Following western blot analysis, only eluent fractions giving a single product of the expected size were combined and dialysed. Handling only with gloves, a 12cm length of dialysis tubing (MWCO 12-14 kDa; Medicall International Ltd., USA) was cut and placed into a beaker with dH<sub>2</sub>O. This was boiled, the dH<sub>2</sub>O replaced, and boiled again (to prepare the tubing). dH<sub>2</sub>O was replaced again and the tubing was left at 4°C to cool. Once suitably chilled, dH<sub>2</sub>O was discarded and one end of the tube clamped. Sample was poured into the bag and the other end was firmly sealed. The dialysis tube was then submerged in 500ml cold Tris buffer (23mM, pH 7.5) and kept at 4°C. Buffer was replaced every 20-30 min for 4 h. After this time the buffer was discarded and dialysed protein was poured into a sterile 15ml tube. The sample was analysed by BCA assay (section 2.8.7.3) to determine the concentration and the purity was confirmed by SDS-PAGE. The fusion construct was then ready for commercial production into an antibody.

# 2.8.8 Generation of a peptide anti-GABA<sub>A</sub> receptor $\gamma$ 4-subunit antibody

In addition to the fusion protein being produced, a peptide antibody was also generated. In this case, a region of nucleotide sequence encoding a portion of the large variable intracellular loop between TM3 and TM4 of the GABA<sub>A</sub> receptor  $\gamma$ 4 subunit gene was selected and all the subsequent production was outsourced commercially to Fisher Scientific. The sequence was homologous between the zebra finch and chicken to enable production of an antibody which could target either species. Due to mis-matches between the species, only an 11AA sequence could be found which was suitable (Fig. 2.1).

#### (a) Nucleotide alignment

	<u>TM 3</u>
Zebra Finch $\gamma4$	GCTCTCATGGAGTACGCCACGCTCAACTACCTGGTGGGAAACAAGAAACCACTGGAGCAC
Chicken $\gamma 4$	GCTCTCATGGAGTATGCTACACTCAACTACCTGGTGGGAAACAAGAAGCCACTGGAGCAC
	************* ** ** ** ****************
Zebra Finch γ4	AGCCACAGGAG <mark>AGCCAGGCTGCCACCTGCCGGTGCTCAGGTGATGCCA</mark> ACCTTCACCACC
Chicken $\gamma 4$	AGC <mark>AG</mark> CAGGAAAGCCAGACTGCCACCCGCTGGTGCACAGGTGATGCCATCCTTTACTGCT
	*** ***** ****** ******* ** ****** *****
Zebra Finch γ4	ATCAACATCAACCACATCATGCACTGGCCCCCGGAGAACCTTCACCACCATCAA
Chicken $\gamma 4$	ATCAACATTAACATCAACAACATCATGCACTGGCCTCCAGAGATAGAGGAGGATGAA
	****** **** ***************************
Zebra Finch $\gamma4$	CATCAACCACATCATGCACTGGCCCCCGGAGGGGAAGGAGTGTGAGAGG
Chicken $\gamma 4$	GAT-GACGAC <mark>CCTGGCTCTCC</mark> GT <mark>G</mark> CC <mark>T</mark> GGAAGGAAAGGAATGTGAGAGG
	** ** ** ** ** ** *********

#### (b) Peptide alignment



**Figure 2.1**. Nucleotide (a) and peptide (b) alignments of partial zebra finch GABA<sub>A</sub> receptor  $\gamma$ 4-subunit intracellular loop sequence (GenBank accession number: AM086993) between transmembrane domains (TM) 3 and 4 with the analogous chicken sequence (GenBank accession number X73533). (-) denotes gaps in the sequence which have been introduced to maintain the alignment and (\*) indicates perfect alignment. (.) and (:) below the peptide sequence indicate partial codon homology despite a difference of amino acids. Highlighted in grey are single-base mis-matches between the nucleotide sequences of the two species. Highlighted in blue are those single base mis-matches that result in a change to the peptide sequence. In order to create a sequence which was specific to both species the sequence shown in orange was selected as the most appropriate.

The selected sequence was sent to Fisher Scientific and the BioPerformancePlus<sup>TM</sup> antibody production plan used. Initially, the sequence was analysed to verify it was suitable. The check included domain exclusion analysis, hydrophobicity and antigenicity analysis. The peptide was then synthesised and purified. The purified peptide was coupled to a carrier protein and this conjugate was used to immunise the rabbit.

# 2.9 RNA interference (RNAi)

In order to verify the potential functional role of  $\gamma$ 4-subunit-containing GABA<sub>A</sub> receptors in learning and memory in the song system of the male zebra finch, RNAi was employed to silence the expression of this gene. Constructs designed for this could be injected into specific nuclei of the song system and any effect on song acquisition or production could then be observed *in vivo*. This method involves introducing exogenous short-hairpin (sh)RNA into a cellular system whereby it enters the natural RNAi pathway within the cell and RNA expression is suppressed. A more detailed introduction to the technique is included in Chapter 6.

# 2.9.1 Short-hairpin (sh)RNA construct design

The system utilised in this study was the p*Silencer*<sup>TM</sup> 4.1-CMV neo vector kit (New England Biolabs<sup>®</sup> Inc.), which enables the ligation of a shRNA construct into an expression vector. The plasmid offers a powerful promoter in the form of a modified cytomegalovirus (CMV) promoter, which enables constitutive expression of a cloned shRNA template, and a modified simian virus-40 (SV40) polyadenylation signal which terminates transcription. The vector also encodes an ampicillin and neomycin resistance gene to enable antibiotic screening (for full vector map, refer to Appendix).

Prior to silencing the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit gene in the zebra finch *in vivo*, it was necessary to determine that the RNAi reactions would work efficiently. Therefore constructs were initially transfected into primary neurons and the degree of corresponding mRNA 'knock-down' was assayed by real-time RT-PCR. As zebra finches were not readily accessible and no cell lines expressing the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit gene were available, the work was completed with one-day-old chicks (Gallus gallus domesticus), which, phylogenetically, are very similar. Thus three sets of 55bp-constructs were designed, the first to silence the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit gene in the chicken, the second set to silence the same gene in the zebra finch and the third set as controls (Fig. 2.2). It was critical to include positive and negative control constructs. Non-targeting shRNA is a useful negative control that activates RISC and the RNAi pathway, but does not target any chicken/zebra finch genes. This allowed for examination of any potential effects of shRNA transduction on gene expression. Cells transfected with the non-target shRNA also provide a useful reference for interpretation of knockdown. A positive control served to measure the transduction efficiency and aids in optimising shRNA delivery.

Initially, 21-base mRNA targets were selected within the GABA<sub>A</sub> receptor  $\gamma$ 4subunit nucleotide sequence: in regions encoding the extracellular amino-terminus and large intracellular loop between TM3 and TM4 of the chicken (GenBank accession number X73533) and the zebra finch (GenBank accession number AM086993). These regions were selected as they exhibit the least similarity to other GABA<sub>A</sub> receptor subunit sequences (i.e. to avoid cross-silencing). siRNA targets were chosen by use of a target finder on the Ambion<sup>TM</sup> Website (www.ambion.com/techlib/misc/siRNA\_finder.html), which selects suitable targets from an mRNA sequence based on guidelines first specified by Tuschl and colleagues (Elbashir *et al.*, 2001a, b); the guidelines are only available at www.rockefeller.edu/labh eads/tuschl/sirna.htm. In addition to this, each 21-base sequence fulfilled the recommended criteria (Chalk *et al.*, 2004; Reynolds *et al.*, 2004), namely:

1. At least 100-200bp downstream of the AUG codon and upstream of termination codon

- 2. G+C content between 20% and 50%
- 3. No more than 3 consecutive bases (especially G and C)
- 4. Asymmetric (A/T at the 5'end and G/C at the 3' of the sense strand), if possible
- 5. 2 nt overhang

The 21-base target then served as the basis for the design of the 55bp construct (Fig. 2.2). Specifically, a 19-base sense stem is joined to a 21-base antisense version of this stem by a non-coding loop sequence (9 bases) that had been previously proved successful (Yu *et al.*, 2002). At each end of the construct there were single-stranded overhangs encoding the *Bam*HI and *Hind*III restriction sites for ligation into the p*Silencer*<sup>TM</sup> 4.1-CMV neo vector.

Driginal target				
	mRNA Sequence (se	nse) — AACAT	CAACCACATCATGCAC -	Intracellular loop
Top strand	<u>oligonucleotide</u>			
BamHI	Sense (19nt)	Loop	Antisense (21nt)	HindIII
5' GATCC CA	TCAACCACATCATGCAC	TTCAAGAGA	GTGCATGATGTGGTTGATG	<u>A</u> 3'
<u>Bottom stra</u>	<u>nd oligonucleotide</u>			
HindIII	Sense (21nt)	Loop	Antisense (19nt)	BamHI
5' AGCTT <u>AA</u>	CATCAACCACATCATGCAC	TCTCTTGAA	GTGCATGATGTGGTTGA	TG G 3'
ebra finch/C	hick (GABA <sub>A</sub> R $\gamma$ 4 ge	ene):		Amine terminus
nginai target	IIIRINA Sequence (se	nse) – Attga	GGCTGAATGTCTGCTC -	Amino-terminus_
<u>Top strand</u>	<u>oligonucleotide</u>			
BamHI	Sense (19nt)	Loop	Antisense (21nt)	HindIII
5' GATCC TG.	AGGCTGAATGTCTGCTC	TTCAAGAGA	GAGCAGACATTCAGCCTCA	<u>AT</u> A 3'
Bottom stra	nd oligonucleotide			
HindIII	Sense (21nt)	Loop	Antisense (19nt)	BamHI
5' AGCTT <u>AT</u>	IGAGGCTGAATGTCTGCTC	TCTCTTGAA	GAGCAGACATTCAGCCT	CA G 3'
Cebra finch/cl Driginal target Ton strand	h <b>ick (GABA<sub>A</sub> R γ4 ge</b> mRNA Sequence (se	ne): nse) — TTTCC	CAATGGACACTCACTC -	Amino-terminus
<b>Zebra finch/cl</b> Driginal target <u>Top strand</u>	hick (GABA <sub>A</sub> R γ4 ge mRNA Sequence (se oligonucleotide	ne): nse) – TTTCC	CAATGGACACTCACTC -	Amino-terminus
ebra finch/cl Driginal target Top strand BamHI	hick (GABA <sub>A</sub> R γ4 ge mRNA Sequence (se <u>oligonucleotide</u> Sense (19nt)	ne): nse) – TTTCC Loop	CAATGGACACTCACTC – Antisense (21nt)	Amino-terminus HindIII
ebra finch/cl Driginal target <u>Top strand</u> BamHI 5', GATCC TCC	hick (GABA <sub>A</sub> R γ4 ge mRNA Sequence (se oligonucleotide Sense (19nt) CCAATGGACACTCACTC	ne): nse) — TTTCC Loop TTCAAGAGA	CAATGGACACTCACTC – Antisense (21nt) GAGTGAGTGTCCATTGGGA	Amino-terminus HindIII AA A 3'
Zebra finch/cl         Driginal target         Top strand         BamHI         5'         GATCC         Bottom stra	hick (GABA <sub>A</sub> R γ4 ge mRNA Sequence (se oligonucleotide Sense (19nt) CCAATGGACACTCACTC nd oligonucleotide	ne): nse) — TTTCC Loop TTCAAGAGA	CAATGGACACTCACTC – Antisense (21nt) GAGTGAGTGTCCATTGGGA	Amino-terminus HindIII AAA A 3'
Zebra finch/cl         Driginal target         Top strand         BamHI         57         GATCC         Bottom stra         HindIII	hick (GABA <sub>A</sub> R γ4 ge mRNA Sequence (se <u>oligonucleotide</u> Sense (19nt) CCAATGGACACTCACTC <u>nd oligonucleotide</u> Sense (21nt)	ne): nse) — TTTCC Loop TTCAAGAGA	CAATGGACACTCACTC – Antisense (21nt) GAGTGAGTGTCCATTGGGA Antisense (19nt)	Amino-terminus HindIII AA A 3' BamHI
ebra finch/cl priginal target <u>Top strand</u> BamHI <sup>(7)</sup> GATCC TCC <u>Bottom stra</u> HindIII <sup>(7)</sup> AGCTT <u>T</u>	hick (GABA <sub>A</sub> R γ4 ge mRNA Sequence (se oligonucleotide Sense (19nt) CCAATGGACACTCACTC nd oligonucleotide Sense (21nt) FCCCAATGGACACTCACTC	ne): nse) – TTTCC Loop TTCAAGAGA Loop	CAATGGACACTCACTC – Antisense (21nt) GAGTGAGTGTCCATTGGGA Antisense (19nt) GAGTGAGTGTCCATTGG	Amino-terminus HindIII AA A 3' BamHI GA G 3'
Tebra finch/cl         Driginal target         Top strand         BamHI         5'         GATCC         Bottom stra         HindIII         5'	hick (GABA <sub>A</sub> R γ4 ge mRNA Sequence (se oligonucleotide Sense (19nt) CCAATGGACACTCACTC nd oligonucleotide Sense (21nt) FCCCAATGGACACTCACTC	ne): nse) – tttcc Loop ttcaagaga Loop tctcttgaa	CAATGGACACTCACTC – Antisense (21nt) GAGTGAGTGTCCATTGGGA Antisense (19nt) GAGTGAGTGTCCATTGG	Amino-terminus HindIII AAA A 3' BamHI GA G 3'
Zebra finch/cl Driginal target Top strand BamHI 5' GATCC TCC Bottom stra HindIII 5' AGCTT TT	hick (GABA <sub>A</sub> R γ4 ge mRNA Sequence (se oligonucleotide Sense (19nt) CCAATGGACACTCACTC nd oligonucleotide Sense (21nt) FCCCAATGGACACTCACTC	ne): nse) – TTTCC Loop TTCAAGAGA Loop TCTCTTGAA	CAATGGACACTCACTC – Antisense (21nt) GAGTGAGTGTCCATTGGGA Antisense (19nt) GAGTGAGTGTCCATTGG	Amino-terminus HindIII AAA A 3' BamHI GA G 3'
Sebra finch/cl         Driginal target         Top strand         BamHI         5' GATCC TCC         Bottom stra         HindIII         5' AGCTT TT	hick (GABA <sub>A</sub> R γ4 ge mRNA Sequence (se oligonucleotide Sense (19nt) CCAATGGACACTCACTC nd oligonucleotide Sense (21nt) FCCCAATGGACACTCACTC	ne): nse) – tttcc Loop ttcaagaga Loop tctcttgaa	CAATGGACACTCACTC – Antisense (21nt) GAGTGAGTGTCCATTGGGA Antisense (19nt) GAGTGAGTGTCCATTGG	Amino-terminus HindIII AA A 3' BamHI GA G 3'
ebra finch/cl Driginal target <u>Top strand</u> BamHI 5' GATCC TCC <u>Bottom stra</u> HindIII 5' AGCTT TT Chick (GABA,	hick (GABA <sub>A</sub> R γ4 ge mRNA Sequence (se oligonucleotide Sense (19nt) CCAATGGACACTCACTC nd oligonucleotide Sense (21nt) FCCCAATGGACACTCACTC	ne): nse) - TTTCC Loop TTCAAGAGA Loop TCTCTTGAA	CAATGGACACTCACTC – Antisense (21nt) GAGTGAGTGTCCATTGGGA Antisense (19nt) GAGTGAGTGTCCATTGG	Amino-terminus HindIII AA A 3' BamHI GA G 3'
Zebra finch/cl Driginal target <u>Top strand</u> BamHI 5' GATCC TCC <u>Bottom stra</u> HindIII 5' AGCTT <u>TT</u> Chick (GABA, Driginal target	hick (GABA <sub>A</sub> R γ4 ge mRNA Sequence (se oligonucleotide Sense (19nt) CCAATGGACACTCACTC nd oligonucleotide Sense (21nt) FCCCAATGGACACTCACTC	ne): nse) – tttcc Loop ttcaagaga Loop tctcttgaa	CAATGGACACTCACTC – Antisense (21nt) GAGTGAGTGTCCATTGGGA Antisense (19nt) GAGTGAGTGTCCATTGG	Amino-terminus HindIII AA A 3' BamHI GA G 3' Intracellular loop
Zebra finch/cl Driginal target <u>Top strand</u> BamHI 5' GATCC TCC <u>Bottom stra</u> HindIII 5' AGCTT <u>T</u> Chick (GABA, Driginal target <u>Top strand</u>	hick (GABA <sub>A</sub> R γ4 ge mRNA Sequence (se oligonucleotide Sense (19nt) CCAATGGACACTCACTC nd oligonucleotide Sense (21nt) FCCCAATGGACACTCACTC	ne): nse) – TTTCC Loop TTCAAGAGA Loop TCTCTTGAA NSe) – AACAA	CAATGGACACTCACTC – Antisense (21nt) GAGTGAGTGTCCATTGGGA Antisense (19nt) GAGTGAGTGTCCATTGG	Amino-terminus HindIII AA A 3' BamHI GA G 3' Intracellular loop
Zebra finch/cl Driginal target <u>Top strand</u> BamHI 5' GATCC TCC <u>Bottom stra</u> HindIII 5' AGCTT <u>TT</u> Chick (GABA, Driginal target <u>Top strand</u> BamHI	hick (GABA <sub>A</sub> R γ4 ge mRNA Sequence (se oligonucleotide Sense (19nt) CCAATGGACACTCACTC nd oligonucleotide Sense (21nt) FCCCAATGGACACTCACTC	ne): nse) – TTTCC Loop TTCAAGAGA Loop TCTCTTGAA Nse) – AACAA	CAATGGACACTCACTC - Antisense (21nt) GAGTGAGTGTCCATTGGGA Antisense (19nt) GAGTGAGTGTCCATTGG GAAGCCACTGGAGCAC - Antisense (21nt)	Amino-terminus HindIII AA A 3' BamHI GA G 3' Intracellular loop HindIII

	HindIII	Sense (21nt)	Loop	Antisense (19nt)	Bam	HI
5'	AGCTT	AACAAGAAGCCACTGGAGCAC	TCTCTTGAA	GTGCTCCAGTGGCTTCTTG	G	3'

# Chick (GAPDH gene): Original target mRNA Sequence (sense) – адатодосадатосадотост – Intracellular loop

	<u>Top str</u>	and oligonucleotide			
	BamHI	Sense (19nt)	Loop	Antisense (21nt)	HindIII
5'	GATCC	ATGGGCAGATGCAGGTGCT	TTCAAGAGA	AGCACCTGCATCTGCCCAT	<u>гт</u> а 3'
	<u>Bottom</u>	strand oligonucleotide			
	HindIII	Sense (21nt)	Loop	Antisense (19nt)	<u>Bam</u> HI
5'	AGCTT	AAATGGGCAGATGCAGGTGC	T TCTCTTGAA	AGCACCTGCATCTGCCC	AT G 3'

#### Chick (non-coding negative control):

Top strand oligonucleotide

	BamHI	Sense (18nt)	Loop	Antisense (21nt)	HindIII
5'	GATCC	ACTACCGTTGTTAAGGTG	TTCAAGAGA	CACCTATAACAACGGTAGT	<u>тт</u> <u>а</u> 3'
	<u>Bottom</u>	strand oligonucleotide			
	HindIII	Sense (21nt)	Loop	Antisense (18nt)	<b>Bam</b> HI
5'	AGCTT	<u>AA</u> ACTCAAGTTGTTATAGGTC	G TCTCTTGAA	CACCTTAACAACGGTAG	т G З'

**Figure 2.2.** The 55bp shRNA constructs designed for ligation into the p*Silencer*<sup>TM</sup> 4.1-CMV neo vector. Several constructs were designed to target the chicken/zebra finch GABA<sub>A</sub> receptor  $\gamma$ 4-subunit gene to ensure selection of the optimal sequence. Negative control sequence was modified from Rumi *et al.* (2006) and the positive-control chick glyceraldehyde-3-phosphate dehydrogenase (GAPDH) construct was based on the sequence (GenBank accession number NM\_204305). The first two bases of the oligonucleotide sequence that are discarded in the sense strand when using a top strand oligonucleotide template, but they are included in the antisense strand where they form the 2-nucleotide overhang (underlined). The non-coding loop sequence is highlighted in yellow, *Bam*HI and *Hind*III restriction sites were included for ligation into the p*Silencer*<sup>TM</sup> 4.1 CMV-neo vector. Top and bottom strands were annealed together to form a double stranded construct which was ligated into the p*Silencer*<sup>TM</sup> 4.1-CMV neo vector.

# 2.9.2 Ligation of shRNAs into pSilencer <sup>™</sup>4.1-CMV neo vector

Single-stranded shRNA constructs (Fig. 2.2) were synthesized by Eurofins MWG operon (Germany) and upon arrival were diluted to  $1\mu g/\mu l$  in H<sub>2</sub>O. The annealing mix was assembled as follows: sense oligonucleotide (top strand;  $2\mu g$ ); antisense oligonucleotide (bottom strand;  $2\mu g$ ) and  $46\mu l$  1x DNA annealing solution (Ambion<sup>TM</sup>). The mixture was heated for 3 min at 90°C and then incubated for 1 h

at 37°C. 5µl of the annealed shRNA construct was added to  $45\mu$ l H<sub>2</sub>O to a final concentration of 8ng/µl.

Annealed oligonucleotides were ligated into the vector in the following mix: annealed shRNA insert (8ng), 1µl 10x T4 DNA ligase buffer (Promega), 0.1µg linearised p*Silencer*<sup>TM</sup> 4.1-CMV neo vector (Ambion<sup>TM</sup>), T4 DNA ligase (5U; Promega) and 6µl H<sub>2</sub>O. The ligation reaction was left overnight at 4°C. XL1-Blue cells were then transformed with the ligation reaction (section 2.8.4.1) and shRNA-containing plasmids were extracted as per sections 2.5.4-2.5.5. Due to problems with the efficiency of a double restriction digest with *Hind*III and *Bam*HI, insert-containing plasmids were sent for automated DNA sequencing immediately after the minipreparation, to confirm presence of an insert with no prior diagnostic digest. Suitable constructs were 'bulked up' with a plasmid maxi kit (Qiagen<sup>®</sup>), which is a replica of the minipreparation kit used in section 2.5.5 only on a larger scale. Yield was determined by UV photospectrometry as per section 2.2.2; plasmids were stored at -20°C ready for use.

# 2.9.3 Primary neuronal culture

# 2.9.3.1 Animals

Fertilised chicken eggs (Leghorn) were obtained from Henry Stewart & Co. Ltd., Lincolnshire, UK. Eggs were initially incubated at 37°C at 20% relative humidity (INCA 50 incubator; Fiem, Italy). After two days the humidity was increased to 50-60% where it was maintained until embryonic day (E) 20, at this point the humidity was increased to 70-80% to encourage hatching. Eggs were subjected to a 12:12 h light: dark cycle and were constantly turned to avoid the embryo 'sticking' to the side of the shell, affecting development. Eggs were candled at regular intervals to ensure healthy development and any embryos showing no signs of growth were removed from the incubator.

# 2.9.3.2 Establishment of primary neuronal culture

Once hatched, at one day old, chicks were sacrificed by dislocation of the neck, their brains quickly excised and the cerebellum dissected out and stored in cold 1x Hank's balanced salt solution (HBSS; 400mg/l KCl; 60mg/l KH<sub>2</sub>PO<sub>4</sub>; 8g/l

NaCl; 350mg/l NaHCO<sub>3</sub>; 90mg/l NaH<sub>2</sub>PO•7H<sub>2</sub>O; 1g/l glucose; 20mg/l phenol red; Lonza, Switzerland). Cerebellums were minced with a McIllwain automatic tissue chopper to 1mm<sup>3</sup> pieces (Vibratome<sup>TM</sup>, USA) and then trypsinised with 0.04% (v/v) trypsin (Sigma-Aldrich<sup>®</sup>) in 5ml HBSS at 37°C for 30 min with gentle agitation. The reaction was terminated by the addition of 20ml growth medium (Neurobasal medium supplemented with 20% (v/v) fetal bovine serum; FBS and 1% (v/v) penicillin/streptomycin and 0.5mM L-glutamine; all Lonza). Tissue was triturated by pipetting up and down ~10 times with a 10ml pipette and poured through a 70µl mesh into a sterile 50ml tube. A further 10ml medium was washed through the mesh (2x 5ml rinses) and the mesh discarded. Cells were left to settle for a few minutes between each rinse of the mesh. Filtrate was centrifuged at 200*g* for 5 min at 4°C (5702R; Eppendorf, UK) and the supernatant aspirated and discarded. Cells were resuspended in 5ml fresh growth medium (as above) and were ready for seeding; 20µl of cells were removed and reserved for counting.

 $20\mu$ l of cell suspension was aliquotted into a 0.5ml sterile microfuge tube with  $20\mu$ l Trypan blue (0.4% w/v in 0.81% (w/v) sodium chloride; 0.06% (w/v) potassium phosphate; Sigma-Aldrich<sup>®</sup>). Prior to counting, cell aggregates were broken up with a 0.6mm syringe, the cells were loaded onto a haemocytometer and cells which excluded the trypan blue were counted.

Cells were seeded at 200,000 cells/well in a 24-well plate with 1ml growth medium. They were gently mixed and then left for 24 h without disruption to allow them to establish in a humidified  $37^{\circ}C/5\%$  CO<sub>2</sub> incubator (Sanyo, Japan). After 24 h, half (500µl) of the medium was changed and after 48 h, all the medium was replaced with fresh medium. This was left for 24 h and then replaced with fresh medium, on average the neuronal cultures survived 11-14 days. The day prior to transfection the growth medium was removed and replaced with fresh medium containing no antibiotics

#### 2.9.3.3 Transfection of cells

Individual shRNA constructs were co-transfected with pRK5-Clomeleon vector (kindly provided by Thomas Kuner (Kuner and Augustine, 2000) in order to verify the transfection efficiency. Each construct was transfected in six

samples/wells ( $\gamma$ 4-targetting shRNA, GAPDH-targetting shRNA, non-specific targeting shRNA and the Clomeleon vector alone). Per sample (well), 200ng of shRNA construct and 200ng of pRK5-Clomeleon vector were diluted in 50µl OptiMEM<sup>®</sup> I reduced serum medium (Gibco<sup>TM</sup>, Invitrogen) and gently mixed. Concurrently, 2µl Lipofectamine<sup>TM</sup> 2000 was diluted in 50µl OptiMEM<sup>®</sup> I reduced serum medium (final concentration 1µg/25µl), mixed and incubated for 5 min at room temperature. Following this, the diluted DNA and Lipofectamine<sup>TM</sup> 2000 were combined (total volume 100µl) and incubated at room temperature for 20 min to allow DNA-Lipofectamine<sup>TM</sup> 2000 complexes to form; these were then added to the wells containing cells and medium and very gently mixed. Cells were incubated at 37°C/5% CO<sub>2</sub> for 24 h, after which the transfection medium was removed and fresh growth medium was added, 72 h after this the cells were harvested.

# 2.9.4 RNA extraction from cells

Medium was aspirated from each well and discarded, 500µl PBS was added per well and gently agitated to wash the cells of any remaining medium. PBS was aspirated and  $25\mu$ l Tri<sup>®</sup> (Ambion<sup>TM</sup>) reagent was added to each well. Cells were harvested with a cell scraper and the replicates of each treatment group were pooled into 0.5ml sterile microfuge tubes. 30µl chloroform was added to each tube and vigorously mixed. Following 10 min incubation on ice, cells were centrifuged at 12,000*g* for 15 min at 4°C. The top aqueous layer was aspirated and combined with 150µl isopropanol, mixed and centrifuged at 12,000*g* for 10 min at 4°C. The supernatant was removed and discarded and remaining RNA was ethanol precipitated (section 2.3.3) and solubilised in 300µl H<sub>2</sub>O. Subsequently, first-strand cDNA was synthesised as per section 2.3.4

# 2.9.5 Real-time RT-PCR

This was a slightly different method to that described in section 2.7 as these experiments were completed in our laboratory. Real-time RT-PCR primers were designed to amplify partial regions of the chicken GABA<sub>A</sub> receptor  $\gamma$ 4-subunit and GAPDH genes in order to quantify the silencing efficiency of the shRNA

constructs. In addition, primers were designed for various chicken-specific housekeeping genes in order to validate the integrity of the RT-PCR reaction (detailed in Table 2.6) and allow normalisation of the data during analysis. All primers were designed adhering to the criteria outlined in section 2.4.1.

Target gene	Forward: 5'→3'	Reverse: 5'→3'	Amplicon (bp)
GABA <sub>A</sub> receptor γ4 subunit	GGTGTGAGAGCACAGAGGAG	GACAACAAACTGAGGCCTGA	142
GAPDH	AGTCCAAGTGGTGGCCATC	AGTCCAAGTGGTGGCCATC	154
β-actin	GAGAAATTGTGCGTGACATCA	TGGCAATGAGAGGTTCAGG	149
18S Ribosomal RNA	TCAGATACCGTCGTAGTTCC	AACTTAAAGGAATTGACGGAA	154
β-2-Microglobulin	AAGGAGCCGCAGGTCTAC	GTATGACGGCAAAGAGCAAG	151

**Table 2.6**. Sequences and amplicon sizes of real-time RT-PCR primers targeting the chicken GABA<sub>A</sub> receptor  $\gamma$ 4-subunit and GAPDH genes to validate the effects of RNAi. House-keeping primer sequences were taken from Li *et al.*, (2005) and designed from published chicken sequences. GenBank accession numbers as follows;  $\beta$ -actin (L08165); 18S Ribosomal RNA (AF173612);  $\beta$ -2-Microglobulin (Z48922).

Real-time RT-PCR was performed in a Rotor-gene 6000 (Qiagen, formerly Corbett Life Science) using the QuantiTect<sup>®</sup> SYBR<sup>®</sup> Green PCR system (Qiagen). Master mix comprised of (per 25µl reaction) 12.5µl 2x QuantiTect<sup>®</sup> SYBR<sup>®</sup> Green PCR Master Mix (which contained: HotStarTaq<sup>®</sup> DNA polymerase, QuantiTect<sup>®</sup> SYBR<sup>®</sup> Green PCR buffer (Tris-Cl, KCl, (NH<sub>4</sub>)<sub>2</sub>SO<sub>4</sub>, 5mM MgCl<sub>2</sub>, pH 8.7), dNTP mix, SYBR<sup>®</sup> Green I and ROX and RNase-free H<sub>2</sub>O), 100ng each of forward and reverse primer and 0.5µl cDNA template, made up to 25µl with H<sub>2</sub>O. Cycling details were as follows; an initial incubation step of 15 min at 95°C, which activated the HotStarTaq<sup>®</sup> DNA polymerase thereby reducing the formation of primer-dimers and mis-primed products and then 35 cycles of: 94°C *denaturation* (30 sec), 59.5°C *annealing* (30 sec), 72°C *extension* (30 sec).

During all preliminary real-time RT-PCR reactions a DNA dissociation (melt) analysis was completed in order to optimise the cycling conditions and identify any dimers or non-specific products. All primers gave a single product and following 1% (w/v) agarose gel electrophoresis (section 2.4.4), were deemed suitable for use.

Data was normalised and quantified with the Rotor-gene analysis software (Qiagen, formerly Corbett Life Science), no statistical analysis was completed as the full experiment was only completed twice.

# 3. SEQUENCES ENCODING ZEBRA FINCH GABA<sub>A</sub> AND GABA<sub>C</sub> RECEPTOR SUBUNITS

# **3.1 Introduction**

# 3.1.1 Isolation of GABA<sub>A</sub> receptor subunit sequences

Pentameric GABA<sub>A</sub> receptors are the primary mediators of rapid phasic inhibitorysynaptic neurotransmission and extrasynaptic tonic inhibition in the mammalian brain; this is also true of lower vertebrates such as the zebra finch. Isolation of sequences encoding the individual mammalian GABA<sub>A</sub> receptor subunits began in earnest in the mid-late 1980s after the first subunit cDNAs (encoding  $\alpha 1$  and  $\beta 1$  subunits) were isolated from bovine brain (Schofield *et al.*, 1987). A plethora of mammalian subunits have been uncovered over the last 20 years:  $\alpha 1$ - $\alpha 6$ ,  $\beta 1$ - $\beta 3$ ,  $\gamma 1$ - $\gamma 3$ ,  $\delta$ ,  $\varepsilon$ ,  $\theta$  and  $\pi$ ; and they are classified solely by their sequence similarity (for reviews see Burt and Kamatchi, 1991; Barnard *et al.*, 1998; Olsen and Sieghart, 2008, 2009). Members of the same class typically share 60-80% identity (at a nucleotide level) and subunits from different classes share 30-40% similarity. More recently a complete analysis of the human genome has revealed no further GABA<sub>A</sub> receptor subunit sequences exist and confirmed that each subunit polypeptide is encoded for by a separate gene (Simon *et al.*, 2004).

# 3.1.2 GABA<sub>A</sub> receptor subunit sequences

Each subunit gene encodes for a large extracellular hydrophilic N-terminus of ~230 AA which contains the putative 15 AA Cys-loop domain, characteristic of the LGIC superfamily. The amino-terminal is also the site of ligand binding and houses two highly conserved residues,  $\text{Ser}^{171}$  and  $\text{Tyr}^{172}$ , which are positioned close to the Cys-loop sequence and are purported to be involved in receptor association and cell-surface expression (Jin *et al.*, 2004), but not ligand binding (Whiting *et al.*, 1997b). Following this, there are four highly conserved hydrophobic transmembrane (TM)  $\alpha$ -helices of ~20 AA, the second of which forms the lining of the intrinsic chloride pore and contains a highly conserved octapeptide sequence presumably critical to pore formation (discussed in further detail in section 3.4; Imoto *et al.*, 1986; Xu and Akabas, 1996; Horenstein *et al.*, 2001). It is also purported that TM1 may also contribute to the channel formation (Sine and Engel, 2006). The small intracellular loop connecting TM1 and TM2 is relatively conserved within mammalian GABA<sub>A</sub> receptor subunit sequences

and mutations of this pentapeptide sequence of the  $\beta$ 3 (not  $\alpha$ 2 or  $\gamma$ 2) subunit revealed that it plays a role in ion selectivity (where mutations resulted in a cation-selective channel), but the pharmacology remained mostly unchanged (Jensen *et al.*, 2002). The long intracellular loop linking TM3 and TM4 is the most variable part between GABA<sub>A</sub> receptor subunit sequences and is speculated to be involved in receptor trafficking (Chen *et al.*, 2005) and influence ion conduction (Peters *et al.*, 2005); this is by the far the largest loop within the subunit complex (discussed later in further detail). Each subunit terminates in a short extracellular carboxyl domain.

# 3.1.3 GABA<sub>A</sub> receptor subunit splice variants

Following the isolation of cDNAs encoding 16 mammalian GABA<sub>A</sub> receptor subunits, heterogeneity of this already complex family was furthered by the detection of splice variants of many GABA<sub>A</sub> receptor subunits. The first of which was reported for the chicken  $\beta$ 4-subunit gene where there was a presence/absence of 4 AA within the large intracellular loop between TM3 and TM4; the variant arose due to the use of a different 5'-donor site (Bateson *et al.*, 1991). Two polypeptides have also been identified for the chicken  $\beta$ 2-subunit gene, which differ by a 17 AA presence ( $\beta$ 2L) or absence ( $\beta$ 2S) in the large variable intracellular loop. The 17 AA insert/deletion encodes a potential PKC phosphorylation site and was reported to be encoded for by a separate, additional exon (Harvey *et al.*, 1994) which, via alternative splicing is present in  $\beta$ 2L. Although a slightly longer insertion (38AA) was found for the human version of the gene, the splice site remained the same (McKinley *et al.*, 1995; Simon *et al.*, 2004). Rat and human  $\beta$ 3-subunit isoforms exist due to alterative splicing of exon 1 or 1a, resulting in two alternative signal peptides which are the same length but differ substantially in sequence (Kirkness and Fraser, 1993).

Within the class of  $\alpha$  subunits, the  $\alpha$ 2-subunit gene demonstrates the greatest number of isoforms (in rat), six in total due to three 5'-UTR (untranslated region) exons which can be variably spliced (Fuchs and Celepirovic, 2002). In the human genome 12 exons have been identified for the  $\alpha$ 2-subunit gene, resulting six potential isoforms (Simon *et al.*, 2004). Alternative splicing of human and mouse  $\alpha$ 4-subunit pre-mRNA results in a truncated mature mRNA and an inactive protein (Mu *et al.*, 2002) due to deletion of exons 3-8 (Simon *et al.*, 2004). In humans and rats, the  $\alpha$ 5-subunit gene has the largest

number of exons (13) with four 5'-UTR exons which can be variably deleted, resulting in three mRNA isoforms (Kim *et al.*, 1997; Simon *et al.*, 2004). Furthermore, alternative splicing is evident in the  $\alpha$ 6-subunit gene in rat where there is 10 AA deletion in the large intracellular loop (between TM3 and TM4); resulting in a nonfunctional polypeptide, attributed to incorrect splicing at a 3'-acceptor site within the exon (Korpi *et al.*, 1994).

It is the  $\gamma$ 2-subunit splice variants which have attracted the most attention. They differ by the presence ( $\gamma$ 2L) or absence ( $\gamma$ 2S) of an 8 AA insertion in the large variable intracellular loop (between TM3 and TM4) bearing a putative PKC site (Ser<sup>343</sup>) in bovine, human, rat (Whiting et al., 1990) and mouse brain (Kofuji et al., 1991). y2L accumulates at post-synaptic sites with the highest efficiency. Efficiency is further increased after short-term PKC activation, indicative of a role for phosphorylation of the Ser<sup>343</sup> site in the accumulation of GABA<sub>A</sub> receptors at inhibitory sites (Meier and Grantyn, 2004). More recently a third  $\gamma$ 2-subunit splice variant was uncovered ( $\gamma$ 2XL) which contains a 40 AA insert in the N-terminus created by a weak splice site within intron 5 by an Alu-J sequence (Jin et al., 2004). This primate-specific family of short interspersed elements has also been identified in intronic sequences of human  $\alpha$ 1-,  $\alpha$ 3-,  $\alpha$ 5-,  $\beta$ 1-,  $\gamma$ 2-,  $\rho$ 1- and  $\pi$ -subunit genes, whereby alternative subunit isoforms may be generated that are not detectable by examination of the exon content (Simon et al., 2004).  $\gamma$ 3-subunit splice variants have been identified in human. There is a presence  $(\gamma 3L)$  or absence  $(\gamma 3S)$  of 6 AA in the large intracellular loop (between TM3 and TM4) of the  $\gamma$ 3-subunit gene which encodes an additional PKC phosphorylation site (Poulsen et al., 2000). The *\varepsilon*-subunit gene is composed of 9 coding exons (Sinkonnen et al., 2000; Simon et al., 2004). Deletions in the first 3 exons and part of the fourth exon give rise to four expressed  $\varepsilon$ -subunit variants, including the full length transcript (Davies *et al.*, 1997; Garret et al., 1997; Whiting et al., 1997; Wilke et al., 1997).

# 3.1.4 GABA<sub>C</sub> receptor subunit splice variants

The human  $\rho 1$  subunit of the GABA<sub>C</sub> receptors also has two additional smaller, alternatively spliced variants. The first (named GABA<sub> $\rho 1$ </sub> $\Delta 450$ ) has a 450 nt deletion within the N-terminus and the second (named GABA<sub> $\rho 1</sub>\Delta 51$ ) has a 51 nt deletion from</sub>

the same 5' position (Martínez-Torres *et al.*, 1998). GABA<sub> $\rho 1$ </sub> $\Delta 51$  has lost exon 2 and was found to be functional whereas GABA<sub> $\rho 1$ </sub> $\Delta 450$  is missing exons 2-5 (Simon *et al.*, 2004) and is rendered non-functional due to loss of the GABA-binding site motif (Martínez-Torres *et al.*, 1998). Furthermore, a 300 bp *Alu* sequence was also detected within intronic sequence of the human  $\rho 1$ -subunit gene whereby a splice variant could potentially arise, these were not apparent in the human  $\rho 2$ - and  $\rho 3$ -subunit genes (Simon *et al.*, 2004), and no alternative splicing has been documented in any species for the  $\rho 2$  and  $\rho 3$  subunit polypeptides.

# 3.1.5 GABA<sub>C</sub> receptors

Ionotropic GABA<sub>C</sub> receptors are less well characterised, their structure is identical to that of GABA<sub>A</sub> receptors, with the subunits sharing all the same structural features. Where GABA<sub>A</sub> receptors are the most complicated member of the cys-loop LGIC superfamily, GABA<sub>C</sub> receptors are the simplest, with only three documented subunits  $(\rho_1-\rho_3)$  which are primarily localised to bipolar and horizontal cells of the retina (Qian and Dowling, 1993; Albrecht and Darlison, 1995; Enz et al., 1995; Chebib et al., 1998) and tend to form homomeric Cl<sup>-</sup> channels (Enz and Cutting, 1998; Borman, 2000). They are not well expressed within the mammalian brain (in comparison to the majority of GABA<sub>A</sub> receptor subunit genes) and are insensitive to classical GABA<sub>A</sub> receptor modulators such as barbiturates and benzodiazepines (Qian and Pan, 2002), so have avoided much scrutiny. Consequently there is no evidence of their role within the zebra finch song system. However, some data has reported expression of GABA<sub>C</sub> receptor subunits in mammalian brain (Delaney and Sah, 1999; Ogurusu et al., 1999; Rozzo et al., 2002; Milligan et al., 2004; Alakuijala et al., 2005), which may be able to colocalise with GABA<sub>A</sub> receptor subunits to form functional receptors (Ekema et al., 2002; Qian and Pan, 2002; Milligan *et al.*, 2004). Moreover, relatively high levels of  $\rho$ 1- and ρ2-subunit mRNAs have been detected in the chicken brain (Albrecht et al., 1997), where it was postulated they may be implicated in short-term memory formation (Gibbs and Johnston, 2005). Therefore, it was deemed important for these subunit cDNAs to be amplified and included in the expression studies.

To date 14 GABA<sub>A</sub> receptor subunit polypeptides have been identified in lower vertebrates  $\alpha 1$ - $\alpha 6$ ,  $\beta 2$ - $\beta 4$ ,  $\gamma 1$ ,  $\gamma 2$  and  $\gamma 4$ ,  $\delta$  and  $\pi$ ; where it is accepted that the  $\gamma 4$  and  $\beta 4$ 

subunits are orthologues of the mammalian  $\varepsilon$  and  $\theta$  subunits respectively (Darlison *et al.*, 2005), and three GABA<sub>C</sub> receptor subunit polypeptides ( $\rho$ 1- $\rho$ 3). However, not all of these have been formally identified (i.e. cloned) from zebra finch. Evidence for functional GABA<sub>A</sub> receptors within the song system is purely electrophysiological (Bottjer *et al.*, 1998; Dutar *et al.*, 1998; Luo and Perkel, 1999; Spiro *et al.*, 1999; Farries *et al.*, 2005; Prather *et al.*, 2008) and pharmacological (Carlisle *et al.*, 1998; Luo and Perkel, 1999; Vicario and Raksin, 2000; Farries *et al.*, 2005; Mooney and Prather 2005; Ölveczky *et al.*, 2005). A recent study by Thode *et al.* (2008) isolated  $\gamma$ 4-subunit cDNA from zebra finch and fully documented the expression of the transcript within the song system of the zebra finch brain. Findings demonstrated a robust expression pattern encompassing nearly all the nuclei of the male song system. This was the first time that a GABA<sub>A</sub> receptor subunit transcript had been spatially mapped within the zebra finch brain and results indicated that GABA<sub>A</sub> receptors containing the  $\gamma$ 4-subunit polypeptide may play an important role in song acquisition and production in the zebra finch (further details in Chapters 1 and 5).

Due to the vast repertoire of polypeptide subunits which can assemble into a large variety of receptor subtypes (each which has an individual pharmacological, biochemical and electrophysiological profile), and to understand which particular subtypes exist within zebra finch brain, all subunits must be considered. Work outlined in this chapter aimed to isolate partial cDNAs encoding all GABAA and GABAC receptor subunits from zebra finch so subsequent expression studies could be undertaken. At the time this work was completed, no zebra finch GABAA receptor subunit sequences, aside from the  $\gamma$ 4 subunit (Thode *et al.*, 2008) were published. Some partial cDNAs ( $\alpha 1$ ,  $\alpha 2$ ,  $\alpha 5$ ,  $\beta 2$ ,  $\beta 3$ ,  $\beta 4$ ,  $\gamma 2$ ,  $\gamma 4$ ) had already been isolated in the laboratory (data unpublished), but the remainder (e.g.  $\alpha 3$ ,  $\alpha 4$ ,  $\alpha 6$ ,  $\gamma 1$ ,  $\delta$  and  $\pi$ ) required cloning. Therefore, degenerate primers were based on human, rat and chicken sequences (where available). All GABA<sub>A</sub> and GABA<sub>C</sub> receptor subunit sequences demonstrate high inter-species conservation and high sequence homology between classes of subunits in the regions encoding the four transmembrane domains, but show some variability within the areas encoding the N-terminus and large intracellular loop situated between TM3 and TM4; it was within these regions that degenerate primers were designed and partial cDNAs amplified.

# 3.1.6 Sequencing of the zebra finch genome

The zebra finch song system is a valuable research tool for understanding the mechanisms underlying vocal learning and behaviour, which can be extrapolated to humans. Recognition of this has lead to its selection for whole genome sequencing. It is the second avian genome to be sequenced, after the chicken. Sequencing and assembly carried Washington University Genome were out at Sequencing Centre (http://genome.wustl.edu/genomes/view/taeniopygia\_guttata/) and was released in 2008 (following completion of this section of the study), revealing a genome of 1.2 Gb in length. Although many of the genes have been assigned and annotated they still remain predicted sequences and, as discussed later on, should be validated by experimental evidence. To this end, sequences which were cloned from the zebra finch (prior to the sequencing of the genome), were compared in the analyses to the corresponding, recently released sequences.

# **3.2 Results**

# 3.2.1 Sequences encoding zebra finch GABA<sub>A</sub> receptor subunits

Degenerate primers based on chicken and rat sequences were used to amplify partial sequences of  $GABA_{A/C}$  receptor subunits from zebra finch first strand cDNA (list of primer details in Tables 2.1 and 2.2; Chapter 2). Amplified RT-PCR products of expected size were purified, ligated into the pGEM<sup>®</sup>-T Easy vector and transformed into *E.coli* XL1 blue cells. Plasmids were extracted with minipreparations and digested with *Eco*R1 endonuclease to determine presence of inserts; positive clones were sent for automated sequencing. The process was repeated several times for each subunit clone to ensure consistency. Sequences were subjected to BLAST analyses for identification and were aligned using software available on the internet alongside recently published predicted sequences.



**Figure 3.1.** Schematic diagram depicting  $GABA_A/GABA_C$  receptor subunit topology, highlighting the region amplified by RT-PCR (----). Isolated cDNAs encompassed a portion of the amino-terminus, including the Cys-loop, putative  $\alpha$ -helical transmembrane domains (TM) 1-3 and a section of the large intracellular loop linking TM3 and TM4.

#### (a) Sequence of zebra finch $GABA_A$ receptor $\alpha$ 3 subunit

Zebra finch Chicken $\alpha$ 3 Zebra finch	α3 α3(Ρ)	RQSWRDERLKFDGPMKILPLNNLLASKIWTPDTFFHNGKKSVAHNMTTPNKLLRLVDNGT RQSWRDERLKFDGPMKILPLNNLLASKIWTPDTFFHNGKKSVAHNMTTPNKLLRLVDNGT RQSWRDERLKFDGPMKILPLNNLLASKIWTPDTFFHNGKKSVAHNMTTPNKLLRLVDNGT ************************************
Zebra finch Chicken $\alpha$ 3 Zebra finch	α3 α3 (P)	LLYTMRLTIHAECPHHLEDFPMDVHACPLKFGSYAYTKTEVIYTWTLGKDKSVEVAKGGS LLYTMRLTIHAECPMHLEDFPMDVHACPLKFGSYAYTKTEVIYTWTLGKDKSVEVAKGGS
	u3 (1)	Image: And the second secon
Zebra finch Chicken $\alpha$ 3	α3	RLNQYDLLGHVVGTEMVRSSTGEYVVMTTHFHLKRKIGYFVIQTYLPCIMTVILSQVSFW RLNQYDLLGHVVGTEMVRSSTGEYVVMTTHFHLKRKIGYFVIQTYLPCIMTVILSQVSFW
Zebra finch	α3(P)	RLNQYDLLGHVVGTEMVRSSTGEYVVMTTHFHLKRKIGYFVIQTYLPCIMTVILSQVSFW
Zebra finch Chicken $\alpha$ 3 Zebra finch	α3 α3 (P)	IM 2 LNRESVPARTVFGVTTVLTMTTLSISARNSLPKVAYATAMDWFMAVCYAFVFSALIEFAT LNRESVPARTVFGVTTVLTMTTLSISARNSLPKVAYATAMDWFIAVCYAFVFSALIEFAT LNRESVPARTVFGVTTVLTMTTLSISARNSLPKVAYATAMDWFIAVCYAFVFSALIEFAT ************************************
Zebra finch Chicken $\alpha$ 3 Zebra finch	α3 α3(P)	VNYFTKRSWAWDGKKVLEAQEMKKKEPVALAKKTNN VNYFTKRSWAWDGKKVLEAQEMKKKEPVAL <mark>W</mark> KKTNN VNYFTKRSWAWDGKKVLEAQEMKKKEPVALAKKTNN **********************************

**Figure 3.2a.** Alignment of the partial amino-acid sequence (in single letter code) encoding the zebra finch GABA<sub>A</sub> receptor  $\alpha$ 3 subunit with the corresponding chicken sequence (GenBank accession number XP\_420268) and predicted zebra finch sequence (as no peptide sequence was available, it was translated from the nucleotide sequence; GenBank accession number XR\_053905). Abbreviations for all figure 3.2a-f: (P) denotes predicted sequence, asterisks (\*) denote complete homology between sequences and dots represent low (.) or partial (:) homology, in some cases gaps (-) were introduced to maintain the alignment. Differences between sequences are highlighted in black. Also shown are transmembrane spanning domains (TM); putative Cys-loop consensus sequence (red) and conserved serine (S) and tyrosine (Y) residues (grey).

(b) S	equence	of zebra	finch	$GABA_A$	receptor	α4 subunit
-------	---------	----------	-------	----------	----------	------------

Chicken $\alpha 4$		DTFFRNGKKSVAHNMTAPNKLFRIMRNGTILYTMRLTISAECPMRLVDFPMDGHACPLKF
Zebra finch	α4	DTFFRNGKKSVAHNMTAPNKLFRIMRNGTILYTMRLTISAE <mark>CPMRLVDFPMDGHAC</mark> PLKF
		***************************************
Chicken $\alpha$ 4		G <u>SY</u> AYPKSEMIYTWTKGPEKSVEVPEESSSLVQYDLLGHTVSSETIKSITGEYIVMTVYF
Zebra finch	α4	GSYAYPKSEMIYTWTKGPEKSVEVPEESSSLVQYDLLGHTVSSETIKSITGEYIVMTVYF ************************************
		TM 1 TM 2
Chicken $\alpha 4$		HLRRKMGYFMIQTYIPCIMTVILSQVSFWINKESVPARTVFGITTVLTMTTLSISARHSL
Zebra finch	α4	HLRRKMGYFMIQTYIPCI
		*****
		TM 3
Chicken $\alpha 4$		PKVSYATAMDWFIAVCFAFVFSALIEFAAVNYFTNIQMERAKRKTVKSLLEFPVAPIQRE
Zebra finch	α4	PKVSYATAMDWFIAVCFAFVFSALIEFAAVNYFTNIQME <mark>K</mark> AKRKTVKSLLEFPVAP <mark>V</mark> QRK
		*******
Chicken $\alpha 4$		KCTEETHTSTDANSNVRKRTNATVQSEADGGSRIDTRHSSIQPPSVAQGSSDVTPHSLSA
Zebra finch	α4	RSTEETETSTDANSNVRKRTNATVQSEADGGSRIDTRHSS <mark>V</mark> QPPSVAHGSSD <mark>I</mark> TPHSLSA
		·· ****.*******************************
Chicken $\alpha$ 4		SSPNPFTRI
Zebra finch	α4	SSPNPFTRI
		* * * * * * * *

**Figure 3.2b.** Alignment of the partial amino-acid sequence (in single letter code) encoding the zebra finch GABA<sub>A</sub> receptor  $\alpha$ 4 subunit with the corresponding chicken sequence (GenBank accession number XP\_420724). No predicted zebra finch sequence encoding  $\alpha$ 4 was available

# (c) Sequence of zebra finch $GABA_A$ receptor $\alpha 6$ subunit

FINADC
IGQTVS
EGQTVS
EGQTVS * * * * * *
ARTVFG ARTVFG ARTVFG
*****
QTQRAM
QTQRAM
QTQRAM * * * * * *

**Figure 3.2c.** Alignment of the partial amino-acid sequence (in single letter code) encoding the zebra finch  $GABA_A$  receptor  $\alpha 6$  subunit with the corresponding chicken sequence (GenBank accession number NP\_990389) and predicted zebra finch sequence (GenBank accession number XP\_002193341).

# (d) Sequence of zebra finch $GABA_A$ receptor $\gamma I$ subunit

Zebra finch $\gamma 1$ (F Zebra finch $\gamma 1$ Chicken $\gamma 1$	) HWITTPNRLLRIWSDGRVLYTLRLTINAECYLQLHNFPMDEHSCPLEFSSYGYPRNEIEY HWITTPNRLLRIWSDGRVLYTLRLTINAECYLQLHNFPMDEHSC HWITTPNRLLRIWSDGRVLYTLRLTINAECYLQLHNFPMDEHSCPLEFSSYGYPRNEIEY ***********************************
Zebra finch γ1(P Zebra finch γ1 Chicken γ1	) KWKKTSVEVADPKYWRLYQFAFVGLRNTTEISHTLSGDYIIMTIFFDLSRRMGYFTIQTY KWKKTSVEVADPKYWRLYQFAFVGLRNTTEISHTLSGDYIIMTIFFDLSRRMGYFTIQTY KWKKTSVEVADPKYWRLYQFAFVGLRNTTEISHTLSGDYIIMTIFFDLSRRMGYFTIQTY
	тм 1 тм 2
Zebra finch γ1(F Zebra finch γ1 Chicken γ1	) IPCILTVVLSWVSFWINKDAVPARTSLGITTVLTMTTLSTIARKSLPKVSYVTAMDLFVS IPCILTVVLSWVSFWINKDAVPARTSLGITTVLTMTTLSTIARKSLPKVSYVTAMDLFVS IPCILTVVLSWVSFWINKDAVPARTSLGITTVLTMTTLSTIARKSLPKVSYVTAMDLFVS ************************************
Zebra finch γ1(F Zebra finch γ1 Chicken γ1	) VCFIFVFAALMEYGTLHYFTSNRKGDKGKEKKAKSKPSKPPAIAVRPGSTLIPINNINHL VCFIFVFAALMEYGTLHYFTSNRKGDKGKEKKAKSKPSKPPAIAVRPGSTLIPINNINHL VCFIFVFAALMEYGTLHYFTSNRKGDKGKDKKAKEKPSKPSAIAVRPGSTLIPINNINNL ********************************
Zebra finch γ1(F Zebra finch γ1 Chicken γ1	) PERDD PERDD PERDD *****

**Figure 3.2d.** Alignment of the partial amino-acid sequence (in single letter code) encoding the zebra finch GABA<sub>A</sub> receptor  $\gamma 1$  subunit with the corresponding chicken sequence (GenBank accession number XP\_420725) and predicted zebra finch sequence (GenBank accession number XP\_002194407).

# (e) Sequence of zebra finch $GABA_A$ receptor $\delta$ subunit

Zebra finch $\delta$ Zebra finch $\delta$ (P Chicken $\delta$	NKLIRLQPDGVILYSIRITSTVACDMDLSKYPMDEQECMLDLESYGYSSEDIVYHWSENQ NKLIRLQPDGVILYSIRITSTVACDMDLSKYPMDEQECMLDLESYGYSSEDIVYHWSENQ NKLIRLQPDGVILYSIRITSTVACDMDLSKYPMDEQECMLDLESYGYSSEDIVYHWSENQ ******					
	TM 1					
Zebra finch $\delta$	EEIHGLDKLQLAQFTITNYQFTTEIMNFKS-GQFPRLSLHFHLRRNRGVYIIQSYVPSIL					
Zebra finch $\delta$ (P	) <u>EEIHGLDKLQLAQFTITNYQFTTEIMNFKS-GQFPRLSLHFHLRRNRGVYIIQSYVPSIL</u>					
Chicken δ						
	TM 2 TM 3					
Zebra finch $\delta$	LVAMSWVSFWISQSAVPARVSLGITTVLTMTTLMVSARSSLPRASAIKALDVYFWICYVF					
Zebra finch $\delta$ (P	) LVAMSWVSFWISQSAVPARVSLGITTVLTMTTLMVSARSSLPRASAIKALDVYFWICYVF					
Chicken $\delta$	LVAMSWVSFWISQSAVPARVSLGITTVLTMTTLMVSARSSLPRASAIKALDVYFWICYVF ************************************					
Zebra finch $\delta$ Zebra finch $\delta$ (P Chicken $\delta$	VFAALVEYAFAHFNADYMKKQKNKIKARRQSA ) VFAALVEYAFAHFNADYMKKQKNKIKARRQSA VFAALVEYAFAHFNADYMKKQKNKIKARRQS <mark>G</mark> ************************************					



At a peptide level the cloned zebra finch  $\alpha$ 3-subunit sequence shared 95.7% and 99.3% homology with the predicted zebra finch and chicken reference sequences respectively (Fig. 3.2a). Interestingly, the region where there was greatest variability between the

cloned and predicted zebra finch sequences was in the N-terminus within the highly conserved Cys-loop domain. In all other cloned sequences the Cys-loop sequence was identical between the predicted and cloned zebra finch sequences and the chicken reference sequence. The single mis-match in TM3 is potentially a single-nucleotide polymorphism (as it was present in all cloned sequences) whereby a guanine residue was substituted at the final base of the codon changing the amino acid from isoleucine (I) to methionine (M). A BLAST search using the cloned  $\alpha$ 3-subunit sequence as a probe against the chicken and zebra finch genomes revealed the gene to be localised to chromosome 4A in the zebra finch genome (GenBank accession number XR\_053905) and chromosome 4 in the chicken genome (GenBank accession number XM\_420268; Fig. 3.8). Available information on the NCBI database indicates that the zebra finch  $\alpha$ 3-subunit gene comprises 9 exons yielding a polypeptide of 48.6 kDa.

The deduced zebra finch  $\alpha$ 4-subunit amino acid sequence demonstrated 97.7% similarity with the chicken reference sequence (Fig. 3.2b). There was no documented or predicted zebra finch  $\alpha$ 4-subunit sequence available on relevant databases. Following sequencing of the  $\alpha$ 4-subunit gene, the cloned sequence was subjected to a BLAST search against the zebra finch genome which yielded no corresponding annotated sequence in the zebra finch, however a region on chromosome 4 was homologous (100% identity, genomic position 45,548,719-45,548,979) and as the  $\alpha$ 4 gene is localised to chromosome 4 in the chicken genome (GenBank accession number XM\_420724), it was assumed that this was the corresponding zebra finch sequence. As expected, greatest variation between the chicken and zebra finch sequences occurred in the intracellular loop region between TM3 and TM4. The complete zebra finch  $\alpha$ 4-subunit gene comprises 10 exons (as opposed to the normal coding pattern of 9 exons) yielding a polypeptide of 61.3 kDa (considerably larger than the  $\alpha$ 3-subunit protein).

Highest sequence identity observed for any of the isolated subunit sequences was for that encoding the zebra finch  $\alpha$ 6-subunit. It presented 100% and 99.6% homology with the predicted zebra finch and chicken reference sequences respectively (Fig. 3.2c). The only difference was the final residue, which lies within the intracellular loop between TM3 and TM4, a naturally variable region. The same residue was present in all the clones from two different RT-PCR products and so could not be attributed to *Taq* 

polymerase or sequencing errors and was present in the predicted zebra finch sequence. BLAST analysis showed the  $\alpha$ 6-subunit gene to be localised to chromosome 13 in the zebra finch (GenBank accession number XM\_002193305) and chicken genome (GenBank accession number NM\_205058), where it encodes 9 exons, yielding a polypeptide of 50.5 kDa.

The isolated zebra finch  $\gamma$ 1-subunit amino acid sequence exhibited 100% and 98.4% homology with the predicted zebra finch and chicken reference sequences respectively at a peptide level (Fig. 3.2d). As observed with the  $\alpha$ 4 sequence (Fig. 3.2b), the greatest variability between the sequences occurred within the large intracellular loop connecting TM3 and TM4. The GABA<sub>A</sub> receptor  $\gamma$ 1-subunit gene (GenBank accession number XM\_002194366) is localised to chromosome 4 of the zebra finch and chicken genome (GenBank accession number XM\_420725; Fig. 3.8).

There was 100% and 98.1% homology between the isolated  $\delta$ -subunit sequence and predicted zebra finch and chicken reference sequences respectively (Fig. 3.2e). The two zebra finch sequences were homologous, with some residues different to the chicken sequence within the N-terminal region, but not within any of the highly conserved regions. The GABA<sub>A</sub> receptor  $\delta$ -subunit gene (GenBank accession number XM\_002197169) is localised to chromosome 21 in the zebra finch and the chicken genome (GenBank accession number XM\_001234040; Fig. 3.2e). The complete zebra finch  $\delta$ -subunit gene comprises 9 exons yielding a polypeptide of 48.7 kDa.

A complete alignment of all the  $GABA_A$  receptor subunit genes is presented in Figure 3.5, where regions of interest are highlighted.

# (f) Sequence of the zebra finch $GABA_A$ receptor $\pi$ subunit

```
Zebra finch \pi(L)
                           SLVELLWVPDTYIVESKRSFLHDVTVGNRLVRLFSNGTVLYALRITTTVACNMDLSKYP
Zebra finch \pi(S)
Zebra finch \pi(P)
                           SLVELLWVPDTYIVESKRSFLHDVTVGNRLVRLFSNGTVLYALRITTTVA
                           RLVELLWVPDTYIVESKRSFLHDVTVGNRLVRLFSNGTVLYALRITTTVACNMELSKYPM
RLVELLWVPDTYIVESKRSFLHDVTVGNRL<mark>U</mark>RLFSNGT<mark>U</mark>LYALRITTTVACNMDLSKYPM
Chicken \pi (P)
                                ********
                            DTQTCRLQLESWGYDENDVTFTWLRGNNSVRGIEKLRLSQYTVEHYHTLVSKSQQETGSY
Zebra finch \pi(L)
                            DTQTCRLQLESWGYDENDVTFTWLRGNNSVRGIEKLRLSQYTVEHYHTLVSKSQQETGSY
Zebra finch \pi(S)
                            DTQTCRLQLESWGYDENDVTFTWLRGNNSVRGIEKLRLSQYTVEHYHTLVSKSQQETGSY
Zebra finch \pi(P)
                           Chicken \pi (P)
Zebra finch \pi(L)PRLILQFELRRNVLYFILETYVPSTLLVMLSWVSFWITLDSVPARTCIGVTTVLSMTTLMZebra finch \pi(S)PRLILQFELRRNVLYFILETYVPSTLLVMLSWVSFWITLDSVPARTCIGVTTVLSMTTLMZebra finch \pi(P)PRLILQFELRRNVLYFILETYVPSTLLVMLSWVSFWITLDSVPARTCIGVTTVLSMTTLMChicken \pi(P)PRLILQFELRRNVLYFILETYVPSTLLVMLSWVSFWITLDSVPARTCIGVTTVLSMTTLM
                           TM 3
                                     : . :
                                                             * * * * * * * * * * * * * * * * * * *
                                                                                              * * * * * *
Zebra finch \pi(L)EEMCEVNITNIIKNSITSYKQKISFASIEISSNNVNZebra finch \pi(S)EEMEEVNITNIIKNSITSYKQKISFASIEISSNNVNZebra finch \pi(P)EEMEEVNITNIIKNSITSYKQKISFASIEISSNNVN
                           KEMEEINITSILSSSITSYK<mark>KQ</mark>ISFTGIEISSDNIN
Chicken \pi (P)
                           :**:*:***.*:..******
```

**Figure 3.2f.** Alignment of the partial amino-acid sequence (in single letter code) encoding the zebra finch GABA<sub>A</sub> receptor  $\pi$  subunit with the corresponding chicken sequence (GenBank accession number XP\_414507) and predicted zebra finch sequence (GenBank accession number XP\_002188723). Abbreviations as before. (L) long, (S) short.

Isolation of partial cDNAs encoding the  $\pi$ -subunit yielded two similar size but distinct products, of apparent equal intensity (data not shown). Both products were purified and sent for automated sequencing and BLAST analysis confirmed that they were both DNA sequences encoding the GABA<sub>A</sub> receptor  $\pi$  subunit, but the smaller sequence had a 37 AA deletion; the missing region of sequence encodes part of TM2 and TM3; the reading frame remained unchanged with the deletion (Fig. 3.2f). The predicted zebra finch GABA<sub>A</sub> receptor  $\pi$ -subunit gene (GenBank accession number XM\_002188687) is localised to chromosome 13, as is the predicted corresponding chicken gene (GenBank accession number XM\_414507; Fig. 3.8). The full length predicted zebra finch  $\pi$ subunit gene comprises 8 exons yielding a polypeptide of 48.6 kDa (Fig. 3.3).



**Figure 3.3.** Exonic arrangement of zebra finch GABA<sub>A</sub> receptor  $\pi$ -subunit gene adapted from the Ensembl genome browser (www.ensembl.org). Genomic coordinates (chromosome 13) corresponding to the amplified cDNA sequences are given; there are 8 coding exons within the gene. Exon 7 is partly deleted in the shorter version of the  $\pi$ -subunit mature mRNA. The dashed line denotes the exons within the region amplified by RT-PCR.

Discounting the deletion, there was 99.6% similarity between the two deduced GABA<sub>A</sub> receptor  $\pi$  subunit amino acid sequences cloned from the zebra finch at a peptide level. (Fig. 3.2f) Furthermore, there was 99.6% identity for both the long and short sequences with the predicted zebra finch reference sequence (excluding the deletion). The short sequence demonstrated 90.3% sequence similarity with the chicken reference sequence and the long sequence shared 90.2% identity with the corresponding chicken gene. A Spidey analysis (www.ncbi.nlm.nih.gov/IEB/Research/Ostell/Spidey /index.html) was performed to compare the cloned mRNA sequences to the genomic sequence of the zebra finch (GenBank accession number ABQF01055946) and this revealed that the shorter version was missing an exonic section (7) of the gene (Fig. 3.4). Recognition of splice sites in pre-mRNAs (precursor mRNAs) to remove intronic sequence by spliceosomes is achieved by consensus sequences located at polar termini of the introns (Berget, 1995). There are matrices available for the consensus recognition sequences, for U2 dependent introns (U2 is a small nuclear ribonucleoprotein (snRNP), which along with U1, U4, U5 and U6 compose spliceosomes); these are the major class of introns, and splice sites conform as follows (Zhang et al., 1999):

Intron Exon **3' splice sites:** C<u>AG</u> | G

Intron Exon **5' splice sites:** MAG | <u>GT</u>RAGT (where M is A or C and R is A or G)

These sites can vary slightly dependent on the bases used, but the underlined bases have a 100% usage frequency. By comparison of the cloned sequences with the genomic sequences it was possible to determine the location of the splice sites and proffer an explanation as to the 37 AA deletion in the short version (Fig. 3.4).

	$\leftarrow$ EXON 6
	5' splice site
7529	AGAACCTGCA <mark>TTGgtgagtt</mark> tctg
494	AGAACCTGCATTG
	r exon 7→
	3' splice site
8087	gttttccaagGAGTTACAACAGTGCTTTCTATGACAACTCTGATGGTCGG
507	GAGTTACAACAGTGCTTTCTATGACAACTCTGATGGTCGG
8127	<b>中中でえ ここえ え こ 中中中 ス ご 中中中 こ え え え え こ こ え こ 中 こ 中 中中 マ え 中 ご ス ス こ こ こ ス 中 一 ス 中 ご ス 中 ご ス 中 ご ス 中 ご ス 中 ご ス 中 ご ス 中 ご ス 中 ご ス 中 ご ス 中 ご ス 中 ご ス 中 ご ス 中 ご ス 中 ご ス 中 ご ス 中 ご ス 中 ご ス 中 ご ス 中 ご ス ー ご</b>
0127	
547	TTCACGAAGTTTACTTTCAAAAACCAACTGTTTCATTAAGGCCATTGATG
	3' splice site
81//	TTTATCTTGGCATCTGCTT <mark>CAGC</mark> TTCATCTTTGGTGCCCTTGTGGAGTAT
597	
001	
8227	GCAGTGGCTCACTACAGCTCCTCACAGAAGTGTGCAGCTAAAGTACCAGA
647	GCAGTGGCTCACTACAGCTCCTCACAGAAGTGTGCAGCTAAAGTACCAGA
€	EXON 7
	5' splice site
8277	A <mark>GAGgtaagg</mark> ctcg
607	
091	AGAG

**Figure 3.4.** Spidey alignment of amplified partial cDNA encoding the GABA<sub>A</sub> receptor  $\pi$  subunit with the corresponding region of chromosome 13 of zebra finch genomic sequence (GenBank accession number ABQF01055946). Highlighted in grey are exons (which are also numbered), lowercase lettering denotes intronic sequence and splice junctions occurring in wild type long-form pre-mRNA are highlighted in yellow. Genomic and cloned mRNA coordinates are given. Splicing at genomic coordinates 7538 and 8096 removes intronic sequence between exon 6 and exon 7 and the long version of the cDNA is spliced in this way, thereby preserving all of exon 7. The short cDNA is spliced at the 5' splice site at the terminus of exon 4 but the splice junction at the beginning of exon 5 is skipped and a second splice junction is used which is located within exon 7 (position 8199, genomic coordinates), thereby removing a large portion of exon 7 and the beginning of exon 8 as normal.

The shorter sequence occurs due to a 37 AA deletion removing part of TM2 and TM3 domains and the extracellular loop in between (Fig. 3.4). This is a result of a deletion in exon 7, caused by the skipping of the putative 3' splice junction at the beginning of

exon 7 (genomic position 8097 and cloned mRNA position 507), and instead splicing at an alternative acceptor site located at coordinates 8199 of the genomic sequence (Fig. 3.4). The following 5' splice junction remained the same in the short sequence as in the long sequence; this resulted in a truncated mature mRNA.

			1 L								
zebra	finch	α4	DTFFRNGKKSVAHNMTAPNKLFRIMRNGTILYTMRLTISAECPMRLVDFF								
zebra	finch	α6	NLMVSKIWTPDTFFRNGKKSIAHNMTTPNKLFRIMONGTILYTMRLTINADCPMRLVNFF								
zebra	finch	α1	NLMASKIWTPDTFFHNGKKSVAHNMTMPNKLLRITEDGTLLYTMRLTVRAECPMHLEDFP								
zebra	finch	α2	NLMASKIWTPDTFFHNGKKSVAHNMTMPNKI.I.RIODDGTI.I.YTMRI.TVOAF.COMHI.FDF								
zebra	finch	α3	NLLASKIWTPDTFFHNGKKSVAHNMTTPNKI.T.RI.VDNGTI.I.YTMRI.TIHAF.								
zebra	finch	α5	NLLASKTWTPDTFFHNGKKSTAHNMTTPNKLLRLEDDGTLLYTMRLTTSAF								
zebra	finch	γ1	HWITTPNRLLRIWSDGRVLYTLRLTINAECYLOLHNFF								
zebra	finch	γ2	FFRNSKKADAHWITTPNRMLRIWNDGRVLYTLRLTIDAECQLQLHNF								
zebra	finch	γ4	TNMVSRIWIPDTFFRNSKRADSHWITTPNOLLRIWNDGKVLYTLRLTIEAECLLOLONFF								
zebra	finch	δ	NKLIRLQPDGVILYSIRITSTVACDMDLSKY								
zebra	finch	π	-SLVELLWVPDTYIVESKRSFLHDVTVGNRLVRLFSNGTVLYALRITTTVACNMDLSKYF								
zebra	finch	β2	LRRY								
zebra	finch	β2	IRRYF								
zebra	finch	β2	IRRYF								
			* :*								
zebra	finch	α4	MDGHACPLKFG <mark>SY</mark> AYPKSEMIYTWTKGPEKSVEVPEE-SSSLVQYDLLGHTVSSETIKSI								
zebra	finch	α6	MDGHACPLKFG <mark>SY</mark> AYPKSEIIYTWKKGPLHSVEVPQE-SSSLLQYDLIGQTVSSETIKSN								
zebra	finch	α1	MDAHACPLKFG <mark>SY</mark> AYTRAEVVYEWTREPARSVVVAED-GSRLNQYDLLGQTVDSGIVQSS								
zebra	finch	α2	MDAHSCPLKFG <mark>SY</mark> AYTTSEVTYIWTYNASDSVQVAPD-GSRLNQYDLLGQTIGKETVKSS								
zebra	finch	α3	MDVHACPLKFG <mark>SY</mark> AYTKTEVIYTWTLGKDKSVEVAKG-GSRLNQYDLLGHVVGTEMVRSS								
zebra	finch	α5	MDAHACPLKFG <mark>SY</mark> AYPNSEVIYVWTNSTTTSVVVAED-GSRLNQYHLMGQTVGTENISTS								
zebra	finch	γ1	MDEHSCPLEFS <mark>SY</mark> GYPRNEIEYKWKKTSVEVADPKYWRLYQFAFVGLRNTTEISHTI								
zebra	finch	γ2	MDVHSCPLEFS <mark>SY</mark> GYPREEIIYQWKRSSVEVGDTRSWRLYQFAFTGLRNTTEVVKTT								
zebra	finch	γ4	MDTHSCPLVFS <mark>SY</mark> GYPREEIVYRWRRYSIEVSDQRTWRLYQFDFTGLRNTSEVLRT0								
zebra	finch	δ	MDEQECMLDLE <mark>SY</mark> GYSSEDIVYHWSENQEEIHGLDKLQLAQFTITNYQFTTEIMNFF								
zebra	finch	π	MDTQTCRLQLE <mark>SW</mark> GYDENDVTFTWLRGNNSVRGIEKLRLSQYTVEHYHTLVSKSQQE								
zebra	finch	β2	LDEQNCTLEIESYGYTTDDIEFYWRGGDNAVTGVTKIVLPQFSIVDYKLITKNVVFS								
zebra	finch	β3	LDEQNCTLEIESYGYTTDDIEFYWRGGDNAVTGVERIELPQFSIVEYRLVSKNVVF#								
zebra	finch	β4	LDQQNCTLEIE <mark>SY</mark> GYTVDDIVFFWQGNDSAVTGMEVLELPQFTIIEQRLVSREVVF7								
			:* : * * : * : : * * : : * * : .								
			тм 1 тм 2								
zebra	finch	α4									
zebra	finch	ae	TGEYVINTVYFHLORKMGYFMIOTYTPCINTVILSQVSFWINKESVPARTVFGITTVLT								
zebra	finch	α1	TGEYVVMTTHFHLKRKIGYFVIOTYLPCIMTVILSQVSFWIMESVPARTVFGVTTVLT								
zebra	finch	a2	TGEYTVMTAHFHLKRKIGYFVIOTYLPCIMTVILSQVSFWLMRESVPARTVFGVTTVLT								
zebra	finch	a3									
zebra	finch	α5	TGEYTIMTAHFHLKRKIGYFVIOTYLPCIMTVILSQVSFWLMRESVPARTVFGVTTVLT								
zebra	finch	v1	SGDYTIMTIFFDLSRRMGYFTIOTYTPCTLTWVLSWVSFWINKDAVPARTCLCTTWVLT								
zehra	finch	$\sqrt{2}$	SCDYWWSVFFNLSRRMCYFTIOTYI PCTLTWILSWVSFWINKDAWDARTCI CIT								
zehra	finch	7∸ v4									
zehra	finch	λ <sup>-1</sup>	SCOPPRIST. HEHLRRNRGVYTTOSYVPSTLLVAMSWVGFWTCOCAVDARVCTCTTMVTTM								
zebra	finch	π									
zebra	finch	л В2	TGAYPRI,SI,SFKI,KRNTGYFTI,OTYMPSTI,TATI,SWVSFWITTDSVTAKTOTGVTTVDS								
zebra	finch	р <del>-</del> ВЗ	TGAYPRI,SI,SFRWKRNTGYFTLOTYMPSTI,TTTI,SWVSFWTNVDASAPRVALCT								
zebra	finch	рэ 64	TGSYLRLSLSFRIKRNIGYFTLOTYMPSTLTTTLSWVSFWINVDASAARVALCVTTVLT								
20010		r ·	:* : : * *. : :: * * :* ****: . : .* .* :*:*****								

							TM	3			
zebra	finch	α4	TTLSI <mark>s</mark>	AR <mark>H</mark>	SLPKV	/S-YATAMD	WFIAVCF	AFVFSA	LIEFAAV	/NYFT-N	IIQMEKAKRKTVK
zebra	finch	α6	TTLSI <mark>S</mark>	AR <mark>H</mark>	SLPK	/S-YATAMD	WFIAVCF	AFVFSA	LIESAAV	/NYFT-N	ILQTQRAMRKAAR
zebra	finch	α1	TTLSI <mark>s</mark>	AR <mark>N</mark>	SLPKV	/A-YATAMD					
zebra	finch	α2	TTLSI <mark>S</mark>	AR <mark>N</mark>	SLPK	/A-YATAMD					
zebra	finch	α3	TTLSI <mark>S</mark>	AR <mark>N</mark>	SLPKV	/A-YATAMD	WFMAVCY	AFVFSA	LIEFAT	/NYFT-K	RSWAWDGKKVLE
zebra	finch	α5	TTLSI <mark>S</mark>	AR <mark>N</mark>	SLPKV	/A-YATAMD					
zebra	finch	γ1	TTLST <mark>I</mark>	ar <mark>k</mark>	SLPKV	/S-YVTAMD	LFVSVCF	IFVFAA	LMEYGTI	LHYFTSN	IRKGDKGKEKKAK
zebra	finch	γ2	TTLST <mark>I</mark>	ar <mark>k</mark>	SLPKV	/S-YVTAMD	LFVSVCF	IFVFSA	LVEYGTI	LHYFVSN	IRKPSKDKDKKKK
zebra	finch	γ4	TTLST <mark>I</mark>	SR <mark>k</mark>	HLPR	/S-YITAMD	LFVSVCF	IFVFAA	LMEYATI	LNYLVGN	IKKPLEHSHRRAR
zebra	finch	δ	TTLMV <mark>S</mark>	AR <mark>S</mark>	SLPRÆ	AS-AIKALD	VYFWICY	VFVFAA	LVEYAFA	AHFN	
zebra	finch	π	TTLMV <mark>G</mark>	SR <mark>S</mark>	LLSKI	NCFIKAID	VYLGICF	SFIFGA	LVEYAVA	AHYSSSQ	KCAAKVPEEGPA
zebra	finch	β2	TTINT <mark>H</mark>	LR <mark>e</mark>	TLPKI	P-YVKAID	MYLMGCF	VFVFMA	LLEYAL	VNYIFFG	RGPQRQKKAAEK
zebra	finch	β3	TTINT <mark>H</mark>	LR <mark>e</mark>	TLPKI	P-YVKAID	MYLMGCF	VFVFLA	LLEYAF	VNYIFFG	KGPQRQKKLAEK
zebra	finch	β4	TTINT <mark>H</mark>	LR <mark>E</mark>	TLPKI	P-YVKAID	VYLMGCF	VFVFLA	LLEYAF	VNYIFFG	RGPRQQKKQSER
			**:	*	*.:	.*:*					

**Figure 3.5**. Alignment of all cloned zebra finch GABA<sub>A</sub> receptor subunit amino acid sequences. TM domains are annotated, (\*) denote complete homology between sequences and dots represent low (.) or partial (:) homology, (-) gaps were introduced to maintain the alignment. The 15AA Cys-loop domain is well conserved between subunit classes (red) with the two peripheral cysteines and central PMD/PLD (single-letter code) sequence identical in all sequences ( $\downarrow$ ) where cloned. Conserved Ser<sup>171</sup> and Tyr<sup>172</sup> residues shortly after the Cys-loop consensus sequence are indicated in yellow. Within the TM2 region, channel lining residues are highlighted in grey and positions at which disulphide bonds are formed are indicated in green (Horenstein *et al.*, 2001)

Figure 3.5 illustrates the high homology existing between all zebra finch GABA<sub>A</sub> receptor subunit genes. Regions of strong sequence conservation include the Cys-loop domain, serine and tyrosine residues within the N-terminus and hydrophobic transmembrane domains, suggestive of important physiological roles. Moreover, the intracellular TM1-TM2 loop is well conserved within subunit classes, indicative of a subunit specific function; this is also observed in the extracellular TM2-TM3 linker domain. It is only the  $\pi$ -subunit sequence which does not have the conserved hydrophilic Tyr<sup>172</sup> residue and has a hydrophobic tryptophan residue in its place.

# 3.2.1 Sequences encoding zebra finch GABA<sub>C</sub> receptor subunits

The three subunits of the GABA<sub>C</sub> receptor were isolated from the zebra finch brain. The cloned zebra finch  $\rho$ 1-subunit peptide sequence exhibited 98.4% homology with the chicken reference sequence (Fig. 3.6a). No annotated sequence exists for the zebra finch GABA<sub>C</sub> receptor  $\rho$ 1-subunit gene, a BLAST analysis of the cloned transcript against the zebra finch genome revealed homology with a region on chromosome 3 slightly upstream of the gene encoding the GABA<sub>C</sub> receptor  $\rho$ 2-subunit within the zebra finch
genome. The corresponding gene in the chicken is localised to chromosome 3 (GenBank accession number XM\_426190).

The cloned zebra finch  $\rho$ 2-subunit peptide sequence shared 96.9% and 99.6% identity with the predicted zebra finch and chicken reference sequences respectively (Fig. 3.6b). The zebra finch GABA<sub>C</sub> receptor  $\rho$ 2-subunit gene (GenBank accession number XR\_054611) is localised to chromosome 3 as in the chicken genome (GenBank accession number XM\_419839) close to the  $\rho$ 1-subunit gene in the same transcriptional orientation. Both the  $\rho$ 1- and  $\rho$ 2-subunit genes comprise 9 exons yielding polypeptides of 48.6 and 54.2 kDa respectively.

#### (a) Sequence of the zebra finch $GABA_C$ receptor $\rho 1$ subunit

Chicken $\rho$ 1(P) Zebra finch $\rho$ 1	SMTFDGRLVKKIWVPDMFFVHSKRSFIHDTTTDNVMLRVQPDGKVLYSLRVTVTAMCN SMTFDGRLVKKIWVPDMFFVHSKRSFIHDTTTDNVMLRVQPDGKVLYSLRVTVTAM	MD MD **
Chicken $\rho$ 1(P)	FSRFPLDTQTCSLEIESYAYTEDDLMLYWKNGNDSLKTDERISLSQFLIQEFHTTTKL	AF
Zebra finch $\rho$ 1	FSRFPLDTOTCSLEIESYAYTEDDLMLYWKNGNDSLKTDERISLSQFLIQEFHTTTKL	AF **
	TM 1 TM 2	
Chicken $\rho$ 1(P)	YSSTGWYNRLYINFTLRRHIFFFLLQTYFPATLMVMLSWVSFWIDRRAVPARVPLGIT	ΤV
Zebra finch $\rho$ 1	YSSTGWYNRLYINFTLRRHIFFFLLQTYFPATLMVMLSWVSFWIDRRAVPARVPLGIT	TV **
	TM 3	
Chicken $\rho$ 1(P)	LTMSTIITGVNASMPRVSYIKAVDIYLWVSFVFVFLSVLEYAAVNYLTTVQERK	RD
Zebra finch $\rho$ 1	LTMSTIITGVNASMPRVSYIKAVDIYLWVSFVFVFLSVLEYAAVNYLTTVQERKXRKL	RD **
Chicken $\rho$ 1(P)	KLPCACSLPQPRPMMMDGSY	
Zebra finch $\rho$ 1	KPACACSLPQPRPMMVDGSY * .************	

**Figure 3.6a.** Alignment of the partial amino-acid sequence (in single letter code) encoding the zebra finch GABA<sub>C</sub> receptor  $\rho$ 1 subunit with the corresponding chicken sequence (GenBank accession number XP\_426190). **Abbreviations** for all figure 3.6a-c: (P) denotes predicted sequence, asterisks (\*) denote complete homology between the two sequences and the dots (.), (:), represent partial homology Differences between the sequences are highlighted in black. Also shown are transmembrane spanning domains (TM); putative Cys-loop consensus sequence (red) and the putative Serine (S) and Tyrosine (Y) residues (grey).

#### (b) Sequence of the zebra finch $GABA_C$ receptor $\rho 2$ subunit

Chicken $\rho$ 2(P)	FFVHSKRSFIHDTTTDNIMLRVFPDGHVLYS	SMRITVTAMCNMDFSRFPLDSQTCSLELES
Zebra finch $ ho2$	FFVHSKRSFIHDTTTDNIMLRVFPDGHVLYS	SMRITVTAM <mark>CNMDFSR</mark> FPLDSQTC <mark>SLELE</mark> S
Zebra finch $\rho$ 2(P)	FFVHSKRSFIHDTTTDNIMLRVFPDGHVLYS	SMR <mark>V</mark> TVTAMCNMDFS <mark>E</mark> FPLDSQTCSLELES
	* * * * * * * * * * * * * * * * * * * *	************
Chicken $\rho$ 2(P)	YAYTDEDLMLYWKNGNESLKTDEKISLSQFI	LIQKFHTTSRLAFYSSTGWYNRLYISFTLR
Zebra finch $ ho2$	YAYTDEDLMLYWKNGNESLKT <mark>V</mark> EKISLSQF1	LIQKFHTTSRLAFYSSTGWYNRLYISFTLR
Zebra finch $\rho$ 2(P)	YAYTDEDLMLYWKNGNESLKTDEKNSLSQFI	LIQKFHTTSRLAFYSSTGWYNRLYI <mark>N</mark> FTLR
	TM 1	TM 2
Chicken $\rho$ 2(P)	RHIFFFLLQTYFPATLMVMLSWVSFWIDRR	AVPARVSLGITTVLTMSTIITGVNASMPRV
Zebra finch $ ho2$	RHIFFFLLQTYFPATLMVMLSWVSFWIDRRA	AVPARVSLGITTVLTMSTIITGVNASMPRV
Zebra finch $\rho$ 2(P)	RHIFFFLLQTYFPA <mark>S</mark> LMVMLSWVSFWIDRRA	AVPARVSLGITTVLTMSTIITGVNASMPRV
	******	* * * * * * * * * * * * * * * * * * * *
	TM 3	
Chicken $\rho$ 2(P)	SYIKAVDIYLWVSFVFVFLSVLEYAAVNYL	ITVQERKERKLREKFP
Zebra finch $ ho2$	SYIKAVDIYLWVSFVFVFLSVLEYAAVNYL	TTVQERKERKLREKFP
Zebra finch $\rho$ 2(P)	SYIKAVDIYLWVSFVFVFLSVLEYAAVNYL	ITVQERKERKLRE <mark>R</mark> F <b>S</b>
	*******************************	* * * * * * * * * * * * *

**Figure 3.6b.** Alignment of the partial amino-acid sequence (in single letter code) encoding the zebra finch GABA<sub>C</sub> receptor  $\rho$ 2 subunit with the corresponding chicken sequence (GenBank accession number XP\_419839) and the predicted zebra finch sequence (as no peptide sequence was available it was translated from the nucleotide sequence, GenBank accession number XM\_054611).

#### (c) Sequence of the zebra finch $GABA_C$ receptor $\rho 3$ subunit

Zebra finch $\rho$ 3 Zebra finch $\rho$ 3(P) Chicken $\rho$ 3(P)	VGIDVQVESIDSISEVDMDFTMTLYLRHYWKDERI VGIDVQVESIDSISEVDMDFTMTLYLRHYWKDERI VGIDVQVESIDSISEVDMDFTMTLYLRHYWKDERI ************************************	LSFRSNKNKSMTFDGRLIKKIWVPDV LSFRSNKNKSMTFDGRLIKKIWVPDV LSFRSTKNKSMTFDGRLIKKIWVPDV
Zebra finch $\rho$ 3 Zebra finch $\rho$ 3(P) Chicken $\rho$ 3(P)	FFVHSKRSFIHDTTVENIMLRVYPDGNVLFSLRIT FFVHSKRSFIHDTTVENIMLRVYPDGNVLFSLRIT FFVHSKRSFIHDTTVEN ****************	YVSAM <mark>CFMDFSRFPLDTONC</mark> SLELE <mark>S</mark> YVSAMCFMDFSRFPLDTQNCSLELES YVSAMCFMDFSRFPLDTQNCSLELES
Zebra finch $\rho$ 3 Zebra finch $\rho$ 3(P) Chicken $\rho$ 3(P)	YAYNEDDLMLYWKHGNKSLSTDEHISLSQFFIEEE YAYNEDDLMLYWKHGNKSLSTDEHISLSQFFIEEE YAYNEDDLMLYWKHGN SLSTDEHISLSQFFIEEE *********************************	SASSGLAFYSSTGWYNRLFINFALR SASSGLAFYSSTGWYNRLFINFALR SASSGLAFYSSTGWYNRLFINFALR ************************************
Zebra finch p3 Zebra finch p3(P) Chicken p3(P)	RHIFFFVLQSYFPAMLMVMLSWVSFWIDRRAVPAF RHIFFFVLQSYFPAMLMVMLSWVSFWIDRRAVPAF RHIFFFVLQSYFPAMLMVMLSWVSFWIDRRAVPAF ************************************	VVSLGITTVLTMSTIMTGVSASMPQV RVSLGITTVLTMSTIMTGVSASMPQV RVSLGITTVLTMSTI <mark>I</mark> TGVSASMPQV
Zebra finch $\rho$ 3 Zebra finch $\rho$ 3(P) Chicken $\rho$ 3(P)	SYIKAVDVYLWISFLFVFLSVIEYAAVNYL SYIKAVDVYLWISFLFVFLSVIEYAAVNYL SYIKAVDVYLWISFLFVFLSVIEYAAVNYL ******	

**Figure 3.6c.** Alignment of the partial amino-acid sequence (in single letter code) encoding the cloned zebra finch GABA<sub>C</sub> receptor  $\rho$ 3 subunit with the corresponding chicken sequence (GenBank accession number XP\_428531) and the predicted zebra finch sequence (Genbank accession number XP\_002190668).

Zebra finch amino-acid  $\rho$ 3-subunit sequence exhibited 100% and 98.5% homology with the predicted zebra finch and chicken reference sequences respectively at a peptide level; any mis-matches between zebra finch and chicken sequences were found within the region encoding the N-terminus, although not within any well conserved domains such as the Cys-loop (Fig. 3.6c). The zebra finch GABA<sub>C</sub> receptor  $\rho$ 3-subunit gene (GenBank accession number XM\_00219068) is localised to chromosome 1 as is the corresponding gene in the chicken genome (GenBank accession number XM\_428531; Fig. 3.8), although this is some distance from the GABA<sub>A</sub> receptor gene cluster  $\gamma$ 3 $\alpha$ 5 $\beta$ 3 on the same chromosome. On the NCBI database the  $\rho$ 3-subunit gene is annotated with 9 exons translating into a polypeptide of 49.5kDa.

			$\downarrow \qquad \downarrow \downarrow \downarrow \downarrow \downarrow$	
zebra	finch	ρ1	FFVHSKRSFIHDTTTDNVMLRVQPDGKVLYSLRVTVTAMCNMDFSRFPLDTQTCSLEI	Ξ <mark>S</mark>
zebra	finch	ρ2	FFVHSKRSFIHDTTTDNIMLRVFPDGHVLYSMRITVTAMCNMDFSRFPLDSQTCSLEL	Ξ <mark>S</mark>
zebra	finch	ρ3	FFVHSKRSFIHDTTVENIMLRVYPDGNVLFSLRITVSAMCFMDFSRFPLDTQNCSLEL	Ξ <mark>S</mark>
			***************************************	* *
zebra	finch	ρ1	YAYTEDDLMLYWKNGNDSLKTDERISLSQFLIQEFHTTTKLAFYSSTGWYNRLYINFT	LR
zebra	finch	ρ2	<mark>Y</mark> AYTDEDLMLYWKNGNESLKTDEKISLSQFLIQKFHTTSRLAFYSSTGWYNRLYISFTI	LR
zebra	finch	ρ3	<mark>Y</mark> AYNEDDLMLYWKHGNKSLSTDEHISLSQFFIEEFSASSGLAFYSSTGWYNRLFINFA	LR
			***.::*******:**.**.***:***************	* *
			TM 1TM 2	
zebra	finch	ρ1	TM 1 TM 2 RHIFFFLLQTYFPATLMVMLSWVSFWIDRRAVPARVPLGITTVLTMSTIITGVNASMPI	٦V
zebra zebra	finch finch	ρ1 ρ2	TM 1 TM 2 RHIFFFLLQTYFPATLMVMLSWVSFWIDRAVPARVPLGITTVLTMSTIITGVNASMPI RHIFFFLLQTYFPATLMVMLSWVSFWIDRAVPARVSLGITTVLTMSTIITGVNASMPI	RV RV
zebra zebra zebra	finch finch finch	ρ1 ρ2 ρ3	TM 1 TM 2 RHIFFFLLQTYFPATLMVMLSWVSFWIDRAVPARVPLGITTVLTMSTIITGVNASMPI RHIFFFLLQTYFPATLMVMLSWVSFWIDRAVPARVSLGITTVLTMSTIITGVNASMPI RHIFFFVLQSYFPAMLMVMLSWVSFWIDRAVPARVSLGITTVLTMSTIMTGVSASMPQ ******:***	2V 2V 2V
zebra zebra zebra	finch finch finch	ρ1 ρ2 ρ3	TM 1 TM 2 RHIFFFLLQTYFPATLMVMLSWVSFWIDRAVPARVPLGITTVLTMSTIITGVNASMPI RHIFFFLLQTYFPATLMVMLSWVSFWIDRAVPARVSLGITTVLTMSTIITGVNASMPi RHIFFFVLQSYFPAMLMVMLSWVSFWIDRAVPARVSLGITTVLTMSTIMTGVSASMP( ******:**:**** ***********************	RV RV 2V :*
zebra zebra zebra zebra	finch finch finch	ρ1 ρ2 ρ3 ρ1	TM 1     TM 2       RHIFFFLLQTYFPATLMVMLSWVSFWIDRAVPARVPLGITTVLTMSTIITGVNASMPR       RHIFFFLLQTYFPATLMVMLSWVSFWIDRAVPARVSLGITTVLTMSTIITGVNASMPR       RHIFFFVLQSYFPAMLMVMLSWVSFWIDRAVPARVSLGITTVLTMSTIMTGVSASMPC       ************************************	RV RV 2V :*
zebra zebra zebra zebra zebra	finch finch finch finch finch	ρ1 ρ2 ρ3 ρ1 ρ2	TM 1     TM 2       RHIFFFLLQTYFPATLMVMLSWVSFWIDRAVPARVPLGITTVLTMSTIITGVNASMPI       RHIFFFLLQTYFPATLMVMLSWVSFWIDRAVPARVSLGITTVLTMSTIITGVNASMPI       RHIFFFVLQSYFPAMLMVMLSWVSFWIDRAVPARVSLGITTVLTMSTIMTGVSASMPC       ******:******************************	RV RV 2V :* 7D
zebra zebra zebra zebra zebra zebra	finch finch finch finch finch finch	ρ1 ρ2 ρ3 ρ1 ρ2 ρ3	TM 1 TM 2 RHIFFFLLQTYFPATLMVMLSWVSFWIDRAVPARVPLGITTVLTMSTIITGVNASMPR RHIFFFLLQTYFPATLMVMLSWVSFWIDRAVPARVSLGITTVLTMSTIITGVNASMPR RHIFFFVLQSYFPAMLMVMLSWVSFWIDRAVPARVSLGITTVLTMSTIMTGVSASMPC ******:**:***************************	RV RV 2V :*

**Figure 3.7**. Alignment of all cloned zebra finch  $GABA_C$  receptor subunit amino acid sequences. TM domains are annotated, (\*) denote complete homology between sequences and dots represent low (.) or partial (:) homology, (-) gaps were introduced to maintain the alignment. The 15AA Cys-loop domain is well conserved (red) with the two peripheral cysteines and central PMD/PLD sequence identical in all sequences ( $\downarrow$ ). Conserved Ser<sup>171</sup> and Tyr<sup>172</sup> residues shortly after the Cys-loop consensus sequence are indicated in yellow. Within the TM2 region, potential channel lining residues are highlighted in grey.

The high sequence conservation that exists between all the  $GABA_C$  receptor subunits is illustrated in Figure 3.7. As with the  $GABA_A$  receptor subunits, regions of strong sequence conservation include the Cys-loop domain, serine and tyrosine residues within the N-terminus and hydrophobic transmembrane domains. There is high similarity between all the zebra finch and chicken  $GABA_A$  and  $GABA_C$  receptor subunit sequences. When compared to mammalian sequences the identity remains high at ~90% (data not shown), indicative of common ancestry and an evolutionary conservation of function. This is further validated by the chromosomal arrangement of the genes within the zebra finch genome, which is almost identical to that of humans (Fig. 3.8).



**Figure 3.8**. Schematic depicting chromosomal arrangements and transcriptional orientation of isolated  $GABA_A$  and  $GABA_C$  receptor subunit genes within the zebra finch genome. Data has been adapted from information available on Ensembl and NCBI databases following BLAST analyses and data obtained from Tsang *et al.* (2007). Arrows denote forward ( $\rightarrow$ ) and reverse ( $\leftarrow$ ) transcriptional orientation. GABA<sub>A</sub> receptor subunit genes are labelled in black and GABA<sub>C</sub> receptor subunit genes are labelled in red.

Genes encoding the GABA<sub>A</sub> receptor subunits within the zebra finch genome are localised in clusters similar to those of the human genome (Fig. 3.8). There are four primary gene clusters, three of which comprise  $\alpha\beta\gamma$ -subunit encoding genes; in all cases the  $\beta$ -subunit gene is in an opposing transcriptional orientation to the  $\alpha$ - and  $\gamma$ -subunit genes. Subunit-encoding genes outside the  $\alpha\beta\gamma$  class ( $\pi$ ,  $\delta$  and  $\rho$ 1-3) are not found in clusters, although  $\rho$ 1- and  $\rho$ 2-subunit genes are in close proximity on chromosome 3. The zebra finch  $\delta$ -subunit gene is the only gene located on a separate chromosome from other GABA<sub>A</sub> receptor genes. Although the zebra finch  $\pi$ -subunit gene is located on chromosome 13 with genes encoding  $\alpha$ 1,  $\alpha$ 6,  $\beta$ 2,  $\gamma$ 2 subunits it is some distance from the cluster, as in the human genome (Simon *et al.*, 2004).

Chapter Three

## **3.4 Discussion**

Partial cDNAs encoding GABA<sub>A</sub> receptor  $\alpha 3$ ,  $\alpha 4$ ,  $\alpha 6$ ,  $\gamma 1$ ,  $\delta$  and  $\pi$  subunits and GABA<sub>C</sub> receptor p1-p3 subunits were amplified from zebra finch. Subsequently, discriminative real-time RT-PCR primers and *in situ* hybridisation probes could be designed in order to systematically map and quantify expression of all subunit genes within the zebra finch song system. Peptide alignments with corresponding chicken sequences (which are phylogenetically similar) presented high sequence similarity (> 97%, except for  $\pi$ subunit sequences which exhibited only 90% similarity), thus confirming their identity (Fig. 3.2a-f) and indicating a strong conservation of function. At least five clones were sequenced from two independent RT-PCR experiments for each subunit; ensuring any Taq DNA polymerase and sequencing errors could be easily identified. All RT-PCRs gave multiple products (due to the degeneracy of the primers), but amplicons of expected size were excised, purified, cloned and sequenced. It was only after all the subunit cDNAs (apart from the  $\pi$  subunit) had been amplified that predicted zebra finch GABA<sub>A</sub> receptor subunit sequences were released on NCBI, following the sequencing of the zebra finch genome. These were published as predicted sequences but when aligned with the sequences obtained from the zebra finch clones there was > 95%homology in all cases.

All isolated sequences encoded the extracellular N-terminus (encompassing the Cysloop and conserved serine and tyrosine residues), characteristic putative transmembrane domains TM1-TM3 and their linking intracellular/extracellular loops. All GABA<sub>A</sub> receptor subunit sequences contained the highly conserved octapeptide TTVLTMTT (single-letter code) within the TM2 domain (Fig. 3.5). In GABA<sub>C</sub> receptor subunit sequences there was an amino acid substitution giving the sequence TTVLTSTT, which was identical in all three subunit encoding sequences (Fig. 3.7). This sequence (towards the extracellular end of TM2) incorporates residues which line the anion pore (detailed on Figs. 3.5 and 3.7) and is the region in which disulphide bonds are formed (Horenstein *et al.*, 2001), and is therefore highly conserved. Characterisation of this region in cation-selective members of the LGIC superfamily has deduced that the ion channel gate is located to TM2 or the TM1-TM2 linker, although the precise location remains an issue of some debate (Peters *et al.*, 2005).

The Cys-loop consensus sequence is located within the region encoding the extracellular N-terminus (ligand binding) domain and first transmembrane pore-forming domain. It is a 15AA disulphide-bridged region which is highly conserved across all known vertebrate GABAA receptor sequences (and other members of the LGIC superfamily, Simon et al., 2004). Results indicated that two peripheral cysteine residues and the PMD/PLD residues at positions 9-11 were conserved in all cloned GABA<sub>A</sub> and GABA<sub>C</sub> receptor sequences (Figs. 3.2a-f and 3.6a-c). Due to the location, it is proposed that in nAChRs (and hence other LGICs) the Cys-loop functions as a pivot which effectively stabilises the receptor conformation during the gating process (Unwin et al., 2002; Grutter et al., 2005; Unwin, 2005). Data provided by mutations of the motif indicates the same function in GABA<sub>A</sub> receptors (Tierney et al., 2008). The Cys-loop disulphide bond is essential for receptor assembly (Blount and Merlie, 1990; Fu and Sine, 1996). The highly conserved hydrophobic PMD/PLD domain (at the tip of the Cys-loop) is not required for normal protein folding and expression, but rather is involved in impedance of drug-potentiated channel gating and direct activation of the receptor in a subunit-dependant manner (Tierney et al., 2008). Poor homology between cloned and predicted zebra finch  $\alpha$ 3-subunit sequences within the Cys-loop domain was unanticipated. The isolated zebra finch sequence presents a characteristic Cys-loop domain sequence (Fig. 3.2a), however in the predicted  $\alpha$ 3-subunit sequence (annotated from the published zebra finch genome (http://genome.wustl.edu/genomes/view/ taeniopygia\_guttata/), the initial cysteine residue was missing and the highly conserved central motif was changed. Predicted/annotated sequences are not always accurate (as inferred by the name) and are generated by computational algorithms, leaving much room for error as was the case here. This highlights the requirement for experimental evidence to confirm the integrity of theoretical/predicted sequences.

Six residues downstream from the Cys-loop consensus sequence are located conserved serine (S) and tyrosine (Y) residues (Figs. 3.5 and 3.7). These are both highly conserved within the GABA<sub>A</sub> and GABA<sub>C</sub> receptor subunit sequences with the exception of the  $\pi$ -subunit gene, whereby the hydrophilic tyrosine (Y) is replaced with hydrophobic tryptophan (W). Mutagenesis studies with the  $\gamma$ 2 subunit have demonstrated that substitution of Ser<sup>171</sup> and Tyr<sup>172</sup> with glycine and threonine respectively, prevented receptor association and expression at the cell surface (Jin *et al.*, 2004). Yet this effect

was not evident when the same mutations were introduced to the  $\alpha 2$  and  $\beta 1$  subunits. Furthermore, previous work whereby Tyr<sup>172</sup> was substituted with a serine residue in the  $\gamma 2$ -subunit sequence showed no discernable effects on binding or association (Whiting *et al.*, 1997), thus it may be only the Ser<sup>171</sup> residue which is critical for receptor association.

Another observation was the high sequence homology of the TM1-TM2 loop within different subunit classes; this was indicative of a conserved physiological function that may be specific to each class of subunits (Fig. 3.5). Previous studies have shown mutations of residues within this region of the  $\beta$ 3 subunit alter ion selectivity of the pore from anion to cation without affecting the pharmacology. This was only apparent with the  $\beta$ 3 subunit as the same mutations in the  $\alpha$ 2- and  $\gamma$ 2-subunit polypeptides elicited no such changes (Jensen et al., 2002), further substantiating the theory of subunit-class specific effects. Greatest variability between the cloned sequences and the chicken/zebra finch reference sequences was within regions encoding the large intracellular loop between TM3 and TM4 (Fig. 3.5). This is the most variable region within, otherwise well conserved sequences, and serves to differentiate the subunits. This linker region is well situated for interactions with the intracellular environment, the primary function appears to be trafficking of the receptors via interactions of the loop and associated proteins such as GABAA receptor-associated protein (GABARAP; Chen et al., 2005); golgi-specific zinc-finger protein (Keller et al., 2004); GABA<sub>A</sub> receptor interacting factor-1 (Kittler et al., 2004); phospholipase C-related inactive protein type 1 (Terunuma et al., 2004) and ubiquitin-like protein Plic-1 (Bedford et al., 2001), all of which are subunit specific (due to variability within the loop sequences) and demonstrate an involvement in receptor trafficking (for further details refer to section 1.2.8). Thus variations between subunit sequences are necessary to provide different binding sites for a range of associated proteins fulfilling diverse physiological functions. It is also postulated that the intracellular loop may influence ion conduction (Peters et al., 2005).

Isolation of GABA<sub>A</sub> receptor  $\pi$ -subunit partial cDNA was challenging due to contradictory information in databases (NCBI and Ensembl), which resulted in unreliable sequences with which to design even degenerate RT-PCR primers. Two chicken GABA<sub>A</sub> receptor  $\pi$ -subunit sequences were available (GenBank accession

102

numbers: XM 414507 and XM 426524). The former was reported to be located on chromosome 13 and the latter on chromosome 6. Both were generated from predictive automated computational analysis sharing only 55% identity to each other and, 49% and 56% identity, (at amino-acid level) to corresponding rat and human sequences respectively. This degree of similarity was low compared to other subunits, as GABA<sub>A</sub> receptor subunit sequence identity between vertebrate species is close to 90% (Fig. 3.2af). Degenerate primers based on these sequences did not successfully amplify the  $\pi$ subunit cDNA. However, the published zebra finch genome predicted a  $\pi$ -subunit sequence localised to chromosome 13 sharing 77% homology, at a nucleotide level, with the corresponding gene in rat (GenBank accession numbers: rat, NM\_031029, zebra finch XM\_002188687) and this was the sequence used to design the final set of primers. Long and short forms of the  $\pi$ -subunit cDNA were isolated from the zebra finch brain and both exhibited 90% sequence identity with the chicken reference sequence (Fig. 3.2f). Levels of heterogeneity within the GABA<sub>A</sub> receptor family have been furthered by the identification of splice variants; however, these have not been described for all subunits. A novel splice variant of the GABA<sub>A</sub> receptor  $\pi$ -subunit gene has been identified in this study, which has not been previously reported (Figs. 3.2f and 3.4). Analysis of the human genome has revealed the  $\pi$ -subunit gene to compose 10 exons (9 coding), conforming to normal GABA<sub>A</sub> receptor coding patterns (Simon et al., 2004), and in the zebra finch and chicken genomes there are 8 and 9 coding exons respectively. Interestingly, no splice variants have been reported for the  $GABA_A$ receptor  $\pi$  subunit, in any species. In silico examination of the human genome revealed the presence of an Alu sequence within the intronic sequence which, should a favourable intronic splice site be created, would result in a well expressed splice variant (Simon et al., 2004). However, these short interspersed elements are exclusive to primates, thus a splice variant (similar to that seen with the  $\gamma$ 2-subunit gene) would not be observed within the zebra finch gene. The variant reported here is likely to be a non-functional protein due to the deletion of a large portion (37AA) of exon 7 corresponding to part of TM2 (which forms intrinsic chloride pore lining) and TM3 and the extracellular loop lying between them. Analysis of the deleted sequence revealed no loss of phosphorylation sites.

Another observation was that during amplification of  $\pi$ -subunit cDNA from zebra finch brain; agarose gel bands representing both amplicons (full length and splice variant) were of equal intensity (data not shown). It could therefore, tentatively be speculated that the truncated form appeared at a frequency of ~50%. However, it must be considered that this may have been an artefact of the RT-PCR due to the increased efficiency of amplification of a shorter product. RT-PCRs and subsequent sequencing reactions were performed in triplicate (with cDNA from different zebra finch brains) to ensure the data were consistent and the variant was not an anomaly, all reactions gave the same result.

Furthermore, initial attempts were made to isolate the  $\pi$ -subunit utilised zebra finch cDNA from ovarian, gastrointestinal and lung tissue due to its primarily peripheral localisation (Symmans et al., 2005; Xiang et al., 2007). Following poor amplification of expected products, the primers were tested with brain-derived cDNA and two products were immediately apparent. This was in conflict with published data which suggests that  $\pi$ -subunit mRNA is rarely expressed in the brain (Hedblom and Kirkness, 1997; Jin et al., 2005) rather, it is predominantly expressed in alveolar tissues of the lungs (Jin et al., 2005; Xiang et al., 2007) where receptors containing this subunit are implicated in alveolar fluid homeostasis (Jin *et al.*, 2006). In addition, the  $\pi$ -subunit transcript has also been detected in ovaries and smooth muscle cells of the urinary bladder and uterus in rat, where it has been suggested that  $\pi$ -subunit-containing GABA<sub>A</sub> receptors play a role in the regulation of muscle contraction (Hedblom and Kirkness, 1997; Fujii and Mellon, 2001). Moreover, it has been defined as a potential marker for detection of breast lymph node metastasis (Backus et al., 2005; Symmans et al., 2005), and is overexpressed in pancreatic adenocarcinomas (malignant tumour originating in glandular tissue; Johnson and Haun, 2005). Robust expression in zebra finch brain may be indicative of a physiological role more relevant to avian species than mammalian systems. However, no such differences are documented for any of the other  $GABA_A$ receptor subunit transcripts. Low expression in mammalian CNS may also explain why no splice variants have been previously detected, or the variant may be exclusive to lower vertebrates. It is conceivable that the alternatively-spliced variant is restricted to the brain/CNS and not found in peripheral tissues, akin to the ɛ-subunit transcript variants, which are differentially expressed. The full-length  $\varepsilon$  transcript is well

expressed in heart and lung tissue, and at low levels in the rat brain (Davies *et al.*, 1997; Whiting *et al.*, 1997; Wilke *et al.*, 1997), and two further variants are expressed in peripheral tissues (Wilke *et al.*, 1997). Although no specific function has been assigned to particular isoforms of the GABA<sub>A</sub> receptor subunits; there is evidence of regional-(Kirkness and Fraser, 1993; Harvey *et al.*, 1994; Korpi *et al.*, 1994; Paulson *et al.*, 2000; Fuchs and Celepirovic, 2002; Mu *et al.*, 2002), and developmental-specific (Kirkness and Fraser, 1993; Fuchs and Celepirovic, 2002; Mu *et al.*, 2002) regulation of expression within tissues (most notably the brain), which is suggestive of functionality. Splice variants are important as they have been implicated in disease states (Huntsman *et al.*, 1998; Dredge *et al.*, 2001; Volk *et al.*, 2002; Zhao *et al.*, 2009) and severely truncated polypeptides (such as  $\alpha 4$  and  $\pi$  subunits) may perform a regulatory function in the trafficking and assembly of certain GABA<sub>A</sub> receptors (Mu *et al.*, 2002).

Zebra finch was the second avian species to have its genome sequenced, after that of the chicken. There is almost identical similarity between the chicken and zebra finch genome with the exception of chromosome 4 of which in passerine species there are two chromosomes, denoted 4 and 4A (Derjusheva et al., 2004). Therefore some of the genes present on chromosome 4 and 4A of the zebra finch are present on chromosome 4 in the chicken. Within the NCBI gene database, predicted sequences encoding the zebra finch  $\alpha$ 1- and  $\alpha$ 2-subunit polypeptides have been localised to chromosomes 4 and 13 of the genome respectively. In contrast to the chicken genome, where the  $\alpha$ 1-subunit gene is on chromosome 13 and the  $\alpha$ 2-subunit gene is on chromosome 4. As the remainder of the GABA<sub>A</sub> receptor subunit chromosomal locations are identical to that of the chicken, it was assumed this may be an error in the annotation of the zebra finch sequences. Preliminary BLAST searches probing the chicken genome with the predicted zebra finch  $\alpha$ 1-subunit sequence produced a match within chromosome 4 (location of  $\alpha$ 2subunit gene), furthermore, a second analysis with the zebra finch  $\alpha$ 2-subunit sequence produced a match within chromosome 13 in the chicken genome (location of  $\alpha$ 1-subunit gene). Nucleotide alignments have demonstrated that the zebra finch  $\alpha$ 2-subunit sequence was 93% identical to the  $\alpha$ 1-subunit sequence of chicken, indeed alignments of chicken and zebra finch  $\alpha$ 2-subunit sequences showed only 65% homology and with the  $\alpha$ 1-subunit sequence showed 68% homology, but alignment of zebra finch  $\alpha$ 1subunit and chicken  $\alpha$ 2-subunit sequences gave a sequence identity of 90%. It was therefore concluded that the predicted zebra finch  $\alpha 1$ - and  $\alpha 2$ -subunit sequences (GenBank accession number XM\_002194306 and XM\_002193396 respectively) have been mis-labelled, but the chromosomal locations are correct. This would then be in accordance with the clustering of the same genes within the human genome. Again this highlights the requirement for experimental evidence to confirm the integrity of theoretical/predicted sequences.

Isolated zebra finch sequences were aligned with the whole zebra finch genome and in agreement with the clustering of the GABA<sub>C</sub> receptor subunit genes within the human genome (Fig. 3.8), the zebra finch  $\rho$ 1- and  $\rho$ 2-subunit genes are located together on chromosome 3 (chromosome 6 in humans; Cutting et al., 1992) and the p3-subunit gene is located on chromosome 1 (chromosome 3 in humans; Bailey et al., 1999; Simon et al., 2004). Isolated sequences encoding the GABA<sub>C</sub> receptor  $\rho$ 1- and  $\rho$ 2-subunits exhibit the greatest similarity (84%), with the p3-subunit sequence sharing 74% and 67% homology with sequences encoding  $\rho 1$  and  $\rho 2$  subunits respectively. High identity between the three different subunits is indicative of a common ancestor,  $\rho$ 1- and  $\rho$ 2subunit genes are closely located on the same chromosome, and so are likely to be a consequence of local duplication during evolution (Bailey et al., 1999), whereas it is purported that the  $\rho$ 3-subunit gene may be a product of a duplication of a  $\rho$ 1-/ $\rho$ 2subunit gene ancestor (Bailey et al., 1999). GABA<sub>C</sub> receptor subunits display relatively low sequence identity with their GABA<sub>A</sub> counterparts (28-42% in human homologues); suggesting an early evolutionary divergence from the GABAA receptor subunit genes (Bailey et al., 1999). Native GABA<sub>A</sub> receptors may be composed of their chromosomal partners (Barnard et al 1998). They are not thought to commonly form homomers and subunit genes are found in  $\alpha\beta\gamma$  clusters on chromosomes (Simon *et al.*, 2004; Darlison et al., 2005; Olsen and Sieghart, 2008), with the  $\beta$ -subunit gene in an opposite transcriptional orientation to  $\alpha$ - and  $\gamma$ -subunit genes (Tsang *et al.*, 2007). For example the most prevalent subtype in the mammalian CNS is  $\alpha 1\beta 2\gamma 2$  (Whiting, 2003) and the respective genes are located in a single cluster on chromosome 5 (Simon et al., 2004). Chromosomal positions of the GABA<sub>C</sub> receptor subunit genes both in the human and zebra finch genome do not easily lend themselves to the possibility of co-expression, however there is a small amount of evidence to the contrary (Qian and Pan, 2002; Milligan et al., 2004). Furthermore, confirmed native GABA<sub>A</sub> receptor subtypes such as

 $\alpha 3\beta\gamma 2$ ,  $\alpha 4\beta\gamma 2$ ,  $\alpha 4\beta 3\delta$ ,  $\alpha 6\beta 3$ , $\delta$  (Olsen and Sieghart, 2008), to name but a few, all compose of polypeptides from up to three different chromosomes, in light of this, GABA<sub>A</sub> and GABA<sub>C</sub> receptor subunit association cannot be completely discounted due to chromosomal positioning. However, the GABA<sub>C</sub> receptor subunits more readily form homomeric assemblies (Bormann, 2000), and do so frequently (Olsen and Sieghart, 2008), so there is less requirement for them to be close chromosomal partners.

It has been generally accepted that in avian species,  $\beta$ 1- or  $\gamma$ 3-subunit polypeptides have been lost (Darlison and Albrecht, 1995) or have evolved into non-functioning pseudogenes (Darlison et al., 2005). However, in the recent sequencing of the zebra finch genome, sequences encoding these subunits have been identified, the  $\beta$ 1-subunit gene is localised to chromosome 4A (GenBank accession number XM\_002186787) and the  $\gamma$ 3-subunit gene is on chromosome 1 (GenBank accession number XM\_002197375). Furthermore, predicted sequences encoding these polypeptides are present in the chicken genome where the  $\gamma$ 3-subunit gene is reported to be located on chromosome 1 (GenBank accession number XM\_001233420) and the  $\beta$ 1-subunit gene (GenBank accession number XM 420267) on chromosome 4, in accordance with the chromosomal assignment of the zebra finch genes. Predicted exonic arrangements show 9 coding exons for both chicken and zebra finch  $\gamma$ 3-subunit genes and 9 and 7 exons for the chicken and zebra finch  $\beta$ 1-subunit genes respectively. However, it was observed that the chicken and zebra finch  $\beta 1$  genes were localised on chromosome 4/4A in a cluster with genes encoding  $\alpha 3$  and  $\gamma 4$  subunits. In the human genome (on the X chromosome) the gene cluster is  $\varepsilon$ ,  $\alpha 3$  and  $\theta$ , where  $\varepsilon$  is the orthologue of  $\gamma 4$  and  $\theta$  is the orthologue of  $\beta$ 4 (Darlison *et al.*, 2005), therefore it would be expected that in the zebra finch and chicken genome it would be the  $\beta$ 4-subunit gene occupying this position, not the  $\beta$ 1-subunit gene. Within the predicted zebra finch GABA<sub>A</sub> receptor subunit sequences, there was no gene identified which encodes the  $\beta$ 4-subunit. Therefore, alignments were prepared of the predicted zebra finch *β*1-subunit and *β*4-subunit genes (cloned within the laboratory, data not published) which demonstrated almost perfect identity. High similarity would be anticipated between sequences of two genes from the same class within the same organism, approximately 60-80% identity (Darlison et al., 2005), but not as high as 99% (see Appendix for alignment). Subsequently it appears that the databases have annotated the sequences incorrectly and the  $\beta$ 1-subunit gene is

essentially identical to the previously cloned  $\beta$ 4-subunit gene. This would also mean that the chromosomal allocation of the genes within the zebra finch would be identical to that of the human genome with a cluster of the  $\beta$ 4-,  $\alpha$ 3- and  $\gamma$ 4-subunit encoding genes. In the chicken genome, there is the verified  $\beta$ 4-subunit gene (Bateson *et al.*, 1991) and a predicted  $\beta$ 1-subunit gene, alignment of these two genes displayed > 99% homology, indicating that the predicted  $\beta$ 1-subunit gene was in fact the previously cloned  $\beta$ 4-subunit gene, and now the  $\beta$ 4-subunit gene can be localised to chromosome 4 (see Appendix for alignment). Again, this highlights the need to verify the integrity of predicted sequences and their annotations.

Sequences encoding all zebra finch GABA<sub>A</sub> and GABA<sub>C</sub> receptor subunits have been isolated and their identity verified. A novel splice variant has been described for the  $\pi$ -subunit gene, which is expressed along with the full-length transcript in the avian brain. Further expression studies involving this subunit would be interesting as no splice variants have been previously reported and data pertaining to  $\pi$ -subunit gene expression in brain is scarce. Analysis of the cloned zebra finch GABA<sub>A</sub> and GABA<sub>C</sub> receptor subunit sequences with predicted sequences from the zebra finch genome have revealed some discrepancies, demonstrating that caution must be taken when using such data and sequences should all be validated with experimental evidence. All cloned sequences were subsequently used to design real-time RT-PCR primers and *in situ* hybridisation probes for expression studies as detailed in the following chapter.

# 4. EXPRESSION OF GABA<sub>A</sub> AND GABA<sub>C</sub> RECEPTOR SUBUNIT GENES IN THE SONG SYSTEM

# 4.1 Introduction

Excitatory action elicited by glutamate in the brain is counterbalanced by the inhibitory activity of  $\gamma$ -aminobutyric acid (GABA) acting at post-synaptic receptors, resulting in a finely tuned neuronal network (Hablitz et al., 2009). Heteromeric GABA<sub>A</sub> receptors are by far the most abundantly distributed of the GABA receptors in the central nervous system CNS (Bormann, 2000). As previously mentioned, their arrangement is pentameric and it is the stoichiometry of an individual receptor that dictates its electrophysiological, pharmacological and functional properties (Whiting 2003a, b; Wafford, 2004; Rudolph and Möhler, 2006; Olsen and Sieghart, 2008). Over the past 20 years 16 GABA<sub>A</sub> receptor subunits have been identified in the mammalian CNS:  $\alpha$ 1- $\alpha$ 6;  $\beta 1-\beta 3$ ;  $\gamma 1-\gamma 3$ ;  $\pi$ ,  $\theta$ ,  $\delta$  and  $\varepsilon$ , each encoded for by a separate gene (Simon *et al.*, 2004). However in avian species there are only 14 documented GABA<sub>A</sub> receptor subunits: α1- $\alpha 6$ ;  $\beta 2$ - $\beta 4$ ;  $\gamma 1$ ,  $\gamma 2$ ,  $\gamma 4$ ,  $\delta$  and  $\pi$ ; where the  $\gamma 4$  and  $\beta 4$  subunits are orthologues of mammalian  $\varepsilon$  and  $\theta$  subunits respectively (Darlison *et al.*, 2005). The most frequent combination of subunits is  $2\alpha$ 's,  $2\beta$ 's and a single  $\gamma$ ; with  $\alpha 1\beta 2\gamma 2$  as the predominant subtype in mammals (Whiting, 2003a). The promiscuous nature of the subunits in their associations provides potential for a multitude of receptor subtypes, although preferred assemblies do exist, but after two decades of research only 11 confirmed native subtypes have been identified in mammals (Table 1, Chapter One; Olsen and Sieghart, 2008, 2009). GABA<sub>A</sub> receptors are of considerable clinical interest due to their sensitivity to benzodiazepines, barbiturates, ethanol, volatile anaesthetics and steroids; thus the elucidation of specific receptor subtypes would enable design of more specific therapeutic agents (Sieghart and Ernst, 2005). One of the first steps in identifying GABA<sub>A</sub> receptor subtypes is determining which subunits are co-expressed in vivo. Several previous studies have spatially mapped expression of GABA<sub>A</sub> receptor subunit genes in mammalian brain, both in embryonic and post-natal rat (Laurie et al, 1992a, b; Wisden et al, 1992; Sperk et al., 1997; Pirker et al., 2000; Pöltl et al., 2003) and monkey (Macaca mulatta) brain (Kultas-Ilinsky et al., 1998; Huntsman et al., 1999). Although, none have documented all the GABA<sub>A</sub> receptor subunit genes, or considered

the  $\rho$ -subunit genes of the GABA<sub>C</sub> receptor in parallel, despite considerable speculation that these should be classified as part of the GABA<sub>A</sub> family (Olsen and Sieghart, 2008; Collingridge *et al.*, 2009). All studies have demonstrated discrete, yet overlapping cellular and regional distributions of GABA<sub>A</sub> receptor subunit transcripts in brain which provides a preliminary insight into which subunits may be co-expressed (reviewed by Sieghart and Sperk, 2002). Furthermore, expression studies have indicated differences in GABA<sub>A</sub> receptor subunit gene expression during different developmental stages, demonstrating that subtype composition is highly variable depending on brain region and is also subject to temporal regulation (Laurie *et al.*, 1992a; Huntsman *et al.*, 1999), further details in section 4.3.

#### 4.1.1 Heteromerisation of GABA<sub>A</sub> and GABA<sub>C</sub> receptors

Recent speculation has suggested potential heteromerisation of GABA<sub>A</sub> and GABA<sub>C</sub> receptor subunits (Milligan *et al.*, 2004). All three subunit genes ( $\rho$ 1- $\rho$ 3) are transcribed in rat brain; though  $\rho$ 1 and  $\rho$ 2 at significantly higher levels than the  $\rho$ 3 subunit (Ogurusu & Shingai, 1996; Milligan *et al.*, 2004). Furthermore, Qian and Pan (2002), Ekema *et al.* (2002) and Milligan *et al.* (2004) revealed that the GABA<sub>A</sub> receptor  $\gamma$ 2-subunit polypeptide can associate with the GABA<sub>C</sub> receptor  $\rho$ 1 and  $\rho$ 2 subunits *in vitro* to form functional recombinant receptors, indicating the potential for new 'hybrid' receptor subtypes in brain. Recombinant receptors comprising both GABA<sub>A</sub> and GABA<sub>C</sub> receptor activity (Qian and Pan, 2002), inferring that the combination of subunits from these two receptor families may yield a novel receptor subtype (Milligan *et al.*, 2004). For this reason, expression of GABA<sub>C</sub> receptor subunits has been included in the *in situ* hybridisation experiments (further information concerning the properties of GABA<sub>C</sub> receptors in section 1.3).

#### 4.1.2 Zebra finch song system

The zebra finch (*Taenopygia guttata*) song system has long since been used as a paradigm for studying the underlying molecular mechanisms of learning and memory; due to the discrete nature of song, the song system and the finite period of song development (Brainard and Doupe, 2002). It comprises a set of interconnected telencephalic nuclei which can be subdivided into two relatively distinct pathways

which control the acquisition and production of song (Nottebohm *et al.*, 1976). Firstly, the anterior forebrain pathway (AFP); this is important for the development of normal song in juveniles and adult song plasticity (Mooney, 1999; Woolley, 2004; Funabiki and Funabiki, 2007; Sober and Brainard, 2009). Notable nuclei within this pathway include: Area X, the lateral magnocellular nucleus of the anterior nidopallium (LMAN) and the medial nucleus of the dorsolateral thalamus (DLM). Secondly, the vocal motor pathway (VMP), which is primarily responsible for song production; this comprises the NIf (nucleus interfacialis of the nidoapllium), HVC (formal name) and RA (robust nucleus of the arcopallium; Nottebohm et al., 1976). This pathway links to the syrinx via projections from the RA to the tracheosyringeal portion of the hypoglossal nucleus (nXllts). These forebrain systems which control motor systems are exclusive to vocal learners e.g. humans, songbirds and some cetaceans (Brainard and Doupe, 2002; Bolhius and Gahr, 2006). The avian brain displays many comparable structures and pathways to mammalian systems (Farries and Perkel, 2000; Jarvis et al., 2005; Mooney and Prather, 2005; Bolhuis and Gahr, 2006; Scott and Lois, 2007) and there are striking parallels between birdsong and speech production in humans, most notably in infants; hence the fundamental neuronal mechanisms appear to be similar (Doupe and Kuhl, 1999; Kuhl, 2004); discussed further in section 1.4.8.

#### 4.1.3 GABA<sub>A</sub> receptors in zebra finch song system

GABA is widely distributed in the song system (Pinaud and Mello, 2007), but less work has involved the identification and functionality of GABA<sub>A</sub> receptors within the telencephalon. Pharmacological evidence confirms the presence of GABA<sub>A</sub> receptors within all the major nuclei of the song system (for example, Carlisle *et al.*, 1998; Vicario and Raksin, 2000; Farries *et al.*, 2005; Mooney and Prather, 2005; Ölveczky *et al.*, 2005). In adult zebra finches, GABA<sub>A</sub> receptors have been identified by electrophysiological studies to occur on HVC neurons projecting to Area X as well as on HVC interneurons (Dutar *et al.*, 1998), on projecting neurons of the RA (Spiro *et al.*, 1999) and on small spiny Area X neurons (Farries *et al.*, 2005). However, the role of GABA<sub>A</sub> receptors expressed in the song control system is not well understood. It has been postulated that in HVC, GABA<sub>A</sub> receptors might be involved in the establishment of the auditory-vocal mirror cell function of neurons projecting to Area X neurons (Prather *et al.*, 2008). In RA GABA<sub>A</sub> receptors appear to be involved in the coordinated firing of multiple projection neurons (Spiro *et al.*, 1999) and so contribute to the initiation and temporal pattern of vocalisation (Vicario and Raksin, 2000). In Area X, GABA<sub>A</sub> receptors affect the activity of indirect but not direct innervations of pallidal Area X neurons projecting to DLM (Farries et al., 2005). More recently, in situ hybridisation studies have localised  $\gamma$ 4-subunit mRNA in many of the major song system nuclei, suggesting that GABA<sub>A</sub> receptors containing this polypeptide play a role in the regulation of the acquisition and production of song (Thode et al., 2008). Besides this, no other studies have thoroughly investigated the distribution of the GABA<sub>A</sub> or GABA<sub>C</sub> receptor subunit genes within the song system. Here is provided the first complete study documenting spatially and quantitatively, the expression of all the avian GABA<sub>A</sub> and GABA<sub>C</sub> receptor subunit genes (with the exception of the  $\pi$  subunit; which has a primarily peripheral localisation; Symmans et al., 2005; Xiang et al., 2007) in four nuclei of the adult male zebra finch song system (LMAN, Area X, HVC and RA). In addition, real-time RT-PCR was employed to determine the temporal expression profile of selected GABA<sub>A</sub> receptor subunit genes ( $\alpha$ 1- $\alpha$ 4,  $\beta$ 2,  $\beta$ 3 and  $\gamma$ 4) in the song system (Area X, HVC and RA) throughout different stages of song development (PHD 30, 55, 80 and 100). This data provides a basis for further investigations into subtype composition and the functional role of GABA<sub>A/C</sub> receptors in this learning and memory model.

Chapter Four

# 4.2 Results

Transcript-specific oligonucleotide probes were utilised to spatially and quantitatively investigate the distribution of all documented avian GABA<sub>A</sub> receptor subunit transcripts (namely  $\alpha 1 - \alpha 6$ ,  $\beta 2 - \beta 4$ ,  $\gamma 1$ ,  $\gamma 2$  and  $\gamma 4$  and  $\delta$ ) in selected nuclei of the adult (PHD >100) male zebra finch song system. The  $\pi$  subunit was not included due to its' primarily peripheral localisation (Symmans et al., 2005; Xiang et al., 2007). Initially, two probes targeting the corresponding mRNA of each subunit were used for hybridisation studies (Table 2.5), and in all cases the spatial distribution of the corresponding mRNA was identical for both probes (data not shown). Subsequent, grouped in situ hybridisation experiments for all probes were carried out on parallel 10µm coronal sections which were hybridised, then washed together and opposed for 4 weeks to the same X-ray film to reduce procedural variability. Anatomical and densitometric analysis enabled qualitative and quantitative analysis of the expression of each subunit gene respectively within the four nuclei examined (LMAN, Area X, HVC and RA). Expression of each gene could be quantitatively compared between the four nuclei, due to normalisation with an unrelated internal control region (nidopallium; refer to section 5.2.1). mRNA levels in each nucleus were calculated as a percentage of the internal control signal. One-way ANOVA tests were employed to determine any significant deviations in levels of mRNA from baseline levels (those measured in nidopallium). Cross-analysis of different probes within a single nucleus could not be completed quantitatively as the probes were not normalised to each other, due to different binding affinities.

Spatial distribution patterns of individual GABA<sub>A</sub> receptor subunit mRNAs are shown in Figures 4.1, 4.3 and 4.5. No two GABA<sub>A</sub> receptor subunit genes exhibited identical spatial expression patterns throughout the whole zebra finch brain. Furthermore, although mRNA levels were quantitatively variable, there was no complete absence of mRNA encoding any of the GABA<sub>A</sub> receptor subunits within the four nuclei of the song system examined. Taken together, this implicated that a variety of potential major and minor GABA<sub>A</sub> receptor subtype assemblies may exist within the zebra finch song system.



4.2.1 Expression of  $GABA_A$  receptor  $\alpha$ -subunit genes in the song system

**Figure 4.1.** Inverse autoradiographs highlighting the expression of GABA<sub>A</sub> receptor  $\alpha$ -subunit genes in four nuclei of the adult male zebra finch song system (LMAN, Area X, HVC and RA). **Abbreviations:** A, arcopallium; Cb, cerebellum; HA, apical part of the hyperpallium; HD, densocellular part of the hyperpallium; Hp, hippocampus; HVC (formal name); LSt, lateral striatum; M, mesopallium; MSt, medial striatum; N, nidopallium; NC, caudal nidopallium; LMAN, lateral magnocellular nucleus of the anterior nidopallium; RA, robust nucleus of the arcopallium; TeO, optic tectum; X, Area X.



**Figure 4.2**. Densitometric quantification of GABA<sub>A</sub> receptor  $\alpha$ -subunit mRNAs (a-f) in four selected nuclei (LMAN, Area X, HVC and RA) of the adult male zebra finch song system, relative to the nidopallium. Data is presented as mean  $\pm$  SE (n=5 per group, \*  $p \le 0.05$ ; \*\*  $p \le 0.01$ ). The dashed line indicates gene expression in the nidopallium; these readings were considered as baseline mRNA expression, at 100%. **Abbreviations:** Area X (formal name); HVC (formal name); LMAN, lateral magnocellular nucleus of the anterior nidopallium; RA, robust nucleus of the arcopallium.

Within LMAN, the most predominant gene expression for the GABA<sub>A</sub> receptor  $\alpha$ -subunit class was that encoding the  $\alpha$ 2 subunit (131%, Figs. 4.1 (B1) and 4.2b). All

other  $\alpha$ -subunit transcripts demonstrated a relatively homogenous distributions, whereby the amount of detectable mRNA was similar to that of their established individual baseline levels (Fig. 4.2). In Area X,  $\alpha$ -subunit gene expression levels were much more variable.  $\alpha$ 1- and  $\alpha$ 6-subunit mRNAs were both detected at baseline levels;  $\alpha$ 3- and  $\alpha$ 5subunit mRNAs were abundantly distributed ( $p \leq 0.01$  for both).  $\alpha$ 2- and  $\alpha$ 4-subunit mRNAs showed markedly low signal intensities (56%;  $p \leq 0.05$  and 68%;  $p \leq 0.01$ respectively) relative to their individual baseline mRNA levels. Interestingly akin to LMAN, the HVC illustrated a fairly homogenous distribution of the transcripts encoding the  $\alpha$ -subunit genes (Fig. 4.2), all at basal levels. All  $\alpha$ -subunit (except for  $\alpha$ 2) genes were well expressed throughout the RA (Fig. 1 and Fig. 4); with particular reference to the  $\alpha$ 4- and  $\alpha$ 5-subunit mRNAs where signal intensity was especially high (143% and 119% respectively;  $p \leq 0.01$ ) and  $\alpha$ 6-subunit mRNA levels were twice that of nidopallium.



4.2.2 Expression of  $GABA_A$  receptor  $\beta$ -subunit genes in the song system

**Figure 4.3.** Inverse autoradiographs highlighting the expression of  $GABA_A$  receptor  $\beta$ -subunit genes in four nuclei of the adult male zebra finch song system (LMAN, Area X, HVC and RA). **Abbreviations:** A, arcopallium; Cb, cerebellum; HA, apical part of the hyperpallium; HD, densocellular part of the hyperpallium; Hp, hippocampus; HVC (formal name); LSt, lateral striatum; M, mesopallium; MSt,

medial striatum; N, nidopallium; NC, caudal nidopallium; LMAN, lateral magnocellular nucleus of the anterior nidopallium; RA, robust nucleus of the arcopallium; TeO, optic tectum; X, Area X.



**Figure 4.4**. Densitometric quantification of GABA<sub>A</sub> receptor  $\beta$ -subunit mRNAs (a-c) in four selected nuclei (LMAN, Area X, HVC and RA) of the adult male zebra finch song system, relative to the nidopallium. Data is presented as mean  $\pm$  SE (n=5 per group, \*  $p \le 0.05$ ; \*\*\*  $p \le 0.001$ ). The dashed line indicates gene expression in the nidopallium; these readings were considered as baseline mRNA expression, at 100%. **Abbreviations:** Area X (formal name); HVC (formal name); LMAN, lateral magnocellular nucleus of the anterior nidopallium; RA, robust nucleus of the arcopallium.

The primary  $\beta$ -subunit transcript detected in the LMAN was that encoding the  $\beta$ 3 subunit; this showed reasonably high signal intensity within the nucleus (154%) relative to baseline levels (Fig. 4.3 (B1) and Fig. 4.4b), whereas  $\beta$ 2- and  $\beta$ 4-subunit transcripts were both present in LMAN at a basal level (Fig. 4.3 (A1 and C1) and Fig. 4.4a, c). Predominant  $\beta$ -subunit genes identified in Area X were  $\beta$ 3 and  $\beta$ 4 (Fig. 4.3 (B2 and C2) and Fig. 4.4b, c). The corresponding mRNAs displayed robust hybridisation signals in Area X (355% and 160% respectively) relative to baseline ( $p \le 0.001$  for both); whereas

the level of the  $\beta$ 2-subunit transcript was considerably less (81%; Fig. 4.4a). It was only the  $\beta$ 4 subunit which exhibited pronounced gene expression in HVC (114%  $p \leq 0.05$ ). All  $\beta$ -subunit transcripts were detected in the RA, but the relative abundance of each subunit was quite different (Figs. 4.3 and 4.4).  $\beta$ 2-subunit mRNA was observed at a basal level, similar to that of the nidopallium (Fig. 4.4a).  $\beta$ 3-subunit transcript was present at higher levels than in the nidopallium (137%; Fig. 4.4b), whereas mRNA encoding the  $\beta$ 4 subunit was considerably lower than its counterparts (75%; Fig. 4.4c).



## 4.2.3 Expression of $\gamma$ - and $\delta$ -subunit genes in the song system

**Figure 4.5.** Inverse autoradiographs highlighting the expression of GABA<sub>A</sub> receptor  $\gamma$ - and  $\delta$ -subunit genes in four nuclei of the adult male zebra finch song system (LMAN, Area X, HVC and RA). **Abbreviations:** A, arcopallium; Cb, cerebellum; HA, apical part of the hyperpallium; HD, densocellular part of the hyperpallium; Hp, hippocampus; HVC (formal name); LSt, lateral striatum; M, mesopallium; MSt, medial striatum; N, nidopallium; NC, caudal nidopallium; LMAN, lateral magnocellular nucleus of the anterior nidopallium; RA, robust nucleus of the arcopallium; TeO, optic tectum; X, Area X.



**Figure 4.6**. Densitometric quantification of GABA<sub>A</sub> receptor  $\gamma$ - and  $\delta$ -subunit mRNAs (a-d) in four selected nuclei (LMAN, Area X, HVC and RA) of the adult male zebra finch song system, relative to the nidopallium. Data is presented as mean  $\pm$  SE (n=5 per group, \*  $p \le 0.05$ ; \*\*  $p \le 0.01$ ; \*\*\*  $p \le 0.001$ ). The dashed line indicates gene expression in the nidopallium; these readings were considered as baseline mRNA expression, at 100%. **Abbreviations:** Area X (formal name); HVC (formal name); LMAN, lateral magnocellular nucleus of the anterior nidopallium; RA, robust nucleus of the arcopallium.

The most striking GABA<sub>A</sub> receptor subunit gene expression in LMAN was that of the  $\gamma 1$  subunit (540%;  $p \le 0.001$ ; Fig. 4.6a). This was a considerably higher increase from baseline than that of any of the other subunit genes in any of the song-system nuclei under observation. Abundant expression was also observed for  $\gamma 4$ - and  $\delta$ -subunit genes (148% and 151% respectively; Figs 4.6c, d) within LMAN; only  $\gamma 2$ -subunit mRNA was detected at basal levels (Fig. 4.6b).  $\gamma 4$ -subunit gene expression was robust in Area X (Fig. 4.5 (C2) and Fig. 4.6c), with corresponding mRNA levels significantly higher than baseline (190%;  $p \le 0.01$ ).  $\gamma 1$ - and  $\delta$ -subunit genes displayed nominal expression (61% and 70% respectively; Fig. 4.6a, d) whereas, similar to LMAN, mRNA encoding the  $\gamma 2$ 

subunit in Area X was at a level close to baseline (Fig. 4.5 (B2) and Fig. 4.6b). Within HVC,  $\gamma 1$ - and  $\gamma 2$ -subunit transcripts were present at levels similar to that of their individual baselines (Figs 4.5 (A4 and B4) and 4.6a, b).  $\delta$ -subunit transcript was detected at a level marginally higher (120%; Fig. 4.6d); but it was the  $\gamma 4$ -subunit gene which exhibited the most pronounced expression in this nucleus, nearly 3-fold greater than baseline (285%;  $p \le 0.01$ ; Fig. 4.6c).

	mRNA levels relative to nidopallium				
Nucleus	Higher	Equivalent	Lower		
LMAN	α2, β3, γ1***, γ4, δ	α1, α3-6, β2, β4, γ2	-		
Area X	α1, α3**, α5**, β3***, β4***, γ4**	α6, γ2	α2*, α4**, β2, γ1, δ		
HVC	β4*, γ4**, δ	$\alpha$ 1-6, $\beta$ 2, $\beta$ 3, $\gamma$ 1, $\gamma$ 2	-		
RA	α1, α3, α4**, α5**, α6, β3, γ4**	β2	$\alpha 2, \beta 4, \gamma 1, \gamma 2^*, \delta$		

**Table 4.1**. Summary data illustrating the quantitative distribution of all the GABA<sub>A</sub> receptor subunit mRNAs within selected nuclei of the adult zebra finch song system. Data is grouped according to whether the hybridisation signal was higher, equal to, or below that detected in the nidopallium. ( $p \le 0.05$ ; \*\*  $p \le 0.01$ ; \*\*\*  $p \le 0.001$ ). Abbreviations: Area X (formal name); HVC (formal name); LMAN, lateral magnocellular nucleus of the anterior nidopallium; RA, robust nucleus of the arcopallium.

 $\alpha$ 1-,  $\alpha$ 3- and  $\alpha$ 5-subunit genes demonstrated fairly homogeneous distribution profiles across the four nuclei examined. All were detected at a basal level in LMAN and HVC and at high levels within Area X and RA (Table 4.1). Diametric expression patterns were observed for the  $\alpha$ 2-subunit gene in the song system. Corresponding mRNA was not well represented in Area X or RA (where  $\alpha$ 1-,  $\alpha$ 3- and  $\alpha$ 5-subunit mRNA levels were high), but it was elevated in LMAN where  $\alpha$ 1-,  $\alpha$ 3- and  $\alpha$ 5-subunit mRNAs were detected at low levels.  $\alpha$ 4-subunit mRNA exhibited a variable quantitative distribution pattern with levels most pronounced within the RA, at a basal level within the LMAN and HVC and lowest in Area X. An interesting observation was that  $\alpha$ 6-subunit mRNA was detected at a basal level in LMAN, Area X and HVC but corresponding hybridisation signals were quite prominent in the RA, relative to baseline. Maximum levels of mRNA for this gene was observed in cerebellum (Fig. 4.1, F4), but this was not quantified. Transcripts encoding  $\beta$ 2 and  $\gamma$ 2 subunits were detected at baseline levels throughout song system nuclei (Table 4.1); demonstrating a rather ubiquitous and

120

diffuse mRNA distribution. In contrast  $\beta$ 3- and  $\gamma$ 4-subunit mRNAs gave strong signals in all nuclei, neither one exhibiting a low level in any of the nuclei under examination.  $\beta$ 4 and  $\gamma$ 1 subunits displayed the greatest variability in mRNA levels, ranging from relatively meagre (e.g. RA) to very robust (e.g. LMAN and Area X). This was also evident with  $\delta$ -subunit mRNA, which was at appreciable levels in LMAN and HVC, but was barely detectable in Area X and RA (Fig. 4.6d).

Anatomical and densitometric analysis clearly demonstrated that the  $\gamma$ 4-subunit gene was strongly expressed in all the nuclei examined; a feature which was not apparent with any of the other subunit genes. Where the  $\gamma$ 4-subunit transcript is abundant (Area X, HVC and RA; Table 4.1) the other  $\gamma$ -subunit transcripts are at lower levels, but in LMAN, the region in which the  $\gamma$ 4-subunit transcript levels were weaker relative to the other areas,  $\gamma$ 1-subunit mRNA distribution pronounced (Fig. 4.6). In all selected nuclei (with the exception of the HVC) there was at least 1 $\alpha$ -, 1 $\beta$ - and 1 $\gamma$ - subunit transcript detected at a high level, indicative of complex receptor populations.

Although not quantified, GABA<sub>A</sub> subunit gene expression was also documented in other regions of the zebra finch brain. All GABA<sub>A</sub> receptor subunit genes gave strong signals in cerebellum, except for  $\beta$ 3- and  $\gamma$ 1-subunits (Figs. 4.1, 4.3 and 4.5). In the optic tectum (TeO),  $\alpha 1-\alpha 4$ ,  $\alpha 6$ -,  $\beta 2$ -,  $\gamma 1$ -,  $\gamma 2$ - and  $\gamma 4$ -subunit genes exhibited strong expression and  $\alpha 5$ -,  $\beta 3$ -,  $\beta 4$ - and  $\delta$ -subunit genes were transcribed at basal levels. In the lateral striatum (LSt), all subunit mRNAs were well expressed, except for those encoding the  $\alpha 6$ ,  $\gamma 2$  and  $\delta$  subunits. Prominent expression was also observed in zebra finch medial striatum (MSt, region surrounding Area X) for  $\alpha 1$ -,  $\alpha 3$ -,  $\alpha 5$ -,  $\beta 3$ -,  $\beta 4$ - and  $\gamma$ 4-subunit transcripts, at a lower level  $\alpha$ 6 was expressed and expression for the remaining subunits was virtually absent. Within the arcopallium (area surrounding RA) transcript levels were more variable with  $\alpha 1$ -,  $\alpha 5$ - and  $\beta 3$ -subunit genes being well expressed;  $\alpha 3$ -,  $\alpha 4$ -,  $\alpha 6$ - and  $\beta 2$ -subunit genes expressed at lower levels, and  $\alpha 2$ -,  $\beta 4$ -,  $\gamma$ 1-,  $\gamma$ 2-,  $\gamma$ 4- and  $\delta$ -subunit genes poorly expressed. In the apical part of the hyperpallium (HA) only mRNAs encoding  $\alpha 2$ ,  $\alpha 4$ ,  $\alpha 5$ ,  $\beta 2$ ,  $\beta 3$ ,  $\gamma 1$  and  $\delta$  subunits were well distributed, which differed from the densocellular part of the hyperpallium (HD) where  $\alpha$ 1-,  $\alpha$ 3-,  $\alpha$ 4-,  $\beta$ 2- and  $\gamma$ 4-subunit mRNAs were most prominent.

Furthermore, it should be noted that there was no difference in mRNA expression levels of any subunit within any nucleus between the left and right hemispheres of the zebra finch brain (data not shown).

#### 4.2.4 GABA<sub>A</sub> receptor $\pi$ subunit

 $\pi$ -subunit gene expression was not included in the *in situ* analysis, as it was not considered as important as the other subunit cDNAs due to its peripheral location (Symmans *et al.*, 2005; Xiang *et al.*, 2007). However, RT-PCR experiments indicated that  $\pi$ -subunit mRNA was expressed in zebra finch brain, potentially even at higher levels than in the gastro-intestinal tract and testes (Fig. 4.7).



**Figure 4.7.** Gel electrophoresis (1.5% (w/v) agarose gels) of RT-PCRs detailing amplification of  $\pi$ subunit partial cDNAs from different tissues of the zebra finch. Each reaction was completed in triplicate with tissues from three different adult male zebra finches and run alongside a molecular weight marker Hyperladder IV (Bioline). Two products were amplified, purified and sequenced revealing a long (831bp) and short (718bp) version of the  $\pi$ -subunit cDNA. **Abbreviations:** GI, gastro-intestinal tract; NTC, no template control;  $\pi$ L, GABA<sub>A</sub> receptor  $\pi$ -subunit long version;  $\pi$ S, GABA<sub>A</sub> receptor subunit short version. Numbers represent replicates at 58°C, 59°C and 60°C annealing temperature.

Although the RT-PCR data was not quantitative, it clearly suggests  $\pi$ -subunit gene expression in zebra finch brain, in addition to testes and GI tract, where the products appeared somewhat weaker.

#### 4.2.5 Expression of GABA<sub>C</sub> receptor subunit mRNAs in song system

Expression of GABA<sub>C</sub> receptor subunit genes ( $\rho$ 1,  $\rho$ 2 and  $\rho$ 3) was also assessed by *in situ* hybridisation (Fig. 4.8). Two probes targeting each subunit mRNA were designed and tested and in all cases gave identical binding patterns.



**Figure 4.8.** Inverse autoradiographs highlighting the expression of GABA<sub>c</sub> receptor  $\rho$ 1-,  $\rho$ 2-, and  $\rho$ 3subunit genes in four nuclei of the adult male zebra finch song system, (LMAN, Area X, HVC and RA). **Abbreviations:** A, arcopallium; Cb, cerebellum; HA, apical part of the hyperpallium; HD, densocellular part of the hyperpallium; Hp, hippocampus; HVC (formal name); LSt, lateral striatum; M, mesopallium; MSt, medial striatum; N, nidopallium; NC, caudal nidopallium; LMAN, lateral magnocellular nucleus of the anterior nidopallium; RA, robust nucleus of the arcopallium; TeO, optic tectum; X, Area X.

GABA<sub>C</sub> receptor  $\rho$ 1-subunit gene expression was low and diffuse throughout the entire zebra finch brain, with the striking exception of the cerebellum where there was prominent expression (Fig. 4.8, A4). This was similar to the  $\rho$ 3-subunit gene, which demonstrated very weak expression throughout the whole zebra finch brain of adult males (an identical pattern of expression was observed with juvenile zebra finches < PHD 90, data not shown), and a marginal elevation in expression in cerebellum (Fig 4.8, C4). Although the expression was nominal, data indicated that both  $\rho$ 1- and  $\rho$ 3subunit genes were expressed in zebra finch brain, and thus potentially in the song system. The GABA<sub>C</sub> receptor  $\rho$ 2-subunit gene showed the greatest variability in expression (Fig. 4.8, row B). Again, corresponding  $\rho$ 2-subunit mRNA was present in the cerebellum, but it was also identified in the optic tectum (TeO; Fig. 4.8, B4). None of the subunits were particularly well transcribed in zebra finch brain or specifically in any of the nuclei of the song system, but nonetheless, their corresponding mRNAs were all present.

# 4.2.6 Expression of $GABA_A$ receptor subunit genes in the zebra finch song system during development

Developmental expression patterns of GABA<sub>A</sub> receptor subunit genes in the male zebra finch song system were investigated. Levels of  $\alpha 1$ - $\alpha 4$ ,  $\beta 2$ -,  $\beta 3$ - and  $\gamma 4$ -subunit mRNAs were assessed in the HVC, RA and Area X during the phase of song learning by realtime RT-PCR. Zebra finch GABA<sub>A</sub> receptor subunit-specific PCR-primer sets were designed (Table 2.4, Chapter Two) and mRNA was isolated from laser-microdissected HVC, RA and Area X tissues obtained at PHD 30, 55, 80 and 100 from male zebra finches. mRNA expression levels of the GABA<sub>A</sub> receptor subunits in these age groups were quantified relative to the mRNA expression levels of  $\beta$ -actin, glyceraldehyde 3phosphate dehydrogenase (GAPDH) and hypoxanthine phosphoribosyltransferase (HPRT) that were constantly expressed throughout the phase of song learning in RA ( $\beta$ actin) and HVC/Area X (all three genes). During all preliminary qPCR reactions, a DNA dissociation (melt) analysis was completed in order to optimise the cycling conditions and identify any dimers or non-specific products, all primers gave a single product (data not shown).



**Figure 4.9**. Developmental expression of GABA<sub>A</sub> receptor  $\alpha$ -subunit genes in song control nuclei of male zebra finches during the phase of song learning. Mean normalised expression (MNE) was determined for GABA<sub>A</sub> receptor subunits  $\alpha 1$ ,  $\alpha 2$ ,  $\alpha 3$  and  $\alpha 4$  in HVC, Area X and RA at post-hatching days (PHD) 30, 55, 80 and 100, respectively. For each song control nucleus and receptor subunit MNE values (n=5) were taken within each age group and means and standard deviations are shown as percentage of the results at PHD 30. Expression levels not connected by same letter within the bars (A, B, C) were significantly different (Tukey's post hoc test;  $p \leq 0.05$ ).



**Figure 4.10**. Developmental expression of GABA<sub>A</sub> receptor  $\beta^2$ -,  $\beta^3$ - and  $\gamma^4$ -subunit genes in song control nuclei of male zebra finches during the phase of song learning. Mean normalised expression (MNE) was determined for GABA<sub>A</sub> receptor subunits  $\beta^2$ ,  $\beta^3$  and  $\gamma^4$  in HVC, Area X and RA at post-hatching days (PHD) 30, 55, 80 and 100, respectively. For each song control nucleus and receptor subunit MNE values (n=5) were taken within each age group and means and standard deviations are shown as percentage of the results at PHD 30. Expression levels not connected by same letter within the bars (A, B, C) were significantly different (Tukey's post hoc test;  $p \le 0.05$ ).

Developmental expression patterns of GABA<sub>A</sub> receptor subunit genes:  $\alpha 1$ - $\alpha 4$ ,  $\beta 2$ ,  $\beta 3$  and  $\gamma 4$  determined for HVC, Area X and RA during the phase of song learning are presented in Figs. 4.9 and 4.10. All subunit transcripts were detectable in all examined nuclei at each developmental time point, concurring with the *in situ* hybridisation data. The relative level of individual subunit mRNAs however, varied considerably according to developmental stage and nuclei observed. In both HVC and Area X the expression of all subunit genes (except  $\alpha 3$  in Area X) exhibited developmental regulation (ANOVA; *p* < 0.005 for all subunits in HVC, *p* < 0.05 for  $\alpha 1$ ,  $\alpha 4$ ,  $\beta 3$  and  $\gamma 4$  in Area X and *p* < 0.1

for  $\alpha 2$  and  $\beta 2$  in Area X. For almost all GABA<sub>A</sub> receptor subunit genes, there was a significant peak in expression level in HVC (except  $\alpha$ 2-subunit) and Area X (except  $\alpha$ 3subunit) during and/or at the end of the phase of song learning (PHD 100). However the exact timing of developmental changes in gene expression varied between subunits. In HVC, early increases in expression levels of  $\alpha 1$ -,  $\alpha 3$ -,  $\alpha 4$ -,  $\beta 2$ -,  $\beta 3$ - and  $\gamma 4$ -subunit genes occurred between PHD 30-55 and all subunit genes shared the same expression pattern of increasing levels as development progressed. However, at PHD 55 only increases in  $\alpha$ 1-subunit gene expression were significant; increases in  $\alpha$ 3-,  $\alpha$ 4-, and  $\gamma$ 4subunit gene expression were not significant until PHD 100 (post-crystallisation), and increases in  $\beta_2$ - and  $\beta_3$ -subunit gene expression were significant by PHD 80 (relative to expression at PHD 30). Similarly in Area X, GABA<sub>A</sub> receptor subunits:  $\alpha^2$ -,  $\alpha^4$ - and  $\gamma^4$ subunit genes exhibited an early peak in expression (at PHD 55, sensory acquisition), which then declined thereafter; whereas the expression of  $\alpha 1$ -,  $\beta 2$ - and  $\beta 3$ -subunit genes increased significantly towards the end of the song learning phase. Within RA, not all subunit genes displayed developmentally regulated expression,  $\alpha^2$ -,  $\alpha^4$ - and  $\beta^2$ -subunit mRNA levels remained relatively unchanged. In RA,  $\alpha$ 1- and  $\gamma$ 4-subunit mRNAs were significantly increased by PHD 80 (p < 0.05) which was maintained at a similar level post-crystallisation and  $\alpha$ 3-subunit mRNA was significantly increased slightly later at PHD 100 (p < 0.05). The  $\beta$ 3-subunit was the only example of a developmentallyassociated decrease in corresponding mRNA level, occurring at an early time point of song learning (between PHD 30 and 55). The temporal expression profile of the  $\gamma$ 4subunit gene was unique. There was a peak in mRNA levels at PHD 55 (sensory acquisition) in Area X and a peak at PHD 80 (sensorimotor phase) in RA, no other subunit gene exhibited such developmentally-specific expression.

The  $\alpha$ 1-subunit was the only transcript which demonstrated the same temporal expression pattern in each of the three nuclei.  $\alpha$ 4- and  $\gamma$ 4-subunit mRNAs displayed similar temporal patterns in Area X and HVC, but differed in RA. In fact, no two subunits shared the same temporal-expression profiles in RA.  $\alpha$ 3-subunit mRNA was only increased in HVC and RA post-crystallisation, suggesting no involvement in the song-learning process. In conclusion, during the phase of song learning, gene expression patterns of seven GABA<sub>A</sub> receptor subunits were song-control-nucleus specific. Nevertheless, developmental changes that occurred in the expression levels of

127

these GABA<sub>A</sub> receptor subunit mRNAs during the phase of song learning were more often similar in HVC and Area X compared to RA.

### **4.3 Discussion**

Literature indicates a role for GABAergic neuronal inhibition in the zebra finch song system and although there is data indicating the presence of GABA<sub>A</sub> receptors within this system, the main body of work is focused exclusively on electrophysiological and pharmacological approaches, which gives us little insight into receptor subtypes and their distribution (Livinston and Mooney, 1997; Bottjer et al., 1998; Dutar et al., 1998; Luo and Perkel, 1999; Spiro et al., 1999; Vicario and Raksin, 2000; Farries et al., 2005; Prather et al., 2008). This study was the first to examine the complete spatial distribution of all the GABA<sub>A</sub> receptor subunit mRNAs (with the exception of  $\pi$ ; which has a primarily peripheral distribution, Symmans et al., 2005; Xiang et al., 2007) in four nuclei of the adult male zebra finch song system; as it is only when all relevant subunits are considered that meaningful conclusions can be drawn. Expression of each subunit gene within each nucleus was quantified relative to an internal control region (nidopallium), enabling individual expression profiles for each subunit to be established. Furthermore, although levels of subunit mRNAs cannot be directly compared against each other (Briode et al., 2004), tentative conclusions may be drawn concerning the emergence of specific GABA<sub>A</sub> receptor subtypes within each nucleus. In addition, developmental expression of selected GABA<sub>A</sub> receptor subunit genes was also quantified in selected nuclei of the zebra finch song system at important developmental time points (PHD 30, 55, 80 and 100) during the phase of song learning, enabling regional- and temporal-specific gene expression profiles to be generated.

# 4.3.1 Spatial expression of $GABA_A$ receptor subunit genes in the song system

The molecular heterogeneity of the GABA<sub>A</sub> receptor gene family is much larger than any other members of the ligand-gated ion channel superfamily, so a vast number of potential subtypes are possible. Despite this, it appears that there are preferred assemblies (Olsen and Sieghart, 2008, 2009). All GABA<sub>A</sub> receptor subunit transcripts ( $\alpha$ 1- $\alpha$ 6,  $\beta$ 2- $\beta$ 4,  $\gamma$ 1,  $\gamma$ 2,  $\gamma$ 4 and  $\delta$ ) were detectable in all nuclei observed (LMAN, Area X, HVC and RA), and the spatial distribution profile of each subunit mRNA across the zebra finch brain was unique to each individual transcript. Therefore it can be assumed that numerous receptor populations comprising varying subunit assemblies may exist within different brain regions. This concurs with previous data completed in rat (Laurie *et al.*, 1992a, b; Wisden *et al.*, 1992; Sperk *et al.*, 1997; Pirker *et al.*, 2000) and monkey (Kultas-Ilinsky *et al.*, 1998; Huntsman *et al.*, 1999) brain indicating spatial regulation of individual GABA<sub>A</sub> receptor subunit transcription. Only  $\gamma$ 4-subunit mRNA demonstrated robust signalling within all the major nuclei of the song system (Fig. 4.5), hence its consideration as a suitable molecular marker for this system (Thode *et al.*, 2008).

The  $\alpha$ 1-subunit gene is the most abundantly expressed of its class in the mammalian brain (Sieghart and Sperk, 2002). If this is knocked out then the total GABA<sub>A</sub> receptor content of the brain is reduced by ~50% (Sur et al., 2001). In situ hybridisation experiments showed the  $\alpha$ 1-subunit gene to be well expressed throughout the brain with higher expression in Area X and RA (Figs 4.1, row A and 4.2a), this was confirmed by real-time RT-PCR, which also showed this subunit to be well expressed within these regions. This data indicated a steady increase in  $\alpha$ 1-subunit mRNA levels as development progressed, with highest levels (in Area X, HVC and RA) at the time of song crystallisation (~PHD 90) (Fig. 4.9). The  $\alpha$ 1-subunit polypeptide most commonly associates with  $\beta 2$  and  $\gamma 2$  subunits *in vivo* to comprise the most prevalent mammalian GABA<sub>A</sub> receptor subtype (Whiting et al., 2003a). This was apparent in HVC and LMAN, where all three subunit mRNAs gave hybridisation signals similar to those in nidopallium (Table 4.1) and increases in β2-subunit mRNA levels in HVC during development correlated with those encoding the  $\alpha 1$  subunit (Fig. 4.10). The generally low, but diffuse mRNA distribution pattern of these three subunits was indicative of a generalised function within the brain, not specific to the song system and thus potentially not playing a pivotal role in learning and memory (refer to Chapter 5). When considering temporal expression, mRNA distribution patterns of  $\alpha 2$  and  $\beta 2$  subunits in Area X and RA were similar, both remaining relatively unchanged in the RA during song development and showing small peaks at PHD 55 and PHD 100 in Area X (Figs. 4.9 and 4.10), in contrast to the steadily-increasing expression of  $\alpha$ 1-subunit gene. A complementary distribution was also observed in *in situ* hybridisation studies whereby in Area X and RA, when the  $\alpha$ 1-subunit gene was highly expressed, the  $\alpha$ 2-subunit

gene was poorly expressed and in LMAN, when the  $\alpha$ 2-subunit gene was highly expressed,  $\alpha$ 1-subunit mRNA was at level equivalent to baseline (Table 4.1). This data indicated the presence of the well documented  $\alpha 2\beta\gamma 2$  subtype in addition to the  $\alpha 1\beta 2\gamma 2$ complex (both are present in mammalian systems; Olsen and Sieghart, 2008). Generally,  $\alpha$ 1-subunit transcripts are more abundant in post-natal mammalian brain than those encoding the  $\alpha$ 2 subunit (in embryonic brain  $\alpha$ 2-subunits prevail; Laurie *et al.*, 1992a; Huntsman *et al.*, 1999; Sieghart and Sperk, 2002); this was reflected in the steady increase in  $\alpha$ 1-subunit mRNA levels as the song system underwent maturity (Fig. 4.9).

 $\alpha$ 1-,  $\alpha$ 3- and  $\alpha$ 5-subunit genes demonstrated overlapping expression profiles which indicated a variety of GABA<sub>A</sub> receptor subtypes may be present within an individual nuclei (Fig. 4.1). On the other hand, it is certainly possible that more than one type of  $\alpha$ subunit polypeptide may comprise a single GABA<sub>A</sub> receptor (Nusser *et al.*, 1998; Pöltl et al., 2003; Benke et al., 2004; Minier and Sigel, 2004), this has most commonly been investigated with the  $\alpha 1$  and  $\alpha 6$  subunits (Benke *et al.*, 2004; Minier and Siegel, 2004). Olsen and Sieghart (2008) have concluded there is sufficient evidence for the  $\alpha 1 \alpha 6 \beta \gamma$  subtype to exist in brain. Elevated expression of  $\alpha 1$ - and  $\alpha 6$ -subunit genes in RA indicated this receptor species may indeed be present in the avian song system. Interestingly, previous data has shown the  $\alpha$ 5-subunit gene to be poorly expressed in brain with the exception of the hippocampus in rat (Laurie *et al.*, 1992; Sperk *et al.*, 1997; Pirker et al., 2000), where  $\alpha$ 5-subunit-containing receptors constitute ~20% of the GABA<sub>A</sub> receptor population (Sur et al., 1998, 1999). Conversely, in situ hybridisation studies herein have demonstrated high levels of  $\alpha$ 5-subunit mRNA in a variety of brain regions, most notably in Area X and RA, relative to baseline levels (Fig. 4.1 and 4.2). Strong expression was also detected in the hippocampus, cerebellum and arcopallium (although this was not quantified), which correlates with data found in rats.  $\alpha$ 5-subtype receptors play an important role in spatial learning in rats (Cavallaro *et al.*, 2002; Collinson et al., 2002; Crestani et al., 2002; Chambers et al., 2004), due to their specific location in the hippocampus (a region associated with spatial learning; Wanatabe and Bischof, 2004), thus the same theory may be extrapolated to explain the robust expression of the  $\gamma$ 4-subunit gene within the song system.

 $\alpha$ 4-subunit-encoding mRNA was abundant in the cerebellum (Fig. 4.1), which disagrees with previous studies (Laurie et al., 1992a, b; Wisden et al., 1992). Although natural variation amongst different species is to be expected, for example, Pöltl et al. (2003) demonstrated differences in GABA<sub>A</sub> receptor subtype assemblies in the cerebellum of mice and rats. The  $\alpha$ 4-subunit gene was also highly expressed in RA exhibiting a distribution somewhat reciprocal to that of the  $\alpha 6$  subunit (Fig. 4.1), although a subtype containing these two subunits has not been previously postulated, it is recognised that ~50% of  $\alpha$ 4-subunit-containing GABA<sub>A</sub> receptors do not posses a  $\gamma$  or  $\delta$  subunit and are  $\alpha/\beta$  pentamers (Bencsits *et al.*, 1999); these are localised extrasynapically on rat hippocampal pyramidal neurons (Mortensen and Smart, 2006). Similar to the  $\alpha 4$ subunit, the distribution of  $\alpha$ 6-subunit transcript found in this study disagrees somewhat with previous literature which reports expression of this gene to be restricted to the cerebellum (Laurie et al., 1992a, b; Persohn et al., 1992; Pirker et al., 2000). Although signal intensity was highest in the cerebellum (Fig. 4.1), there was also evidence that the  $\alpha$ 6-subunit mRNA was present at low levels in all nuclei of the adult song system and at marginally increased levels in RA (the most dorsally examined nucleus; Fig. 4.2, Table 4.1). Thus it appeared that  $\alpha$ 6-subunit mRNA levels increase towards the posterior regions of the brain. Although it must be acknowledged that as the levels of mRNA were relative to baseline; therefore, as the expression of  $\alpha$ 6-subunit gene was fairly weak in the nidopallium (internal control area) and any small increase in a nucleus would be appreciable.

There is no concrete evidence to indicate that two different  $\beta$  subunit isoforms coassemble into a heteromeric receptor although it has been postulated (Li and De Blas, 1997; Jechlinger *et al.*, 1998), thus presence of high levels of mRNA encoding  $\beta$ 3 and  $\beta$ 4 subunits in Area X points at two different receptor species. Although  $\beta$ 2-subunit mRNA was well distributed, it was lower and more diffuse (relative to baseline) than observed for other  $\beta$  subunits (Fig. 4.3).  $\beta$ 3-subunit transcript was the most abundant in the song system relative to the nidopallium, with the exception of the HVC, where  $\beta$ 4subunit mRNA was well represented (Table 4.1). The  $\beta$ 4-subunit gene is localised in a cluster with  $\gamma$ 4- and  $\alpha$ 3-subunit genes on chromosome 4A of the zebra finch genome (Fig. 3.8), analogous to the  $\theta$ ε $\alpha$ 3 cluster on the X chromosome of the human genome
(Simon *et al.*, 2004). Coordinate chromosomal expression would suggest an  $\alpha 3\beta 4\gamma 4$  subtype in the zebra finch brain, but although expression was overlapping, only in Area X, were all three subunit genes expressed at a high level relative to their individual baselines. Thus this combination was not emerging as a strong candidate for a major receptor subtype in the song system.

Levels of  $\gamma$ 1-subunit mRNA in LMAN were striking, suggesting that there were receptors containing this subunit in this nucleus; such a robust distribution is not seen anywhere within the mammalian brain (Laurie et al., 1992a, b; Wisden et al., 1992; Sperk et al., 1997). There is no definitive evidence suggesting that  $\gamma$ 1-subunit polypeptides assemble into receptors in mammalian systems (Olsen and Sieghart, 2008), although this can primarily be attributed to a paucity of studies. Immunoprecipitation studies have indicated that all three mammalian  $\gamma$  subunits are able to associate with  $\alpha 1$ ,  $\alpha 2$ ,  $\alpha 5$  (not  $\gamma 1$ ),  $\beta 2/3$  subunits, and in addition, the  $\gamma 2$  subunit was found to co-precipitate with  $\gamma$ 3 (Benke *et al.*, 1996). This evidence suggested  $\gamma$ 1 may well be able to associate with any of the subunits also detected within LMAN, and the highly confined distribution ( $\gamma$ 1-subunit mRNA was low in all other song system nuclei), may be indicative of a specialised role within the song system, particularly as LMAN was the only nucleus where  $\gamma$ 4-subunit mRNA was not significantly higher than baseline. The most intriguing expression pattern of all was that of the  $\gamma$ 4-subunit gene (Fig. 4.5). Robust expression of the  $\gamma$ 4-subunit gene in the song system of the zebra finch has been recently demonstrated by Thode *et al.* (2008), and accordingly, its role as a molecular song system marker has been proposed. In light of this, it is interesting to speculate that the  $\gamma$ 4-subunit polypeptide is able to co-assemble with a variety of other GABA<sub>A</sub> receptor subunits (as with other  $\gamma$  subunits) as no other subunit mRNA profile was similar to that generated for the  $\gamma$ 4 subunit, both spatially or developmentally. Thus it may form the basis of a variety of receptor subtypes which would account for the relative abundance of the transcript.

The  $\delta$ -subunit gene was well expressed in the telencephalic HVC and LMAN nuclei, and in accordance with previous data, it was most abundant in the cerebellum (Fig. 4.5 (D4); Laurie *et al.*, 1992a, b).  $\delta$ -subunit containing receptors are exclusively extrasynaptic where they mediate tonic inhibition (Kaur *et al.*, 2009). They are generally thought to co-assemble with  $\alpha 4$  or  $\alpha 6$  subunits (Jones *et al.*, 1997; Sur *et al.*, 1999), but have more recently been shown to assemble with the  $\alpha 1$  subunit in hippocampus (Glykys et al., 2007). There was no obvious indication of this partnership in the song system (but both were in cerebellum). Moreover, upon knockout of  $\alpha$ 5-subunit containing receptors (which are also extrasynaptic) in mice, those comprising the  $\delta$  subunit were upregulated in compensation in hippocampal regions (Glykys and Mody, 2006). This agrees with data presented here (Table 4.1) which showed the  $\alpha$ 5-subunit gene to be highly expressed in Area X and RA (where  $\delta$  was poorly expressed) and low expression in LMAN and HVC where the  $\delta$ -subunit gene was highly expressed. However, when  $\delta$ -subunit-containing receptors are knocked out, a small residual tonic current is still detected (not attributed to by  $\alpha$ 5-subunit containing receptors) in regions of the hippocampus (Glykkys et al., 2007), and this is speculated to be due to GABA<sub>A</sub> receptors containing the  $\varepsilon$  subunit. As afore mentioned the  $\varepsilon$  subunit is the mammalian orthologue of the avian  $\gamma 4$  subunit and *in situ* hybridisation studies (including this one) have shown  $\gamma$ 4-subunit mRNA to be well expressed in the hippocampus (Thode *et al.*, 2008). This suggests that the  $\gamma$ 4 subunit, like its' mammalian counterpart, may play a role in tonic inhibition within the zebra finch brain.

### 4.3.2 Expression of GABA<sub>A</sub> receptor $\pi$ -subunit gene in avian brain

Although it is regarded as a bona-fide GABA<sub>A</sub> receptor subunit, the  $\pi$ -subunit is different in that it is generally expressed in peripheral tissues such as lung (Jin *et al.*, 2005; Xiang *et al.*, 2007), ovaries and uterus (Hedblom & Kirkness, 1997; Fujii and Mellon, 2001) in rat. It has been suggested that  $\pi$ -subunit-containing GABA<sub>A</sub> receptors play a role in the regulation of muscle contraction and alveolar fluid homeostasis in the lungs (Jin *et al.*, 2006), where they may act in an excitatory capacity (Xiang *et al.*, 2007).  $\pi$ -subunit-containing receptors are significantly more sensitive to the potent progesterone metabolite allopregnanolone (which is at high circulating levels during pregnancy; Concas *et al.*, 1999) and there is a drastic decrease in the level of  $\pi$ -subunit mRNA in uterine tissues just prior to labour, hence these receptors could help regulate uterine contractions (Fujii and Mellon, 2001). More recently, the  $\pi$ -subunit has been established as a molecular marker for detecting breast cancer metastasis (Backus *et al.*, 2005; Symmans *et al.*, 2005) and tumorigensis in pancreatic cancer (Johnson and Haun,

2005) but there is little data documenting its expression within brain. However, some recent data has shown that not only is the  $\pi$ -subunit transcript present in rat brain, but it demonstrates a learning-associated increase in expression in hippocampus following training with the Morris water maze (Cavallaro *et al.*, 2002). Coupled with the seemingly robust product amplification from brain cDNA (Fig. 4.7), it would appear that  $\pi$ -subunit mRNA is present in the avian brain and this merits further investigation.

# 4.3.3 Spatial expression of $GABA_C$ receptor subunit genes in the song system

Although not quantified densitometrically, spatial expression of GABA<sub>C</sub> subunit genes was investigated within the male zebra finch brain. GABA<sub>C</sub> receptor subunit genes are predominately expressed in the bipolar and horizontal cells of the retina, where they mediate synaptic inhibition (Albrecht and Darlison, 1995; Chebib and Johnston, 1999; Rozzo et al., 2002). Thus, unlike the majority of GABA<sub>A</sub> receptor-subunit genes, expression of GABA<sub>C</sub> receptor subunit genes was low and diffuse; with no notable regions of discrete expression except in cerebellum. This was in agreement with previous data which initially suggested p-subunits were not transcribed in brain, but now purports expression to be present at low levels in mammalian brain (Ogurusu and Shingai, 1996, Boue-Grabot et al., 1998; Rozzo et al., 2002; Milligan et al., 2004; Alakuijala et al., 2006) and chicken brain (Albrecht et al., 1997). They are also expressed in mammalian ovary, testes (Rozzo et al., 2002) and gut (Jansen et al., 2000). Data indicated that GABA<sub>C</sub> receptor p3-subunit mRNA was present at very low levels in zebra finch brain compared to  $\rho 1$  and  $\rho 2$  (Fig. 4.8), concurring with northern blotting analysis which initially illustrated that the  $GABA_C$  receptor  $\rho$ 3-subunit gene is expressed at lower levels in the retina compared to  $\rho$ 1- and  $\rho$ 2-subunit genes (Ogurusu and Shingai, 1996). Real-time RT-PCR has confirmed these findings and further revealed that although mRNA encoding the  $\rho$ 3 subunit is present in the cerebellum and hippocampus, it is at significantly lower levels than the other two GABA<sub>C</sub> receptor subunit mRNAs (Milligan et al., 2004). All subunits were well expressed in cerebellum, where they could potentially associate into heteromers (Enz and Cutting, 1999), but generally GABA<sub>C</sub> receptors tend to form homooligomeric complexes (Enz and Cutting 1998, 1999). In recombinant receptor systems, p-subunits have been reported to

associate with the  $\gamma^2$  subunit of the GABA<sub>A</sub> receptor into functional receptors (Ekema *et al.*, 2002; Milligan *et al.*, 2004; Pan and Qian, 2005). Coupled with the more recent identification of GABA<sub>C</sub> subunits in brain and the suggestion that GABA<sub>C</sub> receptors are indeed a sub-class of GABA<sub>A</sub> receptors (Olsen and Sieghart, 2008; Collingridge *et al.*, 2009), it was of interest to observe if there were any regions of co-expression. Their expression in cerebellum was overlapping with that of GABA<sub>A</sub> receptor subunit genes; so this could be a site for potential co-assembly between the two receptor classes, however due to the variety of cell types that exist within cerebellum, it is equally likely that the GABA<sub>C</sub> receptors could be localised to different cells (or even on the same cell in different receptor populations) as there is no documented evidence of native GABA<sub>A/C</sub> complexes existing in brain. None of the GABA<sub>C</sub> receptor subunit genes were well expressed in the song system of zebra finches (not quantified) so analysis was not taken any further.

### 4.3.4 Developmental expression of $GABA_A$ receptor subunit genes in the song system

All GABA<sub>A</sub> receptor subunit mRNAs ( $\alpha$ 1,  $\alpha$ 2,  $\alpha$ 3,  $\alpha$ 4,  $\beta$ 2,  $\beta$ 3 and  $\gamma$ 4) were detected in each nuclei studied (Area X, HVC and RA) of the zebra finch song system, at each developmental stage (Figs. 4.9 and 4.10), which concurred with findings from the *in situ* hybridisation experiments. In HVC, transcription of all GABA<sub>A</sub> receptor subunits appeared to be developmentally regulated. All subunits exhibited the highest level of gene expression in HVC at PHD 100 (post-crystallisation) after the song-learning phase was finished.  $\alpha 1$ ,  $\alpha 4$ ,  $\beta 2$ ,  $\beta 3$  and  $\gamma 4$  subunits all demonstrated steady increases in gene expression as development progressed (perhaps correlating with the growing HVC which does not reach full size until ~PHD 60; Ward et al., 1999), only a2- and a3subunit genes exhibited lower expression during the sensorimotor phase, but following crystallisation, levels increased once more (Figs. 4.9 and 4.10). This was in agreement with previous data which showed that  $\alpha^2$ -,  $\alpha^3$ -,  $\alpha^5$ - and  $\beta^3$ -subunit genes were well expressed in embryonic cortex and thalamus of rats and decreased in adult brain and the reverse was true for  $\alpha 1$ -,  $\alpha 4$ -,  $\beta 2$ - and  $\delta$ -subunit genes (Laurie *et al.*, 1992a). Overall,  $\alpha$ 1- and  $\alpha$ 4-subunit genes demonstrated increased expression in the developing song system whereas  $\alpha^2$ - and  $\alpha^3$ -subunit genes remained at relatively low and unchanging levels. The HVC is involved in both the anterior forebrain pathway (AFP) and vocal

motor pathway (VMP) and thus is important in both sensory acquisition and the sensorimotor phase (Nottebohm et al., 1976; Wild et al., 2005; Mooney and Prather, 2008). This steady increase during song development was also observed for  $\alpha$ 1- and  $\beta$ 3subunit mRNAs in Area X, whereas  $\alpha 2$ ,  $\alpha 3$ ,  $\alpha 4$  and  $\gamma 4$  subunits all displayed discontinuous temporal mRNA distributions with the highest levels during sensory acquisition (~PHD 35-60) and a decrease by the end of the sensorimotor phase (PHD 80). Area X forms part of the AFP and thus is primarily involved in sensory acquisition during song learning (Bottjer et al., 1984; Scharff and Nottebohm, 1991) and so would be most active during this stage (up to PHD 60) and cell number within Area X increases between ~PHD 10-50 (Tomaszycki et al., 2009). RA housed the smallest number of developmentally-regulated subunit genes with levels  $\alpha^2$ -,  $\alpha^4$ - and  $\beta^4$ -subunit mRNAs all remaining relatively unchanged over time. Interestingly, in previous studies, RA displays the lowest number of genes regulated by song production (Wada et al., 2006). β3 was the only subunit to show a decrease in transcript expression following the onset of song learning and y4-subunit mRNA again showed a discontinuous profile with highest levels towards the end of song learning, indicating a rapid increase in gene transcription during the whole period of song learning, followed by a small decrease post-crystallisation. Previous findings using *in situ* hybridisation suggested that  $\gamma$ 4subunit mRNA only appears in RA at ~PHD 35 (coinciding with establishment of the VMP; Nordeen and Nordeen, 1988; Wang et al., 1999; Scott and Lois, 2007; Thode et al., 2008). This was echoed in the real-time RT-PCR data which indicated a large increase of y4-subunit mRNA in RA occurred after PHD 30 (Fig. 4.10). The RA forms part of the VMP and is the output of the song system, thus it only becomes fully active during the sensorimotor phase of song learning. Thus  $\gamma$ 4-subunit mRNA was highest in Area X during sensory acquisition, highest in RA during the sensorimotor phase and highest in HVC following crystallisation. Correlation of gene expression with specific developmental time points is highly indicative of a functional role for receptors containing this subunit in the learning of song; no other subunit displayed such a defined temporal profile. The level of  $\gamma$ 4-subunit mRNA is developmentally regulated in the chicken (Harvey et al., 1993; Enomoto et al., 2001) and is down-regulated in response during imprinting training in the one-day-old chicken (Harvey et al., 1998). Coupled with the robust expression observed here and previously (Thode et al., 2008), and coordinated expression during song development in the zebra finch, it can be

concluded that  $GABA_A$  receptors containing the  $\gamma$ 4-subunit may play an important role in learning and memory systems.

### 4.3.5 Concluding remarks

The heterogeneity of the GABA<sub>A</sub> receptor family is both its redeeming feature and greatest encumbrance.  $GABA_A$  receptors mediate the majority of inhibitory neurotransmission within the brain and the plethora of subunits enables temporal- and regional-specific expression of many different subtypes (demonstrated herein), each with its own individual pharmacological and electrophysiological signature, which potentially fulfil different physiological functions (Wafford et al., 2004; Olsen and Sieghart, 2008). However, the extreme promiscuity of the subunits causes problems in elucidating defined native GABA<sub>A</sub> receptor subtypes, this is frequently observed in coprecipitation studies (Sieghart and Sperk, 2002) and expression studies such as this are notoriously hard to decipher as all subunits are present and may be assembling into many combinations. As GABA<sub>A</sub> receptors are targets of many clinically important drugs (benzodiazepines, barbiturates, steroids and volatile anaesthetics; Reynolds, 2008), it is important to resolve specific receptor subtypes involved in mediating different physiological functions in an effort to generate novel subtype-specific therapeutic strategies (such as those being developed for  $\alpha$ 5-subunit containing receptors; Chambers et al., 2004). GABAA receptors have long been implicated in modulation of learning and memory (for review, see Chapouthier and Venault, 2002; Maubach, 2003; McNally et al., 2008). In the chicken (Gallus gallus domesticus), which is phylogenetically similar to the zebra finch, augmentation of GABA<sub>A</sub>-receptormediated inhibition with muscimol produced amnesia and bicuculline improved memory formation (Clements and Bourne, 1996). A later study specifically identified a significant decrease in levels  $\gamma$ 4-subunit mRNA (but not  $\gamma$ 2-subunit mRNA) in learningrelevant regions of chicken brain 10 h following imprinting training (further details in Chapter Five; Harvey *et al.*, 1998). Following this, the  $\gamma$ 4-subunit gene was mapped in the zebra finch brain and a unique, robust expression profile was revealed (Thode *et al.*, 2008). The avian song system is a well established paradigm for studying neuronal mechanisms of cognitive processes and development of sophisticated motor skills. Due to the numerous parallels with human speech production (Doupe and Kuhl, 1999; Brainard and Doupe, 2002; Kuhl, 2004; Bolhius and Gahr, 2006), defining GABAA

137

subtypes and their functional roles in this system may extrapolate to human behaviour and drugs can be designed to target cognitive defects and give insight into diseases which affect motor coordination. Robust expression of the  $\gamma$ 4-subunit gene and developmental regulation of transcription in the song system points to an important role for GABA<sub>A</sub> receptors containing this subunit. Electrophysiological evidence shows that  $\gamma$ 4-subunit-containing receptors demonstrate some different pharmacological characteristics to those comprising the other  $\gamma$  subunits, including a novel benzodiazepine pharmacology and potent blocking of currents by Zn<sup>2+</sup> ions (Forster *et al.*, 2001), thus creating an attractive therapeutic target.

# 5. THE EFFECT OF AUDITORY LEARNING ON THE EXPRESSION OF THE GABA<sub>A</sub> RECEPTOR $\gamma$ 4-SUBUNIT GENE IN THE SONG SYSTEM

### **5.1 Introduction**

Previous data (Chapter Three) illustrated a robust, region-specific distribution of mRNA encoding the GABA<sub>A</sub> receptor  $\gamma$ 4 subunit, within the zebra finch song system. The spatial gene expression and the strength of hybridisation signals was exclusive to the  $\gamma$ 4 subunit and was indicative of an important role for GABA<sub>A</sub> receptors containing this subunit in song acquisition and production in the zebra finch. This was further compounded by the correlation of gene expression with specific developmental time points, which again was exclusive to the  $\gamma$ 4-subunit gene. Such findings warranted further investigation.

The zebra finch song system is an important paradigm for investing the underlying mechanisms of cognitive function, which can be extrapolated to humans due to the numerous parallels between the two systems (Brainard and Doupe, 2002; Kuhl, 2004; Bolhuis and Gahr, 2006; refer to section 1.4.8). This is exemplified by studies of FOXP2 (which encodes the forkhead box transcription factor). It is documented that genetic aberrations can cause developmental verbal dyspraxia in humans. Studies using twins have eluded that despite equal intelligence and opportunity, developmental speech impairments can occur (Bishop, 1995). This can be attributed to, at least in part, an inherited genotype. Further work by Lai et al. (2001, 2003) and MacDermot et al. (2005), identified a mutation in FOXP2 in human subjects, which caused speech and language impairments; this was the first gene to be implicated in such a disorder. RNA interference (RNAi) studies with the same gene within the zebra finch song system have revealed that when FOXP2 gene expression was 'knocked-down', vocalisations were affected in the manner of poor imitation of tutor song and production of song with abnormally variable syllables (Haesler et al., 2007). Due to the complexity of neuronal networks, these effects cannot be attributed to a single gene and thus the search for other genetic links is paramount. As previously mentioned, songbirds are one of the closest accessible models for neuronal circuitry controlling learned vocalisations similar to that of humans (Brainard and Doupe, 2002; Kuhl, 2004; Bolhuis and Gahr, 2006); this was therefore a relevant system to observe  $\gamma$ 4-subunit gene expression and how it is implicated in the acquisition and production of song.

Birds live in heterogeneous populations where they are exposed to a variety of songs from different species. It is a matter of survival that they learn a song specific to their species as song is principally used for attraction of a mate, individual identification (which is especially relevant in zebra finches which live in groups of up to 1000 members) and defence of territory (Zann, 1996; Brenowitz et al., 1997) and thus needs to be interpreted by conspecifics. The auditory template model of song learning has been proposed Marler, 1970). It suggests that song birds, from birth, have a genetically-encoded, innate template which pre-disposes them to sing a song representative of their own species. It is the songs that they hear from their conspecifics (tutors) during the sensory acquisition phase that 'trigger' this song learning process (Konishi, 1965, 1985, 2004; Marler, 1970, 1976). Evidence supporting this arises from the fact that young birds and female birds respond more to conspecific song than heterospecific song both at a behavioural (Dooling and Searcy, 1980; Clayton, 1989; Nelson and Marler, 1993) and molecular level (Nick and Konishi, 2005; Tomaszycki et al., 2006); and will learn conspecific song in preference to heterospecific song (Marler and Peters, 1988). Although a tutor is required during sensory acquisition, when the bird enters the sensorimotor phase of song learning, the tutor is no longer necessary. At this stage the bird must be able to hear its own vocal output to compare to the acquired template (Konishi, 2004) and this need for auditory feedback extends into adulthood, even after crystallisation (Nordeen and Nordeen, 1992; Leonardo and Konishi, 1999; Woolley, 2004; Funabiki and Funabiki, 2007; Sober and Brainard, 2009). Deprivation experiments have highlighted that birds raised in isolation are able to produce songs that include some species-specific features, which are enough to evoke a reaction from other adult male or female conspecifics (Marler and Sherman, 1985; Searcy and Marler, 1987; Zann, 1996; Konishi, 2004; Kojima and Doupe, 2007). Deafened birds produce a song which is more abnormal than isolate song (Konishi, 2004), yet even these songs retain some basic features of natural song, but not enough to elicit a reaction from conspecifics (Marler and Sherman, 1983, 1985; Searcy and Marler, 1987). Disruption of auditory feedback in adult birds results in a gradual deterioration of song quality (Nordeen

and Nordeen, 1992; Woolley et al., 1997; Leonardo and Konishi, 1999; Funabiki and Konishi, 2003; Zevin et al., 2004). However, despite all of the previous behavioural work regarding the isolation and deafening of song birds, the underlying molecular mechanisms are still not properly understood. With regards to gene expression and song behaviour, the majority of work has concentrated on mapping expression of genes within the song system in response to song production. Some of the first genes to be studied were immediate early genes (IEG)s, which are characterised by their rapid and transient response to a stimulus and consequently are often used as endogenous markers of neuronal activity (Chaudhuri, 1997; Tischmeyer and Grimm, 1999). ZENK (a.k.a. zif-268, NGFI-A, egr-1 or Krox-24; Jarvis and Nottebohm, 1997), c-fos (Kimpo and Doupe, 1997), BDNF (brain-derived neurotrophic factor; Li et al., 2000), Arc/Arg3.1 (activity-regulated cytoskeleton-associated protein/activity-regulated gene; Tarcisco et al., 2005), SytIV (synaptotagmin IV; Poopatanapong *et al.*, 2006) and *UCHL1* (ubiquitin carboxyl-terminal hydrolase; Lombardino et al., 2005) all show a rapid increase in gene expression in nuclei of the zebra finch song system at the onset of song production. Continuing along this theme, more recently, Wada et al. (2006) considered 33 of the many genes expressed within four nuclei of the song system; these genes encode for proteins such as transcription factors (e.g. *c-fos*, cellular-fos; *c-jun*, cellular-jun; *Atf4*, activating transcription factor 4), actin-binding proteins (e.g.  $\beta$ -actin; Tagln2, transgelin2), cell signalling molecules (e.g. Penk, Proenkephalin; Stard7, START domain containing 7) and many more. All observed genes were regulated by the onset of singing. Of the four regions, Area X showed the greatest variability of gene expression with 94% of genes regulated by singing; the majority of which were up-regulated in response to song production. In contrast, within the RA, only 33% of the genes examined demonstrated song-driven expression (Wada et al., 2006), which is interesting as RA is heavily involved in the motor production of song. In addition to regional-specific gene expression, the zebra finch song system also displays contextual gene expression. During undirected singing (in the absence of females) ZENK expression is high in HVC, Area X and elevated in RA, whereas when song is directed at females, ZENK expression is only high in HVC and is virtually absent in Area X and RA (Jarvis et al., 1998). Conversely, c-fos, another immediate early gene is better expressed in RA during directed singing (Kimbo and Doupe, 1997). Despite numerous gene expression studies in zebra finch, there still remains little data available documenting gene expression in relation to the learning of song (with the exception of *FOXP2*). Studies such as these are necessary for understanding the complex molecular mechanisms underlying the acquisition and production of learned song.

In addition to the zebra finch song system, imprinting in the one-day-old chicken is another well established learning paradigm (Horn, 1998). Gene expression levels of various IEGs, which are involved in the formation of long-term memories, have been measured in response to auditory imprinting. Results revealed significantly increased expression of ZENK and Arc/Arg3.1 within regions involved in visual and auditory imprinting (Bock et al., 2005; Thode et al., 2005). Similar experiments utilising naive one-day-old chicks trained on auditory and visual stimuli, indicated a regional-specific down-regulation of y4subunit transcript within numerous brain areas associated with auditory and visual imprinting (Harvey et al., 1998), which suggested a role for receptors containing this subunit in learning and memory. Decreases between trained and untrained chicks were apparent 5 h post-training but only became significant 10 h post-training. Brain regions significantly affected were the entopallium (E) and the intermediate medial mesopallium (IMM) - an area involved in recognition memory during imprinting (Horn, 1998). The authors postulated that a down-regulation of GABAergic transmission ameliorated learning and memory processes (Harvey et al., 1998). Furthermore, when compared to the embryonic (E18) chicken brain (Harvey et al., 1993), y4-subunit gene expression in the optic tectum of the one-day-old chicken was very high. This area is involved in processing visual information which is necessary to some extent during embryogenesis (where light passing through the shell modulates brain development; Rogers, 1982), but is critically important following hatching; the increase in  $\gamma$ 4-subunit transcript correlated with this.

As aforementioned, GABA is widely distributed in the avian song system (Pinaud and Mello, 2007). Although electrophysiological and pharmacological evidence confirms the presence of GABA<sub>A</sub> receptors (Livingston and Mooney, 1997; Bottjer *et al.*, 1998; Dutar *et al.*, 1998; Luo and Perkel, 1999; Spiro *et al.*, 1999; Vicario and Raksin, 2000; Farries *et al.*, 2005; Prather *et al.*, 2008), these studies give no insight as to the subtype composition. A

recent study by Thode *et al.* (2008) highlighted the wide-spread distribution of the  $\gamma$ 4subunit transcript in the majority of song-system nuclei and the interesting observation that the  $\gamma$ 4-subunit mRNA only appears in the RA in birds ~PHD 35 or older (coinciding with the onset of singing), suggesting a role for receptors containing this subunit in control of motor production of song, indicative of developmentally-regulated transcription. Developmental expression of the  $\gamma$ 4-subunit gene was also observed in the embryonic chicken brain (Gallus gallus domesticus - where cDNA encoding the GABA<sub>A</sub> receptor  $\gamma 4$ subunit was initially identified). Northern blot analysis revealed a developmentallyregulated pattern of gene expression in the brain whereby the  $\gamma$ 4-subunit transcript was first expressed at embryonic day (E) 13 and continued to increase in expression during embryogenesis (Harvey et al., 1993), correlating with expression in the brainstem which is not detected until E8, some time after many other GABAA receptor subunit mRNAs are detected (Enomoto et al., 2001). In situ hybridisation studies of chicken brain at E18 demonstrated discrete, robust expression of the  $\gamma$ 4-subunit transcript in many areas of the brain most notably the entopallium, nucleus rotundus (both of which are involved in the processing of visual information), mesopallium and low levels of transcript were also detected in the optic tectum (Harvey et al., 1993).

Down-regulation of GABAergic transmission appears to play a role in learning and memory processes (Harvey *et al.*, 1998; Cavallaro *et al.*, 2002; Collinson *et al.*, 2002; Maubach, 2003). GABA<sub>A</sub> receptors are the major mediators of inhibition within the brain and much evidence suggests modulation of GABA<sub>A</sub> receptors significantly affects learning and memory behaviour (reviewed by Chapouthier and Venault, 2002; McNally *et al.*, 2008). This concurs with evidence that augmentation of GABAergic function via GABA<sub>A</sub> receptors with muscimol results in amnesia and antagonism with bicuculline improves memory formation in the chick (Clements and Bourne, 1996), rat (Zarrindast *et al.*, 2006) and mouse (Brioni and McGaugh, 1988). Due to the unusually strong expression of the  $\gamma$ 4subunit transcript the zebra finch song-system (Thode *et al.*, 2008), another learning and memory paradigm; it was interesting to characterise the expression of receptors containing this subunit at a behavioural level and determine how  $\gamma$ 4-subunit gene expression is affected by auditory input (e.g. from a tutor). As birds that are raised in auditory isolation from a tutor are unable to produce normal song (Immelmann, 1969; Marler and Sherman, 1983; Eales, 1987; Searcy and Marler, 1987; Tchernichovski and Nottebohm, 1998), it can be assumed that the molecular mechanisms involved in song acquisition and production would be affected. Exploitation of this paradigm was used to highlight the importance of the GABA<sub>A</sub> receptors containing the  $\gamma$ 4-subunit in four nuclei of the zebra finch song system, namely, LMAN, Area X, HVC and RA. Male zebra finches were raised in the presence of a tutor or in auditory isolation until PHD 80 where their songs were recorded (in the presence of a female) and analysed and the expression of the  $\gamma$ 4-subunit mRNA was quantified densitometrically within the four nuclei and compared between the experimental groups. This assisted in determining the role of the  $\gamma$ 4-subunit-containing GABA<sub>A</sub> receptors in learning and memory processes (i.e. acquisition and production of song).

### **5.2 Results**

Two rounds of behavioural experiments were completed. In preliminary experiments, male zebra finches were reared in the presence of a tutor and conspecifics (i.e. in an aviary) or in tutor-isolation (i.e. in an individual, sound-proof isolation cage) to determine if tutorisolation and thus a disruption to song acquisition and production would affect the expression of the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit gene within selected nuclei of the song system (LMAN, Area X, HVC and RA). Following in situ hybridisation for the y4-subunit transcript, experiments were repeated with a  $\gamma$ 2-subunit selective probe to ascertain if effects were subunit specific. To ensure that environmental and social factors were not affecting gene expression, in the subsequent set of experiments, three sets of male zebra finches were raised. The first group were raised in an aviary with tutors and conspecifics, as before. A second group were raised in individual isolation cages with their parents (so a tutor was present), a third group were raised in individual isolation cages, but instead of being alone (as in the preliminary work), they were accompanied by a female zebra finch (which does not sing). Again, y4-subunit mRNA levels were measured in LMAN, Area X, HVC and RA and compared between groups. At PHD 80, the song of each bird was recorded in the presence of an unfamiliar female (directed song), then sacrificed, their brains excised and immediately shock-frozen. Sequential 10µm coronal sections were taken (through the entire brain) and hybridised with a radiolabelled  $\gamma$ 4-specific oligonucleotide probe; sections were washed and exposed to the same film together to reduce procedural variability. Autoradiographic images were digitally acquired with a light box and CCD camera and captured using Easygrab Picolo<sup>TM</sup> software (Euresys<sup>TM</sup>). Gene expression in individual regions (nuclei) and internal control regions (nidopallium), was quantified by densitometric analysis of three consecutive sections using Scion image software (Scion Corporation, USA). Autoradiographic signal in areas of interest was calculated as pixel density per brain area measured. Gene expression was determined by subtracting background hybridisation signals from both the nuclei and internal control readings and then mRNA expression in each nucleus was calculated as a percentage of the internal control. One-way ANOVA tests were employed to determine any significant deviations in gene expression from baseline levels (nidopallium).

### 5.2.1 Internal control region (Nidopallium)

To quantify the autoradiographic signal (of  $\gamma$ 4- or  $\gamma$ 2-subunit transcript), an internal control region was required. It was essential to be able to normalise the results, to enable quantitative comparison between levels of mRNA in different nuclei of the brain. The internal control area selected was the nidopallium, which has no documented involvement with learning and memory processes and is a continuous structure throughout the brain (Fig. 5.1). Furthermore, it was ascertained that the genes were expressed in this region and that the nidopallium remained unaffected by the parameters of the experiment, i.e. gene expression was unchanged in nidopallium of tutor-isolated and non-isolated birds.



**Figure 5.1**. Sagittal schematic illustrating layers within the zebra finch brain. Nidopallium (N) is highlighted in orange and encompasses HVC (formal name) and lateral magnocellular nucleus of anterior nidopallium (LMAN); it forms a continual structure throughout the brain. Area X (X) lies within the striatum (St) and robust nucleus of the arcopallium (RA) lies within the arcopallium (A). **Abbreviations:** Cb, cerebellum; DM, dorsal medial nucleus of the midbrain; H, hyperpallium; M, mesopallium; MLd, Mesencephalic lateral dorsal nucleus; NIf, interfacial nucleus of the nidopallium; P, pallidum; St, striatum; X, Area X.

When the measurement of mRNA expression was taken in each nucleus, a simultaneous reading was also taken from the corresponding region of the nidopallium. Figure 5.2 illustrates the areas of the nidopallium selected for internal controls at each observed stereotactic level of the zebra finch brain (i.e. in parallel to each nucleus of interest). Each stereotactic level is denoted by a letter (Fig. 5.2):

- Stereotactic level A corresponds to nidopallium at level of LMAN
- Stereotactic level B corresponds to nidopallium at level of area X
- Stereotactic level C corresponds to nidopallium at level of HVC
- Stereotactic level D corresponds to nidopallium at level of RA



**Figure 5.2**. Coronal (column 1) and sagittal (column 2) schematics detailing the location of the selected nidopallial internal control areas (orange), which lie in parallel to each song-system nucleus (highlighted in grey). Stereotactic levels are hereafter termed a, b, c and d corresponding to each level sampled. Abbreviations: A, arcopallium; Cb, cerebellum; Hb, habenula; HA, apical part of the hyperpallium; HD,

densocellular part of the hyperpallium; Hp, hippocampus; HVC (formal name); LSt, lateral striatum; M, mesopallium; MSt, medial striatum; N, nidopallium; NC, caudal nidopallium; LMAN, lateral magnocellular nucleus of the anterior nidopallium; RA, robust nucleus of the arcopallium; TeO, optic tectum; X, Area X. Images have been modified from Thode *et al.* 2008.

Nidopallial regions immediately surrounding the nuclei were not measured to avoid any accessory regions being included in the analyses, such as the HVC shelf (which is present in all birds; Jarvis *et al.*, 2002); LMAN shell and RA cup (Bottjer, 1989; Iyengar and Bottjer, 2002). Software enabled coordinates of the internal control regions to be mapped to ensure consistent sampling of regions throughout the experiments.

### 5.2.2 Spatial expression of $GABA_A$ receptor $\gamma$ 4-subunit gene in aviaryreared and tutor-isolated male zebra finch brains

Preliminary results showed discrete expression of the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit gene was apparent in all four selected nuclei within the song system (LMAN, Area X, HVC and RA), both in the aviary-reared birds and those that were isolated from a tutor (data not shown). Spatial gene expression in both groups of birds was identical to that demonstrated in Chapter Four (Fig. 4.5) with the  $\gamma$ 4-subunit gene well transcribed in many important structures of the avian brain. In addition to those mentioned above, strong hybridisation signals were detected in dorsolateral thalamus (DLM) which forms part of the AFP; field L (formal name), a primary auditory area; optic tectum (TeO) which is involved in visual processing; pallial regions including medial magnocellular nucleus of the nidopallium (MMAN), densocellular part of the hyperpallium (HD) and mesopallium (M); medial and lateral striatum (MSt and LSt); habenula (Hb) which lies adjacent to the LSt; hippocampus (Hp) and cerebellum (Cb) in the brains of both aviary-reared and tutor-isolated zebra finches (refer to Fig. 5.2 for anatomical locations). Signal intensity was analysed densitometrically to clarify and quantify the mRNA levels in each song-system nucleus between the two groups of birds.

### 5.2.3 Quantitative expression of $GABA_A$ receptor $\gamma$ 4-subunit gene in aviaryreared and tutor-isolated male zebra finch brains

#### 5.2.3.1 Internal control (y4-subunit gene expression in nidopallium)

mRNA levels encoding the GABA<sub>A</sub> receptor  $\gamma$ 4 subunit remained largely unchanged by the parameters of the experiment in the nidopallium of aviary-reared and tutor-isolated zebra finches at each stereotactic level (Fig. 5.3). There were elevated mRNA levels in the nidopallium of aviary-reared birds at the equivalent levels for LMAN (A), HVC (C) and RA (D), but these differences were marginal. There was no relationship between the acquisition and production of structured song and GABA<sub>A</sub> receptor  $\gamma$ 4-subunit mRNA expression in the nidopallium; it was therefore acceptable to use this area as an internal control to normalise further results (i.e. expression of  $\gamma$ 4- and  $\gamma$ 2-subunit genes in song nuclei).

NIDOPALLIUM



Cage w /o tutor Z Aviary w /tutor

**Figure 5.3.** Densitometric quantification of GABA<sub>A</sub> receptor  $\gamma$ 4-subunit mRNA in the nidopallium at the stereotactic levels of each song-system nucleus (A, B, C and D in parallel to LMAN, Area X, HVC and RA respectively; refer to Fig. 5.2). Units are arbitrary, and were quantified by subtraction of any background (i.e. non-specific hybridisation signals from the film). Gene expression was compared between birds which were raised in an aviary with a tutor (birds produced normal song) and those raised in individual isolation cages, without a tutor (birds produced abnormal song); all birds were the same age (PHD 80).

GABA<sub>A</sub> receptor  $\gamma$ 4-subunit mRNA was expressed throughout the nidopallium and in all examined nuclei of the song system. Transcript levels remained unchanged in the nidopallium between the aviary-reared and tutor-isolated birds (Fig. 5.3).

#### 5.2.3.2 GABA<sub>A</sub> receptor $\gamma$ 4-subunit mRNA expression in song-system nuclei

Expression of the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit mRNA in LMAN, Area X, HVC and RA was quantified as per section 5.2, the results are shown in Figure 5.4.



Cage w/o tutor Aviary w/tutor

**Figure 5.4**. Densitometric quantification of GABA<sub>A</sub> receptor  $\gamma$ 4-subunit mRNA levels within four nuclei of the zebra finch song system relative to the nidopallium. The dashed line indicates mRNA expression in the nidopallium; these readings were considered as baseline, at 100%. Data is presented as mean ±SE, (n=5 per group; \*\*\*  $p \leq 0.001$ ). Transcript expression was compared between birds which were raised in an aviary with a tutor (birds produced normal song) and those raised in individual isolation cages, without a tutor (birds produced normal song); all birds were the same age (PHD 80). Abbreviations: Area X (formal name); LMAN, lateral magnocellular nucleus of the anterior nidopallium; HVC (formal name); RA, robust nucleus of the arcopallium.

Detailed densitometric analysis of the results demonstrated a highly significant reduction in levels of GABA<sub>A</sub> receptor  $\gamma$ 4-subunit mRNA in HVC (24%) and RA (38%; both  $p \leq$  0.001), in brains of aviary-reared birds relative to tutor-isolated birds (Fig. 5.4). Interestingly HVC and RA form part of the VMP (vocal motor pathway), which is responsible for the production of song. However, there was no significant difference in the level of the  $\gamma$ 4-subunit transcript, between tutor-isolated and aviary-reared birds, in LMAN and Area X, which form part of the AFP (anterior forebrain pathway); which is involved in the memorisation of song. In fact, in Area X, gene expression was marginally higher in brains of tutor-isolated animals compared to their aviary-reared counterparts. In brains of tutor-isolated zebra finches, the highest levels of  $\gamma$ 4-subunit gene expression were detected in the RA followed closely by the HVC (Fig. 5.4) – also the sites of the greatest differences in expression following song-learning. GABA<sub>A</sub> receptor  $\gamma$ 4-subunit gene expression levels across nuclei varied between tutor isolated birds (where RA > HVC > LMAN > Area X) and aviary-reared animals (where HVC > Area X > RA > LMAN). Area X was the only nucleus where  $\gamma$ 4-subunit gene expression was higher in the aviary-reared birds. In summary, data illustrated a significant, regional-specific down-regulation of GABA<sub>A</sub> receptor  $\gamma$ 4-subunit gene expression within the nuclei of the VMP pathway (HVC and RA) in response to acquisition and production of normal song.

Levels of  $\gamma$ 4-subunit mRNA was also measured in left and right hemispheres of the zebra finch brain to determine lateralisation of expression. Data revealed (Fig. 5.5a and b) that there were no significant differences between the  $\gamma$ 4-subunit gene expression in nuclei located in the left and right side of the brain. In tutor-isolated birds, the nuclei of the right hemisphere showed slightly higher expression (Fig. 5.5b). Aviary-reared birds mirrored this pattern with the exception of the RA where the nucleus located within the left hemisphere had marginally higher expression (Fig. 5.5a).



**Figure 5.5.** Densitometric quantification of GABA<sub>A</sub> receptor  $\gamma$ 4-subunit mRNA levels in four selected nuclei of the song system relative to the nidopallium. Gene expression was compared between nuclei within the left and right hemispheres in the brains of aviary-reared (a) and cage-reared, tutor-isolated (b) zebra finches. Data is presented as mean ± SE (n=5 per group). The dashed line indicates mRNA levels in the nidopallium; these readings were considered as baseline, at 100%. Abbreviations: Area X (formal name); LMAN, lateral

magnocellular nucleus of the anterior nidopallium; HVC (formal name); RA, robust nucleus of the arcopallium.

### 5.2.4 Spatial expression of $GABA_A$ receptor $\gamma$ 2-subunit gene in aviaryreared and tutor-isolated male zebra finch brains

It was speculated that the observed decreases in expression within the HVC and RA following normal song learning may not have been exclusive to the GABA<sub>A</sub> receptor  $\gamma$ 4 subunit so the experiment was repeated, with sections from the same brains as before with a probe selective to the GABA<sub>A</sub> receptor  $\gamma$ 2-subunit transcript (same probe as in Chapter 3). This subunit was selected as previous work involving imprinting in the one-day-old chicken showed no increase/decrease of the corresponding mRNA following training (Harvey *et al.*, 1998) and it forms the basis of the most prevalent subtype in the mammalian brain (Whiting *et al.*, 2003a).

Distribution of mRNA encoding the  $\gamma$ 2 subunit was mapped within the zebra finch brain (Fig. 5.6) and autoradiographs illustrated that the spatial distribution of mRNA was identical within brains of aviary-reared and tutor-isolated birds. This concurred with previous data found (for a full description of the  $\gamma$ 2-subunit distribution consult Chapter 3). No song-system nucleus demonstrated particularly robust gene expression (i.e. above basal), with the exception of Area X where  $\gamma$ 2-subunit mRNA levels were higher than the surrounding striatum (Fig. 5.6C and D). Due to the diffuse distribution of  $\gamma$ 2-subunit transcripts were co-expressed, most notably in Area X. It was only the RA where expression of the  $\gamma$ 2-subunit gene expression between tutor-isolated and aviary-reared zebra finches.



**Figure 5.6.** Inverse autoradiographs highlighting the distribution of the GABA<sub>A</sub> receptor  $\gamma$ 2-subunit transcript in four nuclei of the male zebra finch song system, (LMAN, Area X, HVC and RA). Gene expression was compared between birds reared in an aviary with their parents and other conspecifics (column one; A, C, E and G) and birds reared in individual isolation cages without a tutor (column two; B, D, F and H. All birds were sacrificed at PHD 80. **Abbreviations:** A, arcopallium; Cb, cerebellum; Hb, habenula; HA, apical part of the hyperpallium; HD, densocellular part of the hyperpallium; Hp, hippocampus; HVC (formal name); LSt, lateral striatum; M, mesopallium; MSt, medial striatum; N, nidopallium; NC, caudal nidopallium; LMAN, lateral magnocellular nucleus of the anterior nidopallium; RA, robust nucleus of the arcopallium; TeO, optic tectum; X, Area X.

### 5.2.5 Quantitative expression of $GABA_A$ receptor $\gamma$ 2-subunit gene in tutor isolated and non-tutor isolated male zebra finch brains

### 5.2.5.1 Internal control (y2-subunit gene expression in nidopallium)

Hybridisation signals were quantified in the same way as for the  $\gamma$ 4-subunit transcript (section 5.2). Initially levels of  $\gamma$ 2-subunit mRNA were measured at the four stereotactic levels of the nidopallium (A, B, C, and D, refer to Fig. 5.2) the results are given in Fig. 5.7, below.



Cage w/o tutor Aviary w/tutor

**Figure 5.7.** Densitometric quantification of GABA<sub>A</sub> receptor  $\gamma$ 2-subunit mRNA levels in the nidopallium at the stereotactic levels of each song-system nucleus (A, B, C and D in parallel to LMAN, Area X, HVC and RA respectively; refer to Fig. 5.2). Units are arbitrary, and were quantified by subtraction of any background (i.e. non-specific hybridisation signals from the film). Gene expression was compared between birds which were raised in an aviary with a tutor (birds produced normal song) and those raised in individual isolation cages, without a tutor (birds produced abnormal song); all birds were the same age (PHD 80).

Data demonstrated no significant differences in GABA<sub>A</sub> receptor  $\gamma$ 2-subunit mRNA levels in the nidopallium of aviary-reared and tutor-isolated birds at any stereotactic level (Fig. 5.7). In contrast to the results obtained for the  $\gamma$ 4-subunit transcript, there were slightly higher mRNA levels in the nidopallium of tutor-isolated birds compared to aviary-reared birds, with the exception of the RA, but these differences were negligible and in no way significant. Again, the nidopallium was resolved to be a suitable internal control region.





■ Cage w/o tutor 
Aviary w/tutor

**Figure 5.8**. Densitometric quantification of GABA<sub>A</sub> receptor  $\gamma$ 2-subunit mRNA within four nuclei of the zebra finch song system relative to the nidopallium. The dashed line indicates the mRNA level in the nidopallium; these readings were considered as baseline, at 100%. Data is presented as mean ±SE, (n=5 per group). Gene expression was compared between birds which were raised in an aviary with a tutor (birds produced normal song) and those raised in individual isolation cages, without a tutor (birds produced abnormal song); all birds were the same age (PHD 80). **Abbreviations:** Area X (formal name); LMAN, lateral magnocellular nucleus of the anterior nidopallium; HVC (formal name); RA, robust nucleus of the arcopallium.

Data clearly showed that there were no discernable differences between the expression level of the GABA<sub>A</sub> receptor  $\gamma$ 2-subunit gene in any of the nuclei between aviary-reared and tutor-isolated birds (Fig. 5.8). Highest levels of mRNA expression were detected in Area X; this was also visually apparent in the autoradiographs (Fig. 5.6C and D), but generally the distribution was relatively low and diffuse throughout the song system (and the remainder of the brain; data not shown), of both groups of birds. mRNA levels within LMAN, HVC and RA were similar to that of the nidopallium. Results therefore confirmed that the differences observed in  $\gamma$ 4-subunit gene expression were likely to be exclusive to that subunit. As before,  $\gamma$ 2-subunit mRNA distribution was analysed within the nuclei of the left and right hemispheres (Fig. 5.9a and b).



☑ Left hemisphere □ Right hemisphere



**Figure 5.9.** Densitometric analysis of GABA<sub>A</sub> receptor  $\gamma$ 2-subunit gene expression in the left and right hemispheres of four selected nuclei of the zebra finch song system within the brains of aviary-reared (a) and cage-reared, tutor-isolated (b) zebra finches. Signal was quantified by subtracting background signals and expressed as an increase compared to the internal control area. Baseline is defined as the signal intensity in the nidopallium (internal control) minus any background signal (100%). Data is expressed as mean ±SE, n=5 per group. **Abbreviations:** Area X (formal name); LMAN, lateral magnocellular nucleus of the anterior nidopallium; HVC (formal name); RA, robust nucleus of the arcopallium.

Similar to  $\gamma$ 4-subunit mRNA, there were no significant differences in levels of  $\gamma$ 2-subunit encoding transcript between nuclei located within the two sides of the avian brain.  $\gamma$ 2-subunit mRNA was generally, marginally higher within the nuclei of the right hemisphere with the exception of the RA in the tutor-isolated birds (Fig. 5.9b).

### 5.2.6 Effects of tutor-isolation on song production

Following the results of the preliminary experiments, the hypothesis that  $\gamma$ 4-subunit containing GABA<sub>A</sub> receptors play a role in song acquisition and production in zebra finch needed to be further validated. It was so far ascertained that the  $\gamma$ 4-subunit mRNA was reduced in the song system of birds (with the exception of Area X) which learned normal song and that this was specific to the  $\gamma$ 4 subunit, as no such changes in distribution were visible for  $\gamma$ 2-subunit mRNA. It was considered that the effects on gene expression were not entirely due song learning but may have been an artefact of the experimental

conditions. To combat this, three further sets of male zebra finches were raised; the first group were raised with their parents in an aviary with other conspecifics (as in the preliminary experiments), the second group were raised with their parents in an individual cage (this ensured the difference in surroundings was not affecting the results) and the third group were raised in individual cages by their parents until PHD 5, then by their mother until PHD 30 and then by >2 unrelated females for the remainder of the experiment. Presence of females (who do not sing) ensured that any effects of gene expression could not be attributed to social isolation, only tutor-song isolation. Furthermore, the songs of each bird were recorded at PHD 80 in the presence of an unfamiliar female and analysed with a sonograph. All these precautions were in place to ensure that the effects observed in song learning and gene expression by tutor-isolation could be exclusively linked.

Song produced by the two groups of zebra finches raised in individual cages either with or without a tutor was recorded, in the presence of an unfamiliar female at PHD 80. Data revealed that syllable length was much longer in birds that were reared in isolation. There were no differences in the pitch, quality of the pitch, the frequency or entropy between the two groups of birds. Data confirmed that an adult male (tutor) is necessary during the period of song learning for the production of normal song in zebra finches (data not shown). The next step was to determine any differences in the expression of the  $\gamma$ 4-subunit transcript in response to birds which learned song and those which did not.

### 5.2.7 Spatial expression of the GABA<sub>A</sub> receptor $\gamma$ 4-subunit gene in aviaryreared, cage-reared and tutor-isolated zebra finches

Autoradiographs demonstrated that GABA<sub>A</sub> receptor  $\gamma$ 4-subunit transcript was detected in all four nuclei examined (LMAN, Area X, HVC and RA), in all three sets of brains (Fig. 5.10). Unlike in the preliminary experiments, it was apparent from visually comparing the images that there was an increase of transcript in LMAN and Area X in the brains of tutor-isolated birds (Fig. 5.10, column three, C and F). If differences existed between HVC and RA these were not immediately apparent by eye. Spatial mRNA distribution patterns were identical in the brains of all three groups of birds and were comparable to previously obtained data.



**Figure 5.10.** Inverse autoradiographs highlighting the distribution of the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit transcript in four nuclei of the male zebra finch song system (LMAN, Area X, HVC and RA). Gene expression was compared between birds reared in an aviary with their parents and other conspecifics (column one; A, D, G and J); reared in individual cages with their parents (column two; B, E, H and K); or reared in individual isolation cages in the presence of the mother until PHD 30 and then > 2 unrelated females (column three; C, F, I and L). All birds were sacrificed at PHD 80. **Abbreviations:** LMAN, lateral magnocellular nucleus of the anterior nidopallium; X, Area X; HVC, formal name; RA, robust nucleus of the arcopallium.

# 5.2.8 Quantitative expression of the GABA<sub>A</sub> receptor $\gamma$ 4-subunit gene in aviary-reared, cage-reared and tutor-isolated zebra finches

### 5.2.8.1 Internal control (y4-subunit gene expression in nidopallium)

Distribution of GABA<sub>A</sub> receptor  $\gamma$ 4-subunit transcript within the zebra finch song system was quantified to verify any differences between groups. Firstly the levels of transcript were determined within the internal control regions (nidopallium); the results are shown in Figure 5.11.



■ Aviary w/tutor I Cage w/tutor I Cage w/o tutor

**Figure 5.11.** Densitometric quantification of GABA<sub>A</sub> receptor  $\gamma$ 4-subunit mRNA levels in the nidopallium at the stereotactic levels of each song-system nucleus (A, B, C and D in parallel to LMAN, Area X, HVC and RA respectively; refer to Fig. 5.2). Units are arbitrary, and were quantified by subtraction of any background (i.e. non-specific) hybridisation signal. Transcript expression was compared between birds which were raised in an aviary with a tutor and other conspecifics (produced normal song), those raised in individual isolation cages with their parents (produced normal song) and birds raised in individual isolation cages with only females (produced abnormal song). All birds were the same age (PHD 80).

Analysis of the nidopallium again demonstrated that this area was not significantly affected by the parameters of the experiment (Fig. 5.11). There was a nominal decrease in the level of GABA<sub>A</sub> receptor  $\gamma$ 4-subunit mRNA in cage-reared, tutor-isolated birds but was not significant. Therefore cage-rearing (as opposed to in an aviary), had little effect on the gene expression in the internal control region.

#### 5.2.8.2 GABA<sub>A</sub> receptor $\gamma$ 4-subunit mRNA expression in song-system nuclei

In accordance with results obtained from the preliminary experiments, there were higher levels of mRNA encoding the GABA<sub>A</sub> receptor  $\gamma$ 4 subunit in tutor-isolated zebra finches in all four examined nuclei of the song system, compared to aviary-reared birds (Fig. 5.12;  $p \le$ 0.001), potentially demonstrating a decrease of  $\gamma$ 4-subunit mRNA in response to acquisition and production of structured song. Strength of response was far greater relative to preliminary data (Fig. 5.4). Decreases in  $\gamma$ 4-subunit encoding mRNA following song learning were highly significant for all four nuclei; LMAN (40%), Area X (30%), HVC (70%) and RA (100%), whereas in preliminary experiments increases were only observed in RA and HVC (Fig. 5.4). Although it was within these nuclei where the most pronounced effects were measured (Fig. 5.12).



Aviary w/tutor Cage w/tutor Cage w/o tutor

**Figure 5.12**. Densitometric quantification of GABA<sub>A</sub> receptor  $\gamma$ 4-subunit mRNA within four nuclei of the zebra finch song system relative to the nidopallium. The dashed line indicates gene expression in the nidopallium; these readings were considered as baseline, at 100%. Data is presented as mean ±SE, (n=5 per group; \*\*\*  $p \le 0.001$ ). Expression was compared between birds which were raised in an aviary with a tutor and other conspecifics (produced normal song), those raised in individual isolation cages with their parents (produced normal song) and birds raised in individual isolation cages with only females (produced abnormal song). All birds were the same age (PHD 80). **Abbreviations:** Area X (formal name); LMAN, lateral magnocellular nucleus of the anterior nidopallium; HVC (formal name); RA, robust nucleus of the arcopallium.

There were no significant differences in gene expression in brains of birds which were reared normally in an aviary (produced normal song) and those reared in isolation cages with their parents (produced normal song; Fig. 5.12), thereby demonstrating that rearing in a cage environment or absence of conspecifics had no effect on the outcome of the experiment. Presence of female birds (which do not sing) in cages of tutor-isolated birds ensured that none of the effects could be attributed to the stress of social isolation, thus the decrease of  $\gamma$ 4-subunit expression in all four observed nuclei was more likely linked to the acquisition and production of song. In these results, gene expression in nuclei within both the AFP and VMP pathway were significantly affected (although HVC and RA of the VMP pathway were the regions exhibiting the greatest decreases in gene expression; Fig. 5.12), thereby demonstrating significant decreases in GABA<sub>A</sub> receptor  $\gamma$ 4-subunit mRNA in response to both song-acquisition and production in zebra finch.





subtracting background signal and expressed as an increase compared to the internal control area. Baseline is defined as the signal intensity in the nidopallium (internal control) minus any background signal (100%). Data is expressed as mean  $\pm$ SE, n=5 per group. **Abbreviations:** Area X (formal name); LMAN, lateral magnocellular nucleus of the anterior nidopallium; HVC (formal name); RA, robust nucleus of the arcopallium.

As seen previously (Fig. 5.5) there were no significant differences in  $\gamma$ 4-subunit transcript expression between nuclei occupying the left or right hemispheres of the brain (Fig. 5.13) showing no evidence of lateralisation of gene expression.

### **5.3 Discussion**

It has been recently documented that the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit is well transcribed in virtually all nuclei of the zebra finch song system (Thode et al., 2008). Gene expression was so robust and specific that is was proposed as a potential molecular marker for the song system. Following these observations, the work outlined in this chapter quantitatively mapped expression of the  $\gamma$ 4-subunit gene and determined the effect of song learning on transcript levels. The data revealed several key findings. Firstly, birds isolated from a conspecific tutor during song development (up to PHD 80) produced abnormal song. Birds raised both in an aviary or separate cages, with a tutor, produced normal vocalisations. Secondly, *in situ* hybridisation revealed that adult male zebra finches which learned song and those that did not, both shared identical spatial gene expression profiles where the corresponding transcript was detected within all the major nuclei of the song system involved in the acquisition and production of song. Thirdly, although spatially similar, levels of GABA<sub>A</sub> receptor y4-subunit mRNA were significantly different between normally-, (aviary)-reared and tutor-isolated zebra finches. Acquisition and production of normal song was associated with a down-regulation of GABA<sub>A</sub> receptor  $\gamma$ 4-subunit gene expression in four relevant nuclei of the song system. LMAN and Area X which form part of the AFP and HVC and RA which form part of the VMP. Finally, the decrease in expression in response to normal song learning was specific to the  $\gamma$ 4-subunit gene as no learning-associated decreases were observed for the  $\gamma 2$  subunit transcript.

There is a widespread distribution of GABAergic interneurons within the zebra finch song 2007). GABA<sub>A</sub> receptors have been detected system (Pinaud and Mello, electrophysiologically (for example, Bottjer et al., 1998; Dutar et al., 1998; Spiro et al., 1999; Farries et al., 2005; Mooney and Prather, 2005; Prather et al., 2008) and pharmacologically (for example, Carlisle et al., 1998; Vicario and Raksin, 2000; Farries et al., 2005; Mooney and Prather, 2005; Ölveczky et al., 2005), but little information concerning the subtypes present or their physiological roles is available. Only the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit has been unequivocally identified (Thode *et al.*, 2008). The GABA<sub>A</sub> receptor family is highly diverse with a large number of promiscuous subunits associating to form a currently unknown number of receptor subtypes. Each receptor subtype exhibits slightly different biophysical, electrophysiological and pharmacological properties (conferred by the particular combination of subunits; Olsen and Sieghart, 2008; 2009). Furthermore, in mammalian brain, there is regional-specific expression of all the subunit genes indicative of different GABA<sub>A</sub> receptor subtypes fulfilling specific functions (Laurie et al, 1992; Wisden et al, 1992; Pirker et al., 2000; Pöltl et al., 2003; Wafford et al., 2004). This diversity of expression has now been observed in zebra finch brain (refer to Chapter 4) and so the stoichiometry and functionality of different GABA<sub>A</sub> receptor subtypes can begin to be uncovered in this well established learning and memory paradigm.

### 5.3.1 Effects of tutor-isolation on song

Songs of all birds were recorded at PHD 80 in the presence of an unfamiliar female (directed song). Data from the resulting sonographs illustrated that zebra finches raised in tutor isolation produced abnormal song, which concurs with much previous data (Thorpe, 1958; Immelmann, 1969; Marler and Sherman, 1983; Eales, 1987; Searcy and Marler, 1987; Tchernichovski and Nottebohm, 1998; Kojima and Doupe, 2007; Fehér *et al.*, 2009). Most notably, the syllable duration was considerably longer in songs of tutor-isolated birds relative to their aviary-reared counterparts. There were no discernable differences in the pitch, quality of the pitch, the frequency or entropy between the two groups of birds. An increase in syllable length is a feature of the song of birds raised in the absence of a tutor (Kojima and Doupe, 2007), potentially associated with a disruption of neuronal processing in the HVC (Long and Fee, 2008). As this nucleus projects to both pathways of the song

system and is the entry point of auditory information into the song system, it is of critical importance to both the acquisition and production of learned song. Slowing of neuronal processing in HVC results in songs with longer syllable durations and longer periods between syllables (but the song remains stable; Long and Fee, 2008). Generally untutored song is less complicated with reduced syllable variety and longer pauses between syllables (Price, 1979; Laulay *et al.*, 2004). However, interestingly, isolate song also incorporates some species-specific features (Marler and Sherman, 1985; Zann, 1996; Konishi, 2004; Kojima and Doupe, 2007), perhaps attributable to the proposed intrinsic auditory template, present form birth which pre-disposes birds to sing a song selective to their species (Marler, 1970; Konishi, 2004).

### 5.3.2 Effects of environment on song production

Aviary- and cage-reared (both with tutor) zebra finches produced normal song. With regards to gene expression, levels of  $\gamma$ 4-subunit mRNA were slightly higher in all four nuclei of cage-reared birds relative to aviary-reared, although the differences were not significant (Fig. 5.12). Birds raised in a richer auditory environment with much more complex stimuli (i.e. in an aviary with many other conspecifics and tutors) produce much more complex songs (Adar *et al.*, 2008). Therefore, learning of structured song was associated with a decrease in GABA<sub>A</sub> receptor  $\gamma$ 4-subunit gene expression and correspondingly, exposure to a richer auditory stimuli during song learning (and perhaps learning of a more complex song – although this was not tested) was associated with a further decrease in  $\gamma$ 4-subunit mRNA within relevant regions of the song system. Pertaining to this, only songs of zebra finches raised in cages (with parents) were compared with those of tutor-isolated birds.

Zebra finches are social birds living in large groups (up to 1000 members in the wild). To ensure the stress associated with social-isolation was not a factor in the experiments, tutor-isolated animals were raised in the presence of conspecific females. Female zebra finches do not sing (Nottebohm and Arnold, 1977); both males and females produce vocalisations after hatching, but these are begging calls (Zann, 1996) and are not learned (Price, 1979). Females cannot produce structured song; they can make whistling noises and calls and thus

cannot create a sufficient auditory template for juveniles to learn (Nottebohm and Arnold, 1977; Price 1979; Zann, 1996; Reibel, 2003). However, tutor-isolated zebra finches can incorporate some female call notes when permanently isolated from male song (Eales, 1987). In this context, it could be argued that the tutor-isolated birds reared with females may have acquired some rudimentary form of an auditory template (although nowhere near as reinforced as in those raised with tutors), resulting in activation of the AFP. Although the role of female interaction is currently unknown, it may be that by some mechanism the female is able to indicate positive reinforcement during the song-learning process (King and West, 1983; Adret, 2004). Certainly in normally-reared birds, those exposed to a tutor and females learn better than those reared with a tutor alone (West and King, 1988; Adret, 2004).

Another interesting feature of tutor-isolated birds is that even after the period of song learning (post-crystallisation), they prefer (Braaten and Reynolds, 1999), and are able to learn (to an extent), species-specific song (Eales, 1987; Slater *et al.*, 1988; Livingston *et al.*, 2000). Inferring that the song-learning phase in close-ended learners such as zebra finches may remain open after sexual maturity until sufficient, suitable stimuli has been provided. Deafening, muting, perturbation of auditory feedback or isolation from conspecific song, post-crystallisation can result in deterioration of adult song (Nordeen and Nordeen, 1992; Leonardo and Konishi, 1999; Williams *et al.*, 2003; Funabiki and Funabiki, 2007; Sober and Brainard, 2009), which in some cases can be corrected following re-exposure to tutor song (Funabiki and Funabiki, 2007). In birds isolated during song learning, such as in this study, a normal sensory template could not have been established (as reflected by the production of abnormal song). This was accompanied by high levels of  $\gamma$ 4-subunit mRNA within the song system, which would normally be reduced following the acquisition of song.

# 5.3.3 Spatial expression of $\gamma$ 4- and $\gamma$ 2-subunit genes in aviary-reared and tutor-isolated birds

Spatial mRNA distribution profiles for both the GABA<sub>A</sub> receptor  $\gamma$ 2- and  $\gamma$ 4-subunits were identical to those identified in Chapter Four (Figs. 5.6, 5.10 and 4.5) and in agreement with

previous findings (Thode et al., 2008). Thus, despite the birds being raised in different aviaries, of various ages and collected at different seasons of the year (for the individual experiments), gene expression patterns were consistent, further confirming that the distribution in the song system of both subunit transcripts shown herein was accurate. Spatially, the expression patterns of both the  $\gamma$ 2- and  $\gamma$ 4-subunit genes were identical in the song system of aviary-reared and tutor-isolated zebra finches. The GABA<sub>A</sub> receptor  $\gamma$ 2subunit gene exhibited low, but diffuse expression throughout the song system (and the remainder of the brain). This was anticipated as in mammals, this subunit associates with  $\alpha$ 1- and  $\beta$ 2-subunit polypeptides to form the most abundant (~50%), GABA<sub>A</sub> receptor subtype in the mammalian brain (Whiting et al., 2003). The relative abundance of the  $\alpha 1\beta 2\gamma 2$  subtype is indicative of a generalised function, whereas the restricted yet robust expression of the  $\gamma$ 4-subunit gene points to a more specialised role, perhaps in learning and memory processes. Widespread, diffuse expression of the  $\gamma$ 2-subunit gene meant that in many regions of the avian brain (including the song-system nuclei), there was overlapping expression with the  $\gamma$ 4 subunit; this was most notable in Area X. Despite a small amount of evidence suggesting that two different  $\gamma$ -subunit polypeptides may assemble into a functioning receptor (Khan et al., 1994; Quirk et al., 1994; Benke et al., 1996), the general consensus is that native GABA<sub>A</sub> receptors contain a single  $\gamma$  subunit (Olsen and Sieghart, 2008, 2009). However, it is most likely that specific brain regions contain more than one subtype of GABA<sub>A</sub> receptor (Laurie et al, 1992a, b; Wisden et al, 1992; Pirker et al., 2000; Pöltl et al., 2003; Wafford et al., 2004). In mammalian systems, γ2-subunit-containing receptors may be synaptic or extrasynaptic, whereas those lacking a  $\gamma$ 2-subunit are extrasynaptic (Olsen and Sieghart, 2008). If avian systems mirror that of mammals, then it can be postulated that  $\gamma$ 4-subtype receptors are extrasynaptic and thus would occur on the same neurons as the synaptic  $\gamma$ 2-subunit-containing receptors. Furthermore, expression of  $\gamma$ 2- and  $\gamma$ 4-subunit mRNAs seem unlikely to be linked as  $\gamma$ 4-subunit mRNA was downregulated in response to learning and  $\gamma$ 2-subunit mRNA expression was unaffected.

In addition to the four nuclei under scrutiny, expression of the  $\gamma$ 4-subunit gene was robust in many other regions of the brain, including the cerebellum, an area responsible for control of complex behaviour which has recently been implicated in cognitive functions (Spence *et al.*, 2009). The hippocampus also displayed prominent levels of  $\gamma$ 4-subunit mRNA; this region is involved in processing of spatial memory (Patel *et al.*, 1997; Wanatabe and Bischof, 2004). Robust  $\gamma$ 4-subunit gene expression in these two areas may indicate a role in other important memory systems for GABA<sub>A</sub> receptors containing this subunit.

# 5.3.4 Down-regulation of $GABA_A$ receptor $\gamma$ 4-subunit transcript in response to song learning

Although the role of GABA<sub>A</sub> receptors in learning and memory has been well documented (reviewed by Chapouthier and Venault, 2002; Maubach, 2003; McNally et al., 2008), the precise mechanisms of their action are not clarified. The majority of data pharmacologically confirms the involvement of GABA<sub>A</sub> receptors in these processes (reviewed by Brioni, 1993). GABA<sub>A</sub> antagonists such as bicuculline (Brioni and McGaugh, 1988; Clements and Bourne, 1996) and inverse agonists such as FG7142 and L-655 708 improve memory formation (Chambers et al., 2004; McNally et al., 2008) and accordingly, agonists such as muscimol promote amnesia and impair performance in memory tasks (Clements and Bourne, 1996; Ramirez et al., 2005; Zarrindast et al., 2006; Yamamoto et al., 2007). Due to the plethora of GABAA receptor subtypes, their regional-specific distribution and their individual pharmacological profiles (indicative of specialised physiological functions), it is paramount to identify the role of individual subtypes. Data showed a regional-specific decrease in GABA<sub>A</sub> receptor y4-subunit mRNA within the zebra finch song system following acquisition and production of normal song (Figs. 5.4 and 5.12). This was in agreement with previous work which reported a down-regulation of  $\gamma$ 4subunit mRNA in the entopallium and intermediate medial mesopallium of one-day-old chicks (Gallus gallus domesticus), 10 h following imprinting training (another established learning and memory paradigm - Horn, 1998; Harvey et al., 1998). y2-subunit mRNA levels in the song system of tutor-isolated birds did not change relative to aviary-reared birds (Fig. 5.8). This concurred with previous data which demonstrated that  $\gamma$ 2-subunit gene expression remained unchanged in the brain by imprinting in the one-day-old chick (Harvey *et al.*, 1998). In contrast, following water-maze training,  $\gamma$ 2-subunit mRNA was

one of four GABA<sub>A</sub> receptor subunits down-regulated in the hippocampus of adult rats (along with  $\alpha 4$ -  $\alpha 5$ - and  $\beta 2$ -subunits) – interestingly,  $\pi$ -subunit mRNA was up-regulated in response to this training (Cavallaro et al., 2002). Such conflicts in data may arise as this was in mammalian systems (which do not have the  $\gamma$ 4 subunit; Simon *et al.*, 2004), within a completely different region of the brain and a different learning paradigm. Although the decreases in GABA<sub>A</sub> receptor subunit expression is response to learning were not quantified, the results show an overall down-regulation of GABA<sub>A</sub> receptor subunit expression (Cavallaro et al., 2002). The work also served to highlight the differential expression of GABA<sub>A</sub> receptor subtypes in the brain and the presence of subtype-specific mediated responses to learning and memory processes, as seen herein. However, despite localisation of different GABA<sub>A</sub> receptor subtypes to specific regions of the brain and the different physiological properties of these receptors, there is a paucity of data relating individual subtypes to specific learning and memory processes. An exception to this is the  $\alpha$ 5-subtype of the GABA<sub>A</sub> receptors. Receptors comprising the  $\alpha$ 5 subunit are primarily located in the hippocampus where they constitute ~20% of the GABA<sub>A</sub> receptor population in this region (Sur et al., 1998, 1999). This regional-specific localisation of a specific receptor subunit gene in an area of the brain associated with spatial learning (Wanatabe and Bischof, 2004) was indicative for a role of  $\alpha$ 5-subunit receptors in these processes (much like the discrete distribution of  $\gamma$ 4-subunit mRNA within the nuclei of the song system). Homozygous  $\alpha$ 5-subunit knockout mice demonstrated improved performance in hippocampal-dependent spatial learning tasks (Collinson et al., 2002) and point mutations within the  $\alpha$ 5-subunit gene facilitated trace fear conditioning in mice (Crestani *et al.*, 2002). This lead to identification of a partial inverse agonist (L-655 708), an imidazobenzodiazepine with a high affinity for  $\alpha$ 5-subunit-containing receptors, which enhances performance of rats in the Morris water maze (Chambers *et al.*, 2004). Again this data confirms a down-regulation of GABAA receptor-mediated inhibition is associated with learning and memory processes and exemplifies the pharmacological benefits of identifying specific receptors and their function in the brain.

Expression of the  $\gamma$ 4-subunit gene in nuclei of the zebra finch song system has been thoroughly documented (Thode *et al.*, 2008), but it remained unknown as to whether
GABA<sub>A</sub> receptors containing this subunit played a functional role in song learning. This work has indicated a role for  $\gamma$ 4-subunit-containing receptors in both the acquisition and production of song. It may be theorised that as the birds were in isolation from a tutor, they did not acquire an auditory template and so gene expression within nuclei controlling auditory template acquisition would be most affected (LMAN and Area X), with these structures not being as active as those in the brains of birds stimulated by a tutor. Moreover, it may be speculated changes in gene expression in nuclei controlling the motor production of song (HVC and RA) may be less prone to tutor-isolation as both isolated and deafened birds proceed to make vocalisations, even if they are abnormal (Immelmann, 1969; Eales, 1987; Searcy and Marler, 1987; Tchernichovski and Nottebohm, 1998; Konishi, 2004; Kojima and Doupe, 2007; Fehér et al., 2009). Yet it was the HVC and RA (which lie within the VMP) which exhibited the greatest decreases in y4-subunit mRNA following normal song learning. This can be explained by the inextricable links between the two pathways. The origin of both pathways is the HVC. Separate classes of neurons project to RA and Area X; the connections to RA are only established at ~PHD 35 in parallel to onset of the sensorimotor phase (Akutagawa and Konishi, 1985; Mooney and Rao, 1994; Scott and Lois, 2007). The HVC $\rightarrow$ RA pathway (VMP) is involved in the motor production of song (Nottebohm et al., 1976; Simpson and Vicario, 1990), where the RA is the output of the song system projecting to hyperglossal motor neurons which innervate the syrinx (nXIIts) and Ram (nucleus retroambigualis) which is the site of vocal and respiratory muscle coordination (Nottebohm et al., 1976). The AFP is primarily involved in the acquisition of song (Bottjer et al., 1984; Scharff and Nottebohm, 1991), however, more recently, it has been shown to be involved in adult song plasticity, although the exact mechanisms remain unknown (Nordeen and Nordeen 1992; Brainard and Doupe, 2000; Funabiki and Konishi, 2003; Williams et al., 2003; Sober and Brainard, 2009). Projections from HVC link it to Area X, which subsequently connects to LMAN via the DLM (refer to Fig. 1.2, Chapter 1). Premotor neurons of RA receive input from LMAN (as well as those from the HVC). Thus, the involvement of HVC and RA in both pathways may account for the exacerbated effects (on  $\gamma$ 4-subunit expression) described in these particular nuclei. Furthermore, the lessened effects on y4-subunit expression within LMAN and Area X (AFP) of tutor-isolated zebra finches may be attributable to the formation of some rudimentary auditory template by

incorporation of some female call notes (Eales, 1987) and the presumed presence of a genetically-encoded intrinsic template (Marler, 1970; Konishi, 2004), resulting in production of abnormal song which still retained some species-specific features.

It is estimated that > 100 genes are regulated (up or down) by singing in the zebra finch brain (Wada et al., 2006), giving some insight into the complex underlying molecular mechanisms (Jarvis et al., 2002). Immediate-early genes such as ZENK, c-fos and c-jun have all been identified in the song system and their expression is rapidly increased in relevant regions when birds hear species-specific song (Mello et al., 1992; Mello and Clayton, 1994, Nastiuk et al., 1994; Bolhuis et al., 2001). Furthermore, (as detailed in section 5.1) many genes have been be implicated in song-driven gene expression (Jarvis and Nottebohm, 1997; Kimpo and Doupe, 1997; Li et al., 2000; Lombardino et al., 2005; Tarcisco et al., 2005; Poopatanapong et al., 2006; Wada et al., 2006). It is now proposed that due to genes being differentially expressed within song nuclei in variable combinations and at different basal levels, generalised neural activity cannot be the only mechanism implemented in regulation of gene expression within the song system (Wada et al., 2004). In the avian brain, imprinting studies in one-day-old chick have demonstrated regionalspecific increases of mRNA encoding Arc/Arg3.1 (Bock et al., 2005) and ZENK (Thode et al., 2005) in response to learning; in similar experiments to where  $\gamma$ 4-subunit mRNA was found to be down-regulated in response to imprinting training (Harvey et al., 1998). However, in contrast to data presented here, ZENK mRNA is transcribed in all the telencephalic nuclei of the song system in response to song production, in deafened or isolated zebra finches at the same levels as in normally reared birds (Johnson and Whitney, 2005) confirming its status as a motor-driven gene, unlike  $\gamma$ 4-subunit mRNA (Jarvis and Nottebohm, 1997). At a protein level, ZENK mRNA was not faithfully translated in RA. In birds which had been deafened or isolated (despite high levels of mRNA), there was a significant decline in ZENK protein production when compared to normally reared birds. These findings suggested a less common, post-transcriptional regulation of the ZENK gene during motor behaviour in RA (Johnson and Whitney, 2005). Whereas differential expression of the  $\gamma$ 4-subunit mRNA in differently reared birds suggests transcriptional level gene regulation. Further levels of complexity are introduced when considering the type of

song, direct song (in the presence of females) results in an induction of FOS but an absence of ZENK in the AFP and in indirect song ZENK is induced (Kimbo and Doupe, 1997; Jarvis et al., 1998). Findings such as these can link behaviour with regulation of gene expression. Moreover, in the characterisation of genes that were differentially expressed in HVC and the surrounding shelf region, in order to determine the specialisation of molecular and biochemical pathways in the functioning of the HVC, the  $\varepsilon$  subunit of the GABA<sub>A</sub> receptor was tentatively identified as one of the differential markers (Lovell et al., 2008). As lower vertebrates do not posses this subunit and instead have the orthologue  $\gamma 4$  it is likely to have been this that was detected (sequence details were not given). It was postulated that the expression was therefore potentially similar to ZENK in that during song production there is high mRNA expression in the HVC (Jarvis and Nottebohm, 1997) and when listening there is significantly higher gene expression in the shelf region relative to HVC (Mello and Clayton, 1994). This may be interpreted as different GABA<sub>A</sub> receptor subtypes or perhaps the same subtype (but both including the  $\gamma$ 4-subunit) playing region-specific physiological roles within nuclei of the song system. mRNA levels in HVC shelf region were not quantified in this study.

#### 5.3.5 Temporal expression of $GABA_A$ receptor $\gamma$ 4-subunit gene

Another avenue of evidence suggesting a critical physiological role for GABA<sub>A</sub> receptors containing the  $\gamma$ 4-subunit polypeptide is that the corresponding mRNA has been shown to exhibit a developmentally-regulated pattern of expression. It is well expressed in LMAN and Area X (AFP) in juvenile zebra finches but is absent from RA in birds of PHD 22. However, at ~PHD 35 the  $\gamma$ 4-subunit mRNA is highly expressed in the RA and these high levels are continued into adulthood (Thode *et al.*, 2008). The rapid increase in mRNA levels in the RA was simultaneous to establishment of the VMP pathway by innervation of the RA by projections from the HVC (Nordeen and Nordeen, 1988; Wang *et al.*, 1999). A similar response was observed for the ZENK protein, whereby it was only detected in the RA at the stage of plastic or crystallised song. Interestingly this was not seen at mRNA level (where it was high in the RA in all stages of development), indicating that transcription was driven by singing but translation was conditional upon behavioural development (Whitney *et al.*, 2000). As aforementioned, in the case of the  $\gamma$ 4 subunit gene, regulation appears to be at an mRNA level. Coupling this with the profound decrease in  $\gamma$ 4-subunit mRNA levels in nuclei of the VMP following acquisition and production of normal song found here, the evidence strongly implies a functional role for these receptors.

#### 5.3.6 Regulation of gene expression

As aforementioned, the FOXP2 gene was the first gene to be implicated in a human communication disorder (Lai et al., 2001; 2003). Like the GABAA receptor y4-subunit gene, FOXP2 is well expressed within the adult zebra finch song system, and in Area X of canaries, expression of FOXP2 is seasonally regulated, with highest mRNA expression levels when song is plastic (Scharff and Haesler, 2005). A study with adult zebra finches demonstrated that the expression of *FOXP2* was significantly down-regulated in Area X in male birds which produce normal song in comparison to non-singing zebra finches (Teramitsu and White, 2006), mirroring data obtained here for the  $\gamma$ 4-subunit gene. Interestingly, comparison of sequence data between humans, songbirds and non-songbirds demonstrated no appreciable differences at a peptide level of FOXP2 between singing and non-singing birds and in addition no residue similarities between songbird and human FOXP2 (both of which produce vocalisations; Webb and Zhang, 2005). Thus, it is proposed that the evolutionary differences can be attributed to regulatory elements, rather than coding sequence evolution. Both GABA<sub>A</sub> receptors and FOX molecules show regionalspecific expression (Laurie et al., 1992a, b; Wisden et al., 1992; Haesler et al., 2004; Thode et al., 2008) suggestive of multiple regulatory elements and with high sequence homology between the genes of different species. The evolutionary changes (i.e. producing vocalisations) are therefore potentially due to mechanisms that control gene expression rather than sequence changes in proteins (Carroll, 2005). Transcriptional mechanisms controlling both these sets of genes are not well understood and whilst the GABAA receptors and FOXP2 appear to play an important role in song acquisition and production in the zebra finch, their expression is ultimately controlled by regulatory elements and it is these that should be more closely scrutinized (discussed in further detail in section 7.3). It was postulated by Thode *et al.* (2008) that the high levels of  $\gamma$ 4-subunit gene expression in the male zebra finch brain may be attributable to steroid hormones (and not learning and memory processes), due to the strong sexual dimorphism of the song system and its' sensitivity to steroids (Nottebohm and Arnold, 1976; Brenowitz et al., 1997; Thompson et al., 2007). Neurosteroids (potent allosteric, positive modulators of GABA<sub>A</sub> receptors; Belelli and Lambert, 2005; Hosie et al., 2006, 2007, 2009; Li et al., 2009), can accumulate in the brain after being locally synthesised or after metabolism of peripheral steroids. They can modulate GABA<sub>A</sub> receptors by direct activation (Hosie et al., 2009), or via hormoneactivated nuclear receptors which function as transcription factors and can interact with specific sites within GABA<sub>A</sub> receptor subunit gene promoter regions. Such steroid hormone response elements have been identified in the majority of subunit sequences, which interact with ERs (estrogen receptors), PRs (progesterone receptors), GRs (glucocorticoid receptors) and AR (androgen receptors). Mammalian  $\gamma 1$ -,  $\gamma 2$ - and  $\varepsilon$ - (the orthologue of  $\gamma 4$ ) subunit gene promoters contain AR-consensus motifs (Steiger and Russek, 2004). Due to the presence of testosterone (and its metabolites) in the song system during development (Nottebohm 1989), a surge of circulating testosterone just prior to song crystallisation (Pröve 1983) and fluctuations in circulating testosterone during seasonal plasticity (Brenowitz et al., 1997), it may modulate transcription of GABA<sub>A</sub> receptor subunit genes. On the other hand, previous research indicated that in deafened birds endogenous circulating testosterone levels remained unaffected (Woolley and Rubel, 2002; Brenowitz et al., 2007), thus it is possible that isolated zebra finches had the same levels of neurosteroids as their aviary-reared counterparts and that changes in y4-subunit mRNA levels were unlikely to be simply a consequence of hormonal regulation and certainly appear to play a role in learning and memory mechanisms.

As previously mentioned, song is a behaviour which is influenced by early auditory and social experience and development of the neural substrate is also subject to these factors. The development of the song system in zebra finches is on-going throughout development with synaptogenesis and loss of neurons all contributing to the establishment of the nuclei of the vocal pathways (Fig. 1.4; Nordeen, 2006). The process of song learning significantly reduces spine density in LMAN. If the sensory period of song learning is extended, then so is the length of time in which the dendritic spines are lost (Wallhäusser-Franke *et al.*, 1995). Anatomical modification of synapses are integral to the functional development of the nervous system and it appears that the effects of learning are presented as dendritic

pruning (Bock and Braun, 1999; Nordeen, 2006), perhaps eradicating superfluous synapses as a means of information storage (Scheich *et al.*, 1991). This is exemplified by a significant reduction of dendritic spine frequencies in the Dorsocaudal epistriatal pallium (formerly dorsocaudal neostriatum) following filial imprinting in the one-day-old chicken (Bock and Braun, 1999). Such anatomical synaptic changes such as decrease in synaptic connections following learning correlate well with the observed decrease of  $\gamma$ 4-subunit mRNAs. However, there is some evidence to the contrary. Deafened birds cannot learn and produce normal song, but neuronal number and morphology within nuclei remains the same (Burek *et al.*, 1991). Moreover, in late-isolated birds (after PHD 30), dendritic spine density remained relatively unaffected even when the zebra finches memorised new songs (Heinrich *et al.*, 2005). It is possible that it is the neuroendocrine system which is ultimately responsible for synaptic remodelling, but this requires further investigation, for a more comprehensive review see Nordeen (2006).

#### 5.3.7 Lateralisation of function

The data indicated that there was no significant lateralisation of function. There was a marginal increase in  $\gamma$ 4-subunit mRNA expression within the nuclei of the right side of the brain (Fig. 5.13), which is consistent with previous findings (Williams *et al.*, 1992). This is in contrast to many other songbirds in which it is the nuclei of the left hemisphere which are more important in the production of song (Nottebohm and Nottebohm, 1976, 1977) and in some species there is no evidence of lateralisation at all (Suthers *et al.*, 1990). The slight increase in mRNA expression in nuclei of the right hemisphere of the zebra finch brain was also observed for the  $\gamma$ 2-subunit; although, gene expression is not necessarily a definitive indication of lateralisation of function.

#### 5.3.9 Concluding remarks

An understanding of the biology of human behaviour starts with the recognition that there are many parallels between humans and many other species and human-associated tendencies are recognisable throughout the animal kingdom (Adkins-Regan, 2002); an excellent example of this is the parallel between human vocalisations and the acquisition and production of song by songbirds (Kuhl, 2004).

The discrete and proliferate distribution of GABA<sub>A</sub> receptor y4-subunit transcript within nuclei of the avian song system is indicative of an important physiological role. Moreover, zebra finches unable to produce normal song demonstrate higher mRNA levels encoding the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit in nuclei contained in both the AFP and VMP, thereby confirming the theory that down-regulation of GABAergic mechanisms plays a role in learning and memory systems (Clements and Bourne, 1996; Harvey et al., 1998; Cavallaro et al., 2002; Collinson et al., 2002; Zarrindast et al., 2006). This effect appears to be exclusive to the  $\gamma 4$  subunit, no such effects were observed with the  $\gamma 2$ -subunit transcript and coupled with the widespread, diffuse expression was indicative of a more generalised functional role in the CNS as opposed to the robust, specific expression of the  $\gamma$ 4-subunit gene which suggested a more specific functional role. Given that GABAA receptors that contain this subunit have a relatively unique pharmacology (Forster et al., 2001); specificsubtype-selective drugs could be used to ameliorate brain function in learning and memory processes. This information may be extrapolated to human subjects due to the many parallels between human and avian telencephalic systems and the presence of the  $\gamma$ 4subunit orthologue,  $\varepsilon$  in the CNS of mammals.

# 6. PRELIMINARY WORK TO CONFIRM THE ROLE OF $\gamma$ 4-SUBUNIT-CONTAINING GABA<sub>A</sub> RECEPTORS IN THE SONG SYSTEM

# 6.1 Introduction

Although real-time RT-PCR and *in situ* hybridisation were useful tools in providing a qualitative and quantitative insight into expression levels and spatial distribution of GABA<sub>A</sub> receptor subunit-mRNAs, it was important to consider if these results corresponded at a protein level. mRNA expression studies have proven invaluable in the development of markers e.g. for cancer and other diseases (Wallace et al., 2001; Cossette et al., 2002; Edenberg et al., 2004; Backus et al., 2005; Symmans et al., 2005; Torrey et al., 2005). However, these results only represent a 'pattern' as it is the proteins they encode and their interactions within cells which are the causative agents of a phenotype (Greenbaum et al., 2003). It is generally accepted that mRNA is eventually translated into protein, thus a correlation between the two would be expected; however this is not necessarily always the case (Greenbaum et al., 2003). In the field of proteomics, mathematical models have been employed with varying levels of success in an attempt to provide a solution to nonlinear cellular dynamics (Hatzimanikatis et al., 1999; Yu et al., 2007). The primary problem lies in that there are numerous considerations such as mRNA degradation, differing transcription rates and gene sequence lengths to name but a few. It has been demonstrated that in order to create a model it is necessary to have expression information both at mRNA and protein levels. Looking at either level individually to understand gene networks can lead to erroneous conclusions about the genes involved in a particular phenotype (Hatzimanikatis and Lee, 1999; Rhodes and Trimmer, 2006). To this end, investigations into the potential role of  $\gamma$ 4-subunit-containing GABA<sub>A</sub> receptors required analysis at a protein level.

### 6.1.1 Immunohistochemical/immunocytochemical studies

The primary problem with GABA<sub>A</sub> receptor subunit-specific antibodies is that they are not always as specific as required. High levels of sequence similarity between individual subunits of the same class (as demonstrated in Chapter Three), means that targeting a single epitope is very difficult; this is most notable with the  $\beta$  subunit group (McKernan and Whiting, 1996). Despite this, a plethora of immunohistochemical, immunocytochemical and immunoprecipitation studies have been undertaken in an effort to determine the possible co-assembly of the GABA<sub>A</sub> receptor subunits into the pentameric subtype (for review see Sieghart and Sperk, 2002).

Some studies have employed immunohistochemical techniques to localise the distribution of 13 GABA<sub>A</sub> receptor subunits (not  $\pi$ ,  $\theta$ , or  $\varepsilon$  subunits) within the rat hippocampus (Sperk et al., 1997); whole brain (Pirker et al., 2000); cerebellum (Pöltl et al., 2003), basal ganglia and associated limbic areas (Schwarzer et al., 2001). These studies all concluded a regional-specific distribution of corresponding subunit polypeptides and regional co-localisation was also apparent. Other studies have examined protein distribution of only a selection of subunits in the rat brain (Zimprich et al., 1991; Fritschy et al., 1992; Fritschy and Möhler, 1995; Nusser et al., 1996; Scotti and Reuter, 2001). The classical  $\alpha 1\beta 2/3\gamma 2$  combination was found within single neurons across many brain areas (Fritschy et al., 1992). α6-subunit-containing receptors were localised primarily to the cerebellum (agreeing with previous findings concerning mRNA distribution, Laurie et al., 1992), where they were predominantly extrasynaptic (Nusser *et al.*, 1996). More recent work has detected GABA<sub>A</sub> receptor clusters ( $\beta 2/3$ and  $\gamma 2$  subunits) in rat hippocampal cells before the establishment of functional presynaptic nerve terminals (Scotti and Reuter, 2001). One study purported significant co-localisation of the  $\gamma$ 2S and  $\gamma$ 2L splice variants (Miralles *et al.*, 1994), indicative of a single subtype, when mapped by immunoprecipitation in the rat brain. The mammalian ε subunit was only recently identified (Davies et al., 1997; Whiting et al., 1997) and consequently has not undergone the scrutiny of the other receptor subunits. However, *ε*subunit mRNA (mammalian orthologue of  $\gamma$ 4) has been localised to various nuclei of the hypothalamus (e.g. in septum and preoptic areas, and paraventricular and supraoptic nuclei; Moragues et al., 2003), locus coeruleus, dorsal raphe, (Moragues et al., 2000). In situ hybridisation studies suggest the  $\varepsilon$  subunit may associate in various regions of the rat brain with  $\alpha 3$  and  $\theta$  subunits, its chromosomal partners (Moragues *et al.*, 2002). Although, no co-precipitation studies or immunohistochemical studies have validated this co-localisation at a protein level.

Immunohistochemical and immunocytochemical studies have revealed that some GABA<sub>A</sub> receptor-subunit proteins are widely distributed throughout the brain (such as

 $\alpha 1$ ,  $\beta 1$ ,  $\beta 2$ ,  $\beta 3$  and  $\gamma 2$ ), whereas other subunits show a more restricted distribution pattern (including  $\alpha 2$ ,  $\alpha 3$ ,  $\alpha 4$ ,  $\alpha 5$ ,  $\alpha 6$ ,  $\gamma 1$ ,  $\varepsilon$ ,  $\theta$  and  $\delta$ ): which concurs with data concerning mRNA distribution (refer to Sieghart and Sperk, 2002). Although immunohistochemical techniques allow more precise localisation of protein at a subcellular level within any given brain region, there are some caveats such as false positives generated by non-specific antibody binding to tissues (Rhodes and Trimmer, 2006) or false-negatives which may represent an absence of immunoreactivity (due to methodical reasons) rather than an absence of the protein of interest. To this end, immunohistochemical data should only be interpreted alongside complementary data documenting mRNA expression such as *in situ* hybridisation (Fritschy, 2008).

#### 6.1.2 Co-precipitation studies

Co-localisation does not necessarily represent co-assembly, at best it is indicative. In other efforts to determine which subunits co-localise, many studies have used immunoprecipitation techniques using GABA<sub>A</sub> receptor subunit-specific antibodies to purify native receptors containing a particular subunit from solubilised brain membrane extracts which can be subsequently quantified. In this manner subunit associations were found for  $\gamma 2$  with  $\alpha 1$ ,  $\alpha 2$ ,  $\alpha 5$  and  $\beta 2/\beta 3$  subunits in rat brain, demonstrating that potentially more than one type of  $\alpha$  subunit may be present in a single receptor complex (Benke *et al.*, 1991, 1996); this has also been shown for  $\alpha$ 6-subunit-containing receptors in rat cerebellum (Khan et al., 1994; Pöltl et al., 2003) and in co-purification of βsubunits (Li and De Blas, 1997; Jechlinger et al., 1998). However, conflicts in data exist whereby some studies have shown  $\gamma^2$  to co-precipitate with  $\gamma^3$  (Quirk *et al.*, 1994a; Benke et al., 1996) and others find no such co-purification (Mossier et al., 1994). No evidence has shown potential association between  $\gamma$ 1-subunit peptides and  $\gamma 2/\gamma 3$ subunits (Mossier *et al.*, 1994; Quirk *et al.*, 1994a). Interestingly,  $\delta$  and  $\gamma$ 2-subunits have been co-precipitated, inferring potential co-assembly (Mertens et al., 1993; Quirk et al., 1994b), which is in conflict with the theory that the  $\delta$ -subunit polypeptide effectively replaces the γ-subunit (Nusser al., 1996). More recent et immunoprecipitation studies have revealed potential major GABA<sub>A</sub> receptor subtypes existing in rat and mouse cerebellum, the majority containing  $\alpha 6$  subunits (Pöltl *et al.*, 2003), which concurs with previous data which localises  $\alpha$ 6-subunit mRNA primarily to the cerebellum (Laurie *et al.*, 1992a). Data generated from this technique can be variable, when considering that an antibody specific to a particular  $\alpha$  subunit is able to co-precipitate with the majority of the other subunits and the same is true of a  $\beta$ - or  $\gamma$ -subunit specific antibodies. This confirms the promiscuity of the subunits and thus the subsequent potential variety of GABA<sub>A</sub> receptor subtypes in the mammalian brain (Sieghart and Sperk, 2002). Due to the fact that one particular subunit may be present in several receptor subtypes (Olsen and Sieghart, 2008), immunoprecipitation data is difficult to interpret as the number of subunits co-purifying may represent an unknown number of receptor subtypes, but this type of work can certainly contribute when attempting to define native GABA<sub>A</sub> receptor subtype assembly (Olsen and Sieghart, 2008), especially if presented alongside comprehensive mRNA data.

#### 6.1.3 Generation of $GABA_A$ receptor $\gamma$ 4-subunit-specific antibodies

It had been recommended that a useful approach is to obtain convergent data, i.e. a combination of in situ hybridisation or real-time RT-PCR studies to examine the mRNA distribution, followed by immunohistochemical techniques of the same product at a protein level (Rhodes and Trimmer, 2006). In this manner, the two independent data sets can be combined and related contextually to each other, leading to less erroneous interpretation of results. This was the technique applied in this study for examination of the mRNA/protein expression of the GABA<sub>A</sub> receptor  $\gamma$ 4 subunit. Two strategies were employed for generation of antibodies specific to the GABA<sub>A</sub> receptor  $\gamma$ 4 subunit; production of a peptide antibody and production of a protein antibody. Peptide antibodies are generated directly from the amino acid sequence of a protein. Although a well chosen design enables the anti-peptide antibody to work well providing specificity and control, the primary disadvantage is that these are linear epitopes. Natural proteins are of course the ideal antigens as these are surface structural epitopes, but these are rarely pure and producing the amount of protein required can be difficult and timeconsuming.

For the production of the peptide antibody, a short region of the peptide sequence, in the non-conserved region of the cytoplasmic loop between the putative transmembrane domains 3 and 4, which was homologous between the zebra finch and the chicken was selected and the whole process was out-sourced commercially.

For the development of a protein antibody, the pMAL<sup>TM</sup> protein fusion and purification system (New England Biolabs<sup>®</sup> Inc.) was employed which enables the expression and purification of a protein from a cloned gene. In brief, a cloned gene (e.g. a region of the GABA<sub>A</sub> receptor  $\gamma$ 4 subunit) was ligated into the vector (pMAL-c2x), downstream and in the same translational reading frame as the *malE* gene (which encodes maltose binding protein). Transformation of the vector in *E.coli* cells resulted in robust expression of a fusion protein (MBP- $\gamma$ 4) which was harvested and affinity purified with an amylose column and eluted with maltose (for further details consult section 2.8, Chapter Two).

An antibody specific to the GABA<sub>A</sub> receptor  $\gamma$ 4 subunit, ensures primarily that mRNA is being translated into protein and immunohistochemical and immunocytochemical localisation of that protein within the zebra finch song system. Furthermore, in the future, the antibody could be utilised in co-precipitation studies. It would also be employed to verify if the observed increase in mRNA expression in tutor isolated birds correlated at a protein level (chapter 4); and also to verify 'silencing' of the GABA<sub>A</sub> receptor subunit mRNA (and thus protein) in the RNA interference experiments (refer to section 6.1.4).

# 6.1.4 Reverse-genetic strategies employed for assessing GABA<sub>A</sub> receptor function

Eliminating/reducing gene expression (and subsequent protein production) and observing any physiological consequences is a common approach in determining the role of a particular gene. The most extreme examples are deletions or knockout animals, where there is a complete absence of gene expression. GABA<sub>A</sub> receptor  $\alpha 1$ -,  $\alpha 2$ -,  $\alpha 5$ -,  $\beta 2$ -,  $\beta 3$ -,  $\gamma 2$ - and  $\delta$ -subunit genes have all been knocked out in mice.  $\beta 3$ - and  $\gamma 2$ -subunit knockouts produced lethal phenotypes (a major caveat of knockout studies; Culiat *et al.*, 1994; Günther *et al.*, 1995).  $\alpha 5$ - and  $\gamma 3$ -subunit knockouts resulted in altered drug responses (as do most GABA<sub>A</sub> receptor knockouts), and  $\alpha 5$ -subunit lacking mice demonstrated improved learning and memory (Culiat *et al.*, 1994; Collinson *et al.*, 2002). However, the majority of GABA<sub>A</sub> receptor knockout experiments induced compensatory up/down regulation of other subunits (another common problem with

knockout techniques). For example  $\delta$ -subunit knockouts showed increases in  $\gamma$ 2-subunit gene expression (Tretter *et al.*, 2001);  $\alpha$ 6-subunit knockouts resulted in decreased  $\delta$ subunit gene expression (Jones *et al.*, 1997);  $\beta$ 2-subunit knockouts correlated with a decrease in expression of all six  $\alpha$ -subunit genes and mice not expressing the  $\alpha$ 1subunit gene exhibited increases in  $\alpha$ 2- and  $\alpha$ 3-subunit gene expression accompanied by a decrease in expression of  $\beta$ 2-,  $\beta$ 3- and  $\gamma$ 2-subunit genes (Sur *et al.*, 2001). A less extreme method of reducing gene expression utilises antisense oligonucleotides, which elicit degradation of complementary mRNA. A plethora of studies successfully show antisense-oligonucleotide-mediated reduction in the expression of GABA<sub>A</sub> receptors, both *in vitro* and *in vivo* (reviewed by Karle, 2002). Another method is ribozyme technology, but this has only been applied to the  $\alpha$ 1 subunit of the GABA<sub>A</sub> receptor. Although it resulted in a 90% reduction in  $\alpha$ 1-subunit protein levels (Subramaniam *et al.*, 2001), this method still requires some refinement. All loss-of-function methods for assessing GABA<sub>A</sub> receptor function are discussed in Burt, 2003.

The relatively new advent of RNA interference (RNAi) enables sequence-specific silencing, without the caveats of complete gene knockout and with advantages over techniques such as antisense-DNA, dominant negative mutants and small molecular inhibitors, all of which exhibit variable efficiency (Scherer and Rossi, 2003; Mocellin and Provenzano, 2004; Cullen, 2005). Knockout animals ensure complete gene silencing, whereas RNAi is a knock-down method, which allows some translation to occur. The primary disadvantage applied to the majority of knockouts is when levels of one gene are completely diminished; other genes can be upregulated in compensation, 'masking' the phenotypic effects of the knockout. This would especially be of concern in a large gene family such as GABA<sub>A</sub> receptors. With the 90% mRNA knock-down induced by RNAi, upregulation is less of a problem and it also reduces the risk of lethal phenotypes (Dykxhoorn et al., 2003). Furthermore, it is a relatively inexpensive and rapid technique in comparison to using transgenic animals (Zeringue and Constantin-Paton, 2004). RNAi has not been widely employed to study GABA<sub>A</sub> receptors, potentially attributable, in part, to the high sequence homology between subunits and also that it is still a relatively new technique. A further caveat is that one subunit may form the basis of a variety of receptor subtypes, so subtype-specific effects are difficult to resolve. In an effort to determine the role of  $\alpha$ 4-subunit-containing GABA<sub>A</sub> receptors in the mechanisms controlling alcohol intake, a recent study by Rewal *et al.* (2009) demonstrated a ~60-75% silencing in GABA<sub>A</sub> receptor  $\alpha$ 4-subunit protein expression following *in vivo* injection of viral vectors expressing siRNAs into the nucleus accumbens of adult rats, and observed an associated behavioural response. Upon silencing of  $\alpha$ 4-subunit gene, no compensatory changes in the levels of associated mRNAs (e.g.  $\alpha$ 1,  $\beta$ 2,  $\gamma$ 2 and  $\delta$  subunits) were observed (Rewal *et al.*, 2009). *In vivo* and *in vitro* silencing of the GABA<sub>A</sub> receptor  $\gamma$ 2-subunit gene in rat brain has confirmed the importance of this subunit in synaptic clustering of GABA<sub>A</sub> receptors without creating a lethal phenotype (Li *et al.*, 2005), unlike some previous attempts (Günther *et al.*, 1995). shRNAs transfected with a lipid based reagent into primary hippocampal cultures showed up to 80% gene silencing. Other than these studies, there is no further work utilising RNAi to determine GABA<sub>A</sub> receptor function.

#### 6.1.5 RNA interference (RNAi)

This method is based on post transcriptional gene silencing (PTGS) which was originally discovered in petunias and was initially referred to as co-suppression (Napoli *et al.*, 1990; Scherer and Rossi, 2003). The phenomenon was later found in many other plants and also in fungi (where it was termed quelling; Romano and Macino, 1992; Cogoni *et al.*, 1996). It was Fire and colleagues who made the critical discovery that although the sense and antisense strands of RNA, when injected, were able to silence target mRNAs in the nematode *Caenorhabditis elegans*, it was annealed double-stranded RNA which caused the greatest silencing, twice as much as the single-stranded RNAs on their own (Fire *et al.*, 1998). Studies carried out with *Drosphilia melanogaster* (fruit fly) confirmed it was a two-step process (Elbashir *et al.*, 2001a, b). Thus, it has been established that RNAi (or PTGS and quelling) is a well conserved system within all eukaryotes. It is currently postulated that RNAi evolved from an inherent immune response within cells to protect against viral attack, transposon proliferation (Scherer and Rossi, 2003; Zamore and Haley, 2005) and is essential for maintaining genomic stability (Ying and Lin, 2004; Wienholds and Plasterk, 2005; Zamore and Haley, 2005).

#### 6.1.5.1 Mechansims of RNAi

RNAi is triggered by the presence of double-stranded (ds)RNA in the cytoplasm of the cell (Fig. 6.1), regardless of the format of the RNA (i.e. as endogenous pre-cursor

hairpin loops (miRNAs) derived from genomic transcripts in the nucleus, long dsRNA (e.g. from a virus) or synthetic short interfering (si)RNAs). The process of RNAi is summarised in Figure 6.1. In all instances, the first step is cleavage of dsRNA molecules by the dsRNA-specific endonuclease Dicer. The 5' end of the dsRNA binds at the PAZ (Piwi/Argonaute/Zwille) domain at the C-terminal of Dicer, a positively charged flat, central domain binds the remainder of the dsRNA molecule and cleavage occurs at the N-terminal where there are two catalytic ribonuclease III (RNase III) domains (Bernstein et al., 2001; MacRae et al., 2006). Different homologs of Dicer exist in different species and in C. elegans and Drosphilia species there are two Dicer enzymes, both of which require ATP for functioning (Shrey et al., 2009). However, in mammalian cells, there is only one version of Dicer (Zhang et al., 2002) and it is localised to the cytoplasm, where RNAi occurs (Billy et al., 2001). Dicer and its associated DNA-binding protein TRBP (human immunodeficiency virus transactivating response RNA-binding protein), which comprises three RNA-binding regions and protein activator of protein kinase PKR (PACT) process dsRNA into short (21-23 nt) interfering RNAs (siRNAs) which have a characteristic 2-3 nt overhangs at the 3' ends and a 5' phosphate group (Elbashir et al., 2001a, b; Kim and Rossi, 2007). Features necessary for incorporation of the siRNA into RNA-induced silencing complex (RISC), which is the second step in the RNAi pathway (Elbashir *et al.*, 2001b; Yang and Mattes, 2008). RISC is a multi-subunit complex (including helicase, exonuclease, endonuclease and homology searching domains), which is able to bind and unwind siRNA duplexes and cleave target mRNA (Sledz and Williams, 2005). The associated Argonaute 2 (AGO2) protein is responsible for cleavage activity (Chendrimada et al., 2005). RISC exists in two forms and is converted to the active form upon binding of an appropriate siRNA molecule and cleavage of the passenger (sense) strand by AGO2 (Nykanen et al., 2001; Matranga et al., 2005; Rand et al., 2005). Asymmetric binding of siRNA molecules via their first unpaired nucleotide at the 5' end allows only one strand (antisense) to become incorporated into the RISC complex (guide strand; Schwartz et al., 2006. The guide strand is selected by the associated AGO2 protein due its lower thermodynamic stability (Preall and Sontheimer, 2005); ATP is required for siRNA unwinding (Nykanen et al., 2001). The guide strand then directs the RISC-siRNA complex towards corresponding mRNA where AGO2 cleaves target mRNA, which is then completely degraded by cellular exonucleases thereby silencing gene expression (Orban and Izaurralde, 2005). The protected guide stand can then potentially enable the

RISC complex to act repeatedly on many copies of the target mRNA amplifying the silencing effect (Shankar *et al.*, 2005).

The miRNA pathway involves the processing of endogenously produced miRNA molecules encoded for within the genome (within intronic sequence of protein-coding genes or independent noncoding RNAs (Fig. 6.1b). Longer (several hundred bp), primary miRNA molecules encoded within intronic sequence are transcribed by RNA polymerase II or III in the nucleus (Kim, 2005) and then cleaved into 70nt pre-cursor miRNA molecules by nuclear RNases, Drosha and Pasha (Lee *et al.*, 2003; Denli *et al.*, 2004). These stem-loop structures are then recognised and actively exported into the cytoplasm by Exportin-5 (Lund *et al.*, 2004), where they are processed by a Dicer-like complex in the same way as siRNAs. The remainder of the pathway is similar to that described for siRNA duplexes with the exception that, in animals, mature miRNA is not completely homologous to the target mRNA sequence (as it is in plants; Pillai *et al.*, 2007) and as a result, degradation of target mRNA does not occur, but rather translation is suppressed (Olsen and Ambros, 1999). miRNAs in animals generally bind to partially complimentary sites within the 3'-UTR (Pillai, 2005) although, silencing via coding regions or the 5'-UTR is also possible (Kloosterman *et al.*, 2004).



Figure 6.1. Mechanisms of RNA interference (RNAi). (a) In non-mammalian systems, dsRNA present in the cytoplasm is detected and cleaved by the Dicer complex into shorter (21-23nt) short-interfering (si)RNA duplexes with characteristic 2nt 3'overhangs, a hydroxyl group and a 5' phosphate group. The less thermodynamically stable antisense strand binds to the RNA-induced silencing complex (RISC) complex where the duplex is unwound and the sense strand degraded by argonaute 2 (AGO2). The guide strand directs the RISC complex to complementary target mRNA to which it binds and degrades. (b) Endogenous regulation of gene expression can also occur via micro (mi)RNAs. These are generally encoded for in non-coding regions of the genome and are transcribed by RNA polymerase II or III forming primary miRNA (pri-miRNA) transcripts which are subsequently processed Drosha into precursor miRNA (pre-miRNA) which are exported to the cytoplasm by Exportin-5 (EXP-5). They are cleaved into siRNAs by the Dicer complex in a manner similar to dsRNAs. Both siRNA and miRNA pathways culminate in the binding of RISC to homologous mRNA molecules. Perfect complementarity between the guide strand and target mRNA elicits degradation (generally siRNAs) and an imperfect complement results in translational arrest (generally miRNAs); but both ultimately suppress protein synthesis. Induction of RNA silencing can be achieved by transfection of a cell with a viral or plasmid vector expressing a short hairpin (sh)RNA, which mimics endogenous miRNA molecules (Abbas-Terki et al., 2002; Kunath et al., 2003) or by direct induction of dsRNA or siRNAs into the cytoplasm (Elbashir et al., 2001a, b). siRNAs are particularly useful in mammalian cells as longer dsRNA elicits and innate immune response. Abbreviations: PACT, protein activator of protein kinase PKR; TRBP, human immunodeficiency virus transactivating response RNA-binding protein.

#### 6.1.5.2 Functions of RNAi

RNAi functions as an endogenous defence mechanism to protect the cell against viral attack and transposon activity (mobile genetic elements; Hannon, 2002). It also serves to regulate gene expression via miRNAs (Scherer and Rossi, 2003; Ying and Lin, 2004; Wienholds and Pasterk, 2005; Zamore and Haley, 2005). Up to 1/3 of genes in humans are potentially regulated by endogenous, evolutionarily conserved miRNA molecules, of which 1000 have been predicted, although precise mRNA targets and functions have only been described for a few (Pillai *et al.*, 2007). miRNA genes are located in intronic sequence and account for ~1-5% of genes in plants, invertebrates and vertebrates (Ying and Lin, 2004). They display regionally- and developmentally-regulated expression which is subject to change in disease states (reviewed by Ambros, 2004; Bartel, 2004; Wienholds and Plasterk, 2005), most notably in cancer whereby their overexpression contributes to tumour development (He *et al.*, 2005).

#### 6.1.5.3 Interferon response

In mammals, long dsRNA molecules induce the interferon response (Elbashir *et al.*, 2001a, b), characterised by a non-specific termination of transcription and translation (Williams, 1997, 1999). This pathway triggers the degradation of mRNA by inducing oligoadenylate synthase which activates the ribonuclease, RNAse L and also activates RNA-dependent protein kinase (PKR) which phosphorylates the essential translation initiation factor eIF2 $\alpha$  leading to a global inhibition of mRNA translation and global RNA degradation (Minks *et al.*, 1979; Williams, 1997, 1999). The interferon response is a host defence mechanism which was named after its ability to 'interfere' with viral replication. However, dsRNAs of less than 30nt are able to avoid induction of the interferon response and enter the RNAi pathway (Elbashir *et al.*, 2001a, b). More recent research has demonstrated that activation of the interferon response is not wholly dependent on duplex length, but can also be affected by cell type (Reynolds *et al.*, 2006) and sequences which are GU rich are also more likely to induce the interferon response (Judge *et al.*, 2005). All these criteria must be considered when constructing synthetic si/shRNAs for silencing in mammalian systems.

#### 6.1.5.4 Induction of RNAi in cells

RNAi technology is most frequently employed for the silencing of individual genes via siRNAs in cell culture (Shan, 2009). Chemically (Elbashir *et al.*, 2001a) and

enzymatically (Gou et al., 2003) synthesised siRNAs induce transient silencing (Tuschl, 2002), with normal gene expression resuming after ~3-5 cell divisions, which is thought to be attributed to dilution of the duplexes rather than their degradation (McManus and Sharp, 2002; Song et al., 2003; Genc et al., 2004). Silencing via siRNAs is 100-1000x more efficient than with antisense oligonucleotides (Bertrand et al., 2002). In efforts to establish greater longevity, plasmid vectors constitutively expressing siRNAs can be transfected into cells which can elicit silencing for up to two months (Brummelkamp et al., 2002; Sui et al., 2002; Choi et al., 2005), although their transfection efficiency is typically lower than for naked siRNAs, most notably in neuronal cultures (McManus and Sharp, 2002). To circumvent this, siRNAs can be electroporated in neuronal cells but cell death then becomes a factor (McManus and Sharp, 2002). Further to this, viral vectors were considered as they don't require transfection and they were shown to be more effective in cells which were resistant to plasmid vectors, such as primary cells (Barton and Medzhitov, 2002; Liu et al., 2004). However, these lack a selective pressure and can present a potential biohazard (reviewed by Mocellin and Provenzano, 2004). More recently research has reported that artificial dsRNAs (< 30nt in mammalian cells) and shRNAs which are subsequently processed by Dicer are far more efficient at silencing than Dicer-independent siRNAs (Kim et al., 2004; Siolas et al., 2004). Thus the use of short hairpin (sh)RNAs (which mimic naturally occurring miRNAs; Cullen, 2005) within expression/viral vectors has become popular for the establishment of stable gene silencing in culture or transgenic mice (Brummelkamp et al., 2002; Paddison et al., 2002; Yu et al., 2002; Li et al., 2005). In addition to constitutive expression of sh/siRNA duplexes, plasmids enable in vitro/vivo tracking by directly linking the si/shRNA expression cassette to a reporter gene (Dai et al., 2005; Das et al., 2006).

#### 6.1.5.5 Design of RNAi experiments

Criteria for the rational design of siRNA/shRNA molecules are continually being improved, for a full description refer to section 2.9.1, but even adhering to each rule when designing a siRNA, does not ensure successful knockdown or indeed predict the degree of silencing. Therefore it is recommended that several siRNAs are synthesised simultaneously, each targeting a different exonic (intronic regions cannot be used as it is a cytoplasmic process; Dykxhoorn *et al.*, 2003) region of the gene. Only two or three base-pair shifts in the positioning of the siRNA can alter silencing efficiency, revealing

that it is likely to be the sequence composition of the siRNA molecule which primarily governs the success of silencing (Holen *et al.*, 2002; Reynolds *et al.*, 2004). Once a successful siRNA sequence has been selected, dose response curves should be produced to ensure the lowest concentration of siRNAs are introduced to the cell to reduce any off-target effects. Although small amounts should not saturate the endogenous RNAi pathway, they are potentially competing with endogenous miRNA molecules for incorporation into RISC (Castanotto *et al.*, 2007; Shan, 2009) and expression of non-target genes can be affected by introduction of exogenous siRNAs (Jackson *et al.*, 2003; Scacheri *et al.*, 2004). Further considerations concern the half life of the target protein and mRNA (Choi *et al.*, 2005) and the G+C content of the siRNA molecule (Reynolds *et al.*, 2004).

Cells must be transfected when using siRNAs or plasmid shRNA vectors, with standard transfection reagents (generally liposomal), as they are negatively charged polymers so cannot enter hydrophobic cell membranes (Kim and Rossi, 2007; Castanotto and Rossi, 2009). Alternatively, peptide-conjugated siRNA molecules enable high transfection rates (Kim *et al.*, 2006; Davidson *et al.*, 2004; Turner *et al.*, 2007). As different cell lines can react differently to the various reagents, suitable transfection reagents should be tested on a trial and error basis. Aptamers (synthetic DNA/RNA molecules which bind to specific molecular targets) conjugated to siRNA duplexes also allow efficient cell-specific delivery (Que-Gewirth and Sullenger, 2007). Another option is electroporation (Dai *et al.*, 2005; Das *et al.*, 2006; Zietelhofer *et al.*, 2007) but as previously mentioned, cell death can be a problem when using this technique (McManus and Sharp, 2002).

Knockdown of mRNA can be measured by northern blot analysis or real-time RT-PCR, the latter proving more popular and the most appropriate technique. If mRNA silencing is successful then subsequent protein knock-down should also be analysed, most commonly by western blotting or immunofluoresence (Shan, 2009). Comparison between decreases in mRNA and protein can ascertain that all observable effects are attributable to the artificially induced RNAi mechanisms. Silencing is usually detected (in cell culture) approximately 18-24 h following transfection, but factors such as target protein turnover rate, transfection efficiency, longevity of the siRNA duplex, dilution of

187

the siRNA and cell type all affect the efficiency of the RNAi reaction (McManus and Sharp, 2002; Genc *et al.*, 2004; Mocellin and Provenzano, 2004).

#### 6.1.5.6 Applications of RNAi

RNAi has been exploited in commercial applications, such as in coffee plants, where the enzymes involved in caffeine biosynthesis were silenced to reduced caffeine content in beans by up to 70% (Ogita et al., 2003). In the molecular biology field, RNAi is utilised as a powerful tool in reverse genetic strategies in order to determine the function of eukaryotic genes; this can be considerably quicker and cheaper than knock-out animals and reduces the chance of a lethal phenotype (Dykxhoorn et al., 2003). Furthermore, consequences of silencing particular proteins within known or unknown pathways can also be explored (Devi, 2006). RNAi may also be employed as an effective therapeutic agent in diseases that involve over-expression of genes, such as cancer, influenza, HIV, SARS and Polio (Cristofaro and Ramratnam, 2006; Devi, 2006; Kumar, 2008). Successful *in vivo* gene silencing with direct injection of siRNAs and shRNAs in many different rodent tissues (such as liver, eye, kidney and lung) has been reported (reviewed by Shankar et al., 2005) and in chicken embryos (Dai et al., 2005; Das et al., 2005). Subsequently there is a profound interest in employing RNAi in the mammalian brain, but much work needs to be completed to ensure efficient, safe and specific *in vivo* delivery (reviewed by Thakker et al., 2004; Fountaine et al., 2005; Sah, 2006; Kim and Rossi, 2008; Castanotto and Rossi, 2009; Shrey et al., 2009). In vitro RNAi methods in primary neurons have proven successful using liposomal transfection reagents or electroporation (Krichevsky and Kosik, 2002; Li et al., 2005; Tönges et al., 2006). Direct stereotactic injections of naked siRNAs, or plasmid/viral vector based si/shRNAs into mammalian brain has shown ~50% decrease in expression of target genes (Makimura et al., 2002; Xia et al., 2002, 2004; Hommel et al., 2003; Van den Haute et al., 2003; Li et al., 2005; Raoul et al., 2005). The relatively short-lived effects induced by siRNAs may be beneficial in treatment of tissues such as liver or neuronal cells of the CNS due to their slow rate of division where they could silence genes for up to two weeks before becoming too diluted (Song et al., 2003; Genc et al., 2004). Moreover, in instances requiring only a short treatment, such as some viral infections, the short-term effects elicited by siRNAs may be advantageous (Shankar et al., 2005). Although, before the introduction of RNAi techniques as gene therapy, safe delivery, off target

effects and functional consequences must be thoroughly investigated (Shrey *et al.*, 2009).

#### 6.1.5.7 RNAi in zebra finch brain

RNAi was employed to functionally determine the role of y4-subunit-containing GABA<sub>A</sub> receptors in the zebra finch song system; with the ultimate aim of knocking down y4-subunit mRNA and protein in specific nuclei of the song system in vivo and observing any effects on song acquisition and production. The only previous example of RNAi technology being harnessed for functional genetic analysis in the zebra finch brain, is the silencing of FOXP2 which encodes forkhead box transcription factor (Haesler et al., 2007). Bilateral stereotactic injections of shRNAs within viral vectors were administered directly into Area X of anesthetised juvenile (PHD 23) male zebra finches. Vocalisations were recorded at PHD 65, 80 and 90-93. Suppression of translation was analysed by immunohistochemical staining and mRNA knock-down was quantified with real-time RT-PCR. Zebra finches with reduced FOXP2 expression exhibited poor imitation of a tutor and their songs contained abnormally variable syllables (Haesler et al., 2007). Concurring with previous work in humans which suggested a mutation in FOXP2 resulted in speech and language impairments (Lai et al., 2001, 2003; MacDermot et al., 2005). Due to the complexity of neural networks these effects cannot be attributed to a single gene and thus other genetic links should be investigated. As the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit gene has exhibited regional-, developmental- and learning-associated expression, both in chicken (Harvey et al., 1998) and zebra finch (Thode et al., 2008; refer to Chapter 5) and coupled with evidence that modulation of GABA<sub>A</sub> receptors affects cognitive function (Chapouthier and Venault, 2002, 2004) it was an ideal candidate for such a study. Prior to *in vivo* injections of shRNAs targeting the  $\gamma$ 4-subunit mRNA, it was first necessary to verify the silencing efficiency of the constructs by in vitro methods (Haesler et al., 2007). As zebra finch brains were scarce, and no commercial cell lines were available which endogenously expressed the  $\gamma$ 4-subunit, primary culture of chicken neurons was utilised. Due to the high degree of conservation between the  $\gamma$ 4-subunit genes of chicken and zebra finch (97% over 271 AA), shRNA constructs were designed that would target both chicken and zebra finch  $\gamma$ 4-subunit mRNAs.



**Figure 6.2**. Annealed single-stranded shRNA constructs were ligated into the p*Silencer*<sup>TM</sup> 4.1-CMV neo vector. The plasmid was transfected into cells with a lipid-based carrier where the powerful CMV (human cytomegalovirus) promoter drives constitutive expression of the shRNA *in vitro/vivo*. Resulting shRNA enters the endogenous RNAi pathway and is cleaved into siRNAs by Dicer, resulting in subsequent silencing of the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit mRNA, therefore reducing corresponding protein expression.

In summary, successful generation of a  $\gamma$ 4-specific antibody and shRNA constructs would enable a comprehensive functional analysis/confirmation of the role of the  $\gamma$ 4-subunit-containing GABA<sub>A</sub> receptors within the zebra finch song system.

## **6.2 Results**

#### 6.2.1 Generation of a γ4-subunit specific antibody

# 6.2.1.1 Amplification of a partial region of the zebra finch $GABA_A$ receptor $\gamma$ 4-subunit transcript

Following design of the DNA construct, the respective cDNA fragment was amplified by RT-PCR from zebra finch brain (Fig. 6.3) and sent for automated sequencing (Fig. 6.4) to ensure an accurate DNA sequence was going to be expressed alongside maltosebinding protein (MBP) as a fusion protein in *E. coli* cells by the pMAL<sup>™</sup>-c2X vector system.



**Figure 6.3.** A single, 120bp DNA fragment encoding a region of the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit transcript, amplified from zebra finch brain by RT-PCR. Product was electrophoresed on a 2% (w/v) agarose gel stained with ethidium bromide and visualised under UV light. RT-PCRs were run alongside 7µl Hyper ladder IV (Bioline). NTC, no template control.

```
(a)
```

```
Zebra finch (γ4 subunit)
CACCTGCCGAAACCACTGGAGCACAGCACAGGAGAGCCAGGCTGCCACC

Zebra finch (γ4 subunit)
CACCTGCCGAAACCACTGGAGCACAGCCACGGAGAGCCAGGCTGCCACC

Zebra finch (γ4 subunit)
TGCCGGTGCTCAGGTGATGCCAACCTTCACCACCATCAACATCAACCACA

Zebra finch (γ4 subunit)
TGCCGGTGCTCAGGTGATGCCAACCTTCACCACCATCAACATCAACCACA

Zebra finch (γ4 subunit)
TCATGCACTG

Zebra finch (γ4 subunit)
TCATGCACTG

Zebra finch (γ4 subunit)
TCATGCACTG

Zebra finch (γ4 construct)
KKPLEHSHRRARLPPAGAQVMPTFTTININHI

Zebra finch (γ4 subunit)
KKPLEHSHRRARLPPAGAQVMPTFTTININHI

Zebra finch (γ4 construct)
KKPLEHSHRRARLPPAGAQVMPTFTTININHI
```

**Figure 6.4.** Alignments of the cloned construct sequence with the corresponding portion of the original zebra finch GABA<sub>A</sub> receptor  $\gamma$ 4-subunit nucleotide (a) and peptide (b) sequences as a reference (GenBank accession number AM086933; Thode *et al.*, 2008), in single letter code. Sequences encode a region of the large variable intracellular loop between transmembrane domains (TM) 3-4. (a). Modifications to the construct nucleotide sequence include a translational stop codon (TAG) and *Eco*R1 restriction sites (GAATTC), highlighted in grey and orange respectively. (b) The asterix at the end of the peptide antibody sequence (highlighted in grey) indicates the presence of the integrated stop codon.

Agarose gel electrophoresis (Fig. 6.3) revealed a single amplicon of the expected size (120 bp) and completely clean negative control reaction. In order to verify the identity of the amplified product it was sent for automated sequencing (Fig. 6.4).

DNA sequencing results confirmed the RT-PCR product to be identical to the designed construct with 100% alignment to the previously documented zebra finch GABA<sub>A</sub> receptor  $\gamma$ 4-subunit mRNA sequence (GenBank accession number AM086933; Thode *et al.*, 2008), with the exception of any of the intentionally modified bases, as shown in Fig. 6.4). From this, the amino acid sequence was translated using online tools (available at; http://www.justbio.com/translator/index.php) demonstrating that no change in peptide sequence had occurred despite modifications to the DNA sequence (except for the translational stop codon).

The molecular weight of the  $\gamma$ 4-subunit peptide construct was calculated using online software (www.biopeptide.com/PepCalc/PepMWCalc2.dll/Calculate) to be 3.6 kDa.

#### 6.2.1.2 Pilot experiment

It was necessary to determine the behaviour of the MBP-fusion protein before experiments were undertaken on a larger scale. This section of work was also useful in determining the optimum conditions (e.g. inoculation volume, IPTG concentration, incubation time and sonication duration), for harvesting the greatest concentration of fusion protein from *E. coli* cells.

Data obtained for the pilot experiment demonstrated that the pMAL<sup>TM</sup>-c2X vector was expressing the  $\gamma$ 4-subunit shRNA construct and this in turn was being translated into a fusion protein within *E. coli* cells. IPTG induction was optimised and 3 h was determined as the ideal time point for harvesting the cells (as the induced band was much larger at this point than after only one hour induction time, Fig. 6.5a and b; lanes 3 and 4). Time-points were tested for up to 5 h following IPTG induction (data not shown), however, levels after 5h induction were no higher than that of 3 h induction.



**Figure 6.5**. Results of pilot experiment with 10% (w/v) SDS-PAGE (a) and western blot analysis (b) to determine activity of the MBP-fusion protein. An induced band was visible at the position corresponding to the molecular weight of the MBP-fusion protein (i.e. ~46 kDa) in each sample (5mg protein loaded per sample). The western blot (b) identifies the target protein (0.5mg protein loaded per sample) and lane 7 indicated that the protein bound successfully to amylose. **Abbreviations: UN**, uninduced cells; **IND (1h)**, cells harvested 1 h following IPTG induction; **IND (3h)**, cells harvested 3h following IPTG induction; **CE**, crude extract; **IM**, insoluble matter; **P** + **Am**, protein bound to amylose. Samples were run concurrently with 7µl pre-stained broad range protein marker (New England Biolabs<sup>®</sup> Inc.) and 15µl MBP2 marker (MW 42.5kDa).

In addition the final IPTG concentration used for induction was 0.375 mM, various concentrations were tested but this gave the greatest yield of target protein (data not shown). Crude extract and insoluble matter samples both contained the target protein (Fig. 6.5a and b, lanes 5 and 6 respectively). It was inevitable that not all the protein could be released and retained in the crude extract, so some remained detectable in the insoluble matter. In all samples, except the uninduced and protein bound to amylose, there was some evidence of protein degradation, highlighted by western blot analysis (Fig. 6.5b). The conditions of a protein extraction are not favourable to the integrity of the protein and so some loss was to be anticipated. However, only a protein that is fairly

intact is able to bind to amylose therefore it would be expected that protein purified by an amylose-affinity column would be in tact. The single product yielded in lane 7 (protein bound to amylose; Fig. 6.5a and b) indicated that this was the only protein present which would bind to amylose and the molecular weight of the band was slightly above that of the MBP2 marker (42.5 kDa; lane 8), which was expected as the MBPfusion protein molecular weight was 46 kDa. western blot analysis confirmed that this was the MBP-fusion protein with a highly specific MBP-targeting primary antibody.



**Figure 6.6**. Analysis of eluted MBP- $\gamma$ 4 fusion protein following affinity purification with an amylose column. (a) Results of the BCA assay demonstrated that protein was eluted within fractions 4-8. (b) A 10% (w/v) SDS-PAGE, stained with Coomassie brilliant blue confirmed protein elution and showed that only a single protein was eluted from a mixed population, i.e. the crude extract (CE). 10µl sample loaded per lane. (UN) unbound protein, collected once the crude extract was run through the amylose column. (c)

Western blot analysis with an alkaline-phosphatase conjugated secondary antibody verified the identity of the eluted protein; the expected MW was ~46 kDa, 1µl sample loaded per lane. Samples were run concurrently with 7µl pre-stained broad range (BR) protein marker and 15µl MBP2 marker (MW 42.5 kDa).

Results conformed successful isolation of the MBP- $\gamma$ 4-subunit fusion-protein from the crude extract. The BCA assay revealed a peak of protein elution between samples 4-8 (Fig. 6.6a). The SDS-PAGE analysis was necessary to determine the purity of eluted protein samples and the gel revealed only a single product was present (Fig. 6.6b), so the sample was verified as pure. Furthermore, western blot analysis was then performed and this confirmed the identity of the protein due to the high specificity of the primary antibody and the product being of the expected molecular weight (Fig. 6.6c).



(b)

y = 0.001x; x = y/0.001 ∴ x = 0.226µg/µl

**Figure 6.7**. Calibration graph and subsequent calculations following a BCA assay determining the final protein concentration of the MBP- $\gamma$ 4 fusion protein following dialysis. (a) Shown is a calibration curve constructed from the mean of triplicate absorbance readings from bovine serum albumin (BSA) standards in TBS. Dialysed protein was diluted 2x, 4x and 8x in TBS and added to the assay in triplicate. (b) Following a correction for the dilution factor, the concentration of MBP-fusion protein was calculated using the equation taken from the calibration curve.

There was some protein loss during the dialysis process, but the final concentration was found to be 0.23 mg/ml, which was enough protein to enable subsequent antibody production in the future.

#### **6.2.2 RNA interference (RNAi)**

In an effort to determine the function of  $\gamma$ 4-subunit-containing GABA<sub>A</sub> receptors within a learning and memory paradigm; RNAi was employed to silence  $\gamma$ 4-subunit mRNA/protein expression, with the ultimate aim of injecting shRNA constructs in vivo into specific nuclei of the zebra finch song system and observing effects on song acquisition and production. Before in vivo work could commence it was essential to verify the efficiency of the shRNA constructs in vitro. As zebra finches were not readily available, primary cerebellar neurons from 1-day-old chicken (Gallus gallus domesticus; which is phylogenetically similar to the zebra finch) were cultured. Single stranded 55 nt palindromic nucleotide sequences were synthesised commercially (full sequence details, Fig. 2.2, Chapter 2), annealed together and ligated into the pSilencer<sup>™</sup> 4.1 -CMV neo vector. Automated sequencing of both strands confirmed presence/absence of annealed inserts in the vector and revealed any inaccuracies in the sequences. shRNAcontaining plasmids ( $\gamma$ 4-subunit targeting, GAPDH targeting and non-specific targeting) were co-transfected with pRK5-Clomeleon vector (a visualisation vector to verify transfection efficiency) into primary chicken cerebellar neuronal cells and after 72 h cells were harvested. RNA was extracted and levels of y4-subunit and GAPDH (positive control) mRNA assessed by real-time RT-PCR.

#### 6.2.2.1 Optimisation of real-time RT-PCR primers

It was first necessary to verify suitable house-keeping genes for real-time RT-PCR reactions. Three house-keeping genes were tested;  $\beta$ -actin (GenBank accession no. L08165); 18S Ribosomal RNA (GenBank accession no.AF173612);  $\beta$ -2-Microglobulin (GenBank accession no. Z48922); along with primers to amplify mRNAs being targeted by shRNAs i.e. GABA<sub>A</sub> receptor  $\gamma$ 4-subunit and glyceraldehyde-3-phosphate dehydrogenase (GAPDH; GenBank accession no. NM\_204305), further details of primer pairs in Table 2.6, Chapter Two. A derivative data plot showing Sybr melt analysis indicated all primer pairs gave a single product at similar melting temperatures

and subsequent gel electrophoresis confirmed this and showed all amplicons to be between 140-155bp (data not shown).

#### **6.2.2.2 Primary culture and transfection**

Optimisiation of one-day-old chicken cerebellar neuronal culture was relatively labour intensive and the protocol outlined in section 2.9.3 was a result of lengthy optimisation concerning medium used, different plate coatings (such as collagen and poly-D-lysine), tituration technique and trypsinisation times. Cultures of neurons alongside glial cells were produced which could be maintained for up to 12 d. Transfection of plasmid-based shRNAs was carried out in triplicate ~5 days after the culture was established, in the following combinations:

(1) 200ng  $\gamma$ 4-subunit-targeting shRNA in p*Silencer*<sup>TM</sup> 4.1 CMV-neo vector + 200ng pRK5-Clomeleon vector

(2) 200ng GAPDH-targeting shRNA in p*Silencer*<sup>TM</sup> 4.1 CMV-neo vector + 200ng pRK5-Clomeleon vector

(3) 200ng Non-targeting shRNA in p*Silencer*<sup>TM</sup> 4.1 CMV-neo vector + 200ng pRK5-Clomeleon vector

(4) 200ng pRK5-Clomeleon vector alone

#### 6.2.2.3 Quantitative real-time RT-PCR

Preliminary real-time RT-PCR data shown in Figure 6.8 revealed that both the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit and GAPDH shRNAs resulted in sequence-specific silencing of the respective mRNAs. Agarose gel electrophoresis indicated a substantial decrease in  $\gamma$ 4-subunit mRNA expression in cells transfected with a  $\gamma$ 4-subunit targeting shRNA, 72h following transfection. No such decease in GABA<sub>A</sub> receptor  $\gamma$ 4-subunit mRNA was observed in cells transfected with GAPDH-targeting or non-targeting shRNAs (Fig. 6.8b and c). Although, some product was still observable (Fig. 6.8a, lanes 2-4), so silencing was not 100% (this was expected). Further confirmation of sequence-specific gene silencing was seen in cells transfected with a shRNA targeting GAPDH (Fig. 6.8b). Corresponding mRNA was noticeably decreased (lanes 5-7) whereas the levels of  $\gamma$ 4-subunit mRNA remained stable (lanes 2-4).



**Figure 6.8**. Silencing of GABA<sub>A</sub> receptor  $\gamma$ 4-subunit and GAPDH mRNAs in primary chicken cerebellar neurons. Cells were harvested 72 h following transfection with plasmid-based shRNAs and cellular mRNA levels assessed by real-time RT-PCR. (a) GABA<sub>A</sub> receptor  $\gamma$ 4-subunit mRNA levels in neurons were considerably reduced after transfection with the corresponding shRNA, whereas levels of GAPDH remained relatively unaffected, along with mRNA expression of the three housekeeping genes (18s rRNA,  $\beta$ 2-microglobulin and  $\beta$ -actin). (b) GAPDH mRNA levels were decreased in cells in response to transfection with a sequence-specific shRNA, but levels of other mRNAs remained unaffected. (c) Neither  $\gamma$ 4-subunit or GAPDH mRNA levels were noticeably affected by transfection with a non-targeting shRNA or (d) pRK5-Clomeleon vector alone. (n = 2). No template controls (for validating RT-PCR reactions) were negative (data not shown).

Expression of neither of the target mRNAs ( $\gamma$ 4-subunit and GAPDH) were greatly affected by the transfection of a non-targeting shRNA, nor with the pRK5-Clomeleon vector alone, confirming that transfection was not affecting mRNA expression. Real-time RT-PCRs also illustrated consistent expression of the housekeeping genes under all different experimental conditions and validated the real-time RT-PCRs themselves. Due

to time restraints and limitations of resources, the *in vitro* RNAi experiment was only completed twice in its entirety and was not fully quantified, but nonetheless it provided a good indication as to the future success of this work.

### **6.3 Discussion**

It is well documented that to understand true gene expression and thus the phenotype of a cell, mRNA and protein data is required; if either of these are overlooked it can lead to erroneous conclusions (Hatzimanikatis et al., 1999; Hatzimanikatis and Lee, 1999; Greenbaum et al., 2003; Yu et al., 2007). It may be assumed that for the most part, mRNA is faithfully translated into protein. However, within the zebra finch song system, some dissociation has been revealed. ZENK (also known as Zif268, Egr-1, NGFI-A, Krox-24 and tis8; Whitney et al., 2000) is an immediate early gene, and corresponding mRNA expression has been well studied within the zebra finch song system (reviewed by Johnson and Whitney, 2005). Within HVC and RA, ZENK mRNA is well expressed, increasing as song development progresses. In HVC, ZENK protein levels increase in correlation with mRNA levels (indicating faithful translation of mRNA into protein. However, in RA, despite increases in mRNA levels, there are no increases in ZENK protein until the birds begins singing plastic song. This is indicative of posttranscriptional regulatory mechanisms and the authors speculate that this is behaviourally, and not developmentally linked (Whitney et al., 2000). Previous work has shown that  $GABA_A$  receptor  $\gamma$ 4-subunit gene expression is down-regulated in response to learning of song (refer to Chapter Five). Akin to ZENK, this perhaps points towards behaviourally-linked regulation of gene expression, although with the  $\gamma$ 4subunit this appears to be transcriptional regulation; but it is necessary to determine if this correlation appears at a protein level. Certainly a decrease in mRNA levels would inevitably lead to a decline in protein levels as the available template for translation is reduced; however, it must be ascertained if the decrease in mRNA levels is due to decreased rate of transcription, increased protein synthesis or increased mRNA degradation.

With regards to determining native  $GABA_A$  receptor stoichiometry, evidence for true subunit association must firstly be established by co-localisation of subunit mRNAs and corresponding proteins in tissues/cells. Subcellular distribution should then be investigated (as a single neuron can express several different  $GABA_A$  receptor

subtypes), followed by a demonstration of subunit interactions e.g. by co-precipitation studies (Olsen and Sieghart, 2008). In order for all of these criteria to be fulfilled, it was imperative to have a  $\gamma$ 4-subunit-selective antibody. However, no such antibody was commercially available, so one had to be synthesised. Generation of a specific antibody would allow analysis as mentioned above, but would also be of use in the RNAi work to enable validation of  $\gamma$ 4-subunit protein silencing.

#### **6.3.1** γ4-subunit-specific antibody generation

The large number of subunits forming the GABA<sub>A</sub> receptor subunit gene family and the high similarity that exists within classes of subunits (>70%) and even between classes  $(\sim 30\%)$  makes generation of highly specific antibodies a difficult task. The majority of antibodies directed against the GABA<sub>A</sub> receptor subunits are polyclonal. This can be problematic as the composition of individual polyclonal antibodies is variable with each donor animal (Olsen and Sieghart, 2008). However, a paucity of well characterised and highly selective monoclonal antibodies has meant that the majority of data is obtained by use of the polyclonal variety. The strategy undertaken herein was to synthesis a protein antibody (specific to the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit) by use of a fusion protein system. This enabled the expression and purification of a protein from a cloned gene. Results showed the  $\gamma$ 4-subunit-MBP fusion protein had a molecular mass of 46 kDa when analysed by SDS-PAGE which correlated with the predicted molecular mass  $(MBP = 42.5 \text{ kDa} + \gamma 4$ -subunit peptide 3.63 kDa). Subsequent immunoblotting with an anti-MBP antibody confirmed the product as the MBP-fusion protein (Fig. 6.5a and b). Following affinity purification with an amylose column, SDS-PAGE and western blotting revealed a single product was purified of the appropriate molecular weight which was identifiable as the MBP-fusion protein with use of the specific anti-MBP antibody (Fig. 6.6b and c). There were no contaminants or evidence of degradation found within the eluted samples (which can sometimes be a problem with this system; de Pieri et al., 2004), which were all pooled (those fractions containing eluted fusion protein) and concentrated. The next stage of the work would be to cleave the fusion protein by hydrolysis of the integrated cleavage site and then the protein would be ready for commercial synthesis into a protein antibody. This is a commonly used technique for the synthesis of protein antibodies and has proven successful in previous studies (Kang et al., 2005; de Pieri et al., 2004). Several GABA<sub>A</sub> receptor subunit-specific antibodies have been generated using this method including those targeting  $\beta$ 3 (Todd *et al.*, 1996),  $\gamma$ 1 (Mossier *et al.*, 1994) and  $\gamma$ 3 (Tögel *et al.*, 1994).

#### 6.3.2 Silencing of the GABA<sub>A</sub> receptor γ4-subunit gene

The wide heterogeneity and relative promiscuity of GABA<sub>A</sub> receptor subunits culminate in a plethora of potential subtypes, each with a slightly different electrophysiological, biochemical and pharmacological profile. Coupled with their differential expression patterns throughout the brain, it suggests they have evolved to fulfil specific physiological functions (Wafford et al., 2004; Olsen and Sieghart, 2008). As aforementioned, various reverse-genetic strategies have been employed in attempts to identify the roles of particular subunits/subtypes (reviewed by Burt, 2003). Work herein has attempted to elucidate the role of  $\gamma$ 4-subunit-containing GABA<sub>A</sub> receptors in song acquisition and production in the male zebra finch, which is a well established learning and memory paradigm. The advent of RNAi technology allows for effective, inexpensive and rapid analysis of gene function. The ultimate aim of the RNAi work was to utilise in vivo bilateral injections of GABAA receptor y4-subunit-targeting shRNAs directly into specific nuclei of the zebra finch song system, silence  $\gamma$ 4-subunit mRNA/protein expression, and observe subsequent effects on song acquisition and production. An experiment of this nature has only been carried out once previously whereby FOXP2 (which encodes the forkhead box transcription factor) was silenced in vivo by direct bilateral injections of lentivirus mediated shRNA into Area X of zebra finch. Silencing persisted throughout the entire song learning phase (PHD 23-~90) and reduced mRNA and protein levels resulted in birds which less accurately mimicked their tutor's song and produced song with abnormally variable syllables (Haesler et al., 2007). Akin to the  $\gamma$ 4-subunit, previous work with the zebra finch had shown *FOXP2* to be of potential importance during song-learning (Haesler et al., 2004) and mutations in the human homologue of the gene resulted in language impairments (Lai et al., 2001; 2003; MacDermot, 2005), although no such work has been carried out with the mammalian orthologue of  $\gamma 4$ ,  $\epsilon$ . In chicken, a learning-associated down-regulation of  $\gamma$ 4-subunit mRNA expression was observed during imprinting in learning-relevant regions of the brain (Harvey et al., 1998). There is high sequence conservation between zebra finch and chicken y4-subunits (97% over 271 AA). Corresponding mRNA is robustly expressed in all nuclei of the zebra finch song system (Thode et al., 2008), so it can be assumed their respective functions are highly conserved. Accordingly, a similar learning-associated down-regulation of  $\gamma$ 4-subunit mRNA was evident in the zebra finch (refer to Chapter five), this gene was therefore an ideal candidate for investigation with RNAi. If the role of  $\gamma$ 4-subunit-containing GABA<sub>A</sub> receptors in song production in zebra finch could be confirmed, data could be extrapolated to humans which have the  $\varepsilon$  subunit, the orthologue of avian  $\gamma$ 4 (Darlison *et al.*, 2005).

Although the degree of silencing was not fully quantified, real-time RT-PCRs indicated a decrease in levels of  $\gamma$ 4-subunit mRNA in cells transfected with the corresponding shRNA. This decrease in expression was not observed in cells transfected with a GAPDH-targeting shRNA, non-targeting shRNA or with the pRK5-clomeleon vector alone. However, silencing was not absolute; some  $\gamma$ 4-subunit cDNA was amplified in the real-time RT-PCR reactions. This is to be expected, as RNAi is not a complete knockout technique, but a knockdown technique, where 75-90% gene silencing may be considered average. However, this is considered advantageous as it reduces the chances of inducing a lethal phenotype and compensatory upregulation of other genes (Dykxhoorn et al., 2003). Another factor impeding the degree of silencing is the transfection efficiency. Primary cell cultures more accurately represent the biology of normal cells but they tend to be far more sensitive to chemical or physical stress and so are notoriously difficult to transfect (Zeitelhofer et al., 2007), where often transfection efficiency can be around 20% (McManus and Sharp, 2002). This is important in RNAi experiments as proteins synthesised by untransfected cells will compromise analysis (Zeitelhofer et al., 2007). In efforts to circumvent this in other studies, electroporation was introduced. This method uses an electrical field which transiently creates pores in cell membranes allowing DNA and RNA molecules to pass directly into the cell. Although this can increase transfection efficiency up to 95%, it also promotes cell death (~50%; McManus and Sharp, 2002). Despite this, electroporation has been widely employed in vitro (Guigetnet and Mayer, 2008; Zeitelhofer et al., 2007) and in vivo shRNA-mediated silencing via electroporation of chicken embryos has documented silencing efficiencies of up to 90% (Pekarik et al., 2003; Dai et al., 2005; Das et al., 2006). Nevertheless, successful transfection with more traditional lipid-based carriers has also been achieved (Dalby et al., 2004; Tönges et al., 2006), however these can prove toxic when utilised long-term (Davidson et al., 2004). In short term in vitro

experiments such as these, where longevity of the culture is not necessary, lipid-based transfection reagents can be used for induction of RNAi (Krichevsky and Kosik, 2002). Therefore, continuation of this work would involve testing different transfection strategies to improve efficiency. It is also important to complete a titration of shRNA constructs to ensure the lowest concentration is used to elicit silencing; too high concentration of shRNAs can result in off target effects and potentially saturate siRNA machinery (Castanotto et al., 2007; Shan, 2009). Also of consideration is that introducing artificial si/shRNAs into cells may cause competition with endogenous miRNAs which are involved in regulation of cellular gene expression (Shan, 2009). Therefore, some off-target effects may be attributable to interference with the miRNA pathway, so this needs to be verified in future work. Furthermore, before commencing *in vivo* work, it is essential to verify corresponding protein is being silenced. In this regard, the turnover rate of the protein should be ascertained so that residual proteins are not masking the effects of the shRNAs (Choi et al., 2005; Zeitelhofer et al., 2007). The half-life of GABA<sub>A</sub> receptors in chicken neuronal culture is ~25 h and subunit-mRNA ~9 h (Lyons et al., 2000). A computer-predicted miRNA precursor has been identified which encodes the GABA<sub>A</sub> receptor  $\varepsilon$  subunit (the mammalian orthologue of  $\gamma$ 4; Dostie et al., 2003). miRNA precursors are highly conserved between fairly divergent species (Dostie et al., 2003) so it is likely that a miRNA precursor may be present to regulate  $\gamma$ 4-subunit gene expression. Taking this into consideration, the levels of  $\gamma$ 4-subunit protein must be carefully quantified across all experimental conditions to ensure that any knockdown is mediated by the synthetic shRNA and not endogenous miRNA molecules.

Silencing of a constitutively expressed gene (at the same dosage as shRNA targeting the gene of interest) indicates successful RNAi-mediated knockdown (Shan, 2009). GAPDH mRNA was successfully silenced (with no observable effects on  $\gamma$ 4-subunit mRNA levels), indicating that the mRNA knockdown was due to induction of RNAi via shRNAs and not endogenous regulatory mechanisms. Thus it can be assumed that the apparent decrease in  $\gamma$ 4-subunit mRNA was also attributable to the shRNAs, although these findings need to be further verified. It was also paramount to include negative controls. In previous work it was suggested that a scrambled control sequence or a sh/siRNA with a single base-pair mismatch was suitable. However, more recent
discoveries involving the miRNA pathway (refer to section 6.1.5.1) reveal that even mis-matching duplexes can cause silencing, or off-target effects (Alvarez *et al.*, 2006; Birmingham *et al.*, 2006). In neurons, inducing RNAi causes competition with the endogenous miRNA pathway and results in phenotypic effects such as a decrease in length and branching of dendritic spines (Alvarez *et al.*, 2006). This is primarily of importance in long-term silencing, but would need to be considered if work is taken to an *in vivo* level in zebra finch. Thus, it is recommended that a non-targeting sequence be employed (Anon, 2003; Shan, 2009). Only seven consecutive complementary bases are necessary to induce protein silencing via translational arrest (Jackson and Linsley, 2004), so a non-targeting construct requires careful selection. The negative control shRNA used here has proven successful in previous RNAi studies in chicken (Rumi *et al.*, 2006) and results herein showed no observable effects on levels of GAPDH or  $\gamma$ 4subunit mRNA. Transfection of pRK5-Clomeleon vector alone ensured no transfectionspecific problems were arising, as this does not utilise the RNAi machinery.

Although the results here were only preliminary (n = 2), and not fully quantified, they indicate that the synthesised shRNA duplexes may efficiently silence the GABAA receptor  $\gamma$ 4-subunit and GAPDH genes. This paves the way for this work to be taken to an in vivo level, which will clearly identify the role that GABAA receptors comprising this subunit may play in learning and memory. The only example to date of RNAi being used to investigate molecular mechanisms of learning and memory is in nymphal crickets (Gryllus bimaculatus) where long-term potentiation (LTP) was disrupted by silencing of the NOS (nitric oxide synthase) gene (Takahashi et al., 2009). It may also be insightful to carry out this work in chicken in imprinting experiments as previous work has indicated a role for  $\gamma$ 4-subunit-containing receptors in this paradigm (Harvey et al., 1998). If y4-subunit mRNA/protein silencing experiments are successful in vivo in zebra finch (with the established protocol), it would be possible to replicate the work with other  $GABA_A$  receptor subunit genes, which could potentially identify a particular subtype which is involved in learning in and memory in zebra finch. This may be subsequently be applied to humans to aid in the pursuit of understanding learning and memory processes and treating pathologies affecting these mechanisms via subtypespecific rational drug design.

### 7. DISCUSSION

The heterogeneity of the GABA<sub>A</sub> receptor family is both its redeeming feature and greatest encumbrance. GABA<sub>A</sub> receptors mediate the majority of inhibitory neurotransmission within the brain and the plethora of available subunits enables temporal- and regional-specific expression of many different subtypes (Laurie *et al.*, 1992a, b; Wisden *et al.*, 1992; Sperk *et al.*, 1997; Kultas-Ilinsky *et al.*, 1998; Huntsman *et al.*, 1999; Pirker *et al.*, 2000; Pöltl *et al.*, 2003). Each subtype has an individual pharmacological and electrophysiological profile (due to the combination of subunits which compose each receptor), which potentially fulfil different physiological functions (Wafford *et al.*, 2004; Olsen and Sieghart, 2008). However, the extreme promiscuity of the subunits causes problems in elucidating defined native GABA<sub>A</sub> receptor subtypes; this is frequently observed in co-precipitation and expression of GABA<sub>A</sub> receptor subunits within a novel paradigm, the zebra finch song system, and determine the role GABA<sub>A</sub> receptor-mediated neurotransmission may play in learning and memory processes.

# 7.1 Isolation of partial cDNAs encoding GABA<sub>A</sub> and GABA<sub>C</sub> receptor subunits

Isolation of partial cDNAs encoding zebra finch GABA<sub>A</sub> and GABA<sub>C</sub> receptor subunits yielded several key findings. Peptide alignments with corresponding chicken and rat sequences demonstrated high sequence identity (>90% and >80% similarity respectively), confirming their identity and indicative of a strong conservation of function. The work has highlighted several errors in the annotation of predicted zebra finch and chicken GABA<sub>A</sub> receptor subunit sequences available online, and confirmed the need to experimentally determine the sequence of cDNAs and not just rely on *in silico* evidence, generated from algorithms (section 3.4). Nonetheless all isolated sequences shared ~95% similarity with predicted zebra finch GABA<sub>A</sub> receptor subunit sequences, released following sequencing of the whole genome. Isolation of  $\pi$ -subunit partial cDNA heralded various interesting points. Firstly, this cDNA was isolated from brain, where levels appeared to be fairly robust (Fig. 4.7, Chapter Four). In contradiction to much previous work which localises  $\pi$ -subunit mRNA to the periphery such as lung (Jin *et al.*, 2005, 2006, Xiang *et al.*, 2007) and reproductive tissues

(Hedblom and Kirkness, 1997; Fujii and Mellon, 2001), and it has shown promise as a molecular marker for tumours (Backus et al., 2005; Johnson and Haun, 2005; Symmans et al., 2005). However, there is one example of  $\pi$ -subunit gene expression in mammalian brain. A learning-associated decrease in  $\pi$ -subunit gene expression was observed in rat hippocampus following training in the Morris water-maze (Cavallaro et al., 2002). Other than this, there is no evidence of  $\pi$ -subunit gene expression in the brain, so the observed expression in zebra finch brain was an unexpected finding. Not only this, but in zebra finch, a novel  $\pi$ -subunit splice variant was identified which contained a 37AA deletion within exon 7 which corresponds to a region of sequence encoding part of TM2 and TM3, and the extracellular loop lying between them (Figs. 3.3 and 3.4). This is the first report of a  $\pi$ -subunit splice variant in any species and mRNA encoding both the long and short forms were present in zebra finch brain. Splice variants add a further layer of complexity to the GABAA receptor family, but are of importance as they have been implicated in disease states (Huntsman et al., 1998; Dredge et al., 2001; Volk et al., 2002; Zhao et al., 2009) and potentially play a role in trafficking and receptor assembly (Mu et al., 2002). Further investigation pertaining to spatial expression of  $\pi$ -subunit mRNA in zebra finch brain is warranted. There is much lower sequence identity between zebra finch and human  $\pi$ -subunit sequences (~80%) over 275 AA), compared to other GABA<sub>A</sub> receptor subunit sequences. Consequently, the functions may be slightly different between mammals and lower vertebrates. In zebra finch,  $\pi$ -subunit-containing receptors may play a more important role in the brain, hence the more prominent expression and the existence of a splice variant.

# 7.2 Spatial distribution of GABA<sub>A</sub> receptor subunit mRNAs in song system

Involvement of GABA<sub>A</sub> receptors in learning and memory processes (i.e. the song system) has been previously studied in zebra finch by pharmacological and electrophysiological techniques (Livinston and Mooney, 1997; Bottjer *et al.*, 1998; Dutar *et al.*, 1998; Luo and Perkel, 1999; Spiro *et al.*, 1999; Vicario and Raksin, 2000; Farries *et al.*, 2005; Prather *et al.*, 2008). However, this is the first study to qualitatively and quantitatively map mRNA expression of an entire gene family within the brain of any species. The zebra finch is a well established learning and memory model and the neural substrate of song learning, the song system, has been extensively characterised

(Nottebohm *et al.*, 1976). Prior to this work, only GABA<sub>A</sub> receptor  $\gamma$ 4-subunit mRNA had been spatially mapped within the song system and findings showed it to be extremely well expressed in nearly all relevant nuclei (Thode *et al.*, 2008).

Expression studies of this nature are notoriously difficult to interpret owing to ubiquitous and overlapping mRNA expression patterns. Although they give a good insight into potential co-expression of GABA<sub>A</sub> receptor subunit mRNAs, the limited resolution of in situ hybridisation cannot confirm co-assembly. As many GABAA receptor subtypes can be present on a single neuron, immunocytochemistry at a subcellular level may give a clearer idea of which individual subunits are assembling. However, even this technique is not fully conclusive as different receptor species may be adjacent to each other at the membrane. It is also difficult to use in vivo electrophysiological methods due to the presence of multiple receptor assemblies on a single neuron or neuronal population. Single-cell RT-PCR may offer a clearer picture regarding cell-specific expression, but the general consensus is that a variety of complementary techniques (including *in situ* hybridisation and real-time RT-PCR) should be applied which can be contextually related in pursuit of native stoichiometry (Olsen and Sieghart, 2008, 2009). This study successfully demonstrates the wide heterogeneity of GABA<sub>A</sub> receptor subunit gene expression within the song system, with each subunit demonstrating a unique expression profile throughout the zebra finch brain. Detection of all subunit mRNAs in each nucleus confirms the documented promiscuity of this family and highlights the complexity of subtype assembly. Moreover, the work provided a platform for further and more specific investigations into GABA<sub>A</sub> receptors and the song system. The most striking expression profile was that exhibited by the  $\gamma$ 4-subunit gene. No other GABA<sub>A</sub> receptor subunit gene was so strongly expressed in all nuclei (LMAN, Area X, HVC and RA), inferring a functional role for receptors comprising this subunit in the acquisition and production of song.

Electrophysiological studies have confirmed that the recombinant  $\alpha 3\beta 2\gamma 4$  receptor is fully functional (Forster *et al.*, 2001). However, these forced combinations are not necessarily representative of native GABA<sub>A</sub> receptor stoichiometry *in vivo*. When considering the spatial distribution of  $\alpha 3$ -,  $\beta 2$ - and  $\gamma 4$ -subunit mRNAs within the song system nuclei,  $\alpha 3$ - and  $\gamma 4$ -subunit genes were well expressed in RA and Area X but the

 $\beta$ 2-subunit gene was not specifically expressed within any nuclei of the song system (Figs. 4.3 and 4.4, Chapter Four). It may be assumed that the most likely native stoichiometry would be  $\alpha 3\beta 4\gamma 4$  as these genes form a cluster on chromosome 4A of the zebra finch genome (Fig. 3.8, Chapter Three). The same cluster is located to chromosome 4 of the chicken genome and the orthologous genes ( $\alpha 3\theta \epsilon$ ) are located in an identical cluster on the X chromosome of the human genome. The order and transcriptional orientation of the genes is conserved between all three species, thus coordinated expression could be anticipated.  $\beta$ 4-subunit gene expression appeared more robust in song-system nuclei when compared to the  $\beta$ 2-subunit gene, most notably in HVC and Area X, so potential for co-assembly into an  $\alpha 3\beta 4\gamma 4$  subtype exists. Interestingly, similar studies in chicken brain have revealed quite distinct distribution patterns of mRNA encoding  $\beta4$  and  $\gamma4$  subunits, suggesting they do not assemble into a single receptor complex (Darlison et al., 2005). It was purported some time ago that native GABA<sub>A</sub> receptors may comprise of their chromosomal partners (Barnard et al., 1998), however there are numerous exceptions to this rule, suggesting chromosomal partnering is not a prerequisite for native receptor assembly (Olsen and Sieghart, 2008). In the case of  $\alpha 1$ -,  $\beta 2$ - and  $\gamma 2$ -subunit genes (which are clustered on chromosome 5 in humans and 6 in zebra finches), expression in zebra finch brain was modest and diffuse, with no discernible expression in any of the song system nuclei (Figs. 4.1, 4.3 and 4.5, Chapter Four). As this is the most prevalent subtype in the mammalian brain (Whiting 2003a), the function may be conserved in zebra finch brain (especially as the sequences are so well conserved). The diffuse distribution implicates a generalised function (as in mammals), in contrast to the highly discrete and specific expression of  $\gamma$ 4-subunit gene which suggests a more specialised function, perhaps exclusive to the song system. The unique expression of the  $\gamma$ 4-subunit gene may point towards homomeric assemblies within song-system nuclei, but this is highly unlikely as the mammalian orthologue,  $\varepsilon$  is unable to form functional homomers (Jones and Henderson, 2007), and generally homomeric assemblies (of any subunits) are retained within the endoplasmic reticulum for degradation (Kittler et al., 2002; Jacob et al., 2008). Thus it can be postulated that instead,  $\gamma$ 4-subunits form the basis of numerous GABA<sub>A</sub> receptor subtypes, within the nuclei of the zebra finch song system, perhaps purveying a unique physiological property not conferred by  $\gamma$ 1- or  $\gamma$ 2-subunit-containing receptors.

208

## 7.3 Potential association of $GABA_C$ and $GABA_A$ receptor subunits

No previous work has considered GABA<sub>C</sub> receptors in the song system. There is considerable debate pertaining to the classification of GABA<sub>C</sub> receptors, with many groups believing they should be grouped as a subdivision of GABA<sub>A</sub> receptors (Olsen and Sieghart, 2008, 2009; Collingridge et al., 2009), due to ~35% sequence similarity with GABA<sub>A</sub> receptor subunits. However, this would assume that they could coassemble with GABA<sub>A</sub> receptor subunits to form functional receptors, which there is some evidence of in vitro (Ekema et al., 2002; Milligan et al., 2004; Pan and Qian, 2005) but not in vivo. Data obtained from recombinant expression studies should be acknowledged with some caution as the polypeptides are often overexpressed, which may cause forced assemblies which are not apparent *in vivo* (Olsen and Sieghart, 2008). Moreover, it must be considered that the properties of native receptors may differ from recombinant receptors due to endogenous phosphorylation or interactions with other endogenous cellular or synaptic proteins, especially as the majority of studies are not carried out in neuronal cultures (Wanamaker et al., 2003; Everitt et al., 2004). Thus, the forced assembly of GABA<sub>A</sub> and GABA<sub>C</sub> receptor subunits in recombinant studies may lead to spurious assumptions. There are no gene expression studies which have comprehensively mapped all the GABA<sub>C</sub> receptor subunits ( $\rho 1-\rho 3$ ) in parallel to GABA<sub>A</sub> receptor subunits (as this would be the first step in determining if there was coexpression and thus the potential for assembly; Olsen and Sieghart, 2008). Data obtained from this study gave little indication of co-expression of GABA<sub>A</sub> and GABA<sub>C</sub> receptor subunit genes within the song system of the zebra finch. p1-p3 subunits were barely detectable by *in situ* hybridisation in any of the song-system nuclei (Fig. 4.8, Chapter 4). However, all three GABA<sub>C</sub> receptor-subunit genes (with particular reference to  $\rho$ 1) were well expressed within cerebellum (Fig. 4.8, Chapter Four) where they may potentially associate with GABA<sub>A</sub> receptor subunits. Although, there are many different cell types within the cerebellum, and so it is equally likely that GABA<sub>C</sub> receptor subunits may be present in completely separate neuronal populations from those harbouring GABA<sub>A</sub> receptors, forming distinct receptor species. Techniques such as subcellular immunocytochemistry (which has a higher resolution than in situ hybridisation), co-precipitation studies and single-cell RT-PCR would be appropriate for further investigations into heteromerisation within cerebellum. But within the

209

parameters of this study there was no appreciable association between  $GABA_A$  and  $GABA_C$  receptor subunits in any nuclei of the zebra finch song system.

Surface expression of GABA<sub>A</sub> receptors (which are constantly being recycled between the synapse and intracellular endocytic locations, either for recycling or degradation; Bedford *et al.*, 2001) should be confirmed, via techniques such as receptorautoradiography. GABA<sub>A</sub> receptors only assemble at an efficiency of ~25%, (Gorrie *et al.*, 1997), so it would be of paramount importance to determine which assemblies are actually expressed at the membrane, determining the phenotype of the cell. mRNA levels are certainly indicative of a trend, but ideally should be complemented with protein data to avoid erroneous interpretation of results (Greenbaum *et al.*, 2003). Nonetheless, data obtained provides interesting insights and implicates GABA<sub>A</sub> receptors, most crucially those comprising the  $\gamma$ 4-subunit in learning and memory systems in zebra finch.

#### **7.4 Importance of the GABA**<sub>A</sub> receptor $\gamma$ **4-subunit**

The GABA<sub>A</sub> receptor  $\gamma$ 4-subunit cDNA was first isolated from the chicken (Gallus gallus domesticus), displaying greatest identity to rat  $\gamma 1$ ,  $\gamma 2$  and  $\gamma 3$  subunits (67%, 69%) and 70% respectively; Harvey et al., 1993). Following this, the mammalian  $\varepsilon$  subunit was identified which exhibited highest similarity to the  $\gamma$ 4-subunit (49%). This degree of similarity is relatively low (when other receptor subunits are compared across species identity is often >90%), however, there is greater evolutionary diversity when considering the  $\varepsilon$  subunit as rat and human  $\varepsilon$  subunit sequences only share 68% identity (Sinkkonen *et al.*, 2000). Unlike  $\varepsilon$ , the  $\gamma$ 4-subunit shares considerably higher interspecies sequence identity (e.g. 97% over 271 AA between zebra finch and chicken), indicating a strong conservation of function in lower vertebrates. Within the embryonic chicken brain,  $\gamma$ 4-subunit mRNA is well distributed.  $\gamma$ 4-subunit transcript is detected at embryonic day (E) 13 and levels continue to increase during embryogenesis (Harvey et al., 1993); correlating with expression in the brainstem which is not detected until E8, some time after many other GABAA receptor subunit mRNAs are detected (Enomoto *et al.*, 2001). Strong expression of the  $\gamma$ 4-subunit gene within one-day-old chicken brain is apparent in areas essential for visual processing and imprinting, including optic tectum, nucleus rotundus and entopallium (Harvey et al., 1998). Subsequent behavioural studies demonstrated a learning-associated down-regulation of  $\gamma$ 4-subunit mRNA 10 h following imprinting in learning-relevant regions of the chicken brain including intermediate medial mesopallium (a sensory sorting centre), entopallium (visual projection area) and posterior medial nidopallium. Such decreases were not observed for y2-subunit mRNA (Harvey et al., 1998), indicative of a role for y4-subunitcontaining GABAA receptors in learning and memory processes. Following this, the y4subunit gene was found to be robustly expressed in almost all the nuclei of the zebra finch song system (Thode et al., 2008, Fig. 4.5, Chapter Four). Interestingly  $\gamma$ 4-subunit mRNA was only detectable in RA in birds at PHD 35 or older, which coincides with innervation of RA by projections from HVC and establishment of the VMP pathway (Akutagawa and Konishi, 1985; Mooney and Rao, 1994; Scott and Lois, 2007), which strongly implicated a role for receptors comprising this subunit in song production (Thode et al., 2008). Data obtained in the current study has further confirmed the relative importance of  $\gamma$ 4-subunit containing GABA<sub>A</sub> receptors in learning and memory. Corresponding mRNA was robustly distributed throughout the song system, where expression levels appeared to be subject to temporal regulation. Robust distribution and coordinated peaks of gene expression at specific developmental time points during song learning, in the relevant nuclei is indicative of a functional role in acquisition and production of song (Fig. 4.10, Chapter Four). A similar situation is observed in mammalian hippocampus (a brain region associated with spatial learning; Wanatabe and Bischof, 2004), where there is robust expression of  $\alpha$ 5-subunit receptors. Subsequent knockout of this gene resulted in improved performance in hippocampal-dependent spatial learning tasks (Collinson et al., 2002). Akin to this, there was a significant learning-associated down-regulation of  $\gamma$ 4-subunit mRNA levels in four nuclei of the song system in both the AFP (LMAN and Area X) and VMP (HVC and RA, Fig, 5.12, Chapter Five). This again exemplifies that a down-regulation of neuronal inhibition is associated with improved cognitive function (Harvey et al., 1998; Chapouthier and Venault, 2002, 2004; Collinson et al., 2002; McNally et al., 2008). The effect was specific to the  $\gamma$ 4-subunit as no such decreases were observed for  $\gamma$ 2-subunit mRNA (Fig. 5.8, Chapter Five), concurring with previous imprinting-training studies in chicken (Harvey et al., 1998). A decrease in mRNA levels may reflect a reduction in the rate of transcription, decreased mRNA stability or an increase in translation. Observed decreases in  $\gamma$ 4-subunit mRNA levels represent a long-term response to learning, most

likely reflected by an alteration in gene transcription rather than reduced mRNA stability. It is generally accepted that a reduction in GABA<sub>A</sub> receptor-mediated neurotransmission is involved in improved learning (Brioni and McGaugh, 1988; Clements and Bourne, 1996; Chapouthier and Venault, 2002, 2004; Collinson *et al.*, 2002; Crestani *et al.*, 2002; Zarrindast *et al.*, 2006), so the subsequent decrease in  $\gamma$ 4-subunit mRNA levels is unlikely to be attributed to an increased rate of translation.

#### 7.5 Regulation of GABA<sub>A</sub> receptor subunit gene expression

Gene expression can be regulated at many different levels including rate of initiation of transcription, splicing or stability of mRNA products, translation rate, post-translational modifications (including phosphorylation, glycosylation), receptor assembly, trafficking of receptors and protein stability, e.g. degradation or recycling (Steiger and Russek, 2004). Regulation at the level of transcription is important (Lemon and Tjian, 2000) and is controlled by a variety of transcription factors which directly interact with specific DNA sequences adjacent to the gene. Within GABA<sub>A</sub> receptor subunit gene promoter sequences, consensus sites for interaction of transcription factors from a variety of families have been identified via computational analysis. Interestingly, the  $\gamma$ -subunit gene class appear to harbour the greatest number of transcriptional elements. Moreover, the majority of transcriptional binding motifs are unique to each subunit class, thus enabling subunit-specific transcription of  $\alpha$ -,  $\beta$ - or  $\gamma$ -subunit genes (Steiger and Russek, 2004). Presence of multiple transcriptional elements, which may work alone or in tandem, facilitates the characteristic spatial expression of GABA<sub>A</sub> receptor subunit genes and responses in gene transcription to temporal and environmental cues. Cisregulatory elements which bind tissue-specific factors also permit regional-specific regulation of gene expression. Genes which have CNS-restricted expression encompass a characteristic 21 bp motif (neuron-restrictive silencer element; NRSE), which binds neuron-restrictive silencing factor (NRSF, a zinc finger repressor); NRSF is predominantly found in non-neuronal tissues where it can silence expression of genes which have an NRSE (Joyce, 2007). In silico analysis has demonstrated the presence of NRSE-like sequences for the majority of GABA<sub>A</sub> receptor subunit genes located in the promoter region ( $\alpha 2$ ,  $\alpha 3$ ,  $\alpha 4$ ,  $\beta 2$ ,  $\gamma 1$  and  $\delta$  subunits), or downstream of the start site ( $\alpha 1$ ,  $\alpha 5$ ,  $\beta 1$ ,  $\beta 3$ ,  $\gamma 2$ ,  $\delta$  and  $\varepsilon$  subunits; Steiger and Russek, 2004). No NRSE domain was identified in the  $\pi$ -subunit gene, thus accounting for its peripheral distribution. More

recently, regulation of gene expression via the micro (mi)RNA pathway is under closer scrutiny. miRNAs are short, non-coding RNAs which can endogenously regulate gene expression via translational arrest (RNAi pathway; Pillai *et al.*, 2005, 2007). A computer-predicated miRNA has been identified for the  $\varepsilon$ -subunit gene, the mammalian orthologue of  $\gamma$ 4 (Dostie *et al.*, 2003), and although there is some evolutionary sequence divergence between  $\gamma$ 4 and  $\varepsilon$ , there remains much identity and it is likely such a molecule exists for regulation of  $\gamma$ 4-subunit gene expression.

Furthermore, endogenous and exogenous molecules are able to influence  $GABA_A$ receptor subunit expression. As detailed in section 1.2.7, neurosteroids are able to potentiate the activity of GABA<sub>A</sub> receptors, but they can also govern at a transcription level by binding to hormone-activated nuclear receptors. These function as transcription factors and can interact with specific sites within GABAA receptor subunit gene promoter regions. Such steroid hormone response elements have been identified in the majority of subunit sequences, which interact with ERs (estrogen receptors), PRs (progesterone receptors), GRs (glucocorticoid receptors) and ARs (androgen receptors). No consensus sites for steroid hormone response elements were found in  $\alpha 2$ - or  $\theta$ subunit sequences (Steiger and Russek, 2004), which correlates with data herein, as the  $\alpha$ 2-subunit gene showed no evidence of developmentally-regulated expression within the song system.  $\gamma$ 1-,  $\gamma$ 2- and  $\epsilon$ -subunit gene promoters contain AR-consensus motifs and the orthologue of  $\varepsilon$ ,  $\gamma 4$ , exhibited highly coordinated gene expression at important developmental stages (Fig. 4.10, Chapter Four). Testosterone is an androgen and thus can modulate transcription of GABA<sub>A</sub> receptor subunits in mammals. Due to the high sequence conservation, this would also be applicable in lower vertebrates, including the zebra finch. The song system is sexually dimorphic (Arnold, 1997) and the sexual differentiation in brain nuclei structures in the zebra finch are purported to be a result of exposure to steroid hormones (Cooke et al., 1998). In fact both male and female zebra finch nuclei are sensitive to testosterone and estrogen. Testosterone is converted to estrogen by aromatase, which is also present in song system nuclei (Gahr and Metzdorf, 1999; Jacobs et al., 1999). More recently, in vitro studies have demonstrated an essential role for autonomously synthesised estrogen in development of the HVC $\rightarrow$ RA connection in zebra finch (Holloway and Clayton, 2001), which may be connected with the large increase in  $\gamma$ 4-subunit gene expression in RA following establishment of this

connection. Testosterone, (and its metabolites) are implicated in the crystallisation and stereotypy of zebra finch song (Pröve, 1983; Korsia and Bottjer, 1991; Bottjer and Hewer, 1992; Williams et al., 2003; Cynx et al., 2005). Furthermore, increased levels of corticosterone, a glucocorticoid (released under stressful conditions) can facilitate or detrimentally affect learning and memory (Coburn-Litvak et al., 2003; Martin et al., 2009; Quirarte et al., 2009). Corticosterone administration during development results in adult zebra finches producing shorter and less complex songs than their untreated counterparts (Spencer *et al*, 2003). Mammalian  $\alpha$ 3-,  $\beta$ 1-,  $\delta$ - and  $\epsilon$ -subunit promoters all contain glucocorticoid receptor consensus sites. Increased levels of corticosterone produced in response to stress would a) potentiate GABA<sub>A</sub> receptor activity, thereby increasing inhibition and reducing memory formation and b) effect transcription of GABA<sub>A</sub> receptor subunit genes. Accordingly, studies in rats have demonstrated that expression and pharmacology of GABA<sub>A</sub> receptors in the hippocampus are affected by stress levels of corticosterone. In some regions of the hippocampus, the  $\beta$ 2- and  $\gamma$ 2subunit mRNAs are increased in response to high corticosterone levels (Orchinik et al., 1995, 2001). As there is a wealth of evidence to suggest the presence of GABAA receptors within the song system and evidence that upregulation of GABA<sub>A</sub> receptors impairs memory; it may be the case that high levels of corticosterone result in deficits in the memorisation and production of tutor song by increasing GABA<sub>A</sub> receptor-mediated neurotransmission. This was also a reason for ensuring the zebra finches raised in auditory isolation were in the presence of females, as the stressful situation of isolation for such sociable animals may have influenced the results. Despite this, direct links between  $GABA_A$  receptor subunit expression and neurosteroids in the song-system have not been investigated. There is also evidence to the contrary. The rufous-sided towhee (*Pipilo erythropthalmus*) has a large repertoire, but is a close-ended learner and shows increased song nuclei size and increased testosterone on a seasonal basis, but no change in song (Brenowitz et al., 2007). Furthermore, there is no alteration in circulating endogenous testosterone levels in birds which have been mechanically or chemically deafened (Woolley and Rubel, 2002, Brenowitz et al., 2007), nor is there a change in nuclei size (Burek et al., 1991; Brenowitz et al., 2007) which highlights a dissociation between production of song and endogenous neurosteroids.

Chapter Seven

#### 7.6 Concluding remarks

As GABA<sub>A</sub> receptors are targets of many clinically important drugs (benzodiazepines, barbiturates, steroids and volatile anaesthetics; Reynolds, 2008), it is important to resolve specific receptor subtypes involved in mediating different physiological functions in an effort to generate novel subtype-specific (and thus function-specific) therapeutic strategies (such as those being developed for  $\alpha$ 5-subunit containing receptors; Chambers et al., 2004). GABAA receptors have long been implicated in modulation of learning and memory (for review, see Chapouthier and Venault, 2002; Maubach, 2003; McNally et al., 2008), and the avian song system is a well established paradigm for studying neuronal mechanisms of cognitive processes and development of sophisticated motor skills. Due to the numerous parallels with human speech production (Doupe and Kuhl, 1999; Brainard and Doupe, 2002; Kuhl, 2004; Bolhius and Gahr, 2006), defining GABA<sub>A</sub> subtypes and their functional roles in this system may extrapolate to human behaviour and subtype-specific strategies can be employed to target cognitive defects and give further understanding into diseases which affect motor coordination. Results obtained from this study complement those obtained by Wada et al. (2004). They observed differential expression of a selection of glutamate receptor subunits/subtypes (which elicit excitatory post-synaptic responses), in various vocal nuclei of songbirds, the functions of which correspond to the properties of each individual nucleus. Data acquired for both inhibitory- and excitatory-eliciting receptors is essential as there is complex interplay between glutamatergic and GABAergic systems in song learning (Bolhius and Gahr, 2006).

This study gives an exclusive insight into the potential role of GABA<sub>A</sub> receptors in a learning and memory paradigm. Concurring with data obtained in the mammalian brain, GABA<sub>A</sub> receptor subunit mRNAs exhibit ubiquitous, yet unique expression profiles thoughout nuclei of the song system, indicative of many potential assemblies. This was the first complete study of the neuronal GABA<sub>A</sub> receptor gene family, and the first study to also consider GABA<sub>C</sub> receptor subunits, as only when all subunits are considered can meaningful conclusions be drawn. In addition to spatial regulation, GABA<sub>A</sub> receptor subunit mRNAs showed developmentally-correlated patterns of expression, which provided further evidence of a functional role. The most interesting subunit was  $\gamma$ 4, which has avoided much scrutiny since its discovery due to its absence

215

in mammalian species, despite being the orthologue of the mammalian ɛ-subunit gene (Darlison *et al.*, 2005). *ɛ*-subunit-containing receptors have also not been thoroughly investigated as they show no sensitivity to benzodiazepines (Davies et al., 1997; Whiting et al., 1997). Yet, the discrete and proliferate distribution of GABA<sub>A</sub> receptor  $\gamma$ 4-subunit transcript within nuclei of the avian song system is indicative of an important physiological role. Moreover, zebra finches unable to produce normal song demonstrate a significant increase in GABA<sub>A</sub> receptor  $\gamma$ 4-subunit transcript expression in nuclei contained in both the AFP and VMP, thereby confirming the theory that downregulation of GABAergic mechanisms plays a role in learning and memory systems. This effect was exclusive to the  $\gamma$ 4 subunit, no such effects were observed with the  $\gamma$ 2subunit transcript. Such discrete distribution of y4-subunit mRNA suggested a specific functional role. In vitro RNA interference experiments indicated that the shRNA constructs were efficient in silencing  $\gamma$ 4-subunit mRNA. Once subsequent protein reduction is validated, work could be taken to an *in vivo* level and the function of y4subunit receptors further verified. RNAi has successfully been employed in the zebra finch to silence FOXP2 (which encodes the forkhead box transcription factor) in vivo, resulting in birds producing abnormal song (Haesler et al., 2007). A mutation in the same gene in humans causes speech and language impairments (Lai et al., 2001, 2003; MacDermot et al., 2005), demonstrating how data can be extrapolated from zebra finch to human subjects. As mammals have the  $\gamma$ 4-subunit orthologue,  $\varepsilon$ , this work is relevant by contributing to our understanding of GABAA receptor involvement in complex cognitive processes.

With regards to determining native GABA<sub>A</sub> receptor stoichiometry, evidence for true subunit association must firstly be established by co-localisation of subunit mRNAs and corresponding proteins in tissues/cells. This work has, for the first time, characterised the expression of all the GABA<sub>A</sub> receptor subunit genes in the song system revealing subunit-specific and time-specific changes in levels of gene expression, which have been correlated with different stages of the song learning process. A novel splice variant of the  $\pi$ -subunit has been identified which was present, along with the full length transcript in zebra finch brain. Furthermore, evidence has been provided for the involvement of  $\gamma$ 4-subunit-containing receptors in learning and memory. Preliminary *in vitro* studies have verified that  $\gamma$ 4-subunit gene expression can be silenced with specific

shRNA constructs and paves the way for future *in vivo* studies to further confirm the role of  $\gamma$ 4-subunit-containing GABA<sub>A</sub> receptors. Subcellular distribution should then be investigated (as a single neuron can express several different GABA<sub>A</sub> receptor subtypes), followed by a demonstration of subunit interactions e.g. by co-precipitation studies. A  $\gamma$ 4-subunit fusion protein has been generated and purified which can be synthesised into an antibody to allow for future analyses at a protein level. Given that GABA<sub>A</sub> receptors containing the  $\gamma$ 4-subunit have a relatively unique pharmacology (Forster *et al.*, 2001); specific subtype-selective drugs could be applied to ameliorate brain function in learning and memory processes. This information may be extrapolated to human subjects due to the many parallels between human and avian telencephalic systems and the presence of the  $\gamma$ 4-subunit orthologue,  $\varepsilon$  in the CNS of mammals.

### **BIBLIOGRAPHY**

- Abbas-Terki, T., Blanco-Bose, W., Déglon, N., Pralong, W. and Aebischer. P. (2002). Lentiviral-mediated RNA interference. *Hum. Gene Ther.* **13:** 2197-201
- Adar, E., Lotem, A. and Barnea, A. (2008). The effect of social environment on singing behavior in the zebra finch (*Taeniopygia guttata*) and its implication for neuronal recruitment. *Behav. Brain Res.* 187: 178-84
- Adret, P. (2004). Vocal imitation in blindfolded zebra finches (*Taeniopygia guttata*) is facilitated in the presence of a non-singing conspecific female. J. Ethol. 22: 29-35
- Akabas, M. H. (2004). GABA<sub>A</sub> receptor structure-function studies: a re-examination in light of new acetylcholine receptor structures. *Int. Rev. Neurobiol.* 62: 1-43
- Alakuijala, A., Alakuijala, J. and Pasternack, M. (2006). Evidence for a functional role of GABA<sub>C</sub> receptors in the rat mature hippocampus. *Eur. J. Neurosci.* 23: 514-20
- Alakuijala, A., Palgi, M., Wegelius, K., Schmidt, M., Enz, R., Paulina, L., Saarma, M. and Pasternack, M. (2005). GABA receptor ρ subunit expression in the developing rat brain. *Dev. Brain Res.* **154**: 15-23
- Albrecht, B. E., Breitenbach, U., Stühmer, T., Harvey, R. J. and Darlison, M. G. (1997). *In situ* hybridisation and reverse transcription-polymerase chain reaction studies on the expression of the GABA<sub>C</sub> receptor  $\rho$ 1- and  $\rho$ 2-subunit genes in avian and rat brain. *Eur. J. Neurosci.* **9**: 2414-22
- Albrecht, B. E. and Darlison, M. G. (1995). Localization of the  $\rho$ 1- and  $\rho$ 2-subunit messenger RNAs in chick retina by *in situ* hybridization predicts the existence of  $\gamma$ -aminobutyric type C receptor subtypes. *Neurosci. Lett.* **189:** 155-8
- Alldred, M. J., Mulder-Rosi, J., Lingenfelter, S. E., Chen, G. and Luscher, B. (2005). Distinct γ2 subunit domains mediate clustering and synaptic function of postsynaptic GABA<sub>A</sub> receptors and gephyrin. J. Neurosci. 25: 594-603
- Alvarez, V. A., Ridenour, D. A. and Sabatini, B. L. (2006). Retraction of synapses and dendritic spines induced by off-target effects of RNA interference. J. Neurosci. 26: 7820-5
- Amato, A., Connolly, C. N., Moss, S. J. and Smart, T. G. (1999). Modulation of neuronal and recombinant GABA<sub>A</sub> receptors by redox reagents. J. Physiol. 517: 35-50
- Ambros, V. (2004). The functions of animal microRNAs. *Nature* **431**: 350-5
- Anon. (2003). Whither RNAi? Nat. Cell Biol. 5: 489-90
- Arancibia-Cárcamo, L. and Kittler, J. T. (2009). Regulation of GABA<sub>A</sub> receptor membrane trafficking and synaptic localization. *Pharmacol. Ther.* **123**: 17-31
- Arnold, A. P. (1997). Sexual differentiation of the zebra finch song system: Positive evidence, negative evidence, null hypotheses, and a paradigm shift. J. Neurobiol. 33: 572-84
- Atack, J. R. (2003). Anxioselective compounds acting at the GABA(A) receptor benzodiazepine binding site. *Curr. Drug Targets CNS Neurol. Disord.* **2:** 213-32
- Atack, J. R., Dawson, G. R., McKernan, R. M., Whiting, P. J. and Rosahl, T. W. (2001). Loss of the major GABA<sub>A</sub> receptor subtype in the brain is not lethal in mice. J. Neurosci. 21: 3409-18
- Backus, J., Laughlin, T., Wang, Y., Belly, R., White, R., Baden, J., Min, C. J., Mannie, A., Tafra, L., Atkins, D. and Verbanac, K. M. (2005). Identification and characterization of optimal gene expression markers for detection of breast cancer metastasis. J. Mol. Diagn. 7: 327-36

- Bailey, D. J. and Wade, J. (2005). FOS and ZENK responses in 45-day-old zebra finches vary with auditory stimulus and brain region, but not sex. *Behav. Brain Res.* **162**: 108-15
- Bailey, M. E. S., Albrecht, B. E., Johnson, K. J. and Darlison, M. G. (1999). Genetic linkage and radiation hybrid mapping of the three human GABA<sub>C</sub> receptor ρ subunit genes: *GABRR1*, *GABRR2* and *GABRR3*. *Biochim. Biophys. Acta* 1447: 307-12
- Baker, M. C., Bottjer, S. W. and Arnold, A. P. (1984). Sexual dimorphism and lack of seasonal changes in vocal control regions of the white-crowned sparrow brain. *Brain Res.* 295: 85-9
- Bali, M. and Akabas, M. H. (2007). The location of a closed channel gate in the GABA<sub>A</sub> receptor channel. *J. Gen. Physiol.* **129**: 145-59
- Barnard, E. A., Skolnick, P., Olsen, R. W., Mohler, H., Sieghart, W., Biggio, G., Braestrup, C. E., Bateson, A. N. and Langer, S. Z. (1998). International Union of Pharmacology. XV. Subtypes of  $\gamma$ -aminobutyric acid<sub>A</sub> receptors: Classification on the basis of subunit structure and receptor function. *Pharm. Rev.* **50**: 291-310
- Bartel, D. P. (2004). MicroRNAs: genomics, biogenesis, mechanism, and function. *Cell* **116:** 281-97
- Barton, G. M. and Medzhitov, R. (2002). Retroviral delivery of small interfering RNA into primary cells. *Proc. Natl. Acad. Sci. USA* **99:** 14943-8
- Bateson, A. N., Lasham, A. and Darlison, M. G. (1991).  $\gamma$ -aminobutyric acid<sub>A</sub> receptor heterogeneity is increased by alternative splicing of a novel  $\beta$ -subunit gene transcript. *J. Neurochem.* **56**: 1437-40
- Baumann, S. W., Baur, R. and Sigel, E. (2002). Forced subunit assembly in  $\alpha 1\beta 2\gamma 2$ GABA<sub>A</sub> receptors: Insight into the absolute arrangement. *J. Biol. Chem.* **277**: 46020-5
- Beck, M., Brickley, K., Wilkinson, H. L., Sharma, S., Smith, M., Chazot, P. L., Pollard, S. and Stephenson, F. A. (2002). Identification, molecular cloning and characterisation of a novel GABA<sub>A</sub> receptor-associated protein, GRIF-1. *J. Biol. Chem.* 277: 30079-90
- Bedford, F. K., Kittler, J. T., Muller, E., Thomas, P., Uren, J. M., Merlo, D., Wisden, W., Triller, A., Smart, T. G. and Moss, S. J. (2001). GABA<sub>A</sub> receptor cell surface number and subunit stability are regulated by the ubiquitin-like protein Plic-1. *Nat. Neurosci.* 4: 908-16
- Belelli, D. and Lambert, J. J. (2005). Neurosteroids: endogenous regulators of the GABA<sub>A</sub> receptor. *Nat. Rev. Neurosci.* **6:** 565-75
- Belujon, P., Baufreton, J., Grandoso, L., Boué-Grabot, E., Batten, T. F. C., Ugedo, L., Garret, M. and Taupignon, A. I. (2009). Inhibitory transmission in *locus coeruleus* neurons expressing GABA<sub>A</sub> receptor epsilon subunit has a number of unique properties. *Neurophysiol.* **102**: 2312-25
- Benarroch, E. E. (2007). GABA<sub>A</sub> receptor heterogeneity, function, and implications for epilepsy. *Neurol.* **68:** 612-14
- Benke, D., Honer, M., Michel, C. and Mohler, H. (1996). GABA<sub>A</sub> receptor subtypes differentiated by their γ-subunit variants: Prevalence, pharmacology and subunit architecture. *Neurpharmacol.* **35**: 1413-23
- Benke, D., Mertens, S., Trzeciakt, A., Gillessent, D. and Mohler, H. (1991). GABA<sub>A</sub> receptors display association of  $\gamma$ 2-subunit with  $\alpha$ 1- and  $\beta$ 2/3-subunits. *J. Biol. Chem.* **266:** 4478-83

- Benson, J. A., Löw, K., Keist, R., Mohler, H. and Rudolph, U. (1998). Pharmacology of recombinant  $\gamma$ -aminobutyric acid<sub>A</sub> receptors rendered diazepam-insensitive by point-mutated  $\alpha$ -subunits. *FEBS Lett.* **431:** 400-4
- Bentley, G. E., Spar, B. D., MacDougall-Shackleton, S. A., Hahn, T. P. and Ball, G. F. (2000). Photoperiodic regulation of the reproductive axis in male zebra finches, *Taeniopygia guttata. Gen. Comp. Endocrinol.* **117**: 449-55
- Berget, S. M. (1995). Exon recognition in vertebrate splicing. J. Biol. Chem. 270: 2411-4
- Bernstein, E., Caudy, A. A., Hammond, S. M. and Hannon, G. J. (2001). Role for a bidentate ribonuclease in the initiation step of RNA interference. *Nature* 409: 363-6
- Bertrand, J. R., Pottier, M., Vekris, A., Opolon, P., Maksimenko, A. and Malvy, C. (2002). Comparison of antisense oligonucleotides and siRNAs in cell culture and *in vivo. Biochem. Biophys. Res. Commun.* **296**: 1000-4
- Billy, E., Brondani, V., Zhang, H., Muller, U. and Filipowicz, W. (2001). Specific interference with gene expression induced by long, double-stranded RNA in mouse embryonal teratocarcinoma cell lines. *Proc. Natl. Acad. Sci. USA* 98: 14428-33
- Birmingham, A., Anderson, E. M., Reynolds, A., Ilsley-Tyree, D., Leake, D., Fedorov, Y., Baskerville, S., Maksimova, E., Robinson, K., Karpilow, J., Marshall, W. S. and Khvorova, A. (2006). 3' UTR seed matches, but not overall identity, are associated with RNAi off-targets. *Nat. Methods* 3: 199-204
- Bishop, D. V. M., North, T. and Donlan, C. (1995). Genetic basis of specific language impairment: Evidence from a twin study. *Dev. Med. Child Neurol.* **37:** 56-71
- Blount, P. and Merlie, J. P. (1990). Mutational analysis of muscle nicotinic acetylcholine receptor subunit assembly. *J. Cell Biol.* **111**: 2613-22
- Bock, J. and K. Braun. (1999). Filial imprinting in domestic chicks is associated with spine pruning in the associative area, dorsocaudal neostriatum. *Eur. J. Neurosci.* **11:** 2566-70
- Bock, J., Thode, C., Hannemann, O., Braun, K. and Darlison, M. G. (2005). Early socio-emotional experience induces expression of the immediate-early gene Arc/arg3.1 (activity-regulated cytoskeleton-associated protein/activity-regulated gene) in learning-relevant brain regions of the newborn chick. *Neurosci.* **133**: 625-33
- Bocquet, N., Nury, H., Baaden, M., Le Poupon, C., Changeux, J. P., Delarue, M. and Corringer, P. J. (2009). X-ray structure of a pentameric ligand-gated ion channel in an apparently open conformation. *Nature* **457**: 111-4
- Bogdanov, M., Zhang, W., Xie, J. and Dowhan, W. (2005). Transmembrane protein topology mapping by the substituted cysteine accessibility method (SCAM<sup>TM</sup>): Application to lipid-specific membrane protein topogenesis. *Methods* **36**: 148-71
- Böhner, J. (1983). Song learning in the zebra finch (*Taeniopygia guttata*): selectivity in the choice of a tutor and accuracy of song copies. *Anim. Behav.* **31**: 231-7
- Bolhuis, J. J. and Gahr, M. (2006). Neural mechanisms of birdsong memory. *Nat. Rev. Neurosci.* **7:** 347-57
- Bolhuis, J. J. and Macphail, M. (2001). A critique of the neuroecology of learning and memory. *Trends Cogn. Sci.* **5:** 426-33
- Bolhuis, J. J., Van Mil, D. P. and Houx, B. B. (1999). Song learning with audiovisual compound stimuli in zebra finches. *Anim. Behav.* **58**: 1285-92

- Bollan, K. A., Baur, R., Hales, T. G., Sigel, E. and Connolly, C. N. (2008). The promiscuous role of the epsilon subunit in GABA<sub>A</sub> receptor biogenesis. *Mol. Cell. Neurosci.* 37: 610-21
- Bonnert, T. P., McKernan, R. M., Farrar, S., Bourdellès, B. L., Heavens, R. P., Smith, D. W., Hewson, L., Rigby, M. R., Sirinathsinghji, D. J. S., Brown, N., Wafford, K. A. and Whiting, P. J. (1999). θ, a novel γ-aminobutyric acid type A receptor subunit. *Neurobiol.* 96: 9891-6
- Bormann, J. (2000). The 'ABC' of GABA receptors. Trends Pharmacol. Sci. 21: 16-9
- Bormann, J. and Feigenspan, A. (1995). GABA<sub>C</sub> receptors. Trends Neurosci. 18: 515-9
- Bottjer, S. W., Brady, J. D. and Walsh, J. P. (1998). Intrinsic and synaptic properties of neurons in the vocal-control nucleus IMAN from *in vitro* slice preparations of juvenile and adults zebra finches. *J. Neurobiol.* **37**: 642-58
- Bottjer, S. W., Halsema, K. A., Brown, S. A. and Miesner, E. A. (1989). Axonal connections of a forebrain nucleus involved with vocal learning in zebra finches. *J. Comp. Neurol.* **279:** 312-26
- Bottjer, S. W. and Hewer, S. J. (1992). Castration and antisteroid treatment impair vocal learning in male zebra finches. *J. Neurobiol.* **23:** 337-53
- Bottjer, S. W., Miesner, E. A. and Arnold, A. P. (1984). Forebrain lesions disrupt development but not maintenance of song in passerine birds. *Science* **224**: 901-3
- Boué-Grabot, E., Roudbaraki, M., Bascles, L., Tramu, G., Bloch, B. and Garret, M. (1998). Expression of GABA receptor ρ subunits in rat brain. *J. Neurochem.* **70**: 899-907
- Braaten, R. F. and Reynolds, K. (1999). Auditory preference for conspecific song in isolation-reared zebra finches. *Anim. Behav.* **58**: 105-11
- Brainard, M. S. and Doupe, A. J. (2000). Interruption of a basal ganglia-forebrain circuit prevents plasticity of learned vocalizations. *Nature* **404**: 762-6
- Brainard, M. S. and Doupe, A. J. (2002). What songbirds teach us about learning. *Nature* **417:** 351-8
- Brejc, K., van Dijk, W. J., Klaassen, R. V., Schuurmans, M., van Der Oost, J., Smit, A.
  B. and Sixma, T. K. (2001). Crystal structure of an ACh-binding protein reveals the ligand-binding domain of nicotinic receptors. *Nature* 411: 269-76
- Brenowitz, E. A., Lent, K. and Rubel, E. W. (2007). Auditory feedback and song production do not regulate seasonal growth of song control circuits in adult whitecrowned sparrows. *J. Neurosci.* 27: 6810-4
- Brenowitz, E. A., Margoliash, D. and Nordeen, K. W. (1997). An introduction to birdsong and the avian song system. *J. Neurobiol.* **33:** 495-500
- Brenowitz, E. A., Nails, N., Wingfield, J. C. and Kroodsma, D. E. (1991). Seasonal changes in avian song nuclei without seasonal changes in song repertoire. *J. Neurosci.* **11**: 1367-74
- Brioni, J. D. (1993). Role of GABA during the multiple consolidation of memory. *Drug Dev. Res.* **28:** 3-27
- Brioni, J. D. and McGaugh, J. L. (1988). Post-training administration of GABAergic antagonists enhances retention of aversively motivated tasks. *Psychopharmacol.* 96: 505-10
- Brown, N., Kerby, J., Bonnert, T. P., Whiting, P. J. and Wafford, K. A. (2002). Pharmacological characterization of a novel cell line expressing human  $\alpha_4\beta_3\delta$ GABA<sub>A</sub> receptors. *Br. J. Pharmacol.* **136**: 965-74
- Brumm, H., Zollinger, S. A. and Slater, P. J. B. (2009). Developmental stress affects song learning but not song complexity and vocal amplitude in zebra finches. *Behav. Ecol. Sociobiol.* 63: 1387-95

- Brummelkamp, T. R., Bernards, R. and Agami, R. (2002). A system for stable expression of short interfering RNAs in mammalian cells. *Science* **296**: 550-3
- Buddhala, C., Hsu, C-. C., and Wu, J-. Y. (2009). A novel mechanism for GABA synthesis and packaging into synaptic vesicles. *Neurochem. Int.* **55**: 9-12
- Burek, M. J., Nordeen, K. W. and Nordeen, E. J. (1991). Neuron loss and addition in developing zebra finch song nuclei are independent of auditory experience during song learning. J. Neurobiol. 22: 215-23
- Burt, D. R. (2003). Reducing GABA receptors. *Life Sci.* 73: 1741-58
- Burt, D. R. and Kamatchi, G. L. (1991). GABA<sub>A</sub> receptor subtypes: from pharmacology to molecular biology. *Fed. Am. Soc. Exp. Biol.* **5**: 2916-23
- Calero, C. I. and Calvo, D. J. (2008). Redox modulation of homomeric  $\rho_1$  GABA<sub>C</sub> receptors. *J. Neurochem.* **105:** 2367-74
- Canady, R. A., Burd, G. D., DeVoogd, T. J. and Nottebohm, F. (1988). Effect of testosterone on input received by an identified neuron type of the canary song system: A golgi/electron microscopy/degeneration study. J. Neurosci. 8: 3770-84
- Carlisle, H. J., Hales, T. G. and Schlinger, B. A. (1998). Characterization of neuronal zebra finch GABA<sub>A</sub> receptors: Steroid effects. *J. Comp. Physiol.* **182**: 531-8
- Carroll, S. B. (2005). Evolution at two levels: on genes and form. PLoS Biol. 3: 1159-66
- Castanotto, D. and Rossi, J. J. (2009). The promises and pitfalls of RNA-interferencebased therapeutics. *Nature*. **457**: 426-33
- Castanotto, D., Sakurai, K., Lingeman, R., Li, H., Shively, L., Aagaard, L., Soifer, H., Gatignol, A., Riggs, A. and Rossi, J. J. (2007). Combinatorial delivery of small interfering RNAs reduces RNAi efficacy by selective incorporation into RISC. *Nucl. Acids Res.* 35: 5154-64
- Catchpole, C. K. and Slater, P. J. B. (1995). Birdsong: biological themes and variations. Cambridge. Cambridge University Press. pp 10-11
- Cavallaro, S., D'Agata, V., Manickam, P., Dufour, F. and Alkon, D. L. (2002). Memory-specific temporal profiles of gene expression in the hippocampus. *Proc. Natl. Acad. Sci. USA* 99: 16279-84
- Chalk, A. M., Wahlestedt, C. and Sonnhammer, E. L. L. (2004). Improved and automated prediction of effective siRNA. *Biochem. Biophys. Res. Commun.* **319**: 264-74
- Chapouthier, G. and Venault, P. (2002). GABA-A receptor complex and memory processes. *Curr. Top. Med. Chem.* **2**: 841-51
- Chapouthier, G. and Venault, P. (2004). GABA<sub>A</sub> receptor complex and memory processes. *Med. Chem. Rev.* **1**: 91-9
- Chaudhuri, A. (1997). Neural activity mapping with inducible transcription factors. *NeuroRep.* **8:** 5-9
- Chebib, M., Hanrahan, J. R., Kumar, R. J., Mewett, K. N., Morriss, G., Wooller, S. and Johnston, G. A. R. (2007). (3-Aminocyclopentyl) methylphosphinic acids: Novel GABA<sub>C</sub> receptor antagonists. *Neuropharmacol.* **52**: 779-87
- Chebib, M. and Johnston, A. R. (1999). The 'ABC' of GABA receptors: A brief review. *Clin. Exp. Pharmacol. Physiol.* **26:** 937-40
- Chebib, M. and Johnston, G. A. (2000). GABA-activated ligand gated ion channels: Medicinal chemistry and molecular biology. *J. Med. Chem.* **43**: 1427-47
- Chebib, M., Mewett, K. N. and Johnston, G. A. R. (1998). GABA<sub>C</sub> receptor antagonists differentiate between human ρ1 and ρ2 receptors expressed in *Xenopus* oocytes. *Eur. J. Pharmacol.* 357: 227-34

- Chen, Z-. W., Chang, C.-S. S., Leil, T. A., Olcese, R. and Olsen, R. W. (2005). GABA<sub>A</sub> receptor-associated protein regulates GABA<sub>A</sub> receptor cell-surface number in *Xenopus laevis* oocytes. *Mol. Pharmacol.* 68: 152-9
- Chendrimada, T. P., Gregory, R. I., Kumaraswamy, E., Norman, J., Cooch, N., Nishikura, K. and Shiekhattar, R. (2005). TRBP recruits the Dicer complex to Ago2 for microRNA processing and gene silencing. *Nature* **436**: 740-44
- Choi, I., Cho, B-. R., Kim, D., Miyagawa, S., Kubo, T., Kim, J. Y., Park, C.-G., Hwang, W. S., Lee, J. S. and Ahn, C. (2005). Choice of the adequate detection time for the accurate evaluation of the efficiency of siRNA-induced gene silencing. *J. Biotechnol.* 120: 251-61
- Clayton, G. H., Owens, G. C., Wolff, J. S. and Smith, R. L. (1998). Ontogeny of cation-Cl<sup>-</sup> cotransporter expression in rat neocortex. *Dev. Brain Res.* **109**: 281-92
- Clayton, N. S. and Pröve, E. (1989). Song discrimination in female zebra finches and Bengalese finches, *Anim. Behav.* **38:** 352-62
- Clements, M. P. and Bourne, C. (1996). Passive avoidance learning in the day-old chick is modulated by GABAergic agents. *Pharmacol. Biochem. Behav.* **53**: 629-3
- Coburn-Litvak, P. S., Pothakos, K., Tata, D. A., McCloskey, D. P. and Anderson, B. J. (2003). Chronic administration of corticosterone impairs spatial reference memory before spatial working memory in rats. *Neurobiol. Learn. Mem.* **80**: 11-23
- Cogoni, J. T., Irelan, T., Schumacher, M., Schmidhauser, T. J., Sekler, E. U. and Macino, G. (1996). Transgene silencing of the al-1 gene in vegetative cells of *Neurospora* is mediated by a cytoplasmic effector and does not depend on DNA– DNA interactions or DNA methylation. *Eur. Mol. Biol. Assoc. J.* 15: 3153-63
- Collingridge, G. L., Olsen, R. W., Peters, J. and Spedding, M. (2009). A nomenclature for ligand-gated ion channels. *Neuropharmacol.* **56**: 2-5
- Collinson, N., Kuenzi, F. M., Jarolimek, W., Maubach, K. A., Cothliff, R., Sur, C., Smith, A., Otu, F. M., Howell, O., Atack, J. R., McKernan, R. M., Seabrook, G. R., Dawson, G. R., Whiting, P. J. and Rosahl, T. W. (2002). Enhanced learning and memory and altered GABAergic synaptic transmission in mice lacking the α5 subunit of the GABA<sub>A</sub> receptor. *J. Neurosci.* 22: 5572-80
- Concas, A., Follesa, P., Barbaccia, M. L., Purdy, R. H. and Biggio, G. (1999). Physiological modulation of GABA(A) receptor plasticity by progesterone metabolites. *Eur. J. Pharmacol.* 375: 225-35
- Connolly, C. N. and Wafford, K. A. (2004). The Cys-loop superfamily of ligand-gated ion channels: the impact of receptor structure on function. *Biochem. Soc. Trans.* **32:** 529-34
- Cossette, P., Liu, L., Brisebois, K., Dong, H., Lortie, A., Vanasse, M., Saint-Hilaire, J..
  M., Carmant, L., Verner, A., Lu, W-. Y., Wang, Y. T. and Rouleau, G. A. (2002).
  Mutation of *GABRA1* in an autosomal dominant form of juvenile myoclonic epilepsy. *Nat. Gen.* 31: 184-9
- Costa, C., Leone, G., Saulle, E., Pisani, F., Bernardi, G. and Calabresi, P. (2004). Coactivation of GABA<sub>A</sub> and GABA<sub>B</sub> receptor results in neuroprotection during *in vitro* ischemia. *Stroke* **35**: 596-600
- Cowie, R. and Douglas-Cowie, E. (1992). Postlingually acquired deafness: speech deterioration and the wider consequences. *Trends Ling. Stud.Monographs* **62:** 19-53
- Cristofaro, P. and Ramratnam, B. (2006). RNAi tackles a sexually transmitted disease. *Nat. Biotech.* **24:** 48-9

- Culiat, C. T., Stubbs, L., Montgomery, C. S., Russell, L. B. and Rinchik, E. M. (1994). Phenotypic consequences of deletion of the γ3, α5, or β3 subunit of the type A γaminobutyric acid receptor in mice. *Proc. Natl. Acad. Sci. USA* **91:** 2815-8
- Cullen, B. R. (2005). RNAi the natural way. Nat. Gen. 37: 1163-5
- Curtis, D. R., Duggan, A. W., Felix, D. and Johnston, G. A. R. (1970). GABA, bicuculline and central inhibition. *Nature* **226**: 1222-4
- Cutting, G. R., Curristin, S., Zoghbi, H., O'Hara, B., Seldin, M. F. and Uhl, G. R. (1992). Identification of a putative  $\gamma$ -aminobutyric acid (GABA) receptor rho<sub>2</sub> cDNA and colocalisation of the genes encoding rho<sub>2</sub> (GABRR2) and rho<sub>1</sub> (GABRR1) to human chromosome 6q14-q21 and mouse chromosome 4. *Genomics* **12**: 801-6
- Cynx, J., Bean, J. N. and Rossman, I. (2005). Testosterone implants alter the frequency range of zebra finch songs. *Horm. Behav.* **47:** 446-51
- Dai, F., Yusuf, F., Farjah, G. H. and Brand-Saberi, B. (2005). RNAi-induced targeted silencing in developmental control genes during chicken embryogenesis. *Dev. Biol.* 285: 80-90
- Dalby, B., Cates, S., Harris, A., Ohki, E. C., Tilkins, M. L., Price, P. J. and Ciccarone, V. C. (2004). Advanced transfection with Lipofectamine 2000 reagent: Primary neurons, siRNA, and high-throughput applications. *Methods* 33: 95-103
- Darlison, M. G. and Albrecht, B. E. (1995). GABA<sub>A</sub> receptor subtypes: which, where and why? *Semin. Neurosci.* **7:** 115-26
- Darlison, M. G., Pahal, I. and Thode, C. (2005). Consequences of the evolution of the GABA<sub>A</sub> receptor gene family. *Cell. Mol. Neurobiol.* **25:** 607-24
- Darwin, C. R. (1871). *The descent of man, and selection in relation to sex*. London. John Murray. 1: 1st edition
- Das, R. M., Van Hateren, N. J., Howell, G. R., Farrell, E. R., Bangs, F. K., Porteous, V. C., Manning, E. M., McGrew, M. J., Ohyama, K., Sacco, M. A., Halley, P. A., Sang, H. M., Storey, K. G., Placzek, M., Tickle, C., Nair, V. K. and Wilson, S. A. (2006). A robust system for RNA interference in the chicken using a modified microRNA operon. *Dev. Biol.* 294: 554-63
- Davidson, T. J., Harel, S., Arboleda, V. A., Prunell, G. F., Shelanski, M. L., Greene, L. A. and Troy, C. M. (2004). Highly efficient small interfering RNA delivery to primary mammalian neurons induces micro-RNA-like effects before mRNA degradation. J. Neurosci. 24: 10040-6
- Davies, P. A., Hanna, M. C., Hales, T. G. and Kirkness, E. F. (1997). Insensitivity to anaesthetic agents conferred by a class of GABA<sub>A</sub> receptor subunit. *Nature* **385**: 820-3
- Davies, P. A., Kirkness, E. F. and Hales, T. G. (2001). Evidence for the formation of functionally distinct αβγε GABA(A) receptors. *J. Physiol.* **537**: 101-13
- Davies, P. A., McCartney, M. R., Wang, W., Hales, T. G. and Kirkness, E. F. (2002). Alternative transcripts of the GABA<sub>A</sub> receptor  $\varepsilon$  subunit in human and rat. *Neuropharmacol.* **43**: 467–75
- Davies, P. A., Wang, W., Hales, T. G. and Kirkness, E. F. (2003). A novel class of ligand-gated ion channel is activated by Zn<sup>2+</sup>. *J. Biol. Chem.* **278**: 712-7
- Davis, S., Bozon, B. and Laroche, S. (2003). How necessary is the activation of the immediate early gene *zif* 268 in synaptic plasticity and learning? *Behav. Brain. Res.* 142: 17-30
- De Groof, G., Verhoye, M., Van Meir, V., Balthazart, J. and Van der Linden, A. (2008).
  Seasonal rewiring of the songbird brain: An *in vivo* MRI study. *Eur. J. Neurosci.* 28: 2475-85

- De Pieri, C., Beltramini, L. M., Selistre-de-Araújo, H. S., Vettore, A. L., da Silva, F. R., Arruda, P., Oliva, G. and de Souza, D. H. F. (2004). Overexpression, purification, and biochemical characterization of GumC, an enzyme involved in the biosynthesis of exopolysaccharide by *Xylella fastidiosa. Protein Expr. Purif.* **34**: 223-8
- Delaney, A. J. and Sah, P. (1999). GABA receptors inhibited by benzodiazepines mediate fast inhibitory transmission in the central amygdala. *J. Neurosci.* **19**: 9698-704
- DeLorey, T. M. and Olsen, R. W. (1999). GABA and epileptogenesis: Comparing gabrb3 gene-deficient mice with Angelman syndrome in man. *Epilepsy Res.* 36: 123-32
- Denli, A. M., Tops, B. B., Plasterk, R. H., Ketting, R. F. and Hannon, G. J. (2004). Processing of primary microRNAs by the microprocessor complex. *Nature* **432**: 231-5
- Derjusheva, S., Kurganova, A., Habermann, F. and Gaginskaya, E. (2004). High chromosome conservation detected by comparative chromosome painting in chicken, pigeon and passerine birds. *Chromosome Res.* **12**: 715-23
- Devi, G. R. (2006). siRNA-based approaches in cancer therapy. *Cancer Gene Ther.* 13: 819-29
- DeVoogd, T. J. (2004). Neural constraints on the complexity of avian song. Brain Behav. Evol. 63: 221-32
- DeVoogd, T. J., Krebs, J. R., Healy, S. D. and Purvis, A. (1993). Relations between song repertoire size and the volume of brain nuclei related to sing: comparative evolutionary analyses amongst oscine birds. *Proc. R. Soc. Lond.* **254**: 75-82
- D'Hulst, C., Atack, J. R. and Kooy, R. F. (2009). The complexity of the GABA<sub>A</sub> receptor shapes unique pharmacological profiles. *Drug Discov. Today* **14:** 866-75
- D'Hulst, C. and Kooy, R. F. (2007). The GABA<sub>A</sub> receptor: a novel target for treatment of fragile X? *Trends Neurosci.* **30**: 425-31
- Dooling, R. and Searcy, M. (1980). Early perceptual selectivity in the swamp sparrow. *Dev. Psychobiol.* **13:** 499-506
- Dostie, J., Mourelatos, Z., Yang, M., Sharma, A. and Dreyfuss, G. (2003). Numerous microRNPs in neuronal cells containing novel microRNAs. *RNA* **9**: 180-6
- Doupe, A. J. and Kuhl, P. K. (1999). Birdsong and human speech: common themes and mechanisms. *Ann. Rev. Neurosci.* 22: 567-631
- Dredge, B. K., Polydorides, A. D. and Darnell, R. B. (2001). The splice of life: Alternative splicing and neurological disease. *Nat. Rev. Neurosci.* **2:** 43-50
- Dutar, P., Vu, H. M. and Perkel, D. J. (1998). Multiple cell types distinguished by physiological, pharmacological, and anatomic properties in nucleus HVc of the adult zebra finch. *J. Neurophys.* **80**: 1828-38
- Dykxhoorn, D. M., Norvina, C. D. and Sharp, P.A. (2003). Killing the messenger: Short RNAs that silence gene expression. *Nat. Rev. Mol. Cell Biol.* **4**: 457-67
- Eales, L. A. (1987). Song learning in female-raised zebra finches: another look at the sensitive phase. *Anim. Behav.* **35:** 1356-65
- Eccles, J. C. (1982). The synapse: from electrical to chemical transmission. Ann. Rev. Neurosci. 5: 325-39
- Edenberg, H. J., Dick, D. M., Xuei, X., Tian, H., Almasy, L., Bauer, L. O., Crowe, R. R., Goate, A., Hesselbrock, V., Jones, K., Kwon, J., Li, T-. K., Nurnberger Jr., J. I., O'Connor, S. J., Reich, T., Rice, J., Schuckit, M. A., Porjesz, B., Foroud, T. and Begleiter, H. (2004). Variations in *GABRA2*, encoding the α2 subunit of the

GABA<sub>A</sub> receptor, are associated with alcohol dependence and with brain oscillations. *Am. J. Hum. Gen.* **74**: 705-14

- Ekema, G. M., Zheng, W. and Lu, L. (2002). Interaction of GABA receptor/channel ρ1 and γ2 subunit. *Invest. Ophthalmol. Vis. Sci.* **43**: 2326-33
- Elbashir, S. M., Harborth, J., Lendeckel, W., Yalcin, A., Weber, K. and Tuschl, T. (2001a). Duplexes of 21-nucleotide RNAs mediate RNA interference in cultured mammalian cells. *Nature* **411**: 494-8
- Elbashir, S. M., Lendeckel, W. and Tuschl, T. (2001b). RNA interference is mediated by 21- and 22-nucleotide RNAs. *Genes Dev.* **15**: 188-200
- Enz, R., Brandstätter, J. H., Hartveit, E., Wässle, H. and Bormann, J. (1995). Expression of GABA receptor ρ1 and ρ2 subunits in the retina and brain of the rat. *Eur. J. Neurosci.* **7:** 1495-501
- Enz, R. and Cutting, G. R. (1998). Molecular composition of GABA<sub>C</sub> receptors. *Vis. Res.* **38:** 1431-41
- Enz, R. and Cutting, G. R. (1999). GABA<sub>C</sub> receptor ρ subunits are heterogeneously expressed in the human CNS and form homo- and heterooligomers with distinct physical properties. *Eur. J. Neurosci.* **11**: 41-50
- Everitt, A. B., Luu, T., Cromer, B., Tierney, M. L., Birnir, B., Olsen, R. W. and Gage, P. W. (2004). Conductance of recombinant GABA<sub>A</sub> channels is increased in cells co-expressing GABA<sub>A</sub> receptor-associated protein. *J. Biol. Chem.* 279: 21701-6
- Farrant, M. and Nusser, Z. (2005). Variations on an inhibitory theme: phasic and tonic activation of GABA<sub>A</sub> receptors. *Nat. Rev. Neurosci.* **6**: 215-29
- Farries, M., Ding, L. and Perkel, D. J. (2005). Evidence for "direct" and "indirect" pathways through the song system basal ganglia. *J. Comp. Neurol.* **484:** 93-104
- Farries, M. A. and Perkel, D. J. (2000). Electrophysiological properties of avian basal ganglia neurons recorded *in vitro*. J. Neurophysiol. 84: 2502-13
- Farries, M. A. and Perkel, D. J. (2002). A telencephalic nucleus essential for song learning contains neurons with physiological characteristics of both striatum and globus pallidus. J. Neurosci. 22: 3776-87
- Fatemi, S. H., Reutiman, T. J., Folsom, T. D. and Thuras, P. D. (2009). GABA(A) receptor downregulation in brains of subjects with autism. J. Autism Dev. Disord. 39: 223-30
- Fehér, O., Wang, H., Saar, S., Mitra, P. P. and Tchernichovski, O. (2009). *De novo* establishment of wild-type song culture in the zebra finch. *Nature* **459**: 564-8
- Feigenspan, A., Wässle, H. and Bormann, J. (1993). Pharmacology of GABA receptor Cl<sup>-</sup> channels in rat retinal bipolar cells. *Nature* **361:** 159-62
- Fire, A., Xu, S., Montgomery, M., Kostas, S., Driver, S. and Mello, C. (1998). Potent and specific genetic interference by double-stranded RNA in *Caenorhabditis elegans*. *Nature* **391**: 806-11
- Fitch, W. T. (2009). Birdsong normalized by culture. *Nature* **459**: 519-20
- Forster, I. C., Harvey, R. J., Darlison, M. G. and Benson, J. A. (2001). Functional pharmacology of  $GABA_A$  receptors containing the chicken brain  $\gamma 4$  subunit. *Eur. J. Pharmacol.* **419:** 1–7
- Fountaine, T. M., Wood, M. J. and Wade-Martins, R. (2005). Delivering RNA interference to the mammalian brain. *Curr. Gene. Ther.* **5:** 399-410
- Franks, N. P. (2006). Molecular targets underlying general anaesthesia. *Br. J. Pharmacol.* **147:** 72-81
- Fritschy, J. M. (2008). Is my antibody-staining specific? How to deal with pitfalls of immunohistochemistry. *Eur. J. Neurosci* 28: 2365-70

- Fritschy, J. M., Benke, D., Mertens, S., Oertel, W. H., Bachi, T. and Möhler, H. (1992). Five subunits of type A γ-aminobutyric acid receptors identified in neurons by double and triple innumofluorescence staining with subunit-specific antibodies. *Proc. Natl. Acad. Sci. USA* **89**: 7168-79
- Fritschy, J. M. and Möhler, H. (1995). GABA<sub>A</sub> receptor heterogeneity in the adult rat brain: differential regional and cellular distribution of seven major subunits. J. Comp. Neurol. 359: 154-94
- Fromkin, V., Krashen, S., Curtiss, S., Rigler, D. and Rigler, M. (1974). The development of language in genie: a case of language acquisition beyond the "critical period". *Brain Lang.* **1**: 81-107
- Fu, D-. X. and Sine, S. M. (1996). Assymmetrical contribution of the conserved disulfide loop to subunit oligomerization and assembly of the nicotinic acetylcholine receptor. J. Biol. Chem. 271: 31479-84
- Fuchs, K. and Celepirovic, N. (2002). The 5'-flanking region of the rat GABA<sub>A</sub> receptor α2-subunit gene (*Gabra2*). J. Neurochem. **82:** 1512-23
- Fujii, E. and Mellon, S. H. (2001). Regulation of uterine γ-aminobutyric acid<sub>A</sub> receptor subunit expression throughout pregnancy. *Endocrinol.* **142**: 1770-7
- Fujii, M., Arata, A., Kanbara-Kume, N., Saito, K., Yanagawa, Y. and Obata, K. (2007). Respiratory activity in brainstem of fetal mice lacking glutamate decarboxylase 65/67 and vesicular GABA transporter. *Neurosci.* 146: 1044-52
- Fukushima, Y., Takagi, H. and Aoki, K. (2002). Developmental difference of neural functions in the song and call control system in Bengalese finches. *Neurosci. Lett.* 334: 2140-4
- Funabiki, Y. and Funabiki, K. (2007). Song retuning with tutor model by adult zebra finches. *Dev. Neurobiol.* **68:** 645-55
- Funabiki, Y. and Konishi, M. (2003). Long memory in song learning by zebra finches. J. Neurosci. 23: 6928-35
- Gahr, M. (2000). Neural song control system of hummingbirds: Comparison to swifts, vocal learning (Songbirds) and nonlearning (Suboscines) passerines, and vocal learning (Budgerigars) and nonlearning (Dove, owl, gull, quail, chicken) nonpasserines. J. Comp. Neurol. 426: 182-96
- Gahr, M. and Metzdorf, R. (1999). The sexually dimorphic expression of androgen receptors in the song nucleus hyperstriatalis ventrale pars caudale of the zebra finch develops independently of gonadal steroids. *J. Neurosci.* **19**: 2628-36
- Ganguly, K., Schinder, A. F., Wong, S. T. and Poo, M. (2001). GABA itself promotes the developmental switch of neuronal GABAergic responses for excitation to inhibition. *Cell* **105:** 521-32
- Garret, M., Bascles, L., Boué-Grabot, E., Sartor, P., Charron, G., Bloch, B., and Margolskee, R. F. (1997). An mRNA encoding a putative GABA-gated chloride channel is expressed in the human cardiac conduction system. J. Neurochem. 68: 1382-9
- Genc, S., Koroglu, T. F. and Genc, K. (2004). RNA interference in neuroscience. *Mol. Brain Res.* **132:** 260-70
- Gibbs, M. E. and Johnston, G. A. R. (2005). Opposing roles for GABA<sub>A</sub> and GABA<sub>C</sub> receptor in short-term memory formation in young chicks. *Neurosci.* **131:** 567-76
- Giménez, I. (2006). Molecular mechanisms and regulation of furosemide-sensitive Na-K-Cl cotransporters. *Curr. Opin. Nephrol. Hypertens.* **15:** 517-23
- Glykys, J., Mann, E. O. and Mody, I. M. (2008). Which GABA<sub>A</sub> receptor subunits are necessary for tonic inhibition in the hippocampus? *J. Neurosci.* **28**: 1421-6

- Goller, F. and Daley, M. A. (2001). Novel gestures for phonation during inspiration enhance the acoustic complexity of birdsong. *Proc. R. Soc. Lond. Ser. B* 268: 2301-5
- Gorrie, G. H., Vallis, Y., Stephenson, A., Whitfield, J., Browning, B., Smart, T. G. and Moss, S. J. (1997). Assembly of GABA<sub>A</sub> receptors composed of  $\alpha 1$  and  $\beta 2$  subunits in both cultured neurons and fibroblasts. *J. Neurosci.* **17**: 6587-96
- Gou, D., Jin, N. and Liu, L. (2003). Gene silencing in mammalian cells by PCR-based short hairpin RNA. *FEBS Lett.* **548:** 113-8
- Greenbaum, D., Colangelo, C., Williams, K. and Gerstein, M. (2003). Comparing protein abundance and mRNA expression levels on a genomic scale. *Genome Biol.* **4:** 117 (online)
- Grutter, T., De Carvalho, L. P., Dufresne, V., Taly, A., Edelstein, S. J. and Changeux, J. P. (2005). Molecular tuning of fast gating in pentameric ligand-gated ion channels. *Proc. Natl. Acad. Sci. USA* **102**: 18207-12
- Guignet, E. G. and Meyer, T. (2008). Suspended-drop electroporation for high-throughput delivery of biomolecules into cells. *Nat. Methods* **5**: 393-5
- Günther, U., Benson, J., Benke, D., Fritschy, J. M., Reyes, G., Knoflach, F., Crestani, F., Aguzzi, A., Arigoni, M., Lang, Y., Bluethmann, H., Mohler, H. and Luscher, B. (1995). Benzodiazepine-insensitive mice generated by targeted disruption of the γ2 subunit gene of γ-aminobutyric acid type A receptors. *Proc. Natl. Acad. Sci. USA* 92: 7749-53
- Guzowski, J. F. (2002). Insights into immediate-early gene function in hippocampal memory consolidation using antisense oligonucleotide and fluorescent imaging approaches. *Hippocampus* **12**: 86-104
- Hablitz, J. J., Mathew, S. S. and Pozzo-Miller, L. (2009). GABA vesicles at synapses: are there 2 distinct pools? *Neuroscientist* **15**: 218-24
- Haesler, S., Rochefort, C., Georgi, B., Licznerski, P., Osten, P. and Scharf, C. (2007). Incomplete and inaccurate vocal imitation after knockdown of *FoxP2* in songbird basal ganglia nucleus Area X. *PLoS ONE* 5: 2885-97
- Haesler, S., Wada, K., Nshdejan, A., Morrisey, E. E., Lints, T., Jarvis, E. D. and Scharff, C. (2004). *FoxP2* expression in avian vocal learners and non-learners. J. *Neurosci.* 24: 3164-75
- Hannon, G. J. (2002). RNA interference. Nature. 418: 244-51
- Harding, C. F. (2004). Hormonal modulation of singing: hormonal modulation of the songbird brain and singing behaviour. *Ann. N. Y. Acad. Sci.* **1016**: 524-39
- Harvey, R. J., Chinchetru, M. A. and Darlison, M. G. (1994). Alternative splicing of a 51-nucleotide exon that encodes a putative protein kinase C phosphorylation site generates two forms of the chicken  $\gamma$ -aminobutyric Acid<sub>A</sub> receptor  $\beta$ 2 subunit. *J. Neurochem.* **62:** 10-6
- Harvey, R. J., Kim, H-. C. and Darlison, M. G. (1993). Molecular cloning reveals the existence of a fourth  $\gamma$  subunit of the vertebrate brain GABA<sub>A</sub> receptor. *FEBS Lett.* **331:** 211-6
- Harvey, R. J., McCabe, B. J., Solomonia, R. O., Horn, G. and Darlison, M. G. (1998).
  Expression of the GABA<sub>A</sub> receptor γ4-subunit gene: anatomical distribution of the corresponding mRNA in the domestic chick forebrain and the effect of imprinting training. *Eur. J. Neurosci.* 10: 3024-8
- Hatzimanikatis, V., Choe, L. H. and Lee, K. H. (1999). Proteomics: theoretical and experimental considerations. *Biotech. Prog.* **15**: 312-8
- Hatzimanikatis, V. and Lee, K. H. (1999). Dynamical analysis of gene networks requires both mRNA and protein expression information. *Metab. Eng.* **1**: 275-81

- Hayakawa, K., Kimura, M. and Kamata, K. (2002). Mechanism underlying γaminobutyric acid-induced antihypertensive effect in spontaneously hypertensive rats. *Eur. J. Pharmacol.* **438**: 107-13
- He, L., Thomson, J. M., Hemann, M. T., Hernando-Monge, E., Mu, D., Goodson, S., Powers, S., Cordon-Cardo, C., Lowe, S. W., Hannon, G. J. and Hammond, S. M. (2005). A microRNA polycistron as a potential human oncogene. *Nature* 435: 828-33
- Hedblom, E. and Kirkness, E. F. (1997). A novel class of GABA<sub>A</sub> receptor subunit in tissues of the reproductive system. *J. Biol. Chem.* **272:** 15346-50
- Heinrich, J. E., Nordeen, K. W. and Nordeen, E. J. (2005). Dissociation between extension of the sensitive period for avian vocal learning and dendritic spine loss in the song nucleus IMAN. *Neurobiol. Learn. Mem.* **83:** 143-50
- Herlenius, E. and Lagercrantz, H. (2004). Development of neurotransmitter systems during crucial periods. *Exp. Neurol.* **190:** 8-21
- Hessler, N. A. and Doupe, A. J. (1999). Social context modulates singing-related neural activity in the songbird forebrain. *Nat. Neurosci.* **2:** 209-11
- Hilf, R. J. and Dutzler, R. (2008). X-ray structure of a prokaryotic pentameric ligandgated ion channel. *Nature* **452:** 375-9
- Holen, T., Amarzguioui, M., Wiger, M. T., Babaie, E. and Prydz, H. (2002). Positional effects of short interfering RNAs targeting the human coagulation trigger tissue factor. *Nucl. Acids Res.* **30**: 1757-66
- Hommel, J. D., Sears, R. M., Georgescu, D., Simmons, D. L. and DiLeone, R. J. (2003). Local gene knockdown in the brain using viral-mediated RNA interference. *Nat. Med.* 9: 1539-43
- Horenstein, J., Wagner, D. A., Czajkowski, C. and Akabas, M. H. (2001). Protein mobility and GABA-induced conformational changes in GABA<sub>A</sub> receptor porelining M2 segment. *Nat. Neurosci.* 4: 477-85
- Horn, G. (1998). Visual imprinting and the neural mechanisms of recognition memory. *Trends Neurosci.* **21:** 300-6
- Horn, G., McCabe, B. J. and Bateson, P. P. G. (1979). An autoradiographic study of the chick brain after imprinting. *Brain Res.* 168: 361-73
- Hosie, A. M., Clarke, L., Da Silva, H. and Smart, T. G. (2009). Conserved site for neurosteroid modulation of GABA<sub>A</sub> receptors. *Neuropharmacol.* **56**: 149-54
- Hosie, A. M., Dunne, E. L., Harvey, R. J. and Smart, T. G. (2003). Zinc-mediated inhibition of GABA<sub>A</sub> receptors: discrete binding sites underlie subtype specificity. *Nat. Neurosci.* **6**: 362-9
- Hosie, A. M., Wilkins, M. E., Da Silva, H. M. A. and Smart, T. G. (2006). Endogenous neurosteroids regulate GABA<sub>A</sub> receptors through two discrete transmembrane sites. *Nature* **444**: 486-9
- Hosie, A. M., Wilkins, M. E. and Smart, T. G. (2007). Neurosteroid binding sites on GABA<sub>A</sub> receptors. *Pharmacol. Ther.* **116**: 7-19
- Houtman, E. (1992). Female zebra finches choose extra pair copulations with genetically attractive males. *Proc. R. Soc. Lond. Ser. B Biol. Sci.* **249:** 3-6
- Houx, B. B. and Ten-Cate, C. (1999). Do stimulus–stimulus contingencies affect song learning in zebra finches (*Taeniopygia guttata*)? J. Comp. Psych. **113**: 235-42
- Huntsman, M. M., Munoz, A. and Jones, E. G. (1999). Temporal modulation of GABA(A) receptor subunit gene expression in developing monkey cerebral cortex, *Neurosci.* **91**: 1223-45
- Huntsman, M. M., Tran, B. V., Potkin, S. G., Bunney, W. E. G. and Jones, E. G. (1998). Altered ratios of alternatively spliced long and short γ2 subunit mRNAs of the γ-

amino butyrate type A receptor in prefrontal cortex of schizophrenics. *Proc. Natl. Acad. Sci.* USA **95:** 15066-71

- Immelmann. K. (1969) Song development in zebra finch and other Estrildid finches. In Hinde, R. A. (Ed.) *Bird vocalisations* (pp 61-74) London: Cambridge University Press
- Imoto, K., Methfessel, C., Sakmann, B., Mishina, M., Mori, Y., Konno, T., Fukuda, K., Kurasaki, M., Bujo, H., Fujita, Y. and Numa, S. (1986). Location of a deltasubunit region determining ion transport through the acetylcholine receptor channel. *Nature* **324**: 670-4
- Jackson, A. L., Bartz, S. R., Schelter, J., Kobayashi, S. V., Burchard, J., Mao, M., Li, B., Cavet, G. and Linsley, P. S. (2003). Expression profiling reveals off-target gene regulation by RNAi. *Nat. Biotechnol.* 21: 635-7
- Jackson, A. L. and Linsley, P. S. (2004). Noise amidst the silence: off-target effects of siRNAs? *Trends Genet*. **20:** 521-4
- Jackson, M. B., Lecar, H., Mathers, D. A. and Barker, J. L. (1982). Single channel currents activated by gamma-aminobutyric acid, muscimol, and (–)-pentobarbital in cultured mouse spinal neurons. *J. Neurosci.* **2:** 889-94
- Jacob, T. C., Moss, S. J. and Jurd, R. (2008). GABA<sub>A</sub> receptor trafficking and its role in the dynamic modulation of neuronal inhibition. *Nat. Neurosci.* **9**: 331-43
- Jacobs, E. C., Arnold, A. P. and Campagnoni, A. T. (1999). Developmental regulation of the distribution of aromatase- and estrogen-receptor- mRNA-expressing cells in the zebra finch brain. *Dev. Neurosci.* **21:** 453-72
- Jansen, A., Hoepfner, M., Herzig, K-. H., Riecken, E-. O. and Scherübl, H. (2000). GABA<sub>C</sub> receptors in neuroendocrine gut cells: a new GABA-binding site in the gut. *Eur. J. Physiol.* 441: 294-300
- Jarvis, E. D., Güntürkün, O., Bruce, L., Csillag, A., Karten, H., Kuenzel, W., Medina, L., Paxinos, G., Perkel, D. J., Shimizu, T., Striedter, G., Wild, J. M., Ball, G. F., Dugas-Ford, J., Durand, S. E., Hough, G. E., Husband, S., Kubikova, L., Lee, D. W., Mello, C. V., Powers, A., Siang, C., Smulders, T. V., Wada, K., White, S. A., Yamamoto, K., Yu, J., Reiner, A. and Butler, A. B. (2005). Avian brains and a new understanding of vertebrate evolution. *Nat. Rev. Neurosci.* 6: 151-9
- Jarvis, E. D. and Nottebohm, F. (1997). Motor-driven gene expression. *Proc. Natl.* Acad. Sci. USA 94: 4097-102
- Jechlinger, M., Pelz, R., Tretter, V., Klausberger, T. and Sieghart, W. (1998). Subunit composition and quantitative importance of hetero-oligomeric receptors: GABA<sub>A</sub> receptors containing α6 subunits. *J. Neurosci.* **18**: 2449-57
- Jensen, M. L., Timmermann, D. B., Johansen, T. H., Schousboe, A., Varming, T. and Ahring, P. K. (2002). The  $\beta$  subunit determines the ion selectivity of the GABA<sub>A</sub> receptor. *J. Biol. Chem.* **277:** 41438-47
- Jin, N., Kolliputi, N., Gou, D., Weng, T. and Liu, L. (2006). A novel function of ionotropic γ-aminobutyric acid receptors involving alveolar fluid homeostasis. J. *Biol. Chem.* 281: 36012-20
- Jin, N., Narasaraju, T., Kolliputi, N., Chen, J. and Liu, L. (2005). Differential expression of GABA<sub>A</sub> receptor  $\pi$  subunit in cultured rat alveolar epithelial cells. *Cell Tissue Res.* **321:** 173-83
- Jin, P., Zhang, J., Rowe-Teeter, C., Yang, J., Stuve, L. L. and Fu, G. K. (2004). Cloning and characterization of a GABA<sub>A</sub> receptor  $\gamma 2$  subunit variant. *J. Biol. Chem.* **279**: 1408-14
- Johnson, F. and Whitney, O. (2005). Singing-driven gene expression in the developing songbird brain. *Physiol. Behav.* **86:** 390-8

- Johnson, S. K. and Haun, R. S. (2005). The gamma-aminobutyric acid A receptor  $\pi$  subunit is overexpressed in pancreatic adenocarcinomas. *J. Pancreas* **6**: 136-42
- Johnston, G. A. R. (1996). GABA<sub>C</sub> receptors: relatively simple transmitter-gated ion channels? *Trends Pharmacol. Sci.* **17**: 319-23
- Jones, A., Korpi, E. R., McKernan, R. M., Pelz, R., Nusser, Z., Mäkelä, R., Mellor, J. R., Pollard, S., Bahn, S., Stephenson, F. A., Randall, A. D., Sieghart, W., Somogyi, P., Smith, A. J. H. and Wisden, W. (1997). Ligand-gated ion channel subunit partnerships: GABA<sub>A</sub> receptor α6 subunit gene inactivation inhibits δ subunit expression. *J. Neurosci.* **17**: 1350-62
- Jones, A. E., Ten-Cate, C. and Slater, P. J. B. (1996). Early experience and plasticity of song in adult male zebra finches. *J. Comp. Psychol.* **110**: 354-69
- Jones, A. K. and Sattelle, D. B. (2008). The cys-loop ligand-gated ion channel gene superfamily of the nematode, *Caenorhabditis elegans. Invert. Neurosci.* 8: 41-7
- Jones, B. L. and Henderson, L. P. (2007). Trafficking and potential assembly patterns of ε-containing GABA<sub>A</sub> receptors. *J. Neurochem.* **103**: 1258-71
- Joyce, C. J. (2007). *In silico* comparative genomic analysis of GABA<sub>A</sub> receptor transcriptional regulation. *BMC Genomics* 8: 203 (online)
- Judge, A. D., Sood, V., Shaw, J. R., Fang, D., McClintock, K. and MacLachlan, I. (2005). Sequence-dependent stimulation of the mammalian innate immune response by synthetic siRNA. *Nat. Biotechnol.* 23: 457-62
- Kahle, K. T., Rinehart, J., de Los Heros, P., Louvi, A., Meade, P., Vazquez, N., Hebert, S. C., Gamba, G., Gimenez, I. and Lifton, R. P. (2005). WNK3 modulates transport of Cl<sup>-</sup> in and out of cells: implications for control of cell volume and neuronal excitability. *Proc. Natl. Acad. Sci.* **102:** 16783-8
- Kalueff, A. V. (2007). Neurobiology of memory and anxiety: from genes to behaviour. *Neural Plast.* **2007:** 78171 (online)
- Kang, Q. Z., Duan, G. C., Fan, Q. T. and Xi, Y. L. (2005). Fusion expression of *Helicobacter pylori* neutrophil-activating protein in *E.coli*. World J. Gastroenterol. 11: 454-6
- Kao, M. H., Doupe, A. J. and Brainard, M. S. (2005). Contributions of an avian basal ganglia-forebrain circuit to real-time modulation of song. *Nature* **433**: 638-43
- Karle, J. (2002). Antisense studies of brain GABA<sub>A</sub> receptors. *Dan. Med. Bull.* **49:** 130-44
- Karlin, A. (2002). Emerging structure of the nicotinic acetylcholine receptors. *Nat. Rev. Neurosci.* **3:** 102-14
- Karlin, A. and Akabas, M. H. (1998). Substituted-cysteine accessibility method. *Methods Enzymol.* **293:** 123-45
- Kaufman, D. L., Houser, C. R. and Tobin, A. J. (1991). Two forms of the gammaaminobutyric acid synthetic enzyme glutamate decarboxylase have distinct intraneuronal distributions and cofactor interactions. *J. Neurochem.* **56**: 720-3
- Keller, C. A., Yuan, X., Panzanelli, P., Martin, M. L., Alldred, M., Sassoe-Pognetto M. and Luscher, B. (2004). The gamma2 subunit of GABA<sub>A</sub> receptors is a substrate for palmitoylation by GODZ. *J. Neurosci.* **24**: 5881-91
- Khan, Z. U., Guttiérrez, A., De Blas, A. L. (1994). The subunit composition of a GABA<sub>A</sub>/benzodiazepine receptor from rat cerebellum. *J. Neurochem.* **63**: 371-4
- Kim, D-. H., Behlke, M. A., Rose, S. D., Chang, M. -S., Choi, S. and Rossi, J. J. (2004). Synthetic dsRNA Dicer substrates enhance RNAi potency and efficacy. *Nat. Biotechnol.* 23: 222-6
- Kim, D. H. and Rossi, J. J. (2007). Strategies for silencing human disease using RNA interference. *Nat. Rev. Gen.* 8: 173-84

- Kim, D. H. and Rossi, J. J. (2008). RNAi mechanisms and applications. *Biotechniques* **44:** 613-6
- Kim, V. N. (2005). MicroRNA biogenesis: Coordinated cropping and dicing. Nat. Rev. Mol. Cell. Biol. 6: 376-85
- Kim, W. J., Christensen, L. V., Jo, S., Yockman, J. W., Jeong, J. H., Kim, Y. H. and Kim, S. W. (2006). Cholesteryl oligoarginine delivering vascular endothelial growth factor siRNA effectively inhibits tumour growth in colon adenocarcinoma. *Mol. Ther.* 14: 343-50
- Kim, Y., Glatt, H., Xie, W., Sinnett, D. and Lalande, M. (1997). Human gammaaminobutyric acid-type A receptor α5 subunit gene (GABRA5): characterization and structural organization of the 5' flanking region. *Genomics* **42**: 378-87
- Kimpo, R. R. and Doupe, A. J. (1997). FOS is induced by singing in distinct neuronal populations in a motor network. *Neuron* **18**: 315-25
- King, A. P. and West, M. J. (1983). Epigenesis of cowbird song A joint endeavour of males and females. *Nature* 305: 704-6
- Kirkness, E. F. and Fraser, C. M. (1993). A strong promoter element is located between alternative exons of a gene encoding the human gamma-aminobutyric acid-type A receptor β3 subunit (GABRB3). *J. Biol. Chem.* **268**: 4420-8
- Kirn, J. R. (2009). The relationship of neurogenesis and growth of brain regions to song learning. *Brain Lang*. (Epub ahead of print)
- Kittler, J. T., Delmas, P., Jovanovic, J. N., Brown, D. A., Smart, T. G. and Moss, S. J. (2000). Constitutive endocytosis of GABA<sub>A</sub> receptors by an association with the adaptin AP2 complex modulates inhibitory synaptic currents in hippocampal neurons. J. Neurosci. 20: 7972-7
- Kittler, J. T., McAinsh, K. and Moss, S. J. (2002). Mechanisms of GABA<sub>A</sub> receptor assembly and trafficking: implications for the modulation of inhibitory neurotransmission. *Mol. Neurobiol.* **26**: 251-68
- Kittler, J. T., Rostaing, P., Schiavo, G., Fritschy, J. M., Olsen, R., Triller, A. and Moss, S. J. (2001). The subcellular distribution of GABARAP and its ability to interact with NSF suggest a role for this protein in the intracellular transport of GABA(A) receptors. *Mol. Cell. Neurosci.* 18: 13-25
- Kittler, J. T., Thomas, P., Tretter, V., Bogdanov, Y. D., Haucke, V., Smart, T. G. and Moss, S. J. (2004). Huntingtin-associated protein 1 regulates inhibitory synaptic transmission by modulating γ-aminobutyric acid type A receptor membrane trafficking. *Proc. Natl. Acad. Sci. USA* **101**: 12736-41
- Kleingoor, C., Wieland, H. A., Korpi, E. R., Seeburg, P. H. and Kettenmann, H. (1993). Current potentiation by diazepam but not GABA sensitivity is determined by a single histidine residue. *Neurorep.* **4:** 187-90
- Kloosterman, W. P., Wienholds, E., Ketting, R. F., and Plasterk, R. H. (2004). Substrate requirements for let-7 function in the developing zebrafish embryo. *Nucl. Acids Res.* **32:** 6284-91
- Kofuji, P., Wang, J. B., Moss, S. J., Huganir, R. L. and Burt, D. R. (1991). Generation of two forms of the gamma-aminobutyric acid<sub>A</sub> receptor gamma 2-subunit in mice by alternative splicing. *J. Neurochem.* **56**: 713-5
- Kojima, S. and Doupe, A. J. (2007). Song selectivity in the pallial-basal ganglia song circuit of zebra finches raised without tutor song exposure. J. Neurophysiol. 98: 2099-109
- Konishi, M. (1964). Effects of deafening on song development in two species of Juncos. *Condor* **66:** 85-102

- Konishi, M. (1965). The role of auditory feedback in the control of vocalisation in white-crowned sparrows. Z. *Tierpsychol.* **2**: 770-83
- Konishi, M. (1985). Birdsong: from behaviour to neuron. Ann. Rev. Neurosci. 8: 125-70
- Konishi, M. (2004). The role of auditory feedback in birdsong. Ann. N. Y. Acad. Sci. 1016: 463-75
- Konishi, M. and Akutagawa, E. (1985). Neuronal growth, atrophy and death in a sexually dimorphic song nucleus in the zebra finch brain. *Nature* **315**: 145-7
- Korpi, E. R., Gründer, G. and Lüddens, H. (2002). Drug interactions at GABA<sub>A</sub> receptors. *Prog. Neurobiol.* **67:** 113-59
- Korpi, E. R., Kuner, T., Kristo, P., Köhler, M., Herb, A., Lüddens, H. and Seeburg, P. H. (1994). Small N-terminal deletion by splicing in cerebellar α6 subunit abolishes GABA<sub>A</sub> receptor function. *J. Neurochem.* 63: 1167-70
- Korsia, S. and Bottjer, S. W. (1991). Chronic testosterone treatment impairs vocal learning in male zebra finches during a restricted period of development. J. Neurosci. 11: 2362-71
- Krichevsky, A. M. and Kosik, K. S. (2002). RNAi functions in cultured mammalian neurons. Proc. Natl. Aacd. Sci. 99: 11926-9
- Krystal, J. H., Staley, J., Mason, G., Petrakis, I. L., Kaufman, J., Harris, R. A., Gelernter, J. and Lappalainen, J. (2006). γ-aminobutyric acid type A receptors and alcoholism: intoxication, dependence, vulnerability, and treatment. *Arch Gen Psychiatry*. 63: 957-68
- Kucken, A. M., Teissére, J. A., Seffinga-Clark, J., Wagner, D. A. and Czajkowski, C. (2003). Structural requirements for imidazobenzodiazepine binding to GABA<sub>A</sub> receptors. *Mol. Pharmacol.* 63: 289–96
- Kuhl, P. K. (2004). Early language acquisition: cracking the speech code. *Nature Rev. Neurosci.* **5:** 831-43
- Kultas-Ilinsky, K., Leontiev, V. and Whiting, P. J. (1998). Expression of 10 GABA(A) receptor subunit messenger RNAs in the motor-related thalamic nuclei and basal ganglia of *Macaca mulatta* studied with *in situ* hybridization histochemistry. *Neurosci.* 85: 179-204
- Kumar, A. (2008). RNA interference: a multifaceted antiviral defense. *Retrovirol.* 5: 17
- Kuner, T. and Augustine, G. J. (2000). A genetically encoded ratiometric indicator for chloride: capturing chloride transients in cultured hippocampal neurons. *Neuron* 27: 447-59
- Kunath, T., Gish, G., Lickert, H., Jones, N., Pawson, T. and Rossant, J. (2003). Transgenic RNA interference in ES cell-derived embryos recapitulates a genetic null phenotype. *Nat. Biotechnol.* 21: 559-61
- Lai, C. S. L., Fisher, S. E., Hurst, J. A., Vargha-Khadem, F. and Monaco, A. P. (2001). A forkhead-domain gene is mutated in a severe speech and language disorder. *Nature* 413: 519-23
- Lai, C. S. L., Gerrelli, D., Monaco, A. P., Fisher, S. E. and Copp, A. J. (2003). FOXP2 expression during brain development coincides with sites of pathology in a severe speech and language disorder. Brain 126: 2455-62
- Lane, H. L. (1976). The Wild Boy of Aveyron. Cambridge, MA: Harvard Univ. Press
- Lasham, A., Vreugdenhil, E., Bateson, A. N., Barnard, E. A. and Darlison, M. G. (1991). Conserved organization of  $\gamma$ -aminobutyric acid<sub>A</sub> receptor genes: Cloning and analysis of the chicken  $\beta$ 4-subunit gene. *J. Neurochem.* **57**: 352-5
- Laulay, C., Gerlach, N. M., Adkins-Regan, E. and DeVoogd, T. (2004). Female zebra finches require early song exposure to prefer high-quality song as adults. *Anim. Behav.* 68: 1249-55

- Laurie, D. L., Seeburg, P. H. and Wisden, W. (1992a). The distribution of 13 GABA, receptor subunit mRNAs in the rat brain. II. Olfactory bulb and cerebellum. *J. Neurosci.* **72:** 1063-76
- Laurie, D. J., Wisden, W. and Seeburg, P. H. (1992b). The distribution of thirteen GABA<sub>A</sub> receptor subunit mRNAs in the rat brain. III. Embryonic and postnatal development. *J. Neurosci.* **12**: 4151-72
- Leadbeater, E., Goller, F. and Riebel, K. (2005). Unusual phonation, covarying song characteristics and song preferences in female zebra finches. *Anim. Behav.* **70**: 909-19
- Lee, Y., Ahn, C., Han, J., Choi, H., Kim, J., Yim, J., Lee, J., Provost, P., Rådmark, O., Kim, S. and Kim, V. N. (2003). The nuclear RNase III Drosha initiates microRNA processing. *Nature* 425: 415-9
- Leonardo, A. and Konishi, M. (1999). Decrystallization of adult birdsong by perturbation of auditory feedback. *Nature* **399:** 466-70
- Lemon, B. and Tjian, R. (2000). Orchestrated response: a symphony of transcription factors for gene control. *Genes Dev.* 14: 2551-69
- Lewis, D. A. and Gonzalez-Burgos, G. (2006). Pathophysiologically based treatment interventions in schizophrenia. *Nat. Med.* **12:** 1016-22
- Li, H., Tornberg, J., Kaila, K., Airaksinen, M. S. and Rivera, C. (2002). Patterns of cation-chloride cotransporter expression during embryonic rodent CNS development. *Eur. J. Neurosci.* 16: 2358-70
- Li, K. and Xu, E. (2008). The role and the mechanism of γ-aminobutyric acid during central nervous system development. *Neurosci. Bull.* **24:** 195-200
- Li, M. and De Blas, A. (1997). Coexistence of two  $\beta$  subunit isoforms in the same  $\gamma$ -aminobutyric acid type A receptor. *J. Biol. Chem.* **272:** 16564-9
- Li, P., Bandyopadhyaya, A. K., Covey, D. F., Steinbach, J. H. and Akk, G. (2009). Hydrogen bonding between the 17β-substituent of a neurosteroid and the GABA<sub>A</sub> receptor is not obligatory for channel potentiation. *Br. J. Pharmacol.* **158**: 1322-9
- Li, R. W., Yu, W., Christie, S., Miralles, C. P., Bai, J., LoTurco, J. J. and De Blas, A. L. (2005). Disruption of postsynaptic GABA<sub>A</sub> receptor clusters leads to decreased GABAergic innervation of pyramidal neurons. *J. Neurochem.* **95**: 756-70
- Li, X. C., Jarvis, E. D., Alvarez-Borda, B., Lim, D. A. and Nottebohm, F. (2000). A relationship between behavior, neurotrophin expression, and new neuron survival. *Proc. Natl. Acad. Sci. USA* **97:** 8584-9
- Lin, M. H., Takahashi, M. P., Takahashi, Y. and Tsumoyo, T. (1994). Intracellular calcium increase induced by GABA in visual cortex of fetal and neonatal rats and its disappearance with development. *Neurosci. Res.* **20:** 85-94
- Liu, C. and Reppert, S. M. (2000). GABA synchronizes clock cells within the suprachiasmatic circadian clock. *Neuron* **25**: 123-8
- Liu, C. M., Liu, D. P., Dong, W. J. and Liang, C. C. (2004). Retrovirus vector-mediated stable gene silencing in human cell. *Biochem. Biophy. Res. Commun.* **313**: 716-20
- Livingston, F. S. and Mooney, R. (1997). Development of intrinsic and synaptic properties in a forebrain nucleus essential to avian song learning. *J. Neurosci.* **17**: 8997-9009
- Livingston, F. S., White, S. A. and Mooney, R. (2000). Slow NMDA-EPSCs at synapses critical for song development are not required for song learning in zebra finches. *Nat. Neurosci.* **3**: 482-8
- Loebrich, S., Bahring, R., Katsuno, T., Tsukita, S. and Kneussel, M. (2006). Activated radixin is essential for  $GABA_A$  receptor  $\alpha 5$  subunit anchoring at the actin cytoskeleton. *EMBO J.* **25**: 987-99

- Lombardino, A. J., Li, X. C., Hertel, M. and Nottebohm F. (2005). Replaceable neurons and neurodegenerative disease share depressed UCHL1 levels. *Proc. Natl. Acad. Sci. USA* **102:** 8036-41
- Lombardino, A. J. and Nottebohm, F. (2000). Age at deafening affects the stability of learned song in adult male zebra finches. *J. Neurosci.* **20:** 5054-64
- Long, M. A. and Fee, M. S. (2008). Using temperature to analyse temporal dynamics in the songbird motor pathway. *Nature* **456**: 189-94
- Lovell, P. V., Clayton, D. F., Replogle, K. L. and Mello, C. V. (2008). Birdsong "transcriptomics": neurochemical specialisations of the oscine song system. *PLoS ONE* **3**: 3440 (online)
- Löw, K., Crestani, F., Keist, R., Benke, D., Brünig, I., Benson, J. A., Fritschy, J. M., Rülicke, T., Bluethmann, H., Möhler, H. and Rudolph, U. (2000). Molecular and neuronal substrate for the selective attenuation of anxiety. *Science* 290: 131-4
- Lund, E., Guttinger, S., Calado, A., Dahlberg, J. E., and Kutay, U. (2004). Nuclear export of microRNA precursors. *Science* **303**: 95-8
- Luo, M. and Perkel, D. J. (1999). A GABAergic, strongly inhibitory projection to a thalamic nucleus in the zebra finch song system. *J. Neurosci.* **19:** 6700-11
- Lyons, H. R., Gibbs, T. T. and Farb, D. H. (2000). Turnover and down-regulation of GABA<sub>A</sub> receptor  $\alpha 1$ ,  $\beta 2S$ , and  $\gamma 1$  subunit mRNAs by neurons in culture. *J. Neurochem.* **74:** 1041-8
- MacDermot, K. D., Bonora, E., Sykes, N., Coupe, A-. M., Lai, C. S. L., Vernes, S. C., Vargha-Khadem, F., McKenzie, F., Smith, R. L., Monaco, A. P. and Fisher, S. E. (2005). Identification of FOXP2 truncation as a novel cause of developmental speech and language deficits. *Am. J. Hum. Genet.* **76**: 1074-80
- MacDougall-Shackleton, S. A. and Ball, G. F. (1999). Comparative studies of sex differences in the song-control system of songbirds. *Trends Neurosci.* 22: 432-6
- MacRae, I. J., Zhou, K., Li, F., Repic, A., Brooks, A. N., Cande, W. Z., Adams, P. D. and Doudna, J. A. (2006). Structural basis for double-stranded RNA processing by Dicer. *Science* **311**: 195-8
- Makimura, H., Mizuno, T. M., Mastaitis, J. W., Agami, R. and Mobbs, C. V. (2002).
   Reducing hypothalamic AGRP by RNA interference increases metabolic rate and decreases body weight without influencing food intake. *BMC Neurosci.* 3: 18 (online)
- Marcade, M., Bourdin, J., Loiseau, N., Peillon, H., Rayer, A., Drouin, D., Schweighoffer, F. and Désiré, L. (2008). Etazolate, a neuroprotective drug linking GABA<sub>A</sub> receptor pharmacology to amyloid precursor protein processing. J. Neurochem. 106: 392-404
- Marler, P. (1970). A comparative approach to vocal learning: song development in white-crowned sparrows. J. Comp. Phys. Psych. 71: 1-25
- Marler, P. (1976). Sensory templates in species-specific behavior. In J. Fentress (Ed.) *Simpler Networks and Behavior* (pp 314-29). Sunderland, MA: Sinauer
- Marler, P. (1997). Three models of song learning: Evidence from behaviour. J. Neurobiol. 33: 501-16
- Marler, P. and Peters, S. (1988). The role of song phonology and syntax in vocal learning preferences in the song sparrow, *Melospiza melodia*. *Ethol*. **77**: 125-49
- Marler, P. and Sherman, V. (1983). Song structure without auditory feedback: emendations of the auditory template hypothesis. *J. Neurosci.* **3:** 517-31
- Marler, P. and Sherman, V. (1985). Innate differences in singing behaviour of sparrows reared in isolation from adult conspecific song. *Anim. Behav.* **33:** 57-71

- Marler, P. (2004). Bird calls: a cornucopia for communication. In Marler, P. and H. Slabbekoorn, (Eds.) *Nature's music: the science of birdsong* (pp. 132-77). Amsterdam; Elsevier Academic Press
- Marler, P. and Tamura, M. (1964). Culturally transmitted patterns of vocal behaviour in sparrows. *Science* **146**: 1483-6
- Martin, S., Henley, J. M., Holman, D., Zhou, M., Wiegert, O., van Spronsen, M., Joëls, M., Hoogenraad, C. C. and Krugers, H. J. (2009). Corticosterone alters AMPAR mobility and facilitates bidirectional synaptic plasticity. *PLoS ONE* 4: 4714 (online)
- Martínez-Torres, A., Vazquez, A. E., Panicker, M. M. and Miledi, R. (1998). Cloning and functional expression of alternative spliced variants of the ρ1 γ-aminobutyrate receptor. *Proc. Natl. Acad. Sci. USA* **95:** 4019-22
- Matranga, C., Tomari, Y., Shin, C., Bartel, D. P. and Zamore, P. D. (2005). Passenger strand cleavage facilitates assembly of siRNA into Ago2-containing RNAi enzyme complexes. *Cell* **123**: 607-20
- Maubach, K. (2003). GABA(A) receptor subtype selective cognition enhancers. *Curr. Drug Targets CNS Neurol. Disord.* **2:** 233-9
- McCall, M. A., Lukasiewicz, P. D., Gregg, R. G. and Peachey, N. S. (2002). Elimination of the ρ1 subunit abolishes GABA<sub>C</sub> receptor expression and alters visual processing in the mouse retina. *J. Neurosci.* **22**: 4163-74
- McKernan, R. M. and Whiting, P. J. (1996). Which GABA<sub>A</sub>-receptor subtypes really occur in the brain? *Trends Neurosci.* **19:** 139-43
- McKinley, D. D., Lennon, D. J. and Carter, D. B. (1995). Cloning, sequence analysis and expression of two forms of mRNA coding for the human  $\beta 2$  subunit of the GABA<sub>A</sub> receptor. *Mol. Brain Res.* **28**: 175-9
- McManus, M. T. and Sharp, P. A. (2002). Gene silencing in mammals by small interfering RNAs. *Nat. Rev. Gen.* **3**: 737-47
- McNally, G. P., Augustyn, K. A. and Richardson, R. (2008). GABA<sub>A</sub> receptors determine the temporal dynamics of memory retention. *Learn. Mem.* **15**: 106-11
- Meier, J. and Grantyn, R. (2004). Preferential accumulation of GABA<sub>A</sub> receptor  $\gamma$ 2L, not  $\gamma$ 2S, cytoplasmic loops at rat spinal cord inhibitory synapses. *J. Physiol.* **559**: 355-65
- Mello, C.V. and Clayton, D. F. (1994) Song-induced ZENK gene expression in auditory pathways of songbird brain and its relation to the song control system. J. Neurosci. 14: 6652-66
- Mello, C. V., Vicario, D. S. and Clayton, D. F. (1992). Song presentation induces gene expression in the songbird forebrain. *Proc. Natl. Acad. Sci. USA* **89:** 6818-22
- Ménard, A., Auclair, F., Bourcier-Lucas, C., Grillner, S. and Dubuc, R. (2007). Descending GABAergic projections to the mesencephalic locomotor region in the lamprey *Petromyzon marinus*. J. Comp. Neurol. **501**: 260-73
- Mercado, J. and Czajkowski, C. (2008).  $\gamma$ -Aminobutyric Acid (GABA) and pentobarbital induce different conformational rearrangements in the GABA<sub>A</sub> receptor  $\alpha$ 1 and  $\beta$ 2 pre-M1 regions. *J. Biol. Chem.* **283:** 15250-7
- Mertens, S., Benke, D. and Mohler, H. (1993). GABA<sub>A</sub> receptor populations with novel subunit combinations and drug binding profiles identified in brain by alpha 5- and delta-subunit- specific immunopurification. *J. Biol. Chem.* **268**: 5965-73
- Meyer, G., Kirsch, J., Betz, H. and Langosch, D. (1995). Identification of a gephyrin binding motif on the glycine receptor beta subunit. *Neuron* **15**: 563-72
- Mignot, E., Taheri, S. and Nishino, S. (2002). Sleeping with the hypothalamus: emerging therapeutic targets for sleep disorders. *Nat. Neurosci.* **5:** 1071-5

- Miller, D. B. (1979). Long-term recognition of father's song by female zebra finches. *Nature* **280:** 389-91
- Milligan, C. J., Buckley, N. J., Garret, M., Deuchars, J. and Deuchars, S. A. (2004). Evidence for inhibition mediated by coassembly of GABA<sub>A</sub> and GABA<sub>C</sub> receptor subunits in native central neurons. *J. Neurosci.* **24**: 7241-50
- Minks, M. A., West, D. K., Benvin, S. and Baglioni, C. (1979). Structural requirements of double-stranded RNA for the activation of 2',5'-oligo(A) polymerase and protein kinase of interferon-treated HeLa cells. *J. Biol. Chem.* **254**: 101080-3
- Miralles, C. P., Gutiérrez, A., Khan, Z. U., Vitorica, J. and De Blas, A. L. (1994). Differential expression of the short and long forms of the gamma 2 subunit of the GABA<sub>A</sub>/benzodiazepine receptors. *Brain Res. Mol. Brain Res.* **24:** 129-39
- Mitchell, I. J., Cooper, A. J. and Griffiths, M. R. (1999). The selective vulnerability of striatopallidal neurons. *Prog. Neurobiol.* **59**: 691-719
- Mocellin, S. and Provenzano, M. (2004). RNA interference: learning gene knock-down from cell physiology. J. Transl. Med. 2: 39 (online)
- Mody, I. and Pearce, R. A. (2004). Diversity of inhibitory neurotransmission through GABA<sub>A</sub> receptors. *Trends Neurosci.* 27: 569-75
- Mooney, R. and Konishi, M. (1991). Two distinct inputs to an avian song nucleus activate different glutamatergic receptor subtypes on individual neurons. *Proc. Natl. Acad. Sci. USA* **88:** 4075-9
- Mooney, R. and Prather, J. F. (2005). The HVC microcircuit: the synaptic basis for interactions between song motor and vocal plasticity pathways. *J. Neurosci.* **25**: 1952-64
- Mooney, R. and Rao, M. (1994). Waiting periods versus early innervation: the development of axonal connections in the zebra finch song system. J. Neurosci. 14: 6532-43
- Morrison, R. G. and Nottebohm, F. (1993). Role of a telencephalic nucleus in the delayed song learning of socially isolated zebra finches. *J. Neurobiol.* **8:** 1045-64
- Mossier, B., Togel, M., Fuchs, K. and Sieghart, W. (1994). Immunoaffinity purification of  $\gamma$ -aminobutyric acidA (GABA<sub>A</sub>) receptors containing  $\gamma$ 1-subunits. Evidence for the presence of a single type of gamma-subunit in GABA<sub>A</sub> receptors. *J. Biol. Chem.* **269**: 25777-82
- Mu, W., Cheng, Q., Yang, J. and Burt, D. R. (2002). Alternative splicing of the GABA<sub>A</sub> receptor α4 subunit creates a severely truncated mRNA. *Brain Res. Bull.* 58: 447-54
- Muller, P. Y., Janovjak, H., Miserez, A. R. and Dobbie, Z. (2002) Processing of gene expression data generated by quantitative real-time RT-PCR. *Biotechniques* **32**: 1372–8
- Murata, Y., Woodward, R. M., Miledi, R. and Overman, L. E. (1996). The first selective antagonist for a GABA<sub>C</sub> receptor. *Bioorg. Med. Chem. Lett.* **6:** 2073-6
- Napoli, C., Lemieux, C. and Jorgensen, R. (1990). Introduction of a chimeric chalcone synthase gene into petunia results in reversible co-suppression of homologous genes *in trans. Plant Cell* **2:** 279-89
- Nastiuk, K. L., Mello, C. V., George, J. M. and Clayton, D. F. (1994). Immediate- early gene responses in the avian song control system: cloning and expression analysis of the canary c-jun cDNA. *Mol. Brain Res.* **27**: 299-309
- Nelson, D. A. and Marler, P. (1993). Innate recognition of song in white-crowned sparrows: a role in selective vocal learning? *Anim. Behav.* **46**: 806-8
- Nemeroff, C. B. (2003). Anxiolytics: past, present, and future agents. *J. Clin. Psychiatry* **64:** 3-6

- Nick, T. A. and Konishi, M. (2005). Neural song preference during vocal learning in the zebra finch depends on age and state. *J. Neurobiol.* **62**: 231-42
- Nixdorf-Bergweiler, B. E. (2001). Lateral magnocellular nucleus of the anterior neostriatum (LMAN) in the zebra finch: neuronal connectivity and the emergence of sex differences in cell morphology. *Microsc. Res. Tech.* **54**: 335-53
- Nordeen, K. W. (2006). Neural correlates of sensitive periods in avian song learning. *Ann. N. Y. Acad. Sci.* **807:** 386-400
- Nordeen, K. W. and Nordeen, E. J. (1988). Projection neurons within a vocal motor pathway are born during song learning in zebra finches. *Nature* **334**: 149-51
- Nordeen, K. W. and Nordeen, E. J. (1992). Auditory feedback is necessary for the maintenance of stereotyped song in adult zebra finches. *Behav. Neural Biol.* **57**: 58-66
- Nottebohm, F. (1980). Testosterone triggers growth of brain vocal control nuclei in adult female canaries. *Brain Res.* **189:** 429-36
- Nottebohm, F. (1981). A brain for all seasons: cyclical anatomical changes in song control nuclei of the canary brain. *Science* **214**: 1368-70
- Nottebohm, F. (1989). From bird song to neurogenesis. Sci. Am. 260: 74-9
- Nottebohm, F. and Arnold, A. P. (1976). Sexual dimorphism in vocal control areas of the songbird brain. *Science* **194**: 211-3
- Nottebohm, F., Stokes, T. M. and Leonard, C. M. (1976). Central control of song in the canary, *Serinus canarius. J. Comp. Neurol.* **165**: 457-86
- Nusser, S., Sieghart, W., Stephenson, A. F. and Somogyl, P. (1996). The α6 subunit of the GABA<sub>A</sub> receptor is concentrated in both inhibitory and excitatory synapses on cerebellar granule cells. *J. Neurosci.* **16**: 103-14
- Nykanen, A., Haley, B. and Zamore, P. D. (2001). ATP requirements and small interfering RNA structure in the RNA interference pathway. *Cell* **107**: 309-21
- Ogita, S., Uefuji, H., Yamaguchi, Y., Koizumi, N. and Sano, H. (2003). RNA interference: producing decaffeinated coffee plants. *Nature* **423**: 823
- Ogurusu, T., Yanagi, K., Wanatabe, M., Fukaya, M. and Shingai, R. (1999). Localization of GABA receptor  $\rho 2$  and  $\rho 3$  subunits in rat brain and functional expression of homooligomeric  $\rho 3$  receptors and heterooligomeric  $\rho 2\rho 3$  receptors. *Recept. Channels* **6**: 463-75
- Olsen, P. H. and Ambros, V. (1999). The lin-4 regulatory RNA controls developmental timing in *Caenorhabditis elegans* by blocking LIN-14 protein synthesis after the initiation of translation. *Dev. Biol.* **216:** 671-80
- Olsen, R. W. (1982). Drug interactions at the GABA receptor-ionophore complex. Annu. Rev. Pharmacol. Toxicol. 22: 245-77
- Olsen, R. W., Hanchar, H. J., Meera, P. and Wallner, M. (2007). GABA<sub>A</sub> receptor subtypes: the "one glass of wine" receptors. *Alcohol* **41**: 201-9
- Olsen, R. W. and Sieghart, W. (2008). International Union of Pharmacology. LXX. Subtypes of  $\gamma$ -aminobutyric acid<sub>A</sub> receptors: classification on the basis of subunit composition, pharmacology, and function. Update. *Pharmacol. Rev.* **60**: 243-60
- Olsen, R. W. and Sieghart, W. (2009). GABA<sub>A</sub> receptors: Subtypes provide diversity of function and pharmacology. *Neuropharmacol.* **56:** 141-8
- Ölveczky, B. P., Andalman, A. S. and Fee, M. S. (2005). Vocal experimentation in the juvenile songbird requires a basal ganglia circuit. *PLoS Biol.* **3**: 153 (online)
- Orban, T. I. and Izaurralde, E. (2005). Decay of mRNAs targeted by RISC requires XRN1, the Ski complex, and the exosome. *RNA* **11**: 459-69

- Orchinik, M., Carroll, S. S., Li, Y-. H., McEwen, B. S. and Weiland, N. G. (2001). Heterogeneity of hippocampal GABA<sub>A</sub> receptors: regulation by corticosterone. *J. Neurosci.* **21**: 330-9
- Orchinik, M., Weiland, N. G. and McEwen, B. S. (1995). Chronic exposure to stress levels of corticosterone alters GABA<sub>A</sub> receptor subunit mRNA levels in rat hippocampus. *Mol. Brain Res.* **34**: 29-37
- Paddison, P. K., Caudy, A. A., Bernstein, E., Hannon, G. J. and Conklin, D. S. (2002). Short hairpin RNAs (shRNAs) induce sequence-specific silencing in mammalian cells. *Genes Dev.* 16: 948-58
- Pan, Y. and Qian, H. (2005). Interactions between rho and gamma2 subunits of the GABA receptor. *J. Neurochem.* **94:** 482-90
- Pan, Z. H., Zhang, X. and Lipton, S. A. (2000). Redox modulation of recombinant human GABA<sub>A</sub> receptors. *Neurosci.* **98**: 333-8
- Patel, S. N., Clayton, N. S. and Krebs, J. R. (1997). Hippocampal tissue transplants reverse lesion-induced spatial memory deficits in zebra finches (*Taeniopygia* guttata) J. Neurosci. 17: 3861-9
- Pekarik, V., Bourikas, D., Miglino, N., Joset, P., Preiswerk, S. and Stoeckli, E. T. (2003). Screening for gene function in chicken embryo using RNAi and electroporation. *Nat. Biotechnol.* 21: 93-6
- Peters, J. A., Hales, T. G. and Lambert, J. L. (2005). Molecular determinants of singlechannel conductance and ion selectivity in the Cys-loop family: insights from the 5-HT<sub>3</sub> receptor.*Trends Pharmacol. Sci.* **26**: 587-94
- Phan, M. L., Pytte, C. L. and Vicario, D. S. (2006). Early auditory experience generates long-lasting memories that may subserve vocal learning in songbirds. *Proc. Natl. Acad. Sci. USA* 103: 1088-93
- Pillai, R. S. (2005). MicroRNA function: multiple mechanisms for a tiny RNA? *RNA* **11:** 1753-61
- Pillai, R. S., Bhattacharyya, S. N. and Filipowicz, W. (2007). Repression of protein synthesis by miRNAs: how many mechanisms? *Trends Cell Biol.* **17**: 118-26
- Pinaud, R. and Mello, C. V. (2007). GABA immunoreactivity in auditory and song control brain areas of zebra finches. *J. Chem. Neuroanat.* **34:** 1-21
- Pirker, S., Schwarzer, C., Wieselthaler, A., Sieghart, W. and Sperk, G. (2000). GABA<sub>A</sub> receptors: Immuno-cytochemical distribution of 13 subunits in the adult rat brain. *Neurosci.* 101: 815-50
- Pöltl, A., Hauer, B., Fuchs, K., Tretter, V. and Sieghart, W. (2003). Subunit composition and quantitative importance of GABA<sub>A</sub> receptor subtypes in the cerebellum of mouse and rat. J. Neurochem. 87: 1444-55
- Poopatanapong, A., Teramitsu, I., Byun, J. S., Vician, L. J., Herschman, H. R. and White, S. A. (2006). Singing, but not seizure, induces Synaptotagmin IV in zebra finch song circuit nuclei. J. Neurobiol. 66: 1613-29
- Poulsen, C. F., Christjansen, K. N., Hastrup, S. and Hartvig, L. (2000). Identification and cloning of a  $\gamma 3$  subunit splice variant of the human GABA<sub>A</sub> receptor. *Mol. Brain Res.* **78**: 201-3
- Prather, J. F., Peters, S., Nowicki, S. and Mooney, R. (2008). Precise auditory-vocal mirroring in neurons for learned vocal communication. *Nature* **451**: 305-10
- Preall, J. B. and Sontheimer, E. J. (2005). RNAi: RISC gets loaded. Cell 123: 543-53
- Price, P. H. (1979). Developmental determinants of structure in zebra finch song. J. Comp. Physiol. Psychol. 93: 260-77
- Pritchett, D. B. and Seeburg, P. H. (1990). γ-aminobutyric acid<sub>A</sub> receptor α5-subunit creates novel type I1 benzodiazepine receptor pharmacology. *J. Neurochem.* **54**: 1802-4
- Pritchett, D. B., Sontheimer, H., Shivers, B. D., Ymer, S., Kettenmann, H., Schofield, P. R. and Seeburg, P. H. (1989). Importance of a novel GABA<sub>A</sub> receptor subunit for benzodiazepine pharmacology. *Nature* 338: 582-5
- Pröve, E. (1978). Courtship and testosterone in male zebra finches. Z. Tierpsychol. 48: 47-67
- Purohit, Y. and Grosman, C. (2006). Block of muscle nicotinic receptors by choline suggests that the activation and desensitization gates act as distinct molecular entities. J. Gen. Physiol. 127: 703-17
- Pytte, C. L. and Suthers, R. A. (2000). Sensitive period for sensorimotor integration during vocal motor learning. *J. Neurobiol.* **42:** 172-89
- Qian, H. and Dowling, J. E. (1993). Novel GABA responses from rod-driven retinal horizontal cells. *Nature* **361**: 162-4
- Qian, H. and Pan, Y. (2002). Co-assembly of GABA  $\rho$  subunits with the GABA<sub>A</sub> receptor  $\gamma$ 2 subunit cloned from white perch retina. *Mol. Brain Res.* **103**: 62-70.
- Que-Gewirth, N. S. and Sullenger, B. A. (2007). Gene therapy progress and prospects: RNA aptamers. *Gene Ther.* **14**: 283-91
- Quirarte, G. L., Sofía Ledesma de la Teja, I., Casillas, M., Serafín, N., Prado-Alcalá, R. A. and Roozendaal, B. (2009). Corticosterone infused into the dorsal striatum selectively enhances memory consolidation of cued water-maze training. *Learn. Mem.* 16: 586-9
- Quirk, K., Gillard, N. P., Ragan, C. I., Whiting, P. J. and McKernan, R. M. (1994a).  $\gamma$ -Aminobutyric acid type A receptors in the rat brain can contain both  $\gamma 2$  and  $\gamma 3$ subunits, but  $\gamma 1$  does not exist in combination with another  $\gamma$ -subunit. *Mol. Pharmacol.* **45:** 1061-70
- Quirk, K., Gillard, N. P., Ragan, C. I., Whiting, P. J. and McKernan, R. M. (1994b). Model of subunit composition of  $\gamma$ -aminobutyric acid<sub>A</sub> receptor subtypes expressed in rat cerebellum with respect to their  $\alpha$  and  $\gamma/\delta$  subunits. *J. Biol. Chem.* **269:** 16020-28
- Rand, T. A., Petersen, S., Du. F. and Wang, X. (2005). Argonaute2 cleaves the antiguide strand of siRNA during RISC activation. *Cell* **123**: 621-9
- Raoul, C., Abbas-Terki, T., Bensadoun, J-. C., Guillot, S., Haase, G., Szulc, J., Henderson, C. E. and Aebischer, P. (2005). Lentiviral-mediated silencing of *SOD1* through RNA interference retards disease onset and progression in a mouse model of ALS. *Nat. Med.* **11**: 423-8
- Rewal, M., Jurd, R., Gill, T. M., He, D-. Y., Ron, D. and Janak, P. H. (2009). α4containing GABA<sub>A</sub> receptors in the Nucleus accumbens mediate moderate intake of alcohol. *J. Neurosci.* **29:** 543-9
- Reynolds, A., Anderson, E. M., Vermeulen, A., Fedorov, Y., Robinson, K., Leake, D., Karpilow, J., Marshall, W. S. and Khvorova, A. (2006). Induction of the interferon response by siRNA is cell type- and duplex length-dependent. *RNA* 12: 988-93
- Reynolds, A., Leake, D., Boese, Q., Scaringe, S., Marshall, W. S. and Khvorova, A. (2004). Rational siRNA design for RNA interference. *Nat. Biotech.* **22:** 326-30
- Reynolds, D. S. (2008). The value of genetic and pharmacological approaches to understanding the complexities of GABA<sub>A</sub> receptor subtype functions: The anxiolytic effects of benzodiazepines. *Pharmacol. Biochem. and Behav.* **90**: 37-42
- Rhodes, K. J. and Trimmer, J. S. (2006). Antibodies as valuable neuroscience research tools versus reagents of mass distraction. J. Neurosci. 26: 8017-20

- Riebel, K. (2003). The "mute" sex revisited; vocal production and perception learning in female songbirds. *Adv. Stud. Behav.* **33**: 49-86
- Riebel, K., Smallegange, I. M., Terpstra, N. J. and Bolhuis, J. J. (2002). Sexual equality in zebra finch song preference: evidence for a dissociation between song recognition and production learning. *Proc. R. Soc. Lond. Ser. B Biol. Sci.* 269: 729-33
- Roberts, E. and Frankel, S. (1950). γ-aminobutyric acid in brain. Fed. Proc. 9: 219
- Romano, N. and Macino, G. (1992). Quelling: transient inactivation of gene expression in *Neurospora crassa* by transformation with homologous sequences. *Mol. Microbiol.* 6: 3343-53
- Rosen, M. J. and Mooney, R. (2000). Intrinsic and extrinsic contributions to auditory selectivity in a song nucleus critical for vocal plasticity. *J. Neurosci.* **20**: 5437-48
- Rozzo, A., Armellin, M., Franzot, J., Chiaruttini, C., Nistri, A. and Tongiorgi, E. (2002). Expression and dendritic mRNA localization of GABA<sub>C</sub> receptor ρ1 and ρ2 subunits in developing rat brain and spinal cord. *Eur. J. Neurosci.* **15**: 1747-58
- Rudolph, U., Crestani, F., Benke, D., Brunig, I., Benson, J. A., Fritschy, J. M., Martin, J. R., Bluethmann, H. and Mohler, H. (1999). Benzodiazepine actions mediated by specific gamma-aminobutyric acid<sub>A</sub> receptor subtypes. *Nature* **401**: 796-800
- Rudolph, U. and Möhler, H. (2004). Analysis of GABA<sub>A</sub> receptor function and dissection of the pharmacology of benzodiazepines and general anesthetics through mouse genetics. *Ann. Rev. Pharmacol. Toxicol.* **44:** 475-98
- Rudolph, U. and Möhler, H. (2006). GABA-based therapeutic approaches: GABA<sub>A</sub> receptor subtype functions. *Curr. Opin. Pharmacol.* **6**: 18-23
- Rumi, M., Ishihara, S., Aziz, M., Kazumori, H., Ishimura, N., Yuki, T., Kadota, C., Kadowaki, Y. and Kinoshita, Y. (2006). RNA polymerase II mediated transcription from the polymerase III promoters in short hairpin RNA expression vector. *Biochem. Biophys. Res. Commun.* 339: 540-7
- Sah, D. W. Y. (2006). Therapeutic potential of RNA interference for neurological disorders. *Life Sci.* 79: 1773-80
- Sancar, F., Ericksen, S. S., Kucken, A. M., Teissere, J. A. and Czajkowski, C. (2007). Structural determinants for high affinity zolpidem binding to GABA-A receptors. *Mol. Pharmacol.* **71:** 38-46
- Sanna, E., Busonero, F., Talani, G., Carta, M., Massa, F., Peis, M., Maciocco, E. and Biggio, G. (2002). Comparison of the effects of zaleplon, zolpidem, and triazolam at various GABA<sub>A</sub> receptor subtypes. *Eur. J. Pharmacol.* **451**: 103-10
- Santhakumar, V., Wallner, M. and Otis, T. S. (2007). Ethanol acts directly on extrasynaptic subtypes of GABA<sub>A</sub> receptors to increase tonic inhibition. *Alcohol* **41:** 211-21
- Scacheri, P. C., Rozenblatt-Rosen, O., Caplen, N. J., Wolfsberg, T. G., Umayam, L., Lee, J. C., Hughes, C. M., Shanmugam, K. S., Bhattacharjee, A., Meyerson, M. and Collins, F. S. (2004). Short interfering RNAs can induce unexpected and divergent changes in the levels of untargeted proteins in mammalian cells. *Proc. Natl. Acad. Sci. USA* **101**: 1892-7
- Scharff, C. and Haesler, S. (2005). An evolutionary perspective on FoxP2: strictly for the birds? *Curr. Opin. Neurobiol.* **15:** 694-703
- Scharff, C. and Nottebohm, F. (1991). A comparative study of the behavioural deficits following lesions of various parts of the zebra finch song system: implications for vocal learning. J. Neurosci. 19: 2898-913
- Scheich, H., Wallhäeusser-Franke, E. and Braun, K. (1991). Does synaptic selection explain auditory imprinting? In Squire, L. R., Weinberger, N. M., Lynch, G. and

McGaugh, J. L. (Eds.) *Memory: Organisation and Locus of Change*. (pp 114-59). Oxford: Oxford University Press

- Scherer, L. J. and Rossi, J. J. (2003). Approaches for the sequence specific knock-down of mRNA. *Nat. Biotech.* **21:** 1457-65
- Schofield, P. R., Darlison, M. G., Fujita, N., Burt, D. R., Stephenson, F. A., Rodriguez, H., Rhee, L. M., Ramachandran, J., Reale, V., Glencorse, T. A., Seeburg, P. H. and Barnard, E. A. (1987). Sequence and functional expression of the GABA<sub>A</sub> receptor shows a ligand-gated receptor super-family. *Nature* 328: 221-7
- Schwarz, S., Ding, H., Kennington, L., Moore, J., Schelter, J., Burchard, J., Linsley, P. S., Aronin, N., Xu, Z. and Zamore, P. D. (2006). Designing siRNA that distinguish between genes that differ by a single nucleotide. *PLoS Genet.* 2: 1307-18
- Schwarzer, C., Berresheim, U., Pirker, S., Wieselthaler, A., Fuchs, K., Sieghart, W. and Sperk, G. (2001). Distribution of the major  $\gamma$ -aminobutyric acid<sub>A</sub> receptor subunits in the basal ganglia and associated limbic brain areas of the adult rat. *J. Comp. Neurol.* **433**: 526-49
- Scott, B. B. and Lois, C. (2007). Developmental origin and identity of song system neurons born during vocal learning in songbirds. *J. Comp. Neurol.* **502:** 202-14
- Scotti, A. L. and Reuter, H. (2001) Synaptic and extrasynaptic gamma-aminobutyric acid type A receptor clusters in rat hippocampal cultures during development. *Proc. Natl. Acad. Sci. USA* 98: 3489-94
- Searcy, W. A. and Marler, P. (1987). Response of sparrows to songs of deaf and isolation-reared males: further evidence for innate auditory templates. *Dev. Psychobiol.* **20**: 509-19
- Semyanov, A., Walker, M. C., Kullmann, D. M. and Silver, R. A. (2004). Tonically active GABA<sub>A</sub> receptors: modulating gain and maintaining the tone. *Trends Neurosci.* **27**: 262-9
- Shan, G. (2009). RNA interference as a gene knockdown technique. *Int. J. Biochem. Cell Biol.* (Epub ahead of print)
- Shankar, P., Manjunath, N. and Lieberman, J. (2005). The prospect of silencing disease using RNA interference. J. Am. Med. Ass. 293: 1367-73
- Shimada, S., Cutting, G. and Uhl, G. R. (1992).  $\gamma$ -aminobutyric acid A or C receptor?  $\gamma$ aminobutyric acid  $\rho$ 1 receptor RNA induces bicuculline-, barbiturate-, and benzodiazepine-insensitive  $\gamma$ -aminobutyric acid responses in *Xenopus* oocytes. *Mol. Pharmacol.* **41:** 683-7
- Shrey, K., Nishant, S. M. and Vibha, R. (2009). RNA interference: Emerging diagnostics and therapeutics tool. *Biochem. Biophys. Res. Commun.* **386**: 273-7
- Shu, H-. J., Eisenman, L. N., Jinadasa, D., Covey, D. F., Zorumski, C. F. and Mennerick, S. (2004). Slow actions of neurosteroids at GABA<sub>A</sub> receptors. J. Neurosci. 24: 6667-75
- Sieghart, W. (1995). Structure and pharmacology of gamma-aminobutyric acid<sub>A</sub> receptor subtypes. *Pharmacol. Rev.* **47:** 181-234
- Sieghart, W. and Ernst, M. (2005). Heterogeneity of GABA<sub>A</sub> receptors: revived interest in the development of subtype-selective drugs. *Curr. Med. Chem.* **5**: 217-42
- Sieghart, W., Fuchs, K., Tretter, V., Ebert, V., Jechlinger, M., Höger, H. and Adamiker, D. (1999). Structure and subunit composition of GABA<sub>A</sub> receptors. *Neurochem. Int.* 34: 379-85
- Sieghart, W. and Sperk, G. (2002). Subunit composition, distribution and function of GABA<sub>A</sub> receptor subtypes. *Curr. Top. Med. Chem.* **2:** 795-816

- Simon, J., Wakimoto, H., Fujita, N., Lalande, M. and Barnard, E. A. (2004). Analysis of the set of GABA<sub>A</sub> receptor genes in the human genome. J. Biol. Chem. 279: 41422-35
- Simon, P. (2003). Q-Gene: processing quantitative real-time RT–PCR data. *Bioinformatics* **19**: 1439-40
- Simpson, H. B. and Vicario, D. S. (1990). Brain pathways for learned and unlearned vocalizations differ in zebra finches. *J. Neurosci.* **10**: 1541-56
- Sine, S. M. and Engel, A. G. (2006). Recent advances in Cys-loop receptor structure and function. *Nature* **440**: 448-55
- Sinkkonen, S. T., Hanna, M. C., Kirkness, E. F. and Korpi, E. R. (2000). GABA<sub>A</sub> receptor  $\varepsilon$  and  $\theta$  subunits display unusual structural variation between species and are enriched in the rat Locus Ceruleus. *J. Neurosci.* **20**: 3588-95
- Siolas, D., Lerner, C., Burchard, J., Ge, W., Linsley, P. S., Paddison, P. J., Hannon, G. J. and Cleary, M. A. (2004). Synthetic shRNAs as potent RNAi triggers. *Nat. Biotechnol.* 23: 227-31
- Slater, P. J. B., Eales, L. A. and Clayton, N. S. (1988). Song learning in zebra finches (*Taeniopygia guttata*): progress and prospects. *Adv. Study Behav.* **18:** 1-34
- Sledz, C. and Williams, B. (2005). RNA interference in biology and disease. *Blood* **106**: 787-94
- Smith, G. B. and Olsen, R. W. (1995). Functional domains of GABA<sub>A</sub> receptors. *Trends Pharmacol. Sci.* **16**: 162-8
- Smith, G. T., Brenowitz, E. A. and Wingfield, J. C. (1997). Roles of photoperiod and testosterone in seasonal plasticity of the avian song control system. J. Neurobiol. 32: 426-42
- Smith, S. S. (2002). Withdrawal properties of a neuroactive steroid: implications for GABA<sub>A</sub> receptor gene regulation in the brain and anxiety behaviour. *Steroids* **67**: 519-28
- Sober, S. J. and Brainard, M. S. (2009). Adult birdsong is actively maintained by error correction. *Nat. Neurosci.* **12:** 927-32
- Soma, M., Hiraiwa-Hasegawa, M. and Okanoya, K. (2009). Song-learning strategies in the Bengalese finch: do chicks choose tutors based on song complexity? *Anim. Behav.* 78: 1107-13
- Song, E., Lee, S. K., Wang, J., Ince, N., Ouyang, N., Min, J., Chen, J., Shankar, P. and Lieberman, J. (2003). RNA interference targeting Fas protects mice from fulminant hepatitis. *Nat. Med.* 9: 347-51
- Sossinka, R. and Bohner, J. (1980). Song types in the zebra finch *Poephila guttata* castanotis. Z. Tierpsychol. 53: 123-32
- Spence, R. D., Zhen, Y., White, S., Schlinger, B. A. and Day, L. B. (2009). Recovery of motor and cognitive function after cerebellar lesions in a songbird - role of estrogens. *Eur. J. Neurosci.* 29: 1225-34
- Spencer, K. A., Buchanan, K. L., Goldsmith, A. R. and Catchpole, C. K. (2003). Song as an honest signal of developmental stress in the zebra finch (*Taeniopygia guttata*). *Horm. Behav.* **44:** 132-9
- Sperk, G., Schwarzer, C., Tsunashima, K., Fuchs, K. and Sieghart, W. (1997). GABA<sub>A</sub> receptor subunits in the rat hippocampus I: immunocytochemical distribution of 13 subunits. *Neurosci.* **80**: 987-1000
- Spiro, J. E., Dalva, M. B. and Mooney, R. (1999). Long-range inhibition within the zebra finch song nucleus RA can coordinate the firing of multiple projection neurons. *J. Neurophys.* **81:** 3007-20

- Steiger, J. L. and Russek, S. J. (2004). GABA<sub>A</sub> receptors: building the bridge between subunit mRNAs, their promoters, and cognate transcription factors. *Pharmacol. Ther.* 101: 259-81
- Stein, V., Hermans-Borqmever, I., Jentsch, T. J. and Hübner, C. A. (2004). Expression of the KCl cotransporter KCC2 parallels neuronal maturation and the emergence of low intracellular chloride. J. Comp. Neurol. 468: 57-64
- Subramaniam, J. R., Corsi, L., Vicini, S., Whiting, P. J. and Neale, J. H. (2001). Ribozyme-mediated reduction of the GABA<sub>A</sub> receptor α1 subunit. *Mol. Brain Res.* **92:** 149-56
- Sui, G., Soohoo, C., El Affar, B., Gay, F., Shi,Y., Forrester, W. C. and Shi, Y. (2002). A DNA vector-based RNAi technology to suppress gene expression in mammalian cells. *Proc. Natl. Acad. Sci. USA* 99: 5515-20
- Sur, C., Wafford, K. A., Reynolds, D. S., Hadingham, K. L., Bromidge, F., Macaulay, A., Collinson, N., O'Meara, G., Howell, O., Newman, R., Myers, J., Atack, J. R., Dawson, G. R., McKernan, R. M., Whiting, P. J. and Rosahl, T. W. (2001). Loss of the major GABA<sub>A</sub> receptor subtype in the brain is not lethal in mice. J. Neurosci. 21: 3409-18
- Symmans, W. F., Fiterman, D. J., Anderson, S. K., Ayers, M., Rouzier, R., Dunmire, V., Stec, J., Valero, V., Sneige, N., Albarracin, C., Wu, Y., Ross, J. S., Wagner, P., Theriault, R. L., Arun, B., Kuerer, H., Hess, K. R., Zhang, W., Hortobagyi, G. N. and Pusztai, L. (2005). A single-gene biomarker identifies breast cancers associated with immature cell type and short duration of prior breastfeeding. *Endocr. Relat. Cancer* 12: 1059-69
- Takahashi, T., Hamada, A., Miyawaki, K., Matsumoto, Y., Mito, T., Noji, S. and Mizunami, M. (2009). Systemic RNA interference for the study of learning and memory in an insect. J. Neurosci. Methods 179: 9-15
- Tarciso, A., Velho., F., Pinaud, R., Rodrigues, P. V. and Mello, C. V. (2005). Coinduction of activity-dependent genes in songbirds. *Eur. J. Neurosci.* 22: 1667-78
- Taylor, P. D., Attwood, T. K. and Flower, D. R. (2003). BPROMPT: A consensus server for membrane protein prediction. *Nucl. Acids Res.* **31**: 3698-700
- Tchernichovski, O., Lints, T., Mitra, P. P. and Nottebohm, F. (1999). Vocal imitation in zebra finches is inversely related to model abundance. *Proc. Natl. Acad. Sci. USA* 96: 12901-4
- Tchernichovski, O. and Nottebohm, F. (1998). Social inhibition of song imitation among sibling male zebra finches. *Proc. Natl. Acad. Sci. USA* **95**: 8951-6
- Terunuma, M., Jang, I. S., Ha, S. H., Kittler, J. T., Kanematsu, T., Jovanovic, J. N., Nakayama, K. I., Akaike, N., Ryu, S. H., Moss, S. J. and Hirata, M. (2004).
  GABA<sub>A</sub> receptor phospho-dependent modulation is regulated by phospholipase Crelated inactive protein type 1, a novel protein phosphatase 1 anchoring protein. *J. Neurosci.* 24: 7074-84
- Thakker, D. R., Natt, F., Hüsken, D., Maier, R., Müller, M., van der Putten, H., Hoyer, D. and Cryan, J. F. (2004). Neurochemical and behavioural consequences of widespread gene knockdown in the adult mouse brain by using nonviral RNA interference. *Proc. Natl. Acad. Sci. USA* 101: 17270-5
- Thode, C., Bock, J., Braun, K. and Darlison, M. G. (2005). The chicken immediate early gene zenk is expressed in the medio-rostral neostriatum/hyperstriatum ventrale, a brain region involved in acoustic imprinting, and is up-regulated after exposure to an auditory stimulus. *Neurosci.* **130**: 611-7

- Thode, C., Güttinger, H. R. and Darlison, M. G. (2008). Expression of the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit gene in discrete nuclei within the zebra finch song system. *Neurosci.* **157**: 143-52
- Thompson, J. A., Wu, W., Bertram, R. and Johnson, F. (2007). Auditory-dependent vocal recovery in adult male zebra finches is facilitated by lesion of a forebrain pathway that includes the basal ganglia. *J. Neurosci.* **27**: 12308-20
- Thompson, C. K., Bentley, G. E. and Brenowitz, E. A. (2007). Rapid seasonal-like regression of the adult avian song control system. *Proc. Natl. Acad. Sci.* 104: 15520-5
- Thompson, S. A., Bonnert, T. P., Cagetti, E., Whiting, P. J. and Wafford, K. A. (2002). Overexpression of the GABA<sub>A</sub> receptor ε subunit results in insensitivity to anesthetics. *Neuropharmacol.* **43**: 662-8
- Thompson-Vest, N. M., Waldvogel, H. J., Rees, M. I. and Faull, R. L. (2003). GABA<sub>A</sub> receptor subunit and gephyrin protein changes differ in the globus pallidus in Huntington's diseased brain. *Brain Res.* **994:** 265-70
- Thorpe, W. H. (1958). The learning of song patterns by buds, with special reference to the song of the chaffinch, *Fringilla coelebs. Ibis.* **100:** 535-75
- Thorpe, W. H. (1958). The nature and characteristics of subsong. Br. Birds. 51: 509-14
- Thorpe, W. H. (1961). Bird-song. Cambridge University Press, New York
- Tierney, M. L., Luu, T. and Gage, P. W. (2008). Functional asymmetry of the conserved cystine loops in  $\alpha\beta\gamma$  GABA<sub>A</sub> receptors revealed by the response to GABA activation and drug potentiation. *Int. J. Biochem. Cell Biol.* **40**: 968-79
- Tischmeyer, W. and Grimm, R. (1999). Activation of immediate early genes and memory formation. *Cell. Mol. Life Sci.* **55:** 564-74
- Todd, A. J., Watt, C., Spike, R. C. and Sieghart, W. (1996). Colocalisation of GABA, glycine, and their receptors at synapses in the rat spinal cord. *J. Neurosci.* **16**: 974-82
- Tögel, M., Mossier, B., Fuchs, K. and Sieghart,W. (1994). γ-aminobutyric acid<sub>A</sub> receptors displaying association of  $\gamma$ 3-subunits with  $\beta$ 2/3 and different  $\alpha$ -subunits exhibit unique pharmacological properties. *J. Biol. Chem.* **269**: 12993-8
- Tomaszycki, M. L., Peabody, C., Replogle, K., Clayton, D. F., Tempelman, R. J. and Wade, J. (2009). Sexual differentiation of the zebra finch song system: potential roles for sex chromosome genes. *BMC Neurosci.* **10**: 24 (online)
- Tomaszycki, M. L., Sluzas, E. M., Sundberg, K. A., Newman, S. W. and DeVoogd, T. J. (2006). Immediate early gene (ZENK) responses to song in juvenile female and male zebra finches: effects of rearing environment. *Inc. J. Neurobiol.* 66: 1175-82
- Tönges, L., Lingor, P., Egle, R., Dietz, G. P. H., Fahr, A. and Bahr, M. (2006). Stearylated octaarginine and artificial virus-like particles for transfection of siRNA into primary rat neurons. *RNA* 12: 1431-8
- Torrey, E. F., Barci, B. M., Webster, M. J., Bartko, J. J., Meador-Woodruff, J. H. and Knable, M. B. (2005). Neurochemical markers for Schizophrenia, Bipolar Disorder, and Major Depression in postmortem brains. *Biol. Psychiatry* 57: 252-60
- Tramontin, A. D., Wingfield, J. C. and Brenowitz, E. A. (1999). Contributions of social cues and photoperiod to seasonal plasticity in the adult avian song control system. *J. Neurosci.* 19: 476-83
- Tretter, V., Hauer, B., Nusser, Z., Mihalek, R. M., Hoger, H., Homanics, G. E., Somogyi, P. and Sieghart, W. (2001). Targeted disruption of the GABA<sub>A</sub> receptor  $\delta$  subunit gene leads to an upregulation of  $\gamma$ 2 subunit-containing receptors in cerebellar granule cells. *J. Biol. Chem.* **276**: 10532-8

- Tretter, V., Jacob, T. C., Mukherjee, J., Fritschy, J. M., Pangalos, M. N. and Moss, S. J. (2008). The clustering of GABA(A) receptor subtypes at inhibitory synapses is facilitated via the direct binding of receptor alpha 2 subunits to gephyrin. J. *Neurosci.* 28: 1356-65
- Tsang, S-. Y., Ng, S-. K., Xu, X. and Xue, H. (2007). The evolution of GABA<sub>A</sub> receptor-like genes. *Mol. Biol. Evol.* **24**: 599-610
- Turenius, C. I., Htut, M. M., Prodon, D. A., Ebersole, P. L., Ngo, P. T., Lara, R. N., Wilczynski, J. J. and Stanley, B. G. (2009). GABA<sub>A</sub> receptors in the lateral hypothalamus as mediators of satiety and body weight regulation. *Brain Res.* 1262: 16-24
- Turner, J. J., Jones, S., Fabani, M. M., Ivanova, G., Arzumanov, A. A. and Gait, M. J. (2007). RNA targeting with peptide conjugates of oligonucleotides, siRNA and PNA. *Blood Cells Mol. Dis.* 38: 1-7
- Tuschl, T. (2002). Expanding small RNA interference. Nat. Biotechnol. 20: 446-8
- Unwin, N. (2005). Refined structure of the nicotinic acetylcholine receptor at 4Å resolution. J. Mol. Biol, **346**: 967-89
- Unwin, N., Miyazawa, A., Li, J., & Fujiyoshi, Y. (2002). Activation of the nicotinic acetylcholine receptor involves a switch in conformation of the alpha subunits. *J. Mol. Biol.* **319:** 1165-76
- Van den Haute, C., Eggermont, K., Nuttin, B., Debyser, Z. and Baekelandt, V. (2003). Lentiviral vector-mediated delivery of short hairpin RNA results in persistent knockdown of gene expression in mouse brain. *Hum. Gene. Ther.* 14: 1799-1807
- Vicario, D. S. and Raksin, J. N. (2000). Possible roles for GABAergic inhibition in the vocal control system of the zebra finch. *Neurorep.* **11**: 3631-5
- Volk, D. W., Pierri, J. N., Fritschy, J. M., Auh, S., Sampson, A. R. and Lewis, D. A. (2002). Reciprocal alternations in pre- and postsynaptic inhibitory markers at chandelier cell inputs to pyramidal neurons in schizophrenia. *Cereb. Cortex* 12: 1063-70
- Wada, K., Howard, J. T., McConnell, P., Whitney, O., Lints, T., Rivas, M. V., Horita, H., Patterson, M. A., White, S. A., Scharff, C., Haesler, S., Zhao, S., Sakaguchi, H., Hagiwara, M., Shiraki, T., Hirozane-Kishikawa, T., Skene, P., Hayashizaki, Y., Carninci, P. and Jarvis, E. D. (2006). A molecular neuroethological approach for identifying and characterizing a cascade of behaviorally regulated genes. *Proc. Natl. Acad. Sci. USA* 103: 15212-7
- Wada, K., Sakaguchu, H., Jarvis, E. D. and Hagiwara, M. (2004). Differential expression of glutamate receptors in avian neural pathways for learned vocalization. J. Comp. Neurol. 476: 44-64
- Wafford, K. A., Burnett, D. M., Leidenheimer, N. J., Burt, D. R., Wang, J. B., Kofuji, P., Dunwiddie, T. V., Harris, R. A. and Sikela, J. M. (1991). Ethanol sensitivity of the GABA<sub>A</sub> receptor expressed in *Xenopus* oocytes requires 8 amino acids contained in the γ2L subunit. *Neuron* **7**: 27-33
- Wafford, K. A., Macaulay, A. J., Fradley, R., O'Meara, G. F., Reynolds, D. S. and Rosahl, T. W. (2004). Differentiating the role of γ -aminobutyric acid type A (GABA<sub>A</sub>) receptor subtypes. *Biochem. Soc. Trans.* **32**: 553-6
- Wallace, R. H., Marini, C., Petrou, S., Harkin, L. A., Bowser, D. N., Panchal, R. G., Williams, A., Sutherland, G. R., Mulley, J. C., Scheffer, I. E. and Berkovic, S. F. (2001). Mutant GABA<sub>A</sub> receptor γ2-subunit in childhood absence epilepsy and febrile seizures. *Nat. Gen.* 28: 49-52

- Wallhäusser-Franke, E., Nixdorf-Bergweiler, B. E. and DeVoogd, T. J. (1995). Song isolation is associated with maintaining high spine frequencies on zebra finch 1MAN neurons. *Neurobiol. Learn. Mem.* 64: 25-35
- Wanamaker, C. O., Christianson, J. C. and Green, W. N. (2003). Regulation of nicotinic acetylcholine receptor assembly. *Ann. N. Y. Acad. Sci.* **998:** 66-80
- Wang, J., Sakagichi, H. and Sokabe, M. (1999). Sex differences in the vocal motor pathway of the zebra finch revealed by real-time optical imaging technique. *Neurorep.* **10**: 2487-91
- Ward, B. C., Nordeen, E. J. and Nordeen, K. W. (2001). Anatomical and ontogenetic factors producing variation in HVc neuron number in zebra finches. *Brain Res.* 2: 318-26
- Watanabe, S. and Bischof, H. J. (2009). Effects of hippocampal lesions on acquisition and retention of spatial learning in zebra finches. *Behav. Brain Res.* **155**: 147-52
- Webb, D. M. and Zhang, J. (2005) *FoxP2* in song-learning birds and vocal-learning mammals. *J. Hered.* **96:** 1-5
- Weiland, N. G. and Orchinik, M. (1995). Specific subunit mRNAs of the GABA<sub>A</sub> receptor are regulated by progesterone in subfields of the hippocampus. *Mol. Brain Res.* 32: 271-8
- West, M. and King, A. (1988). Female visual displays affect the development of male song in the cowbird. *Nature* **334:** 244-6
- Whiting, P. J. (2003a). GABA<sub>A</sub> receptor subtypes in the brain: A paradigm for CNS drug discovery? *Drug Disc. Today* **8:** 445-50
- Whiting, P. J. (2003b). The GABA<sub>A</sub> receptor gene family: New opportunities for drug development. *Curr. Opin. Drug Disc. Dev.* **6**: 648-57
- Whiting, P. J., McAllister, G., Vassilatis, D., Bonnert, T. P., Heavens, R. P., Smith, D. W., Hewson, L., O'Donnell, R., Rigby, M. R., Sirinathsinghji, D. J. S., Marshall, G., Thompson, S. A. and Wafford, K. A. (1997). Neuronally restricted RNA splicing regulates the expression of a novel GABA<sub>A</sub> receptor subunit conferring atypical functional properties. *J. Neurosci.* 17: 5027-37
- Whiting, P. J., McKernan, R. M. and Iversen, L. L. (1990). Another mechanism for creating diversity in  $\gamma$ -aminobutyrate type A receptors: RNA splicing directs expression of two forms of  $\gamma 2$  subunit, one of which contains a protein kinase C phosphorylation site. *Proc. Natl. Acad. Sci. USA* **87**: 9966-70
- Whitney, O., Soderstrom, K. and Johnson, F. (2000). Post-transcriptional regulation of zenk expression associated with zebra finch vocal development. *Mol. Brain. Res.* 80: 297-90
- Wieland, H. A., Lüddens, H. and Seeburg, P. H. (1992). A single histidine in GABA<sub>A</sub> receptors is essential for benzodiazepine agonist binding. *J. Biol. Chem.* **267**: 1426-9
- Wienholds, E. and Plasterk, R. H. (2005). MicroRNA function in animal development. *FEBS Lett.* **579:** 5911-22
- Wild, J. M. (2004). Functional neuroanatomy of the sensorimotor control of singing. Ann. N. Y. Acad. Sci. 1016: 438-62
- Wild, J. M., Kubke, M. F. and Mooney, R. (2009). Avian nucleus retroambigualis: Cell types and projections to other respiratory-vocal nuclei in the brain of the zebra finch (*Taeniopygia guttata*). J. Comp. Neurol. 512: 768-83
- Wild, J. M., Williams, M. N., Howie, G. J. and Mooney, R. (2005). Calcium-binding proteins define interneurons in HVC of the zebra finch (*Taeniopygia guttata*). J. Comp. Neurol. 483: 76-90

- Wilke, K., Gaul, R., Klauck, S. M. and Poustka, A. (1997). A gene in human chromosome band Xq28 (GABRE) defines a putative new subunit class of the GABA<sub>A</sub> neurotransmitter receptor. *Genomics* **45**: 1-10
- Wilkins, M. E., Hosie, A. M. and Smart, T. G. (2002). Identification of a β subunit TM2 residue mediating proton modulation of GABA type A receptors. *J. Neurosci.* **22**: 5328-33
- Wilkins, M. E. and Smart, T. G. (2002). Redox modulation of GABA<sub>A</sub> receptors obscured by Zn<sup>2+</sup> complexation. *Neuropharmacol.* **43**: 938-44
- Williams, B. R. (1997). Role of the double-stranded RNA-activated protein kinase (PKR) in cell regulation. *Biochem. Soc. Trans.* **25:** 509-13
- Williams, B. R. (1999). PKR: a sentinel kinase for cellular stress. Oncogene 18: 6112-8
- Williams, H., Connor, D. M. and Hill, J. W. (2003). Testosterone decreases the potential for song plasticity in adult male zebra finches. *Horm. Behav.* **44:** 402-12
- Williams, H. and McKibben, J. R. (1992). Changes in stereotyped central motor patterns controlling vocalization are induced by peripheral nerve injury. *Behav. Neur. Biol.* 57: 67-78
- Wingrove, P. B., Safo, P., Wheat, L., Thompson, S. A., Wafford, K. A., Whiting, P. J. (2002). Mechanism of alpha-subunit selectivity of benzodiazepine pharmacology at gamma-aminobutyric acid type A receptors. *Eur. J. Pharmacol.* **437**: 31-9
- Wisden, W., Laurie, D. J., Monyer, H. and Seeburg, P. H. (1992). The distribution of 13 GABA<sub>A</sub> receptor subunit mRNAs in the rat brain. I. Telencephalon, diencephalon, mesencephalon. J. Neurosci. 12: 1040-62
- Woolley, S. M. (2004). Auditory experience and adult song plasticity. *Ann. N. Y. Acad. Sci.* **1016:** 208-21
- Woolley, S. M. and Rubel, E. W. (1997). Bengalese finches Lonchura Striata Domestica depend upon auditory feedback for the maintenance of adult song. J Neurosci. 17: 6380-90
- Woolley, S. M. and Rubel, E. W. (2002). Vocal memory and learning in adult Bengalese finches with regenerated hair cells. *J. Neurosci.* **22**: 7774-87
- Xia, H., Mao, Q., Eliason, S. L., Harper, S. Q., Martins, I. H., Orr, H. T., Paulson, H. L., Yang, L., Kotin, R. M. and Davidson, B. L. (2004). RNAi suppresses polyglutamine-induced neurodegeneration in a model of spinocerebellar ataxia. *Nat. Med.* **10**: 816-20
- Xia, H., Mao, Q., Paulson, H. L. and Davidson, B. L. (2002). siRNA-mediated gene silencing *in vitro* and *in vivo*. *Nat. Biotechnol.* **20**: 1006-10
- Xiang, Y-Y., Wang, S., Liu, M., Hirota, J. A., Li, J., Ju, W., Fan, Y., Kelly, M. M., Ye, B., Orser, B., O'Byrne, P. M., Inman, M. D., Yang, X. and Lu, W-. Y. (2007). A GABAergic system in airway epithelium is essential for mucus over production in asthma. *Nat. Med.* 13: 862-7
- Xie, A., Song, X., Ripps, H. and Qian, H. (2008). Cyclothiazide: a subunit-specific inhibitor of GABA<sub>C</sub> receptors. *J. Physiol.* **586**: 2743-52
- Xu., M. and Akabas, M. H. (1996). Identification of channel-lining residues in the M2 membrane-spanning segment of the GABA<sub>A</sub> receptor α1 subunit. J. Gen. Physiol. 107: 195-205
- Yamada, J., Okabe, A., Toyoda, H., Kilb, W., Luhmann, H. J. and Fukuda, A. (2004). Cl- uptake promoting the depolarizing GABA actions in immature rat neocortical neurons is mediated by NKCC1. J. Physiol. 557: 829-41
- Yang, M. and Mattes, J. (2008). Discovery, biology and therapeutic potential of RNA interference, microRNA and antagomirs. *Pharmacol. Ther.* **117**: 194-204

- Ying, S. Y. and Lin, S. L. (2004). Intron-derived microRNAs-fine tuning of gene functions. *Gene* **342**: 25-8
- Yu, E. Z., Counterman Burba, A. E. and Gerstein, M. (2007). PARE: A tool for comparing protein abundance and mRNA expression data. *BMC Bioinformatics* 8: 309 (online)
- Yu, J-. Y., DeRuiter, S. L. and Turner, D. L. (2002). RNA interference by expression of short-interfering RNAs and hairpin RNAs in mammalian cells. *Proc. Natl. Acad. Sci. USA* 99: 6047-52
- Zamore, P. D. and Haley, B. (2005). Ribo-gnome: the big world of small RNAs, *Science* **309**: 1519-24
- Zann, R. A. (1996). *The Zebra Finch: a Synthesis of Field and Laboratory Studies*. Oxford: Oxford University Press
- Zarrindast, M-. R., Noorbakhshnia, M., Motamedi, F., Haeri-Rohani, A. and Rezayof, A. (2006). Effect of the GABAergic system on memory formation and state-dependent learning induced by morphine in rats. *Int. J. Exp. Clin. Pharmacol.* 76: 93-100
- Zeitelhofer, M., Karra, D., Vessey, J. P., Jaskic, E., Macchi, P., Thomas, S., Riefler, J., Kiebler, M. and Dahmet, R. (2007). High-efficiency transfection of mammalian neurons via nucleofection. *Nat. Protoc.* 2: 1692-704
- Zemkova, H. W., Bjelobaba, I., Tomic, M., Zemkova, H. and Stojilkovic, S. S. (2008). Molecular, pharmacological and functional properties of GABA<sub>A</sub> receptors in anterior pituitary cells. J. Physiol. 586: 3097-111
- Zeringue, H. C. and Constantin-Paton, M. (2004). Post-transcriptional gene silencing in neurons. *Curr. Opin. Neurobiol.* **14:** 654-9
- Zevin, J. D., Seidenberg, M. S. and Bottjer, S. W. (2004). Limits on reacquisition of song in adult zebra finches exposed to white noise. *J. Neurosci.* **24**: 5849-62
- Zhang, H., Kolb, F. A., Brondani, V., Billy, E. and Filipowicz, W. (2002). Human dicer preferentially cleaves dsRNAs at their termini without a requirement for ATP. *Eur. Mol. Biol. Assoc. J.* 21: 5875-85
- Zhang, M. Q. (1999). Statistical features of human exons and their flanking regions. *Hum. Mol. Gen.* **7:** 919-32
- Zhao, C., Xu, Z., Wang, F., Chen, J., Ng, S. K., Wong, P. W., Yu, Z., Pun, F. W., Ren, L., Lo, W. S., Tsang, S. Y. and Xue, H. (2009). Alternative-splicing in the exon-10 region of GABA(A) receptor beta(2) subunit gene: relationships between novel isoforms and psychotic disorders. *PLoS One* **4**: 6977 (online)
- Zhu, Y., Ripps, H. and Qian, H. (2007). A single amino acid in the second transmembrane domain of GABA  $\rho$  receptors regulates channel conductance. *Neurosci. Lett.* **418**: 205-9
- Zimprich, F., Zezula, J., Sieghart, W. and Lassmann, H. (1991). Immunohistochemical localisation of the  $\alpha$ 1-,  $\alpha$ 2-, and  $\alpha$ 3-subunits of the GABA<sub>A</sub> receptor in the rat brain. *Neurosci. Lett.* **127**: 125-8

Bibliography

## **APPENDIX**



pGEM<sup>®</sup>-T Easy vector map. The multiple cloning site is flanked by many recognition sites for restriction enzymes. *Eco*RI sites were used to release the cloned sequences. The ampicillin resistance gene enables antibiotic selection and the *lacZ* gene enables blue/white screening for recombinant plasmids.

The following primers were used for sequencing:

<b>SP6:</b>	5´-TATTTAGGTGACACTATAG-3´
T7:	5´-TAATACGACTCACTATAGGG-3´



p*Silencer*<sup>™</sup> 4.1-CMV neo vector map. Double-stranded oligonucleotides are annealed and ligated into the vector. The powerful human cytomegalovirus (CMV) promoter then drives the expression of the shRNA. The ampicillin and neomycin genes allow for antibiotic selection.

The following primers were used for sequencing:

Forward:	5'-AGGCGATTAAGTTGGGTA-3'
<b>Reverse:</b>	5'-CGGTAGGCGTGTACGGTG-3'



pMAL vector map (~6.5kb). The *malE* gene encodes maltose-binding protein, the arrows denote the direction of transcription. The target gene is inserted in the same translational reading frame as the *malE* gene so the DNA will be expressed by the cell as a single protein. The red arrow indicates the position of insertion of the GABA<sub>A</sub> receptor  $\gamma$ 4-subunit antibody construct. This site was selected as it required the least mis-matches to be introduced to the construct and positioning at the start of the polylinker ensures minimum vector-derived sequence in the expressed fusion protein. The factor Xa cleavage site is shown, after amylose-affinity purification the target protein will be cleaved from the MBP and purified.

The following primers were used for sequencing:

Forward:	5'-GGTCGTCAGACTGTCGATGAAGCC-3'
<b>Reverse:</b>	5'-CGCCAGGGTTTTCCCAGTCACGAC-3'

GenBank accession numbers (mRNA and peptide sequences):

Rat (α3):	NM_017069 (SEQ)	NP_058765 (PEP)
Chicken ( $\alpha$ 3):	XM_420268 (SEQ)	XP_420268 (PEP)
Rat (α4):	NM_080587 (SEQ)	NP_542154 (PEP)
Chicken ( $\alpha$ 4):	XM_420724 (SEQ)	XP_420724 (PEP)
Rat (α6):	NM_021841 (SEQ)	NP_068613 (PEP)
Chicken ( $\alpha 6$ ):	NM_205058 (SEQ)	NP_990389 (PEP)
Rat (γ1):	NM_080586 (SEQ)	NP_542153 (PEP)
Chicken (y1):	XM_420725 (SEQ)	XP_420725 (PEP)
Rat (δ):	NM_017289 (SEQ)	NP_058985 (PEP)
Chicken (δ):	XM_001234040 (SEQ)	XP_001234041 (PEP)
Chicken $(\pi)$ :	XM_426542 (SEQ)	XP_426524 (PEP)
Chicken $(\pi)$ :	XM_414507 (SEQ)	XP_414507 (PEP)
Zebra finch ( $\pi$ ):	XM_002188687 (SEQ)	XP_002188723 (PEP)

```
Zebra finch \beta 1 CTGGAGATCGAGAGCTATGGTTACACGGTGGATGACATTGTCTTCTTCTGGCAAGGGAAT
Zebra finch \beta4 CTGGAGATCGAGAGCTATGGTTACACGGTGGATGACATTGTCTTCTTGGCAAGGGAAT
Zebra finch \beta1 GACTCTGCTGTCACGGGGATGGAGGTGCTGGAGCTGCCCCAGTTCACCATCATCGAGCAG
Zebra finch \beta4 GACTCTGCTGTCACGGGGATGGAGGTGCTGGAGCTGCCCCAGTTCACCATCGAGCAG
Zebra finch \beta 1 AGGCTGGTCAGCAGGGAAGTGGTCTTCACCACTGGTTCGTATCTGCGCTTATCCCTGAGT
Zebra finch \beta4 AGGCTGGTCAGCAGGGAAGTGGTCTTCACCACTGGTTCGTATCTGCGCTTATCCCTGAGT
Zebra finch \beta 1 ATCACCATCCTGTCCTGGGTCTCCTTCTGGATCAACTATGATGCTTCTGCAGCACGAGTG
Zebra finch \beta 4 ATCACCATCCTGTCCTGGGTCTCCTTCTGGATCAACTATGATGCTTCTGCAGCACGAGTG
Zebra finch \beta1 GCACTGGGGGTCACCACGGTGCTGACCATGACAACCATCAACACCACCTGCGGGAGACT
Zebra finch \beta4 GCACTGGGGGTCACCACGGTGCTGACCATGACAACCATCAACACCACCTGCGGGAGACT
              Zebra finch \beta1 CTGCCCAAGATCCCCTACGTCAAGGCTATTGATGTTTATCTCATGGGCTGCTTTGTCTTC
Zebra finch \beta4 CTGCCCAAGATCCCCTACGTCAAGGCTATTGATGTTTATCTCATGGGCTGCTTCGTCTTC
Zebra finch \beta1 GTGTTCCTGGCACTCCTGGAATATGCTTTTGTCAACTACATATTCTTCGGGCGAGGGCCG
Zebra finch \beta4 GTGTTCCTGGCACTCCTGGAATATGCTTTTGTCAACTACATATTCTTCGGGCGAGGGCCG
              Zebra finch \beta1 CGGCAGCAGAAGAAGCAGA-GCGAGCGCGTCAGCAAGGCCAACAACGAGCGCCACCGTTA
Zebra finch \beta4 CGGCAGCAGAAGAAGCAGAAGCAGCGGCGCGCAGCAAGAACGAACGAGCGCCACCGTTA
                  * * * * * * * * * * * * * *
Zebra finch \beta 1 CGAGGAGAAGAGGGTGAGAGAGGGTGAGCAGGTTGACCCTTACGGTAACATCCTCCTCAGCACGCT
Zebra finch \beta4 CGAGGAGAAGAGGGTGAGAGAGGAGGAGGTTGACCCTTACGGTAACATCCTCCTCAGCACGCT
Zebra finch \beta_1 = GGAGATGAACAACGAGCTGCTGGCCACGGACATGATGAGCAGCGTGGGCGACTCTCGAAA
Zebra finch eta4 GGAGATGAACAACGAGCTGCTGGCCACGGACATGATGAGCAGCGTGGGCGACTCTCGAAA
Zebra finch B1 CTCTGTCATGTCCTTCGAAGGCTCAGGAATCCAGTTCCGCAAGCCGCTGGCCTCTCGGGA
Zebra finch \beta4 CTCTGTCATGTCCTTCGAAGGCTCAGGAATCCAGTTCCGCAAGCCGCTGGCCTCTCGGAA
Zebra finch \beta1 TGGCTTTGGCCACCACCCCGGCCACGCCACGTCCCGCTGAGCCACCACGCCGCCGC
Zebra finch \beta4 TGGCTTTGGCCACCACCCACCNTGGACCGCCACGTCCCGNTGAGCCACCACGCCGCCGC
                   ******
Zebra finch \beta1 CCGCAACCGCGCCAACTGCCGCCTGCGCCGGCGGTCCTCCAAGCTGAAGCTCAAAATCCC
Zebra finch \beta4 CCGCAACCGCGCCAACTGCCGCCGGCGGCGGCCCCCAAGCTGAAGCTCAAAATCCC
              Zebra finch \beta1 AGACCTGACAGACGTCAGCACCATTGACAAGTGGTCACGGATCATTTTTCCAATCACTTT
Zebra finch \beta4 AGACCTGACAGACGTCAGCACCATTGACAAGTGGTCACGGATCATTTTTCCAATCACTTT
              Zebra finch \beta1 TGGATTCTTCAACCTTGTT
Zebra finch \beta4 TGGATTCTTCATCCTTGTT
              *****
```

Alignment of the predicted zebra finch  $GABA_A$  receptor  $\beta$ 1-subunit nucleotide sequence (single-letter code) with the zebra finch  $\beta$ 4-subunit sequence cloned within the laboratory. There is 99% sequence identity, confirming the predicted  $\beta$ 1-subunit sequence to be that encoding  $\beta$ 4. (\*) denote identity between the sequences.